SILVERTOWN TUNNEL

6.10 Code of Construction Practice

TR010021

APFP Regulation 5(2)(q) Revision 4 Planning Act 2008 Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

April 2017





MAYOR OF LONDON

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

Silvertown Tunnel

Code of Construction Practice 6.10

Planning Act 2008

The Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Document Reference:6.10Internal Code:ST150030-PLN-ZZZ-DSD-ZZ-0071Regulation Number:5(2)(q)Author:Transport for London

Rev.	Date	Approved By	Signature	Description
0	29/04/2016	David Rowe (TfL Lead Sponsor)		For DCO Application
1	15/11/2016	David Rowe (TfL Lead Sponsor)		For DCO Deadline 1

2	14/12/2016	David Rowe (TfL Lead Sponsor)		For DCO Deadline 2
3	06/03/2017	David Rowe (TfL Lead Sponsor)		For DCO Deadline 4
4	03/04/2017	David Rowe (TfL Lead Sponsor)		For DCO Deadline 6

Contents

List of	List of Abbreviations9		
Gloss	ary of Terms	11	
1.		19	
1.1	Purpose of the Code of Construction Practice	19	
1.2	Description of the Scheme	19	
1.3	Approach to Scheme delivery	21	
1.4	Construction environmental management approach	21	
2.	GENERAL SITE OPERATIONS	29	
2.1	Construction process		
2.2	Construction programme		
2.3	Construction hours of work		
2.4	Worksites		
2.5	Advance works		
2.6	Access Arrangements		
2.7	Emergency planning		
3.	CONSTRUCTION TRANSPORTATION	35	
3.1	Construction Traffic Management Plan		
3.2	River transport		
4.	COMMUNICATIONS AND COMMUNITY LIAISON		
5.	AIR QUALITY		
5.1	General		
5.2	Vehicle and plant emissions		
5.3	Dust management	45	
5.4	Monitoring		
5.5	Odour		
6.	CULTURAL HERITAGE	51	
7.	TERRESTRIAL ECOLOGY	53	
7.1	General	53	

7.2	Monitoring	53
8.	MARINE ECOLOGY	55
9.	GEOLOGY, SOILS AND HYDROGEOLOGY	57
9.1	General	57
9.2	Intrusive Groundworks	57
9.3	Geology and soils	57
9.4	Groundwater	58
9.5	Human Health	61
9.6	Unexploded Ordnance	61
10.	GROUND SETTLEMENT	62
10.1	General	62
10.2	Surveys	62
10.3	Repairs	63
10.4	Settlement Deed	64
11.	NOISE AND VIBRATION	66
11.1	General	66
11.2	Detailed measures	66
12.	TOWNSCAPE AND VISUAL AMENITY	70
12.1	Visual disruption	70
12.2	Lighting	70
13.	MATERIAL RESOURCES AND WASTE	72
13.1	General	72
13.2	Materials management	72
13.3	Waste	74
13.4	Asbestos and health risks	75
14.	EFFECTS ON ALL TRAVELLERS	78
14.1	Pedestrians and cyclists	78
14.2	Active traffic demand management	78
14.3	Additional mitigation	78
15.	WATER ENVIRONMENT AND FLOOD RISK	81

15.1	General81
15.2	Site drainage
15.3	Control of pollution
15.4	In-river works
15.5	Water conservation
15.6	Flood risk
16. C	ARBON EMISSIONS
Appendix	A Settlement Assessment and Mitigation Process
Appendix	B. Settlement Deed97
Appendix	C. CD&E Materials Commitments99
Appendix	D. Receptor Site Assessment101
Appendix	E. Site Waste Management Plan103
Appendix	F. Groundwater Monitoring Strategy 105
Appendix	G. Construction Noise and Vibration Mitigation Scheme
Appendix	H. Outline Ecology Management Plan109

List of Tables

Table 1-1 Summary of subsidiary environmental management plans	. 23
Table 5-1 Mitigation measures to reduce dust impacts	. 45
Table A-1 Building Damage Classification	. 90
Table A-2 Scoring for Sensitivity Assessment of Listed Buildings	. 92

List of Figures

Figure 1-1 Order Limits	20
Figure 2-1 Envisaged construction programme	

List of Abbreviations

AQMP	Air Quality Management Plan
BPM	Best Practicable Means
CEMP	Construction Environmental Management Plan
CEP	Community Engagement Plan
CMMP	Construction Materials Management Plan
CoCP	Code of Construction Practice
СТМР	Construction Traffic Management Plan
DCO	Development Consent Order
DLR	Docklands Light Railway
EAL	Emirates Air Line
ЕНО	Environmental Health Officer
EMP	Environmental Management Plan
EOD	Explosive Ordnance Disposal
EP	Emergency Plan
ES	Environmental Statement
FORS	Fleet Operator Recognition Scheme
HGV	Heavy Goods Vehicle

НМР	Heritage Management Plan
LFEPA	London Fire and Emergency Planning Authority
NVMP	Noise and Vibration Management Plan
PP	Passage Plan
RPA	Root Protection Area
RSA	Receptor Site Assessment
SRN	Strategic Road Network
SWMP	Site Waste Management Plan
ТВМ	Tunnel Boring Machine
TfL	Transport for London
WSI	Written Scheme of Investigation
UXO	Unexploded Ordnance

Glossary of Terms

Asset Control Limits	The predefined values, based on assessment, relating to safety and serviceability considerations that instigate a review of risk to third party assets.	
Asset protection	Asset protection is a process by which the impacts to all structures (that is buildings, utilities and highways or other structures) potentially at risk of damage from ground movements, or other effects arising from construction of the Scheme, are kept to acceptable levels.	
Beneficial use	 Ecological benefit or land reinstatement / landscaping: The activity will assist in ecological benefit and/or help to facilitate an approved change/alteration in land use or form. 	
	 Works (linked to a consented planning activity or permit) that aims to restore, enhance or be part of a land management scheme i.e. landfill or quarry. 	
	 Reduce the requirement for alternative material (waste or not) to be used for the purposes of any such Scheme. 	
Black redstart	The black redstart is a small robin-sized bird that has adapted to live at the heart of industrial and urban centres. Its name comes from the plumage of the male, which is grey-black in colour with a red tail. With fewer than 100 breeding pairs in the UK, the black redstart is on the amber list of Birds of Conservation Concern.	
Blackwall Tunnel	An existing road tunnel underneath the River Thames in east London, linking the London Borough of Tower Hamlets with the Royal Borough of Greenwich, comprising two bores each with two lanes of traffic.	
Considerate Constructors Scheme (CCS)	Construction sites, companies and suppliers can voluntarily register with the CCS and agree to abide by the Code of Considerate Practice, designed to encourage best practice beyond statutory requirements. The Code of Considerate	

	Practice commits those sites and companies registered with the CCS to respect the community, protect the environment, secure everyone's safety and value their workforce.
Contractor	The Contractor will be the construction entity through which the Project Company would deliver the design and construction of the Scheme and includes anyone who directly employs or engages construction workers or manages construction work, including sub-contractors, or any individual self-employed worker or business that carries out, manages or controls construction work.
Detailed Design	A finalised design, complete in all aspects and suitable for construction of the Scheme.
Development Consent Order (DCO)	This is a statutory order which provides consent for the project and means that a range of other consents, such as planning permission and listed building consent, will not be required. A DCO can also include provisions authorising the compulsory acquisition of land or of interests in or rights over land which is the subject of an application. http://infrastructure.planninginspectorate.gov.uk/help/glossary- of-terms/
Dewatering	The process of removing groundwater from an aquifer.
Emergency Preparedness Plan	The plan which defines specific actions to be taken in response to observed asset/ground movements in accordance with the Asset Control Limits.
Excavated Material	Ground or other material removed during a construction process, usually by mechanical means.
Heavy Good Vehicle (HGV)	European Union term for any vehicle with a gross combination mass of over 3500kg.
Heritage Asset	A building, monument, site, place, area of landscape identified as having a degree of significance meriting consideration in planning decisions, because of heritage interest. Heritage

	asset includes designated heritage assets and assets identified by the local planning authority (including local listing).
Illustrative Design	An example of how the proposals could be developed at the next stage of design as a result of engagement with the Project Company contractor, planning authority and other relevant stakeholders. This is an example of how the Scheme may look, but it is not the final design.
Instrumentation and Monitoring Plan	The plan which defines the type and location of instrumentation used to monitor an asset, as well as the frequency of recording and reporting.
Jetty	A structure that projects from land out into water for the purposes of marine logistics.
Order Limits	The extent of land and rights over land that will be needed temporarily to construct the Scheme, and permanently to operate, maintain and safeguard the Scheme (often referred to as 'the red line boundary').
Pollution Prevention Guidelines	Best practice guidelines set out by the Environment Agency to advise industry and public on legal responsibilities and good environmental practice.
Project Company	A Project Company is typically a consortium of private sector companies, formed for the specific purpose of providing the services under a private finance contract. This is also technically known as a Special Purpose Vehicle (SPV).
	The Project Company will obtain funding to design and build the new facilities and then undertake routine maintenance and capital replacement during the remainder of the contract period. The total contract period is typically 30 years.
	The Project Company will repay funders from payments received from TfL during the post construction period of the contract. Receipt of payments from TfL will depend on the ability of the Project Company to deliver the services in accordance with the output specified in the contract.

Public health	All organised efforts to improve population health through prevention and promotion activities. The focus of public health is the population and not the individual.
Reference Design	The design proposals for the Scheme that the DCO application refers to, as modified and developed in response to the Statutory Consultation process. The Reference Design has been developed to a concept stage appropriate to prove engineering and construction feasibility and to inform the construction and operational land requirements, environmental impact assessments and Scheme cost estimate.
Settlement	Settlement is the technical term given to the way the ground moves around a hole after it has been dug out.
Site Waste Management Plan	A document that outlines how the Scheme will reduce, manage, and dispose of its solid waste.
Silt	The generic term for particles with a grain size of 4-63 μ m, i.e. between clay and sand.
Suitable Excavated Material	"Suitable Excavated Material" means all bored or excavated material from the tunnel works for which treatment is not technically feasible or which would not require treatment were it to be disposed of to a permitted facility.
Surface Water	Water that appears on the land surface that has not seeped into the ground, i.e. lakes, rivers, streams, standing water, ponds, precipitation.
The Scheme	The construction of a new bored tunnel with cut and cover sections at either end under the River Thames (the Silvertown Tunnel) between the Greenwich peninsula and Silvertown, as well as necessary alterations to the connecting road network and the introduction of user charging at both Silvertown and Blackwall tunnels.
Transportable	Transportable Moisture Limits are defined in The Merchant Shipping (Carriage of Cargoes) Regulations 1999 as the

Moisture Limits	9/10ths (or 90%) of the flow moisture point. This limit sets the standard for accepting cargoes which may liquefy for marine transport. Cargoes with a moisture content above the TML should not be transported by ship unless <i>"appropriate safety arrangements are made to the satisfaction of the Certifying Authority to ensure adequate stability in the case of cargo shifting, and the ship has adequate structural integrity."</i>
Transport for London (TfL)	A local government body responsible for most aspects of the transport system in Greater London. Its role is to implement transport strategy and to manage transport services across London.
	These services include: buses, the Underground network, Docklands Light Railway, Overground and Trams. TfL also runs Santander Cycles, London River Services, Victoria Coach Station and the Emirates Air Line.
	As well as controlling a 580km network of main roads and the city's 6,000 traffic lights, TfL regulates London's private hire vehicles and the Congestion Charge scheme
The Tunnel, Silvertown Tunnel	Proposed new twin-bore road tunnels under the River Thames from the A1020 in Silvertown to the A102 on Greenwich Peninsula, East London.
Tunnel Boring Machine (TBM)	A machine used to excavate tunnels with a circular cross section. There are two main types of closed face TBMs: Earth Pressure Balance (EPB) and Slurry Shield (SS). Please see those terms for further explanation.
Unexploded Ordnance	Unexploded ordnance are explosive weapons (bombs, shells, grenades, land mines, naval mines, etc.) that did not explode when they were employed and still pose a risk of detonation, potentially many decades after they were used or discarded.
Waste	Waste is defined in Article 1(a) of the European Waste Framework Directive 2008/98/EC as 'any substance or object in the categories set out in Annex I which the holder discards or intends to discard or is required to discard'. The term 'holder' is defined as the producer of the waste or the person

	who is in possession of it and 'producer' is defined as anyone whose activities produce waste. Waste can be further classified as hazardous, non-hazardous or inert.
Watching Brief	The watching of a situation by a suitably qualified person to ensure that works (e.g. an excavation) are being done correctly, and in accordance with relevant environmental standards.
Worksite	An area of land within the Scheme Order Limits which is temporarily occupied and used to undertake construction works.

SUMMARY

- S.1.1 The purpose of this Code of Construction Practice (CoCP) is to set a framework to control possible impacts arising from the construction of the Silvertown Tunnel, hereafter referred to as the Scheme. The CoCP covers environmental, public health and safety aspects of the Scheme that may affect the interests of local residents, businesses, the general public and the surroundings in the vicinity of the Scheme.
- S.1.2 Transport for London (TfL) proposes to deliver the Silvertown Tunnel through a private finance contract, as this would best meet the project objectives and constraints, and achieve an appropriate risk balance between the public and private sector. The Contractor would be responsible for the design, construction, financing and maintenance of the tunnel and supporting infrastructure for a period of 25 years.
- S.1.3 The Contractor would be responsible for undertaking the Detailed Design in accordance with the constraints and parameters of the DCO; TfL's specifications; and the requirements, and any other commitments given by TfL.
- S.1.4 The Contractor will also be responsible for compliance with this CoCP, including production of an overarching Construction Environmental Management Plan and any other detailed plans required to be produced. A summary of these subsidiary plans is included in chapter 1 of the CoCP.

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

1. INTRODUCTION

1.1 Purpose of the Code of Construction Practice

- 1.1.1 This document is the Code of Construction Practice (CoCP), for the Silvertown Tunnel (hereafter referred to as the Scheme), which has been prepared to accompany the Development Consent Order (DCO) Application.
- 1.1.2 The purpose of the CoCP is to set a framework to control possible impacts arising from the construction of the Scheme including removal/decommissioning of temporary works. The CoCP covers environmental, public health and safety aspects of the Scheme that may affect the interests of local residents, businesses, the general public and the surroundings in the vicinity of the Scheme.
- 1.1.3 The control measures set out in this CoCP are based on the findings and mitigation measures set out in the Environmental Statement (Document Reference: 6.1) prepared following consultation with stakeholders.
- 1.1.4 The CoCP will apply to all works authorised by the DCO and all works undertaken by the Contractor. The Contractor will also comply with all legislation relating to the construction activities.
- 1.1.5 Compliance with this CoCP is a requirement of the DCO. Non-compliance with the CoCP would therefore be a breach of the terms of the DCO. In the event of a breach, both TfL and the Contractor would be open to enforcement action under Section 161 of the Planning Act 2008.
- 1.1.6 TfL will direct and manage the Contractor's compliance with these controls and will specify that all construction activities shall be carried out in accordance with the terms of the DCO, including the CoCP. Further details are included in the relevant sections of the CoCP.

1.2 Description of the Scheme

1.2.1 The Silvertown Tunnel scheme (the Scheme) involves the construction of a twin bore road tunnel providing a new connection between the A102 Blackwall Tunnel Approach on Greenwich Peninsula (Royal Borough of Greenwich) and the Tidal Basin roundabout junction on the A1020 Lower Lea Crossing/Silvertown Way (London Borough of Newham). The Silvertown Tunnel would be approximately 1.4km long and would be able to accommodate large vehicles including double-deck buses. It would include a

dedicated bus, coach and goods vehicle lane, which would enable TfL to provide additional cross-river bus routes.

- 1.2.2 The Scheme also includes the introduction of free-flow user charging on both the Blackwall Tunnel (northern portal located in London Borough of Tower Hamlets) and at the new Silvertown Tunnel. This measure is intended to play a fundamental role in managing traffic demand and supporting the financing of the construction, maintenance and operation of the Silvertown Tunnel.
- 1.2.3 Main construction works could commence in late 2018 and would last approximately four years with the new tunnel opening in 2022/23. The main construction Worksite would be located at Silvertown, enabling the utilisation of the existing barge facilities at Thames Wharf along with a new temporary jetty, if necessary, for the removal of spoil and delivery of materials by river. A secondary Worksite would be located adjacent to the alignment of the proposed cut and cover tunnel on the Greenwich Peninsula.
- 1.2.4 Figure 1-1 Order Limits, represents the 'envelope' within which the Scheme would be constructed.



Figure 1-1 Order Limits

Page 20 of 110

1.3 Approach to Scheme delivery

- 1.3.1 TfL proposes to deliver the Silvertown Tunnel through a private finance contract, as this would best meet the project objectives and constraints. The contract would be competitively tendered in accordance with EU procurement procedures.
- 1.3.2 The successful tenderer, to be called the Project Company, would be responsible for the detailed design, construction, financing and maintenance of the tunnel and supporting infrastructure for a period. TfL would retain control of traffic management for Silvertown Tunnel and Blackwall Tunnel as part of the strategic road network in London, and would continue to maintain and operate Blackwall Tunnel and approach roads under existing arrangements.
- 1.3.3 The Project Company would be responsible for undertaking the Detailed Design and constructing the works, in accordance with the constraints and parameters of the DCO; TfL's specifications; and any other commitments given to stakeholders by TfL.
- 1.3.4 TfL through the Contractor will ensure that the tunnel is built in accordance with all relevant and current environmental legislation and, where reasonably practicable, with good practice for minimising the environmental effects of construction.

1.4 Construction environmental management approach

- 1.4.1 Contractual arrangements will require the Contractor to provide suitably qualified environmental staff to monitor, manage and execute works for which they are responsible. TfL will require that the Contractor demonstrates an appropriate awareness of local sensitivities, expected codes of conduct, working knowledge of relevant legislation, codes of practice, and guidance relevant to the various construction activities in which they are engaged. TfL would require the Contractor to have an Environmental Management System in accordance with BS EN ISO14001¹ requirements.
- 1.4.2 Following the appointment of the Contractor for the works, it will be the Contractor's responsibility to produce and maintain an overarching Construction Environmental Management Plan (CEMP) for the construction

¹ BSI (2015). BS EN ISO14001 Environmental Management Systems.

works. The CEMP will be developed in consultation with the local authorities and the relevant statutory stakeholders for each topic area, and will be submitted for approval to TfL. The CEMP will set out the Contractor's roles and responsibilities, together with appropriate control measures, training and briefing procedures, risk assessments, stakeholder engagement responsibilities, monitoring systems, and operations to be employed during planning and constructing the works for all relevant environmental topics to demonstrate compliance with the measures and controls set out in the CoCP and the DCO (including its requirements and the Protective Provisions).

- 1.4.3 In addition to the measures specifically mentioned in the CoCP, the Contractor is required to produce and implement a number of subsidiary plans which are specific to particular environmental topics. As set out in the CoCP, these plans have to be produced in consultation with, or approved by, the relevant planning authorities or other stakeholders, and include the measures required for inclusion within these plans by the CoCP.
- 1.4.4 These subsidiary plans will set out the specific protection, mitigation and compensation measures to be taken by the Contractor to control the environmental effects for the individual topic during the construction of the Scheme including removal/de-commissioning of temporary works. A summary of the subsidiary plans that are required to be produced under the CoCP is set out in Table 1-1.
- 1.4.5 Nothing in this CoCP precludes any of the subsidiary plans set out below being amended by the Contractor following their approval or consultation with the relevant bodies set out below. Any amendment to these plans will be required to go through the same consultation and approval mechanisms set out in the table.
- 1.4.6 No approval of any individual plan listed in Table 1-1 will fetter the discretion of any relevant stakeholder in relation to the content of any other plan in which they have a consultative or approval role.
- 1.4.7 An environmental impact assessment was carried out for the Scheme and an Environmental Statement (ES) prepared in accordance with the Infrastructure Planning (Environmental Impact Assessment) Regulations 2009, as amended. Through the assessment process, mitigation identified with respect to construction effects is embedded within the CoCP.
- 1.4.8 The Contractor must ensure that the construction of the Scheme does not give rise to materially new or materially different environmental effects to

those reported in the ES. This will be achieved through the process described in paragraph 1.4.9.

- 1.4.9 The Contractor must identify, before the commencement of the relevant phase of construction, whether its detailed design, construction methodology, programme, site-specific mitigation or other assumptions are different from those assumed during the preparation of the ES. If they are different, the Contractor must assess the likely significant effects of the proposals and identify any design changes or mitigation measures that are necessary to ensure that the construction does not give rise to materially new or materially different environmental effects to those reported in the ES.
- 1.4.10 The results of this assessment must be included in the subsidiary plans set out in Table 1-1 below when these are submitted to the relevant local planning authority and other stakeholders for approval or consultation. Each submission must include evidence, including details of further mitigation where necessary, to demonstrate that the construction method will not give rise to materially new or materially different environmental effects to those reported in the ES.
- 1.4.11 For the purposes of this assessment, the Scheme shall be deemed not to give rise to materially new or materially different environmental effects if the significance of the effect reported for each topic, taking account of any necessary mitigation, is the same as or better than the level of significance reported against that topic in the ES. In this context, 'the ES' means the documents of that description set out in Schedule 14 to the DCO.

	Document	Role	Role of Local
			Authorities and/or
			Stakeholders
1	Construction	The CEMP will set out the	The CEMP will be
	Environmental	Contractor's roles and	produced by the
	Management	responsibilities to demonstrate	Contractor for approval
	Plan (CEMP)	compliance with the measures	by TfL in consultation
		and controls of the CoCP and	with the relevant
		the DCO (including its	planning authorities and
		Requirements and the	other relevant
		Protective Provisions).	stakeholders including
			the PLA.
Emerg	gency Planning		
2	Emergency	The Emergency Plan will relate	The Emergency Plan will
	Plan	to the Order land that is not	be produced by the
	(including	within the River Thames and	Contractor in

rabio i i oanniary of oabolalary official inanagomone plano

Document Reference: 6.10

	Document	Role	Role of Local Authorities and/or Stakeholders
3	Emergency Spill Response Plan) Fire Plan	 will include: notification procedures for Emergency Services in the event of an incident; coordination procedures for TfL Customer Services and the Traffic Control Centre; flood risk emergency procedures; emergency spill response procedures; emergency phone numbers; and a Flood Warning and Evacuation Plan 	consultation with the local Emergency Services, and the relevant local authority emergency planning officer. It will include a Flood Warning and Evacuation Plan to be approved by the relevant local authority emergency planning officer in consultation with the Environment Agency.
		event of fire during construction including details of escape routes, emergency doors, meeting points, and fire training.	Contractor in consultation with the London Fire and Emergency Planning Authority.
Const	ruction Transpor	tation	
4	Construction Traffic Management Plan (CTMP)	 The CTMP will include: details of how logistics will be managed, e.g. lorry routes, diversions, main access/egress points; traffic incidents plan dealing with incidents or severe congestion on agreed construction routes; and construction workers travel plan, developed to encourage the use of sustainable modes of transport (including river transport) to and from the 	A CTMP will be produced by the Contractor for each worksite and approved by the relevant planning authority in consultation with the relevant highway authority.

	Document	Role	Role of Local Authorities and/or
			Stakeholders
		worksite by those	
_		working on the project.	
5	Passage Plan (PP)	To establish cycle times for loading, unloading and both journeys for vessels in relation to tides and will permit an informed decision regarding the number of vessels required to meet the production rates achieved for the TBM and civil works, and will include an updated navigational risk assessment which will reflect the findings and recommendations of the Navigational Issues and Preliminary Risk Assessment submitted with the application. To make provision in respect of the River Thames that is equivalent to the provision for dry land in the Emergency Plan	The Passage Plan will be produced by the Contractor for approval by the PLA.
6	Construction Site River Strategy (CSRS)	The CSRS will include details of the approach adopted by the Contractor to maximise river transport for construction and excavated materials and to meet the commitments in respect of the use of river transport set out in this CoCP.	The CSRS prepared by the Contractor will be submitted to TfL for approval in consultation with the relevant planning authority and the PLA.
Comm	nunication and C	ommunity Liaison	
/	Community Engagement Plan (CEP)	The CEP will identify how communication with stakeholders will be managed and programmed throughout the construction period. It will include steps that will be taken to liaise with specific stakeholders, where they are potentially affected by the works.	Engagement Plan (CEP) will be prepared by the Contractor and submitted to the relevant planning authorities for approval.

Document Reference: 6.10

	Document	Role	Role of Local
			Authorities and/or Stakeholders
Enviro	onmental Control	S	
8	Air Quality Management Plan (AQMP)	The AQMP will contain details of the measures to limit vehicle, plant and dust emissions during construction.	An AQMP will be prepared for each worksite by the Contractor and submitted for approval to the relevant planning authority.
9	Archaeological Written Scheme of Investigation (AWSI)	An AWSI will outline the mitigation measures and recording proposals for dealing with the currently unknown sub- surface archaeological remains that could potentially be affected by the Scheme on both the north and south side of the Scheme.	The AWSI will be prepared by the Contractor in consultation with Historic England and submitted for approval to the relevant planning authority. Where mitigations measures within the river Thames are identified in the AWSI, the Contractor shall consult the MMO and PLA.
10	Ecology Management Plan (EMP)	The Ecology Management Plan will set out measures to manage the risk of adversely affecting flora and fauna on and within the vicinity of the worksites, including method statements in the event invasive species are encountered and details how additional survey requirements would be accommodated in the programme for both the north and south side of the Scheme. A Site Clearance Plan will form part of each Ecology Management Plan.	The Ecology Management Plan will be prepared by the Contractor substantially in accordance with the Draft Ecology Management Plan (Appendix G) and in consultation with Natural England and then submitted for approval to the relevant planning authority.
11	Construction Materials	The CMMP will set out measures to ensure materials	The CMMP will be prepared by the

Document Reference: 6.10

	Document	Role	Role of Local Authorities and/or Stakeholders
	Management Plan (CMMP)	are handled and used in a way that prevents harm to human health and pollution of the environment.	Contractor and will be subject to approval by the relevant planning authorities.
12	Groundwater Monitoring and Verification Plan (GMVP)	The Groundwater Monitoring and Verification Plan will set out monitoring and reporting criteria during pre-construction, construction and post construction.	The GMVP will be prepared by the Contractor and will be subject to approval by the Environment Agency.
13	Noise and Vibration Management Plan (NVMP)	The NVMP will set out measures to control and limit noise and vibration levels in the vicinity of the construction works.	An NVMP will be prepared by the Contractor for each Worksite and will be subject to approval by the relevant planning authority.
14	Lighting Management Plan (LMP)	The Lighting Management Plan will include appropriate industry standard procedures which will be implemented at both worksites.	TheLightingManagementPlanforeachworksitewillbepreparedbytheContractorinconsultationwiththerelevantplanningauthority,theEnvironmentAgency,and the PLA .
15	Site Waste Management Plan (SWMP)	The SWMP submitted with the DCO application will be updated by the Contractor. The updated SWMP will consider how the waste hierarchy will be applied and details of how all wastes will be managed. The SWMP will also provide a framework for checking compliance with waste legislation and the Duty of Care.	No consultation/approval needed.

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

Silvertown Tunnel Code of Construction Practice Document Reference: 6.10

2. GENERAL SITE OPERATIONS

2.1 Construction process

2.1.1 The Scheme is a major construction project and will involve many different types of construction activities. These activities are likely to include but will not be limited to: demolition; site clearance; site investigation; treatment of excavated materials (as necessary); tunnelling; piling; excavation; services diversion and new installations; jetty works; new bridge works, highway works; and below ground and surface building works.

2.2 Construction programme

2.2.1 Main construction activities could commence at both Silvertown and Greenwich in late 2018 and will continue for approximately four years, as shown in Figure 2-1.



Figure 2-1 Envisaged construction programme

2.3 Construction hours of work

2.3.1 Hours of work would be in line with standard good practice for major construction works. Normal working hours for non-tunnel construction works will be from 08:00 to 18:00 Monday - Friday and 08:00 to 13:00 on Saturday with no work taking place on Sundays or bank/public holidays. To maximise productivity within the core hours, the Contractor will require a period of up to one hour before and up to one hour after normal working hours for start-up and close down of activities. Start-up and close down activities can include, but are not limited to, preparation, maintenance, site briefings, meetings and

training. Plant or machinery likely to cause a disturbance to local residents or businesses must not be operated during these start-up and close-down periods. These periods will not be considered an extension of core working hours.

- 2.3.2 Where feasible, operations likely to cause disturbance and/or disruption would be limited to within the core working hours. However, some activities may be required outside these hours, e.g. delivery of abnormal loads. Except where activities are safety critical or in an emergency, any works outside these hours must be subject to agreement with the relevant local Environmental Health Officers (EHO).
- 2.3.3 Some minor activities, such as changes in traffic management operations, may be required out of core working hours on a more frequent basis, but this would not be expected to have a significant impact in the context of the existing movements of traffic. A Construction Traffic Management Plan (CTMP) will be put in place and will include mechanisms to review the changes in traffic management operations and measures to minimise any impacts on the local residents of each borough.
- 2.3.4 Tunnel boring works and associated supporting activities will be undertaken on a 24 hour, seven days per week basis. Associated supporting activities will include any below ground works as well as above ground works which may include, but are not limited to; lifting operations into and out of the TBM launch/reception chambers, movement and manufacture/fabrication of materials required for tunnelling (including segments) on site, movement of tunnel spoil using conveyors, HGVs (in accordance with the CTMP) and barges, production/treatment and transportation of concrete and slurry for use in the tunnels, dewatering activities, and any other works required to ensure the continued safe operation of the tunnelling works (including lighting in accordance with the LMP).

2.4 Worksites

- 2.4.1 There will be two main worksites: the Silvertown worksite to the north of the River Thames and the Greenwich worksite to the south of the River Thames.
- 2.4.2 It is envisaged that the worksite located at Silvertown would likely contain offices, stores, plant maintenance facilities, materials testing laboratory, recycling facilities and potential concrete batching plants, materials stockpiles and a wheel wash. This worksite has been selected as the best location for utilising Thames Wharf for marine logistics.

- 2.4.3 A further worksite would be located adjacent to the alignment of the proposed tunnel on the Greenwich Peninsula. It is envisaged that this would likely comprise site offices, spoil and material storage areas and plant storage areas. The Greenwich site does not have any direct wharf access.
- 2.4.4 The worksites would be established at the commencement of the works and will be removed and the worksites will be reinstated at the end of the construction phase.
- 2.4.5 The Contractor will be registered with, and comply with the principles of the Considerate Constructors Scheme or similar.
- 2.4.6 The layout and appearance of the worksites will be designed using the following principles:
 - the size of worksite to be occupied at each of the locations would provide sufficient space to undertake the works in a safe manner without taking space that is non-essential to the construction;
 - worksites will be secure and screened where necessary;
 - storage sites, fixed plant, machinery, equipment and temporary offices will be located to limit environmental effects, as far as reasonably practicable, and have due regard to adjacent buildings, as far as allowed by the constraints of the worksite(s);
 - site lighting will be located and directed so as not to intrude into occupied residential properties or disturb wildlife on sensitive areas or constitute a road hazard or affect navigation; and
 - fixed site plant and facilities will be powered from mains electrical sources.
- 2.4.7 A helpline service will be set up and the helpline number, and a contact name and address will be displayed at appropriate locations on the boundaries of the worksites. Further details of the communication and community liaison are provided in Chapter 4 of this CoCP.
- 2.4.8 Access to the worksites would be limited to specified entry points only and all personnel entries/exits would be recorded and monitored for both security and health and safety purposes. The worksites boundaries will be secured and constructed such that they minimise opportunities for unauthorised entry.

2.5 Advance works

- 2.5.1 Advance works could be undertaken prior to the main works at both Silvertown and Greenwich worksites. These works may include but are not limited to:
 - utility connections;
 - utilities diversions:
 - geotechnical, obstructions and UXO investigations;
 - ecological surveys;
 - archaeological surveys;
 - site clearance;
 - asset protection condition surveys and establishment of monitoring baselines as necessary;
 - design works for early TBM and segment procurement advance (preconstruction) surveys; and
 - fencing and footway diversions; All advance works will be subject to the controls set out in the CoCP.

2.6 Access Arrangements

- 2.6.1 The Contractor will take measures to maintain reasonable access to premises during the construction works. These will include:
 - Measures to ensure the continuation of access for premises including requirements for complementary measures to ensure that delivery and servicing access can take place.
 - Localised temporary diversions will be sought during the cut and cover tunnel works.

2.7 Emergency planning

2.7.1 The Contractor will ensure that emergency procedures for each worksite are developed. The procedures will be standardised as far as possible across both worksites and will be appropriate to the anticipated hazards and the specific layout. The Contractor will set these procedures out in an Emergency Plan (EP) which must be produced in consultation with the

Emergency Services, the PLA the relevant local authority emergency planning officer and the Environment Agency where appropriate. The EP will include:

- notification procedures for Emergency Services , e.g. Police, in the event of an incident;
- coordination procedures for TfL operational services;
- flood risk emergency procedures, as set out in paragraph 2.7.4 and 2.7.5;
- an emergency spill response procedures that will take into account Environment Agency guidelines²; and
- emergency phone numbers and the method of notifying local authorities and statutory bodies. Contact numbers for the key TfL and the Contractor's staff will also be included.

Emergency access

2.7.2 The Contractor will ensure that the requirements of the London Fire and Emergency Planning Authority (LFEPA) will be followed for the provision of site access points.

Fire prevention and control

2.7.3 A Fire Plan will be prepared, in consultation with LFEPA, and will be updated as necessary, having due regard to relevant current guidance.

Flood risk

2.7.4 A Draft Flood Warning and Evacuation Plan (FWEP) (Document Reference: 6.3.16.3) has been prepared and contains information on flood emergency response actions for both construction and operational phases of the Scheme. A number of actions have been outlined within the plan, including registering both the northern and southern portals of the Scheme with the Environment Agency Floodline Warning Direct service, identifying appropriate access and egress routes and designating evacuation points.

² Environment Agency (2014). Incident Response Planning PPG21.

- 2.7.5 The Contractor will be required to further develop and refine the procedures in the Draft FWEP specifically for the construction phase, to be approved by the relevant local authority emergency planning officer in consultation with the Environment Agency. These flood risk emergency response procedures will be included in the Emergency Plan, as set out in paragraph 2.7.1.
- 2.7.6 The Contractor will ensure that all site operatives are familiar with the emergency arrangements.

3. CONSTRUCTION TRANSPORTATION

3.1 Construction Traffic Management Plan

- 3.1.1 Construction worksites and proposed lorry routes have been developed in consultation with local authorities to minimise the impacts of construction traffic on the road network and local communities. The main access routes at both worksites will be further detailed by the Contractor as the Scheme construction solutions are finalised.
- 3.1.2 The management of construction logistics will be established in a Construction Traffic Management Plan (CTMP) to be prepared by the Contractor before construction commences. The plan will embed contractual requirements, the outcome of consultation with the relevant local authorities, and comprehensive logistics planning.
- 3.1.3 A CTMP will be produced by the Contractor for each worksite for approval by the relevant planning authority in consultation with the relevant highway authority, prior to commencing the relevant part of the authorised development The Contractor may bring forward changes to an approved CTMP during the construction of the Scheme, but changes must be approved by the relevant planning authority in consultation with the relevant highways authority.
- 3.1.4 The CTMP for each worksite will be developed in accordance with relevant best practice including for example TfL's guidance on Construction Logistics Plans (or equivalent). The CTMPs will include information on the following aspects:

• Site information

This section will include details of the construction site locations and main access/egress points for vehicles and pedestrians.

Construction details

This section will set out the works programme, with indicative dates for stages of construction, and information on the level of deliveries required. Detailed construction and delivery traffic routes will be specified and agreed by the relevant planning authority in consultation with the relevant highway authority, with local roads only to be used for immediate access to the worksites or local businesses (including wharves). Constraints and restrictions on road vehicle movements to be included in the CTMP are likely to include:

- days of the week and times of the day when road vehicle movements are not permitted;
- maximum number of vehicle movements permitted at defined periods of the day, e.g. between 08:00 and 09:00, or restrictions on the use of the Blackwall Tunnel by construction lorries at peak times; and
- procedures for abnormal loads.

• Traffic management

This section of the CTMP will detail how non-construction traffic will be managed at each stage of construction, including temporary and permanent road closures and diversions and pedestrian and cycle facilities (pursuant to Article 10 of the DCO). Details of any changes that are required to signage and parking arrangements in the vicinity of the worksites will be set out in this section. Information on the process that will be followed by the Contractor in dealing with traffic incidents or severe congestion on agreed construction and delivery routes will also be set out.

• Policies and procedures

The CTMP will incorporate the outcome of communications undertaken in accordance with the CEP.

The CTMP must include a construction workers travel plan (CWTP) developed to encourage the use of sustainable modes of transport (including river transport) to and from the worksite by those working on the project.

• Monitoring, compliance and reporting

The CTMP will also detail how, when and by whom it will be monitored, including, but not limited to, the ongoing review of the following:

- Freight Operator Recognition Scheme (membership);
- collision reporting;
- driver licence checks;
- vehicle safety equipment audits;
- number of vehicle movements to site;
- vehicle mileage;
- level of vehicle fill;
- CO2 emissions;
- delivery accuracy;
- breaches and complaints; and
- construction workers' travel behaviour to inform the ongoing monitoring of the CWTP.
- 3.1.5 The CTMP will ensure that safety measures are implemented to minimise road-related risks. The Contractor will specify the highest current standards in construction vehicle safety, including visibility. This includes but is not limited to FORS Gold (Fleet Operator Recognition Scheme), CLOCS (Construction Logistics and Cycle Safety), SLS (Safety Lorry Scheme) and WRRR (Work Related Road Risk) scheme. Signs identifying the Silvertown Tunnel project and Contractor contact numbers will be displayed in a prominent position on all construction vehicles. All vehicles working in the construction of the Silvertown Tunnel will be compliant with the Mayor's Direct Vision Standard.
- 3.1.6 The CTMP must provide contact details of key personnel, including key stakeholders and highway / planning authorities. These details must be reviewed and updated as necessary as part of the ongoing CTMP monitoring programme.
- 3.1.7 The CTMP will specify the routes to be used by construction heavy goods vehicles (HGVs) to and from the worksites. Construction HGVs would be routed on the TLRN and principal roads as far as possible, with local roads only used to directly access the worksites, local businesses, and wharves used for the import and/or export of material by river. The principal routes to and from the worksites will be the A12, the A13, the A2 and the A102 with access between these routes and the worksite via Lower Lea Crossing, , Blackwall Lane and Millennium Way. Any deviations from this approach would need to be agreed in advance in the CTMP with the local planning authorities.

3.2 River transport

- 3.2.1 The Scheme will seek to maximise river transport for construction materials and excavated material.
- 3.2.2 The river transport objectives are to:
 - minimise the effects of construction of the Scheme associated with the transport of construction materials and excavated materials;
 - minimise the number and length of construction-related transport movements; and
 - minimise the potential social and environmental impacts arising from construction-related transport associated with the Scheme.
- 3.2.3 Subject to paragraphs 3.2.12 3.2.14 the Contractor shall transport:
 - at least 55% by weight of all materials associated with the Scheme by River; and
 - 100% of suitable excavated material out by River.
- 3.2.4 For the purposes of the commitments given in paragraph 3.2.3, the following materials associated with the Scheme shall be deemed to have been transported by river:
 - Materials which are transported to or from the worksite directly by river;
 - Materials which are re-used on site;
 - Materials which are transported by river to a wharf local to the Scheme, transferred to road vehicles and subsequently delivered to the worksite by road, provided that:
 - o the worksite lies within a 4km radius of the wharf; and
 - the distance over which the materials are carried by road from the wharf to the worksite does not exceed the distance that the materials are transported by river from the point of loading up to the local wharf.
 - Concrete for use in the Scheme which is produced at a batching plant sited at a wharf local to the Scheme and using aggregates delivered to the wharf by river and which is delivered to a worksite by road, provided that:

- the worksite lies within 4km of the batching plant; and
- the distance over which the concrete is carried by road from the batching plant to the worksite does not exceed the distance that the aggregates are transported by river to the local wharf.
- Suitable excavated material which is transported by road from the worksite to a wharf local to the Scheme and subsequently transferred to a receptor site using the river, provided that:
 - o The worksite lies within 4km of the wharf; and
 - the distance over which the suitable excavated material is carried by road from the worksite to the wharf does not exceed the distance over which the material is transported by river.
- 3.2.5 "Suitable Excavated Material" means all bored or excavated material from the tunnel works which would not require treatment were it to be disposed of to a permitted facility.
- 3.2.6 For the purposes of paragraph 3.2.5, the definition of treatment shall be in accordance with the Landfill Directive (1999/31/EC) and the Environmental Permitting Guidance: The Landfill Directive (Defra, 2008).
- 3.2.7 Before commencing construction of the Scheme, the Contractor must prepare a Construction Site River Strategy (CSRS), which must:
 - explain how the Contractor will maximise river transport for construction and excavated materials;
 - explain how the Contractor meet the commitments set out in paragraph 3.2.3;
 - provide a breakdown of the quantities of materials that will be transported by river;
 - provide a baseline programme for when materials proposed to be transported by river will be transported to/from site.
- 3.2.8 Construction of the Scheme must not commence until the CSRS has been submitted to and approved by TfL in consultation with the relevant planning authority(s) and the PLA.

Monitoring

- 3.2.9 The Contractor shall submit monthly monitoring reports to TfL, the relevant planning authority(s) and the PLA on the performance of the Scheme in meeting the commitments in paragraph 3.2.3.
- 3.2.10 The monthly monitoring reports shall show the measured quantities of materials in and out of site by river. These measured quantities shall be compared against the baseline programme contained in the CSRS and the total materials in and out of site.
- 3.2.11 In the event that monitoring demonstrates that the quantities of materials being transported by river are not consistent with the baseline programme in the CSRS, the report must contain an explanation of the reasons for this and detail the measures which the Contractor will take to ensure that the commitments in 3.2.3 are met. Anyuch measures must be undertaken in accordance with the CoCP and other relevant subsidiary plans.

Derogations

- 3.2.12 Where circumstances outside the Contractor's control arise which mean that the commitments in paragraph 3.2.3 cannot be met, the Contractor shall submit an application for a Derogation for approval to TfL and the relevant planning authority(s), who shall consult with the PLA prior to approval. Examples of such circumstances include where the river use is unavailable due to poor weather or damage to the river transport system (conveyors/barges).
- 3.2.13 The Contractor may agree with the relevant planning authority(s), who must consult the PLA, additional circumstances in which the Contractor may apply for a Derogation from the commitments in paragraph 3.2.3. Any such agreement must be given in writing by the relevant planning authority(s) and may only be given if the Contractor has demonstrated to the satisfaction of the relevant planning authority(s) that the Derogation will not give rise to any materially new or materially different environmental effects to those assessed in the ES.
- 3.2.14 An application for a Derogation must:
 - Be submitted at the same time as the monthly monitoring report for the month following the month in which the circumstances which require the derogation occur;

- Demonstrate that the Contractor has undertaken such measures as are reasonably practicable to attempt to meet the commitments in 3.2.3 despite the circumstances outside of the Contractor's control;
- Identify measures to mitigate the effects of the derogation which could include additional traffic management and/or mitigation, or limiting permitted hours for HGV movements;
- Provide a revised breakdown of the quantities of materials that will be transported by river;
- Provide a revised programme for when materials proposed to be transport by river will be transported to/from site.

Navigational risk management

- 3.2.15 The Contractor will be responsible for producing a full Passage Plan (PP). The PP will establish cycle times for loading, unloading and journeys for vessels in relation to tides and will permit an informed decision regarding the number of vessels required to meet the production rates achieved for the TBM and civil works. The PP will be prepared for approval by the PLA.
- 3.2.16 The PP will include an updated navigational risk assessment which will reflect the findings and recommendations of the Navigational Issues and Preliminary Risk Assessment (Document Reference: 6.3.7.1) prepared for the Scheme. Risk control measures will include, but will not be limited to:
 - appointing a berthing co-ordinator for the duration of the Scheme's riverine activities to assist with planning, managing and ensuring that safe berthing, approach and manoeuvring practices are adopted and maintained during the construction period;
 - establishing a permanent construction River Response Team to manage the construction and river user vessel interface; in particular with any recreational users. The River Response Team would consult with the PLA and ensure that any exclusion zones are enforced and that safe distances are maintained between construction plant and construction related vessel movements in particular when and if river conditions change;
 - engagement with the PLA; and
 - employment of suitably qualified staff, Masters of Commercial Vessels and marine operators.

4. COMMUNICATIONS AND COMMUNITY LIAISON

- 4.1.1 Communication with local authorities and all key stakeholders will be undertaken throughout the construction period. Key stakeholders will include, but are not limited to; local residents, local businesses, community resources (such as schools and community centres) and vulnerable groups, where they are potentially affected by the works.
- 4.1.2 The Contractor will appoint a Community Construction Liaison Manager (CCLM) for the duration of the works. This manager will prepare and implement a Community Engagement Plan (CEP) which must be submitted to the relevant planning authority for approval. The CCLM will be the main point of contact for stakeholders on site, providing information and resolving issues of concern.
- 4.1.3 The CEP will identify how communication with stakeholders will be managed and programmed throughout the construction period. It will include steps that will be taken to liaise with specific stakeholders, such as schools, where they are potentially affected by the works. This may cover, but is not limited to, community meetings at key stages of work, one-on-one meetings with key resources, newsletters and letter drops (explaining forthcoming works).
- 4.1.4 The CEP will include the following measures:
 - The Contractor will set up and maintain a 24 hour telephone helpline service. This will act as a first point of contact for information or queries raised by stakeholders. All calls and emails will be actioned within a specified time (set out in the CEP) and will be logged, including action taken. There will also be an agreed complaints procedure which shall include a central complaints log.
 - The Contractor will be required to set up and maintain a website which will include the construction programme, main construction activities, and updates. There will also be a project email address, which will enable stakeholders to contact the construction team direct with any queries or concerns.
 - The Contractor will establish and maintain a Community Liaison Group. This group will meet regularly before and during the construction period. It will comprise representatives from key local groups, user representatives, and the local authorities. Invitations to join the group

will also be sent to all affected landowners for the Scheme. The regular meetings will provide an opportunity for the Community Construction Liaison Manager to brief people about up and coming construction activity and answer questions. This will include, but not be limited to, informing the Group of upcoming construction traffic management measures and noisy works. The terms of reference (which will include provision for communications to the Group from the Community Construction Liaison Manager in between meeting dates), chair, and frequency of this group will be finalised at the first meeting.

- The CEP will explain how the needs of vulnerable groups will be met in terms of use of accessible media, English as a second language and appropriate formats for the visually impaired.
- The Contractor will develop a programme of community involvement through volunteering and educational activity.
- The Contractor shall establish co-ordination and communication meetings with key stakeholders. Such meetings will include regular road traffic meetings with TfL, the relevant highway authority, the relevant planning authority as well as affected businesses & developments, which shall include, but not be limited to, Ansco, Knight Dragon, Quintain, Brenntag, Kloeckner Metals UK, Royal Mail, London City Airport and ExCel London Ltd. Where such meetings already exist the Contractor shall be obliged to attend and contribute.

5. AIR QUALITY

5.1 General

- 5.1.1 The Contractor will, as far as reasonably practicable, seek to control and limit emissions to the atmosphere in terms of gaseous and particulate pollutants from vehicles and plant used on the worksites, and dust from construction, demolition, vehicles and plant activities.
- 5.1.2 For each worksite, the Contractor will develop and implement an Air Quality Management Plan (AQMP) to be approved by the relevant planning authority. The AQMP will contain details of the measures to limit vehicle, plant and dust emissions during construction, including those set out in this chapter.

5.2 Vehicle and plant emissions

- 5.2.1 Vehicle and plant emissions will be controlled by implementing the following measures within the AQMP:
 - all Non-road Mobile Machinery (NRMM) will comply with the standards set within the Greater London Authority's Control of Dust and Emissions During Construction and Demolition Supplementary Planning Guidance3. This outlines that, from 1st September 2015, all NRMM of net power 37 kW to 560 kW used on the site of a major development in Greater London must meet Stage IIIA of EU Directive 97/68/EC (Directive 97/68/EC of the European Parliament and of the Council, 1997) and its subsequent amendments as a minimum. From 1st September 2020 NRMM used on any site within Greater London will be required to meet Stage IIIB of the Directive as a minimum;
 - engines of all vehicles, mobile and fixed plant on site will not be left running/idling unnecessarily;
 - using low emission vehicles and plant fitted with catalysts, diesel particulate filters or similar devices;

³ Greater London Authority (2014). The Control of Dust and Emissions during Construction and Demolition.

- using ultra low sulphur fuels in plant and vehicles;
- plant will be well maintained, with routine servicing of plant and vehicles to be completed in accordance with the manufacturer's recommendations and records maintained for the work undertaken;
- minimising the use of diesel or petrol powered generators and using mains electricity or battery powered equipment where practicable;
- maximising energy efficiency (this may include using alternative modes of transport, maximising vehicle utilisation by ensuring full loading and efficient routing);
- the Contractor's delivery vehicles will be required to comply with any low emission zone applicable to the worksite and delivery routes at the time;
- all members of the Contractor's staff who drive vehicles under the contract will undertake a fuel efficient driver training course; and
- all vehicles working on the construction of the Silvertown Tunnel will be Euro 6 unless otherwise agreed with the GLA and relevant local planning authorities.

5.3 Dust management

5.3.1 The Institute of Air Quality Management (IAQM) guidance⁴ and Mayor's Dust and Emissions Supplementary Planning Guidance⁵ provide a number of potential mitigation measures to reduce dust impacts during the construction phase. The measures to be included in the AQMP are set out, but are not limited to those, in Table 5-1.

Issue	Control measure
Communications	 Develop and implement the CEP to include community engagement before work commences on site;
	 Display the name and contact details of person(s)

Table 5-1 Mitigation measures to reduce dust impacts

⁴ Institute of Air Quality Management (2015). Land-Use Planning & Development Control: Planning For Air Quality

⁵ GLA (2014). The Control of Dust and Emissions Supplementary Planning Guidance.

Code of Construction Practice

Document Reference: 6.10

Issue	Control measure
	accountable for air quality and dust issues on the site boundary of both construction worksites;
	 Hold regular liaison meetings with other high risk construction sites within 500m of the site boundary, to ensure plans are co-ordinated and particulate matter emissions are minimised.
Site Management	 Record all dust and air quality complaints in a complaints log which may be made available to the Local Authority upon request; and
	• Record any exceptional incidents that cause dust/or air emissions, and the action taken to resolve the situation.
Monitoring	Undertake on-site and off-site inspections to monitor dust;
	 Carry out regular site inspections to monitor compliance with the AQMP;
	 Increase frequency of site inspections when activities with a high potential to produce dust are being carried out;
	 Record inspection results in an inspection log; and
	 Make an inspection log available to the Local Authority upon request.
Preparing and maintaining the worksites	 Plan site layout so that machinery and dust causing activities are located away from receptors, as far as is possible;
	 Erect suitable solid screens or barriers around dusty activities or the site boundary;
	 Fully enclose site or specific operations where there is a high potential for dust production and the site is active for an extensive period;
	 Install green walls, screens or other green infrastructure where appropriate to minimise the impact of dust and pollution;
	 Avoid site runoff of water or mud;
	 keep site fencing, barriers and scaffolding clean using wet methods;
	 Use water as dust suppressant where applicable;
	 Remove waste materials that have a potential to produce dust from site as soon as practicable;
	Cover, seed or fence stockpiles to prevent wind whipping;
	 carry out regular dust soiling checks of buildings within 100 m of site boundary and cleaning to be provided if necessary; and

Code of Construction Practice

Issue	Control measure
	 provide showers and ensure a change of shoes and clothes are required before going off-site to reduce transport of dust.
Operating Vehicle/ Machinery and Sustainable Travel	 Well maintained/low emission vehicles and equipment fitted with catalysts, diesel particulate filters or similar devices; All vehicles to switch off engines - no idling vehicles; Avoid the use of diesel or petrol powered generators where practicable; and Impose a maximum-speed-limit of 25kph on surfaced and 15kph on un-surfaced haul roads and work areas.
Operations	 Cutting equipment to use water as dust suppressant or suitable local extract ventilation; Ensure an adequate water supply on the site for effective dust/particulate matter suppression, using recycled water where possible and appropriate; Use enclosed chutes and covered skips; Minimise drop heights; and Ensure equipment is readily available on site to clean any spillages.
Waste Management	 Reuse and recycle waste to reduce dust from waste materials; and No bonfires and burning of waste materials.
Demolition	 Soft strip inside buildings before demolition (retaining walls and windows in the rest of the building where possible, to provide a screen against dust); Ensure water suppression is used during demolition operations; Avoid explosive blasting, using appropriate manual or mechanical alternatives; and Bag and remove any biological debris or damp down such material before demolition.
Earthworks and Construction	 Re-vegetate earthworks and exposed areas; Use hessian, mulches or trackifiers where it is not possible to re-vegetate; Only remove the cover in small areas during work and not all at once; Avoid removing a thin layer of concrete from structures by compressed air powered machines;

Code of Construction Practice

Document Reference: 6.10

Issue	Control measure
	 Ensure sand and other aggregates are stored and not able to dry out; and
	 Ensure bulk cement and other fine power materials are delivered and stored to prevent escape.
Trackout	 Use water-assisted dust sweepers on the access and local roads;
	 Avoid dry sweeping of large areas;
	 Ensure vehicles entering and leaving worksites are covered to prevent escape of materials;
	 Inspect on-site routes for integrity, instigate necessary repairs and record in site log book;
	 install hard surfaced haul routes on site, which are regularly damped down with fixed or mobile sprinkler systems or mobile water bowsers, and regularly cleaned;
	 Implement a wheel washing system at a suitable location near site exit;
	 ensure there is an adequate area of hard surfaced road between the wheel wash facility and the site exit, wherever site size and layout permits;
	Access gates at least 10m from receptors where possible;
	• Apply dust suppressants to locations where a large volume of vehicles enter and exit the construction site.

5.4 Monitoring

- 5.4.1 Automatic continuous PM₁₀ monitoring will be required as part of the AQMP. It may be appropriate to determine the existing (baseline) pollution levels before work begins. Baseline monitoring will commence three months before main construction works begin. The AQMP will set out appropriate PM₁₀ monitoring procedures and timescales.
- 5.4.2 Automatic continuous dust monitors will be installed across the site and checked regularly. The locations of the dust monitors will be included in the AQMP and agreed with the Local Authority. Baseline monitoring will commence three months before the construction phase begins. The AQMP will set out appropriate dust monitoring procedures and timescales.

5.5 Odour

- 5.5.1 As contaminated materials may be excavated at Greenwich and Silvertown, the excavated materials could contain volatiles that may have a bad smell. The following mitigation measures will be included in the AQMP and implemented by the Contractorto ensure that there is no significant effect to local residents:
 - Contaminated and non-contaminated materials will be stockpiled separately following excavation;
 - Early identification of contaminated material which could generate an odour issue;
 - Covering up of any odorous materials;
 - Locating contaminated materials as far away from residential receptors as possible; and
 - Ensuring odorous material is prioritised for removal from the worksites.

6. CULTURAL HERITAGE

- 6.1.1 The following measures in relation to Cultural Heritage will form part of Archaeological Written Scheme of Investigations (AWSI) to be prepared by the Contractor in consultation with Historic England ..
- 6.1.2 The AWSIs will provide the framework through which archaeological mitigation will be managed and monitored. The AWSI will be prepared for each work site to be approved by the relevant planning authority (in consultation with Historic England and GLASS) prior to commencing construction. Where mitigations measures within the river Thames are identified in the AWSI, the Contractor shall consult the MMO and PLA before submitting to the relevant planning authority for approval.
- 6.1.3 The Archaeological AWSIs will detail a programme of archaeological evaluation as part of the advance works. The programme of archaeological works will include adequate archaeological evaluation in order to properly characterise and quantify further archaeological mitigation works. The Contractor will consult with Historic England / GLAAS on the scope and methodologies for the programme of archaeological mitigation works prior to construction works commencing and ongoing consultations will be held during the archaeological works.
- 6.1.4 Detailed measures to protect any known or unknown archaeological assets and an outline method of approach during the construction of the Scheme will be incorporated into the AWSI and will include but not be limited to:
- updated baseline to ensure the most up-to-date Archaeological Priority Areas information is taken into account;
- rationale of locations of various interventions based on the known assets that may be affected;
- requirement to consider the significance of unknown and known heritage assets,
- monitoring measures of any in-river works;
- contingency measures, including measures in the event that unexpected, highly significant remains are encountered; and

- method for preservation and the approach to post-construction phase assessment, analysis, local knowledge sharing and public dissemination of the results of the programme of archaeological work.
- 6.1.5 If any heritage assets are found, the Contractor must liaise with Historic England and GLASS to determine the need for any consents, or licences and whether a Heritage Management Plan (HMP) is required.
- 6.1.6 The HMP, if required, will set out how the Contractor will protect the heritage assets that have been identified in a consistent and integrated manner during the works. It will include general standards of good practice across the project and specific measures, in relation to individual worksites.
- 6.1.7 Impacts of dredging within the area of the proposed jetty on currently unknown archaeological remains will be mitigated through the monitoring of dredged material in order to identify and record any archaeological materials that are recovered. If any remains are recorded, the Contractor will liaise with Historic England/ GLASS as to any scour protection measures required.
- 6.1.8 Prior to construction, an assessment of the likely effects of settlement on the Grade II listed Blackwall Tunnel building will be undertaken in accordance with Chapter 10: (Settlement) of this Code of Construction Practice.

7. TERRESTRIAL ECOLOGY

7.1 General

- 7.1.1 An Outline Ecology Management Plan (OEMP) has been prepared and can be found at Appendix G to this Code of Construction Practice.
- 7.1.2 Detailed Ecology Management Plans will be developed by the Contractor to be in substantial accordance with this OEMP, but taking into account the detailed design and local issues at each worksite and consultation with Natural England. These detailed Ecology Management Plans will be submitted for approval to the relevant planning authority for both the north and south sides of the Scheme. As can be seen in the OEMP, the detailed Ecology Management Plans will specify measures to manage the risk of adversely affecting flora and fauna on and within the vicinity of the worksites.
- 7.1.3 This CoCP includes dust attenuation measures to prevent pollution, as described in detail in Chapter 5 (Air Quality), and pollution prevention measures following Environment Agency Guidelines, as described in detail in Chapter 9 (Geology, Soils and Hydrogeology).

7.2 Monitoring

7.2.1 The detailed EMPs will state the programme for ecological monitoring during the construction of the Scheme. The monitoring programme should be in substantial accordance with the OEMP.

8. MARINE ECOLOGY

- 8.1.1 The Contractor will employ the following measures to minimise any adverse effects from the construction and demolition of a temporary jetty and any inriver construction activities:
 - the application of EA pollution prevention guidance throughout the construction phase;
 - the development of a non-native species risk assessment and management plan (in liaison with the MMO, PLA and EA as necessary);
 - the use of soft start procedures during piling for a minimum of 20 minutes. Should piling cease for a period greater than 10 minutes the soft start procedure must be repeated;
 - percussive piling will be limited to November-March inclusive (unless otherwise agreed with the MMO, PLA and EA);
 - the lighting on the jetty will be designed to minimise light levels in the marine environment. The lighting on the jetty head would have the lamps facing out to the watercourse, to facilitate unimpeded loading and unloading operations. Reflectors, that avoid excessive light pollution to surrounding areas, will be used.
- 8.1.2 The decommissioning programme of the jetty will adhere to the same seasonal restrictions for piling if hydraulic vibratory methods are used.
- 8.1.3 Any planned (i.e. non-emergency) dredging work must avoid the period of June-August inclusive. Any dredging within the months of June-August inclusive shall only be undertaken with the approval of the Environment Agency in consultation with the PLA.

9. GEOLOGY, SOILS AND HYDROGEOLOGY

9.1 General

- 9.1.1 The Contractor will prepare a Construction Materials Management Plan CMMP) before construction commences which must be approved by the relevant planning authority. The CMMP will set out measures to ensure materials are handled and used in a way that prevents harm to human health and pollution of the environment.
- 9.1.2 Furthermore, the Contractor will prepare a Groundwater Monitoring and Verification Plan (GMVP) which must be approved by the Environment Agency. The GMVP will set out monitoring and reporting criteria during preconstruction, construction and post-construction.

9.2 Intrusive Groundworks

Contamination Protocol

- 9.2.1 For works on the Greenwich peninsula, the Contractor will follow measures outlined within the "The Greenwich Peninsula Environmental Method Statement (EMS)", which details area specific development / construction measures to manage mobilisation potential of existing contamination.
- 9.2.2 Specific pollution prevention measures will be identified and adopted for the removal of historical (gasworks) sub-structures (where they exist) and any retained contaminated soils and liquids that have the potential to become mobilised as a result of construction.
- 9.2.3 With the possibility that unknown hydrocarbon contamination within soils and groundwater may yet still be encountered beneath the Greenwich site during excavation works, the Contractor will employ a watching brief to be maintained to cover the eventuality of unknown contamination. If visually contaminated or odorous material is encountered during those works, the assistance of a suitably qualified and experienced person (a geo-environmental engineer) would be sought.

9.3 Geology and soils

Construction Materials Management Plan (CMMP)

9.3.1 To limit any potential adverse impacts upon geology, soils, and hydrogeology, the Contractor will prepare a CMMP setting out measures to ensure excavated materials are handled and used in a way that prevents harm to human health and pollution of the environment. The CMMP must be approved by the relevant planning authority prior to commencing the relevant part of the authorised development.

- 9.3.2 The CMMPmust comply with the Good Practice Guide for Handling Soils. Any contaminated soils encountered during excavation works will be screened, treated if necessary and either re-used on site or removed from site.
- 9.3.3 In the event that on-site treatment of excavated material is required, the regulator would be consulted and any requisite licences (e.g. Mobile Treatment Permit) sought.

9.4 Groundwater

Detailed Measures

- 9.4.1 When developing the final construction methodology, the Contractor will consider suitable control methods to manage groundwater ingress and dewatering. The Contractor will liaise with the Environment Agency to determine the need for detailed assessment of dewatering impacts.
- 9.4.2 The Contractor will ensure that good practice is employed to establish ground and groundwater conditions, including verifying the presence of geotechnical hazards such as scour features in the vicinity of the Scheme alignment, and that the most appropriate groundwater exclusion or management method is adopted to minimise risks.
- 9.4.3 In particular, the removal of historical piled foundations can be disruptive to ground in immediate contact with the pile shafts. The Contractor will develop a method for pile removal which will prevent potential groundwater flow between aquifers.
- 9.4.4 Some activities with the potential to affect watercourses or groundwater may require either consent or authorisation under the Environmental Permitting (England and Wales) Regulations 2010, and the Contractor will obtain these consents prior to commencing these activities.

- 9.4.5 The Contractor will control the abstraction of potentially contaminated water from excavations through the adoption of mitigation measures as outlined in the Environment Agency's PPG5⁶.
- 9.4.6 To prevent additional land and groundwater contamination the Contractor will adhere to the EA Pollution Prevention Guidelines, including (but not be limited to) the following pollution prevention measures:
 - Silt pollution potentially produced when working in surface water will be minimised by keeping water out of work areas, using appropriate isolation techniques or through operation of special excavation plant. Any water generated from such activities will be disposed of by following guidance set out in the Environment Agency's document PPG 5 – Works and maintenance in or near water;
 - Surplus waste slurry / water produced during tunnel construction will likely require filtration prior to its disposal, because of the anticipated quality. Where necessary, a permit and licence will be obtained;
 - Polymers used for the TBM operation will be bio-degradable and nonhazardous to the water environment. Agreement with the Environment Agency will be sought prior to their use and any licences obtained (where necessary). Discharges which contain polymers will be tested to show that they are bio-degradable and low risk to the water environment;
 - Provision of wheel washing facilities and defined clean down areas for vehicles and equipment;
 - Regular cleaning of site access points;
 - Defined areas for loading / unloading of plant and materials;
 - Defined areas for the storage of plant and materials used during construction;
 - Production of and adherence to an emergency spill response plan (as part of the Emergency Plan developed in consultation with the

⁶ Pollution Prevention Guidance Notes, Works and maintenance in or near water, EA, 2014.

Environment Agency), and ready availability of associated equipment and materials;

- Groundwater and dewatering control measures (e.g. during concrete pouring, where necessary);
- Sediment control measures (compliant with the Site Waste Management Plan); and
- Methods for the removal and reinstatement of obstructions (e.g. piles).
- 9.4.7 The Contractor will apply for any permits and licences that may be needed for excavation and dewatering purposes. These are outlined below:
 - Environmental Permit a Discharge Licence (e.g. for discharge of excavation water into a watercourse, should waste water meet the Environmental Quality Standards (EQS) for discharge);
 - Environmental Permit a Mobile Treatment Plant Licence (e.g. for treatment of contaminated soils prior to their re-use on site)
- 9.4.8 The DCO will dis-apply the need for an abstraction (dewatering) licence. Abstraction will be controlled through the measures set out in a Groundwater Monitoring and Verification Plan to be prepared by the Contractor and approved by the Environment Agency.

Groundwater Monitoring and Verification Plan (GMVP)

- 9.4.9 The GMVP will contain the following measures for the monitoring of baseline, construction and post construction effects:
 - Pre-construction baseline monitoring will commence as soon as practicable and continue until the commencement of construction or the implementation of the construction phase of the GMVP. The preconstruction monitoring will be used to establish a baseline which will inform the setting of alert and trigger levels, for both water quality and groundwater elevations, against which the construction phase monitoring will be compared.
 - Monitoring will be undertaken throughout the construction and post construction of the relevant part of the Scheme, and reported to the Environment Agency. These reporting requirements will be outlined in the Groundwater Monitoring and Verification Plan.

- 9.4.10 Any changes to the GMVP proposed by the appointed Contractor must be approved by the Environment Agency before being adopted.
- 9.4.11 The GMVP will be developed with regard to the Groundwater Monitoring Strategy (Appendix F).

9.5 Human Health

- 9.5.1 Health and safety risks to construction workers from contamination will initially be controlled by the Contractor's responsibility to design out risk, as per the requirements of the Construction Design and Management (CDM) Regulations⁷.
- 9.5.2 The Contractor will be required to produce health and safety risk assessments that specify appropriate precautionary measures during works. These shall be completed by a suitably qualified person appointed by the Contractor.
- 9.5.3 The Contractor will employ appropriate health and safety measures which will be incorporated within the general construction site safety standards at a level sufficient to protect both members of the public and site workers.
- 9.5.4 The Contractor will also employ specific measures to manage the risks of asphyxiation caused by the potential release of deoxygenated air where excavation is required within the Thanet Sand Formation.

9.6 Unexploded Ordnance

- 9.6.1 Linear UXO surveys of the Scheme have previously been undertaken to identify the risk within the Order Limits. The Contractor will carry out further assessment prior to undertaking any intrusive investigation works on the Scheme within areas identified as medium and high risk.
- 9.6.2 Once the assessment is complete, the Contactor will develop a detailed UXO mitigation strategy for the project prior to construction commencing.

⁷ Construction, Design and Management Regulation, 2015, HSE

10. GROUND SETTLEMENT

10.1 General

- 10.1.1 The construction of the Silvertown Tunnel, comprising bored tunnels, cutand-cover sections and retained cuttings, will lead to settlement at the ground surface. The amount of settlement will depend on a number of factors including the depth and volume of the works below ground, soil conditions and the presence and nature of building foundations. The amount of settlement will vary across the area affected and for some buildings the magnitude of settlement may vary across the building; this is known as differential settlement.
- 10.1.2 The Contractor will design and undertake construction of the Scheme in a manner that will avoid or minimise the damage to land and property as a result of settlement. TfL has carried out an initial assessment of potential settlement attributable to the Scheme that assessed the risk of damage to all buildings and structures potentially affected by settlement.
- 10.1.3 The Contractor shall investigate the potential for ground movement (including settlement) associated with the construction methods. This investigation will be carried out in accordance with the Settlement Assessment and Mitigation Process appended to the CoCP (Appendix A).
- 10.1.4 Depending on the level of damage risk identified by the investigation, either no action will be required, or buildings and structures will be monitored during construction, or measures will be implemented to protect the buildings and structures.

10.2 Surveys

10.2.1 Defect surveys will be undertaken on all assets (including buildings, structures, bridges, tunnels and service media) assessed to experience 1mm or more settlement in the assessment carried out in accordance with Appendix A. These will capture the condition of those assets immediately prior to tunnel construction commencing in an area. A defects survey will comprise a written and photographic record of existing cracking and the state of the finishes and structures. They will be carried out by a reputable firm of chartered building surveyors or chartered engineers commissioned by the Contractor on behalf of TfL but in joint names with the asset owner and any other persons as TfL may determine. Asset owners are free to commission their own survey but this will be at their own cost since the

survey undertaken by the Contractor is an objective survey/record of preexisting defects and is not intended to draw any conclusions as to the cause.

- 10.2.2 An electronic copy of the report will be available to asset owners on request.
- 10.2.3 Following the construction of the Scheme in the vicinity of the asset, a second survey will then be undertaken by the appointed professional to record changes from the first survey. The owner may request his own surveyor to attend when the second survey is undertaken and to comment on the draft survey report produced.

10.3 Repairs

- 10.3.1 The Contractor will reimburse asset owners for the reasonable costs they incur in remedying material physical damage arising from ground movement (including settlement) caused by the Scheme, provided:
 - the damage is caused by the Scheme;
 - the owner gives not less than 28 days' notice in writing to the Contractor of the proposal to carry out the repair work;
 - the owner takes reasonable steps to obtain 3 competitive quotes for the repairs beforehand where required by the Contractor; and
 - any claim is made within two years of the opening of the Silvertown Tunnel.
- 10.3.2 The Contractor may, on receiving the advanced notice of the proposal to carry out the repair work, elect to undertake the repair work itself.
- 10.3.3 If there are any pre-existing defects which have worsened as a result of the Scheme then the recoverable loss will be limited to the additional cost of repair over and above that which would have been required to deal with existing defects.
- 10.3.4 If it can be demonstrated that the undertaking to assess the compensation claim based on the reasonable cost of repairs does not compensate the claimant fully for the reduction in value of his interest in the property the reimbursement of repair costs will not prejudice a further claim for compensation in accordance with the national compensation code within the normal limitation period applying to such claims.

10.4 Settlement Deed

- 10.4.1 A Settlement Deed will be offered to asset owners in a standard form, setting out the procedures to determine the need for the monitoring of buildings and other structures and, if necessary, the carrying out of protective and/or remedial works. This is a formal legal undertaking concerning ground movement (including settlement), giving effect to the matters set out in this CoCP.
- 10.4.2 The Deed, the terms of which are subject to change during the DCO examination, is attached as Appendix B. Subject to paragraphs 10.4.3 and 10.4.4 below, the Contractor will enter into a deed in substantially the form of the final version of those terms.
- 10.4.3 To enter into a Settlement Deed the Owner must meet the legal definition of 'owner' in the Acquisition of Land Act 1981 which is as follows:

"a person, other than a mortgagee not in possession, who is for the time being entitled to dispose of the fee simple of the land, whether in possession or reversion, and includes also a person holding or entitled to the rents and profits of the land under a lease or agreement, the unexpired term of which exceeds three years [and a person who would have power to sell and convey or release the land to the acquiring authority if a compulsory purchase order were operative]."

- 10.4.4 The requirement for the Contractor to enter into the Deed does not apply to new buildings that receive planning permission after the date on which the Silvertown Tunnel Order comes into force.
- 10.4.5 It is not necessary to enter into the deed in order to benefit from the process set out in this Chapter.

11. NOISE AND VIBRATION

11.1 General

11.1.1 The Contractor will control and limit noise and vibration levels to minimise any disturbance to the environment and communities in the vicinity of the construction works.

11.2 Detailed measures

- 11.2.1 Pre-construction noise monitoring surveys will be undertaken and agreed with the relevant planning authority to establish a pre-construction baseline for monitoring compliance with construction noise limits. Baseline monitoring will commence three months beforeconstruction works begin.
- 11.2.2 The Contractor will then carry out an updated construction noise assessment aganst this baseline based on the detailed design and construction methodology. This will be carried out in accordance with BS 5228 using specific manufacturer's data and position of equipment. This assessment will inform a Noise and Vibration Management Plan (NVMP), which will be prepared by the Contractor for each work site, setting out the proposed specific mitigation measures to be approved by the relevant planning authority prior to commencing construction. This assessment will also inform the eligibility of any persons eligible for the Construction Noise and Vibration Mitigation Scheme which the Contractor will deliver, set out at Appendix H of this Code of Construction Practice.

11.2.3

11.2.4 Where appropriate the Contractor will obtain consents from the relevant local authority under Section 61 of the Control of Pollution Act 1974 (which will include noise and vibration limits where relevant) for the proposed construction works. Any Section 61 consent that is obtained may contain site specific management and mitigation requirements for noise and vibration. The details of this consent will be included in the NVMP. The Section 61 process may not be appropriate in the case of some types of tunnelling operations.

- 11.2.5 Best Practicable Means⁸ (BPM) as defined under Section 72 of the Control of Pollution Act 1974 will be employed during the construction phase and included in the NVMP. These will include, but not be limited to:
 - installing appropriate fencing around the construction areas likely to generate noise;
 - providing contact details for a site representative in the event that disturbance due to noise or vibration from the construction works occurs; ensuring that any complaints are dealt with pro-actively and that subsequent resolutions are communicated to the complainant;
 - keeping site access routes in good condition and well maintained with no potholes or other significant surface irregularities;
 - turning off plant machinery when not in use;
 - maintaining all vehicles and mobile plant such that loose body fittings or exhausts do not rattle or vibrate;
 - using silenced equipment where possible, in particular silenced power generators and pumps;
 - using the most modern equipment available where possible and maintaining and operating equipment properly by trained staff;
 - locating static noisy plant, including generators, as far away from noise sensitive receptors as is feasible for the particular activity;
 - speed limits of to reduce the effect of construction traffic noise;
 - monthly condition assessments on site to inspect for defects such as pot holes which could cause an increase in noise levels. Indentations of greater than 20mm to be repaired when identified. Existing potholes would need to be considered by a condition assessments prior to the commencement of works;
 - ensuring that the quietest plant and equipment, techniques and working practices available are selected and used; and

⁸ Control of Pollution Act1974.

- no music or radios would be played on site.
- 11.2.6 The NVMP will require that noise and vibration from conveyor systems will be minimised through the implementation of a maintenance programme which includes regular inspection of the conveyor equipment.
- 11.2.7 The Contractor will be responsible for notifying the local residents of particularly noisy work prior to commencement. The mechanisms for notification will be detailed in the Community Engagement Plan. Effective communication should be established, keeping local residents informed of the type and timing of works involved, paying particular attention to potential evening and night time works and activities which may occur in close proximity to receptors.
- 11.2.8 During the construction phase, day time and night time noise and vibration monitoring will be undertaken at key sensitive receptors to ensure that the mitigation measures suggested are working effectively. The location and duration of the monitoring will be set out in the NVMP.

12. TOWNSCAPE AND VISUAL AMENITY

12.1 Visual disruption

- 12.1.1 The Contractor will employ construction good practices to minimise townscape and visual disruption, for example protection of existing vegetation and targeted use of hoarding to screen construction worksites.
- 12.1.2 The visual intrusion of construction worksites on nearby residents and users of local facilities will be contained and limited. Signage, decoration or enhancement, for information or aesthetic purposes, on the hoarding will be in accordance with TfL's corporate requirements.

12.2 Lighting

- 12.2.1 For each worksite a Lighting Management Plan (LMP) will be prepared by the Contractor in consultation with the relevant planning authority, the Environment Agency and the PLA. The Contractor will implement the measures set out in the LMP.
- 12.2.2 The LMP will reflect appropriate industry standard procedures which will be implemented at both worksites. Lighting will be designed, positioned and directed so as not to unnecessarily intrude on adjacent buildings and so as to prevent unnecessary interference with local residents, the DLR, passing motorists, the navigation lights for air or water traffic, wildlife breeding seasons and adjacent habitats that may be used by foraging and commuting bats.
- 12.2.3 Site specific lighting measures to minimise the adverse impacts on adjacent buildings, wildlife sites and land uses will be taken from and applied in accordance with the 'Guidance Notes for the Reduction of Obtrusive Light GN01:2011' (published by the Institution of Lighting Professionals).
- 12.2.4 The Contractor will use cowling, reflectors and other measures on the temporary jetty to avoid excessive light pollution to surrounding areas. These measures will be detailed within the LMP.

13. MATERIAL RESOURCES AND WASTE

13.1 General

- 13.1.1 The Contractor will manage Construction Demolition and Excavation (CD&E) materials generated at worksites in accordance with the waste hierarchy to prevent, reduce, re-use, recycle, recover, and dispose of materials and within the relevant regulatory controls⁹. Measures would be implemented to reduce the impacts of material resources use and waste arisings from the Scheme.
- 13.1.2 A Construction Materials Management Plan (CMMP) will be prepared by the Contractor and approved by the local authorities. The CMMP will include appropriate mitigation measures to ensure that materials are handled and used in a way that prevents harm to human health and pollution of the environment.

13.2 Materials management

- 13.2.1 Measures to manage materials as part of the CMMP will include but are not limited to:
 - Materials delivered to the project will be received and controlled by the Contractor's Logistics Team or appointed person.
 - Materials will be stored to minimise the potential of damage or wastage. Measures will include off-ground storage e.g. on pallets, remaining in original packaging, protection from rain or collision by plant or vehicles.
 - The materials storage area will be secured during out of hours to prevent unauthorised access.
 - The Contractor will be encouraged to apply good practice to source construction materials from suppliers with responsible sourcing certification (as far as practicable).
 - Local sources for aggregate supplies will be considered whenever possible.

⁹ Mayor of London (2015). London Plan.
- Materials will be ordered, where possible, in sizes to prevent wastage e.g. in the form of off cuts and waste to be able to be returned to the original supplier e.g. plastic pipe.
- The procurement process shall ensure that materials are ordered so that the timing of the delivery (e.g. 'just in time' deliveries), the quantity delivered and the storage are optimised to reduce opportunity for oversupply and damage on site.
- Wherever possible, standardisation of materials and building elements will be incorporated into the Scheme design in order to minimise required material resources and the production of waste, e.g. the use of prefabricated components.
- Where possible, consideration will be given to the reuse of material (e.g. uncontaminated soils) back into the project.
- 13.2.2 The Contractor must deal with CD&E Materials in accordance with the CD&E Materials Commitments that have been agreed by TfL and are set out at Appendix C.
- 13.2.3 The Contractor will also, as part of measures to be included in the CMMP:
 - where specification allows, utilise at least a 10% portion of construction materials to include reused and recycled content;
 - minimise the use of primary aggregated by the selection of secondary materials, where possible;
 - obtain all timber products from sustainable sources. All timber procured will be obtained from recycled, reclaimed sources or be accredited to meet sustainable forestry standard such as the Forestry Stewardship Council (FSC). Any remaining timber not sourced through the above will target a known temperate source using the Department for Environmental, Food and Rural Affairs (Defra) central point of expertise in timber (CPET);
 - use low embodied carbon materials; and
 - achieve a score of Very Good and ideally Excellent using CEEQUAL, adherence to materials and waste elements.

13.3 Waste

- 13.3.1 A Site Waste Management Plan (Appendix D) has been developed for the Scheme and will be refined and updated by the Contractor as the design and the Scheme progresses. The Contractor will manage waste in accordance with the SWMP. This includes the management of material dredged from the River Thames to facilitate the construction and operation of the temporary jetty and NAABSA at the Silvertown site (if required). All transport of waste will be in accordance with relevant waste legislation.
- 13.3.2 The SWMP will also provide a framework for checking compliance with waste legislation and the Duty of Care¹⁰.
- 13.3.3 Waste on site will be managed through the implementation of the CMMP and SWMP. Waste management measures include but are not limited to:
 - The Contractor will have a Waste Manager or Champion who will oversee the implementation of the waste control strategy and the handling of any waste material, as set out in the SWMP.
 - The Contractor will consider setting off-cut/surplus targets for subcontractors with a positive incentive scheme for on-site waste champions.
 - A waste management compound will be established within the Silvertown site to handle incoming waste from construction activities. This will be designed to facilitate the segregation of key waste streams to maximise the opportunity to re-use, recycle and return wastes generated on site.
 - Excavated materials, such as soils, will be carefully stored in segregated piles for subsequent reuse on the site, where possible. If the material is contaminated then it will be kept separate from clean material and sent for either treatment, recycling or recovery, where appropriate, or disposal at appropriately permitted facilities.

¹⁰ Waste Framework Directive, Directive 2008/98/EC European Waste Framework Directive [2008] OJ L 312/3. Environmental Permitting (England and Wales) Regulations 2010. SI 675. EU Landfill Directive, Council Directive 1999/31/EC of 26 April 1999 on the landfill of waste [1999] OJ L 182/1. The Waste (England and Wales) Regulations 2011.

- The Contractor will be required to divert all vegetation waste from landfill, unless identified as an invasive species and no other options are available. The greatest opportunity for the sustainable management of vegetation waste is through recycling into compost.
- A specific area will be laid out and labelled within the waste management compound to facilitate the separation of materials, where possible, for potential recycling, salvage, reuse and return. Recycling and waste bins/skips are to be kept clean and will be clearly marked/colour coded in order to avoid contamination of materials.
- Shelter will be provided to prevent materials such as cardboard and paper from deteriorating while being sorted or awaiting collection.
 Space will be provided to accommodate skips and the storage of reusable materials.
- For all waste management options on the site compound, consideration will need to be given for identifying whether waste exemptions or permits are required to enable for the storage and treatment of waste materials.
- Waste management options will be supported by the identification of appropriately permitted waste management and recycling facilities in close proximity to the site compound as set out within the SWMP.
- 13.3.4 A Receptor Site Assessment (RSA) (Appendix E) has been developed to provide a transparent process and methodology for the evaluation of worksites that may receive excavated material, including material dredged from the River Thames to facilitate the construction and operation of the temporary jetty and NAABSA at the Silvertown site (if required). The final output from the RSA will be a preferred list of receptor sites as well as a reserve list and their scores against each of the assessment criteria. The assessment criteria consider environmental impacts, the operation of the facilities, the proximity principle, and the impact on the local area.
- 13.3.5 The Contractor will be required to select receptor sites from the preferred list or reserve list of sites or to follow the RSA methodology to identify alternative sites.

13.4 Asbestos and health risks

13.4.1 Some of the materials generated by excavation activities will be contaminated or hazardous.

- 13.4.2 The Contractor will comply with all relevant legislation relating to protection of employees and others who may be affected from health risks within working environments, including COSHH regulations.
- 13.4.3 The risk from release of asbestos during alteration, demolition and excavation works will be managed in accordance with The Control of Asbestos Regulations 2012 and associated codes of practice and guidance. Measures for managing asbestos in excavation works will include but are not limited to:
 - employing competent contractors to carry out the works;
 - contractors implementing a procedure for dealing with potentially suspect materials exposed requiring sampling and analysis by an independent specialist consultant;
 - formal exchange of information before start of work, including relevant information from the Asbestos Register to clearly identify location of asbestos-containing materials; and
 - method statements for any works in the vicinity of asbestos-containing materials to avoid any disturbance to such materials.

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

14. EFFECTS ON ALL TRAVELLERS

14.1 Pedestrians and cyclists

- 14.1.1 To minimise any adverse effects on pedestrian and/or cyclists during construction the following mitigation measures will be implemented as part of the Construction Traffic Management Plan for each worksite to be prepared by the Contractor, and agreed with the relevant planning authority, in consultation with the local highways authority, in accordance with Chapter 3:
 - All existing public rights of way and provision for cyclists and pedestrians that are affected by the construction works will be clearly signed, fenced and diversion routes provided from the outset to ensure that their usage would not be unduly impaired by construction activities. Envisaged diversion routes are provided in the Transport Assessment (Document Reference: 6.5). The Scheme phasing would accommodate all revisions to the alignment of these public rights of way and any changes would occur progressively during the construction period.
 - The alternative routes for pedestrian and cyclists will be kept to a minimum feasible length. Alternative safe routes will be provided for people with reduced mobility to ensure that they can access facilities in a safe manner. On completion of the works the pedestrian and cycle routes will be reinstated to their original or revised alignments.
 - The existing footbridge over the A102 Blackwall Tunnel Approach at Boord Street will be demolished and replaced with a new foot and cycle bridge. Ramped access to the current footbridge or to the replacement foot and cycle bridge will be maintained during all stages of construction.

14.2 Active traffic demand management

14.2.1 Active traffic demand management during the works will be coordinated and controlled by TfL to ensure consistent and accurate information for the network users.

14.3 Additional mitigation

14.3.1 Table 14-1 summarises the additional mitigation measures which are envisaged to be likely to be required to be implemented at specific roads/junctions. The necessity of these measures will be determined by the Contractor following consultation with the relevant highway authorities as part of the development of the Construction Traffic Management Plan. Where these measures have been determined to be necessary, the Contractor will implement them, subject to the approval of the relevant highway authority where they are outside of the Order Limits.

Road/Junction Mitigation measures Lower Lea Crossing Improve lighting and increase effective width between Leamouth Road of footways by cutting back overgrown and Tidal Basin Road vegetation. Provide dropped kerbs at all crossing locations. Provide temporary pedestrian and cycle Temporary realignment of Millennium Way crossing facilities. Provide a shared use path or segregated cycle track. Temporary link road Provide temporary pedestrian and cycle through the worksite crossing facilities. connecting Millennium Way to West Parkside to the south of the current alignment of Edmund Halley Way. Blackwall Lane/A102 At pedestrian/cycle crossings, provide wider junction waiting areas and remove pinch points where feasible. Repaint road markings, repair tactile paving and remove rutting in the carriageway. Remove clutter in footways where feasible. Cut back overgrown vegetation. Improve footway drainage and improve lighting. Millennium Way/John Repaint road markings and improve footway Harrison Way/Bugsby's drainage. Review signal timings for Way/Blackwall Lane pedestrians and cyclists. Junction Tidal Basin Roundabout Improve lighting. Repair tactile paving. Provide safer pedestrian and cycle routes.

Table 14-1 Mitigation measures at roads/junctions

14.3.2 The Contractor will determine whether it is necessary to review signal timings for pedestrians and cyclists at the Blackwall Lane/A102 junction following consultation with the relevant highway authorities as part of the development of the Construction Traffic Management Plan. If this measure has been determined to be necessary, the Contractor must procure that TfL implements such a change to the signal timings.

15. WATER ENVIRONMENT AND FLOOD RISK

15.1 General

- 15.1.1 The Contractor shall implement measures to protect surface water from pollution as well as measures to conserve water and manage flood risk during the construction of the Scheme.
- 15.1.2 The Contractor will employ the appropriate water conservation and pollution prevention measures and site drainage and in river works controls set out below.
- 15.1.3 The Contractor will employ measures to minimise spillage of materials associated with the Scheme being loaded onto, or unloaded from, barges or other vessels;

15.2 Site drainage

- 15.2.1 The Contractor will ensure that the site drainage meets the effluent standards required by the sewerage undertaker, or Environment Agency, as appropriate, and will provide holding or settling tanks, separators, and other measures as may be required. It will be the Contractor's responsibility to ensure that access is provided to the sewerage undertaker so that samples of discharge can be obtained and analysed and the flows verified as required. The relevant sections of BS 6031:2009 Code of Practice for Earthworks¹¹ for the general control of site drainage will be followed.
- 15.2.2 Drainage systems will be inspected regularly and maintained as necessary to ensure they operate to the appropriate standard. Inspection and maintenance will be required more often in areas with a high level of construction activity.
- 15.2.3 Where required, new drainage outfalls, storage and pollution control systems will be built as early in the construction sequence as is practicable.
- 15.2.4 Consideration will be given to protecting any existing drainage when storing fill materials, aggregates and plant to prevent potential drainage and pollution issues.

¹¹ BSI (2009). BS 6031:2009 Code of Practice for Earthworks

- 15.2.5 Temporary site drainage systems will be put in place to retain surface water runoff within the Order Limits, where practicable.
- 15.2.6 Any ingress of water into excavations will be pumped to a suitable settlement lagoon or tank and the clear water discharged into the drainage system in a condition suitable to meet the requirements of the Environment Agency/ Thames Water as applicable. As outlined in paragraph 9.3.6, a Mobile Treatment Plant Licence (e.g. for treatment of dewatering water) will be obtained, if required.

15.3 Control of pollution

- 15.3.1 The Contractor will undertake the works and implement working methods which will be developed to protect surface water from pollution and other adverse impacts including change to flow volume, water levels and quality. This will be completed in accordance with relevant legislative requirements and appropriate industry guidance and in liaison with the Environment Agency.
- 15.3.2 The Contractor will utilise good practice pollution prevention methods for activities such as excavation and dewatering, storage of fuels, chemicals and oils, vehicle washing.
- 15.3.3 All refuelling, oiling and greasing by the Contractor will take place above drip trays or on an impermeable surface which provides protection to underground strata and watercourses and away from drains as far as reasonably practicable. Vehicles will not be left unattended during refuelling.
- 15.3.4 Access to pollution control equipment and spillage clean up facilities will be provided at all worksites and the Contractor must take measures to prevent pollution caused by severe weather.
- 15.3.5 Measures to deal with major pollution incidents at the worksites will be included within the Emergency Plan to be produced in accordance with Chapter 2. The measures will reflect the Environment Agency guidance on pollution incident response planning.
- 15.3.6 The Contactor will carry out an assessment of the mobilisation of off-site contamination, and if necessary, mitigation measures will be implemented to deliver protection of controlled waters.

15.4 In-river works

- 15.4.1 If necessary, a temporary jetty may be constructed to permit the operation of the proposed marine transportation system.
- 15.4.2 In order to mitigate against the potential for increases in turbidity and scour during construction of the jetty, works will be undertaken in accordance with good practice methods for pollution control. Other mitigation measures may include the use of drill water recycling and capture during piling and the deployment of silt curtains, if these are considered necessary at the detailed design stage.
- 15.4.3 Measures to minimise effects on water quality from dredging associated with the temporary jetty will include but are not limited to:
 - the management of spill water decanting from the excavated material transport barge to minimise locally high concentrations of suspended sediment and changes in water quality;
 - the minimisation of the loss of material from the back-hoe by optimising angle of dredge to the prevailing tidal current flow;
 - avoiding summer/low river flow periods if possible; and
 - limiting the areas of dredging to a minimal footprint to allow safe access to the jetty.

15.5 Water conservation

- 15.5.1 The Contractor will implement working methods that control water consumption and ensure water is used efficiently on the construction worksites to support water conservation wherever possible. The measures will include but will not be limited to:
 - water audits that identify all water-using processes, activities and equipment on site (aligned with significant changes in site(s) activities throughout the construction phases);
 - staff engagement and training, to reduce water consumption by all water-using processes, activities and equipment on site;
 - a monitoring regime that assesses the effectiveness of water conservation measures;

- stored water collected by the drainage systems would be used for dust suppression and for other construction phase tasks, such as operation of the TBM; and
- if required, pumps will be provided at each storage lagoon for use in filling water bowsers.
- 15.5.2 Measures to encourage water use efficiency in the worksite offices and canteens will be adopted.

15.6 Flood risk

- 15.6.1 The existing flood defences provide a high standard of flood protection from the River Thames. The construction of the Scheme is not anticipated to affect the integrity of existing river walls.
- 15.6.2 The draft DCO (Document Reference: 3.1) requires that any works carried out under the DCO within 16m of the banks of the River Thames or River Lea, or which might affect flood defences require the prior approval of the Environment Agency, which may be given subject to requirements for the protection of water resources and the prevention of flooding.
- 15.6.3 During the construction phase flood warning and emergency procedures will be in place, as part of the Emergency Plan. Construction site operatives would use the plan to assess the need to put evacuation and Scheme shutdown procedures into action, thereby mitigating the residual risk of flooding in the very unlikely scenario of a breach on the River Thames defences during the construction period.

16. CARBON EMISSIONS

- 16.1.1 The Contactor will take measures to reduce energy consumption and improve energy efficiency onsite during construction, which may include but are not limited to the following:
 - minimising the use of diesel or petrol powered generators and instead using mains electricity or battery powered equipment;
 - powering down of equipment/plant during periods of non-utilisation;
 - ensuring all vehicles and machinery is serviced at recommended intervals to guarantee optimum engine efficiencies and reduce waste energy;
 - using fuel-efficient plant, machinery and vehicles wherever possible;
 - implementing SMART targets for consumption during construction;
 - deploying correctly sized generators for electrical provision onsite, where applicable; and
 - providing appropriate levels of thermal insulation to the relevant areas of site accommodation to reduce energy demand for heating.

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

Appendix A Settlement Assessment and Mitigation Process

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

A.1 Settlement Assessment and Mitigation Process

- A.1.1 The Contractor shall investigate the potential for ground movement associated with the design and possible construction methods using the process defined in paragraph 1.2 for the following purposes:
 - to assess risk of building damage by identifying those zones where the implementation of the design is likely to cause ground movements which may result in Risk Category 2 'Slight' being exceeded (see Table 1);
 - to assess the degree of such damage risk occurring and consider alternative designs as necessary;
 - to undertake an assessment of the potential consequences where there is a significant likelihood that Risk Category 2 'Slight' will be exceeded and identify specifically what the risks are;
 - to design protective measures where necessary to mitigate against the risk of damage exceeding Risk Category 2;
 - to demonstrate that the environmental effects of excavation induced ground movements have been considered and taken account of in the design;
 - to assess the risk of damage to utilities and to design mitigation measures in agreement with the utility owner;
 - to assess the effects of excavation to existing above-ground and underground infrastructure and to design suitable mitigation measures;
 - to indicate where property may require demolition or structural modification; and
 - to enable the preparation of contingency plans to deal with residual risks.
- A.1.2 A phased assessment process shall be followed to identify the risk and impact of construction activities on third parties. Five phases have been identified to assess the impact and risk from pre-construction to completion. The initial phases are aimed at assessing risk, whilst the latter are about limiting, managing and recording the impact of construction. The 5 phases are:

Phase 1 – Scoping

A.1.3 Schedules and plans shall be prepared to identify all third party assets assessed to experience ground movement exceeding 1mm using conservative parameters.

Phase 2 – Initial Assessment

A.1.4 The Contractor shall carry out initial assessment calculations using simple, empirically calibrated methods and moderately conservative parameters to classify the risk of damage to assets. For building structures the risk should be classified in accordance with Table A-1. For non-building infrastructure the level of risk will be determined by ensuring that deformations do not exceed tolerable values determined in consultation with the asset owner. These calculations shall be based on record drawings, where available and an inspection for assessment. Assets estimated to be at risk of damage greater than Risk Category 2 (Slight) or where damage exceeds the agreed tolerable limits will require further detailed assessment at Phase 3. A schedule and plans of predicted damage shall be prepared for that phase.

Building damage classification				
R is k Category	DDescriptio n of degree of damage+	Description of typical and likely forms of repair for typical masonry buildings	Approx. crack width* (mm)	Max. tensile strain %
0	NNegligible	Hairline cracks		< 0.05
• 11	VVery slight	Fine cracks easily treated during normal redecoration. Perhaps isolated slight fracture in building. Cracks in exterior visible upon close inspection	00.1 to 1.0	00.05 to 0.075
• 22	S S light	Cracks easily filled. Redecoration probably required. Several slight fractures inside building. Exterior cracks visible; some repainting may be required for weathertightness. Doors and windows may stick slightly	• 11 to 5	00.075 to 0.15
• 33	MModerate	Cracks may require cutting out and patching. Recurrent cracks can be masked by suitable linings. Tuck pointing and possible replacement of a small amount of exterior brickwork may be required. Doors	5 5 to 15 or a number of cracks greater than 3	00.15 to 0.3

Table A-1 Building Damage Classification

			and windows sticking. Utility services may be interrupted. Weather tightness often impaired		
•	44	S S evere	Extensive repair involving removal and replacement of walls especially over door and windows required. Window and door frames distorted. Floor slopes noticeably. Walls lean or bulge noticeably. Some loss of bearing in beams. Utility services disrupted.	15 to 25 but also depends on number of cracks	> > 0.3
•	55	VVery Severe	Major repair required involving partial or complete reconstruction. Beams lose bearing, walls lean badly and required shoring. Windows broken by distortion. Danger of instability	UUsually > 25 but depends on number of cracks	
+ In assessing the degree of damage, account must be taken of its location in the building or					

* Crack width is only one aspect of damage and should not be used on its own as a direct measure of it.

structure.

Burland, J.P. and Wroth, C.P., Settlement of Buildings and Associated Damage, Proceedings of a Conference on the Settlement of Structures, Cambridge, 1974, pp 611 – 54 and 764 – 810;

- A.1.5 The heritage value of a Listed Building should be considered during the initial assessment by reviewing the sensitivity of the building structure and of any particular features together with the initial assessment calculations. The heritage assessment examines the following:
 - the sensitivity of the building / structure to ground movements and its ability to tolerate movement without significant distress. The potential for interaction with adjacent buildings / structures is also considered. A score within the range of 0- 2 should be allocated to the building/structure in accordance with the criteria set out in Table A-2; and
 - the sensitivity to movement of particular features within the building / structure and how they might respond to ground movements. A score within the range of 0-2 should be allocated to the building in accordance with the criteria set out in Table A-2.

A.1.6 The scores for each of the two categories (a) and (b) should be added to the category determined in paragraph 3.1 to inform the decision making process. In general, Listed Buildings which score a total of 3 or higher should be subject to further assessment as part of the Phase 3 – Detailed Assessment. Buildings that score a total of 2 or less are predicted to suffer a degree of damage which may be easily repairable using standard conservation based techniques and hence no protective measures for the building's particular features should be required. However, ultimately the professional judgement of engineering and historic building specialists should be used to determine whether additional analysis is required.

	Criteria		
Score	Sensitivity of the structure to ground movements and interaction with adjacent buildings	Sensitivity to movement of particular features within the building	
0	Masonry building with lime mortar not surrounded by other buildings. Uniform facades with no particular large openings.	No particular sensitive features	
• 11	Buildings of delicate structural form or buildings sandwiched between modern framed buildings which are much stiffer, perhaps with one or more significant openings.	Brittle finishes, e.g. tight-jointed masonry, which are susceptible to small movements and difficult to repair.	
• 22	Buildings which, by their structural form, will tend to concentrate all their movements in one location.	Finishes which if damaged will have a significant effect on the heritage of the building, e.g. cracks through frescos.	

Table A-2 Scoring for Sensitivity Assessment of Listed Buildings

Phase 3 - Detailed Assessment and Mitigation Design

- A.1.7 The Contractor shall carry out detailed assessments of structures assessed at being at Risk Category 3 or above so that any monitoring works and mitigation works can be designed and implemented.
- A.1.8 The detailed assessment should determine:
 - the influence of the structure and its foundations on the predicted ground movements;

- the volume loss at which the risk of damage to the structure (or any sensitive finishes/features) is 'slight' or better;
- whether this volume loss may be achieved by the proposed excavation design/control measures;
- any special control measures required to reduce the predicted damage to acceptable levels (i.e. Risk Category 2 or below) such as significantly higher face pressures with EPBM tunnelling and the practicality of these;
- the amount of ground movement that the structure (and or any sensitive finishes/features) can accommodate without exceeding Risk Category 2 or any other agreed damage level; and
- the level of residual risk if intrusive mitigation measures are not implemented.
- A.1.9 The detailed assessments should include a number of iterations to determine how the risk of damage to a building may be reduced. Asset-specific empirical models shall be prepared successively using moderately conservative and best estimate parameters. If after these iterations the use of empirical methods do not reduce the risk of building damage to acceptable levels (i.e. Risk Category 2 (slight) and below), the damage assessment shall be refined by increasing the sophistication of the analysis with the aim of reducing the risk of asset damage to acceptable levels and to eliminate the asset from further assessment.
- A.1.10 If the risk of damage cannot be shown to be reduced by detailed assessment to acceptable levels, then mitigation measures shall be designed.
- A.1.11 The primary means of settlement mitigation shall be practical measures to control ground movement by good design and construction practice. This could include staged excavation sequences within sprayed concrete lining (SCL) works, ground treatment, face stabilisation, spiling / face dowels, increasing face pressure when using an tunnel boring machine (TBM), adopting stiffer walls/propping for rectangular shafts etc.
- A.1.12 In the event that physical mitigation measures are still required (i.e. to control building damage to Risk Category 2 or lower or to meet the third party asset owner's requirements), the Contractor shall seek to obtain the owner's approval or may use the powers under the DCO to undertake protective works.

- A.1.13 The Contractor shall also undertake a comparative risk assessment to demonstrate that the risks associated with installation/implementation of any intrusive mitigation measures (such as compensation grouting) are no worse than the risks associated with the base case.
- A.1.14 The relevant Local Authority and Historic England shall be consulted on the results of the Listed Building assessment reports and the proposals for protective measures, if any are required. Historic England shall also be consulted in relation to Listed Buildings where they would normally be notified or consulted on planning applications or listed building consent applications.
- A.1.15 When considering the need and type of protective measures for Listed Buildings, due regard should be given to the sensitivity of the particular features of the building which are of architectural or historic interest and the sensitivity of the structure of the building to ground movement. Where the assessment highlights potential damage to the features of the building which it will be difficult or impossible to repair and/or if that damage will have a significant effect on its heritage value, the assessment may recommend appropriate measures to safeguard those features either in-situ or by temporary removal and storage off-site if those with relevant interest(s) in the building consent.
- A.1.16 The form of monitoring of Listed Buildings should be determined based on the results of the assessment process.
- A.1.17 Where repair works are necessary they will require the consent of those with relevant interest(s) in the building.
- A.1.18 For Docklands Light Railway (DLR) track and track support structures the Contractor shall:
 - review the track surveys (including specifying additional surveys if required) and establish that ground movement can be accommodated without exceeding track standard operational tolerance in conjunction with the DLR Infrastructure Manager;
 - identify locations where fettling of the track is required pre construction and /or during construction to ensure the track geometry and clearances are acceptable.
- A.1.19 The Contractor shall prepare plans and sections showing the zone of influence of the works that is defined by ground movements exceeding 1mm.

- A.1.20 The Contractor shall develop an instrumentation and monitoring plan to validate that ground movements within the zone of influence are in accordance with design assumptions and that the infrastructure remains within acceptable limits. The Contractor shall ensure that there is a clear distinction between parameters measured to confirm the change in any parameter is in accordance with the design and parameters measured to limit damage to the assets. This plan shall identify the minimum period of time required to obtain base line data for each monitoring point. Instrumentation adjacent to the railway, which will remain in place during traffic hours, shall conform to the standards of the asset owner.
- A.1.21 Note: A competent engineer responsible for the works shall consider those factors which may influence the monitoring data and shall determine an appropriate period and frequency for baseline monitoring. This decision making process will include an element of engineering judgement to account for the possible effects of any underlying environmental trends in the assets under consideration.
- A.1.22 The Contractor shall demonstrate that the monitoring system complies with the British Tunnelling Society Monitoring Underground Construction best practice guide.
- A.1.23 Note: A review of the monitoring system against the checklists provided in Appendix B of the BTS Monitoring Underground Construction best practice guide may be used as a tool to demonstrate compliance.
- A.1.24 The detailed assessments shall define the control limits that need to be imposed on the tunnel construction methods in the zone of influence. The Contractor shall state these control measures on drawings and specifications.
- A.1.25 The Contractor shall identify the critical parameters to be monitored and define the Asset Control Limits based on:
 - the ability of the asset or structure to withstand ground movement investigated during the assessments carried out in Phases 2 and 3.
 - the risk to third party operations.
- A.1.26 The Contractor shall link the Asset Control Limits to actions within the Emergency Preparedness Plan.
- A.1.27 The Instrumentation and Monitoring Plan and Emergency preparedness Plan shall be agreed with the relevant third party asset Owner.

Phase 4 – Construction

- A.1.28 Contingency plans shall be developed and agreed with third party asset Owners to cover the risks posed to third parties before commencement of the construction of the Scheme.
- A.1.29 Contingency plans shall be implemented where the results of monitoring or inspection so indicate.
- A.1.30 Ground movement and construction progress records shall be maintained and reported in regular reviews when construction processes are taking place within the zone of influence.
- A.1.31 Predictions and assumptions made during design in respect of both ground movement and the effects which such ground movement will have on adjacent assets shall be verified by measurement during construction.

Phase 5 – Completion and Close-out

- A.1.32 After ground movement has stopped as confirmed by instrumentation the Contractor shall prepare a "Completion Report". This shall include the following:
 - details of any modifications/mitigation measures to the existing structure;
 - graphs that show the ground movement and construction progress over time with at least 3 months duration of readings which show no change;
 - a schedule showing actual movement compared to predicted movement;
 - a schedule of defects recording only the exceptions (changes) identified during the post construction defects survey;
 - details of any remedial works undertaken; and
 - as-built records (including any temporary works remaining in situ).

Appendix B. Settlement Deed

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

DATED 201[X] (1) [PROJECT CO]

and

(2) **[PARTY NAME]**

DEED

Concerning the mitigation of the effects of settlement arising from the construction works undertaken at [Location]

1	INTERPRETATION	3
2	CONDITIONALITY	6
3	GROUND MOVEMENT AND PROTECTIVE WORKS	7
4	MONITORING OF PROTECTED PROPERTY	12
5	SCHEDULE OF DEFECTS	14
6	COMPENSATION FOR DAMAGE	15
7	COSTS	17
8	AS TO PROJECT CO'S LIABILITY IN CERTAIN CASES	18
9	DISPUTES	18
10	SERVICE OF NOTICES	20
11	RIGHTS OF THIRD PARTIES	20

THIS DEED is made on the day of two thousand and [YEAR]

BETWEEN

- (1) [PROJECT CO] (Company Registration No. [xxx] whose registered office is at [ADDRESS] ("PARTY NAME"); and
- (2) [PARTY NAME][REP [X]](Company Registration No. [xxx]) whose registered office is at [ADDRESS] ("PARTY NAME").

RECITALS

WHEREAS:

- (A) TfL applied on 3 May 2016 for an Order under section 37 of the Planning Act 2008 entitled "Silvertown Tunnel Order" ("the Order") to authorise TfL to construct and operate a twin bored road tunnel under the River Thames linking the Greenwich Peninsula and Silvertown together with necessary alterations to the connecting road network ("the Authorised Works").
- (B) The Order was made on [*date*] and came into force on [*date*].
- (C) TfL has appointed Project Co to construct the Authorised Works.
- (D) Under article 58 of the Order TfL has transferred its obligation to carry out the Authorised Works in accordance with the code of construction practice to Project Co.
- (E) The Order requires that the Authorised Works are carried out in accordance with a code of construction practice which includes an obligation to enter into this Deed with property owners who satisfy certain qualifying conditions.
- (F) The Owner satisfies those qualifying conditions and has required that this Deed be entered into.

OPERATIVE PROVISIONS

1. INTERPRETATION

1.1 In this Deed: -

"the Assessment"	means the Assessment described in clause 3.1;
"the Authorised Works"	mans the works authorised by the Order;
"Box"	means an excavated structural box of a depth of 3 metres or more;
"the Building"	means the building [<i>insert address and description of the building concerned</i>] together with any bridge, tunnel or water, sewerage, gas, electricity or other major service media connected thereto in which the Owner has a legal estate or legal interest;
"Conditions Precedent"	means the conditions set out in clause 2;
"Ground Movement"	means movement of the ground in response to the Relevant Construction;
"Listed building"	means a listed building within the meaning of the Planning (Listed Buildings and Conservation Areas) Act 1990;
"the Notice"	means the Notice described in clause 3.6.2;
"the 1 mm settlement contour"	means a prescribed predicted settlement contour undertaken as part of an initial "green field" stage 1 settlement analysis within the Order limits and identified for assessment as set out in the Appendix to this deed;
"the Opening Date"	means the date of opening to the public of such of the Authorised Works that are intended for public use in the vicinity of the Building;
"the Order"	means the Silvertown Tunnel Order 201[];
"the Owner"	includes the Owner's successors in title to and assigns of the Protected Property;
"the Owner's Engineer"	means the Engineer appointed in accordance with

clause 3.6.1;

"Project Co" means [name of company entering into deed with owner]

"the Protectedmeans any part of the Building in which the OwnerProperty"has a legal estate or in respect of which the Owner
has repairing obligations;

"Protective Works" means works for the supporting or strengthening of the Building or its foundations undertaken to minimise damage to the Building arising as a result of the Relevant Construction;

"the Relevant in relation to the Building, means any part of: -

Construction"

- (a) bored tunnels, comprised in the AuthorisedWorks and which is within the 1 mm settlementcontour; or
- (b) excavations comprising a Shaft or Box,
 comprised in the Authorised Works and which
 is within the 1 mm settlement contour;

"Relevant Protective means the Protective Works described in the Report as necessary (whether as consulted on under clause 3.5 or as agreed or deemed to have been agreed or as determined as mentioned in clause 3.6 as the case may be);

"the Report" means the Report for the Building described in clause 3.2;

"Sent" means (in relation to the sending to the Owner of the Report under clause 3 and of the Notice transmitted by facsimile (and confirmed by transmission confirmation slip), delivered by hand or posted by ordinary first-class post or recorded delivery post addressed to the address referred to

	in clause 10.1) and "send" shall be construed
	accordingly;
"Shaft"	means an excavated shaft of a depth of 3 metres or more; and
"the Surveyor"	means the surveyor or engineer appointed by the Owner for the purposes of this Deed;
"TfL"	means Transport for London; and
"Working Day"	means a day other than a Saturday or Sunday or public Holiday in England.

- 1.2 Where in this Deed reference is made to any clause, paragraph or appendix or recital such reference (unless the context otherwise requires) is a reference to a clause, paragraph or appendix or recital in this Deed.
- 1.3 Words importing the singular meaning where the context so admits include the plural meaning and vice versa.
- 1.4 Words of the masculine gender include the feminine and neuter genders and words denoting actual persons include companies, corporations and firms and all such words shall be construed interchangeably in that manner.
- 1.5 TfL may appoint any of its contractors or sub-contractors for any of the Authorised Works as agent in relation to the exercise of any of its functions under this Deed (but without prejudice to any liability of TfL in the event of a failure to comply with the terms of this Deed).

2. CONDITIONALITY

- 2.1 The obligation on Project Co to carry out the Assessment, the Relevant Protective Works and/or monitoring of the Building under this Deed is subject to:-
 - 2.1.1 the obtaining of any necessary consents, agreements or other approvals (which Project Co shall use reasonable endeavours to obtain) required for the purpose, whether required under any provision having effect under statute, or under any undertaking or agreement given or

entered into by the Secretary of State, TfL or Project Co which governs the exercise of the powers given by the Order, or otherwise and the obligation to use reasonable endeavours to obtain such consents, agreements or other approvals includes (where applicable) the obligations at clause 3.8; and

2.1.2 the Owner permitting Project Co to enter the Protected Property at reasonable times and on the giving of reasonable notice to perform its obligations under this Deed.

3. GROUND MOVEMENT AND PROTECTIVE WORKS

Preparation of Assessment and Report

- 3.1 Before it commences the Relevant Construction Project Co shall carry out an assessment (to be undertaken by a competent qualified person with appropriate experience) to predict any Ground Movement to the Building that may result from the Relevant Construction and any damage to the Building which may be caused by such movement.
- 3.2 The Assessment shall be carried out in accordance with the procedures set out in the Appendix to this Deed and following the Assessment Project Co shall prepare the Report in respect of the Building which shall contain such of the following as may be applicable: -
 - 3.2.1 an assessment of predicted Ground Movement to the Building caused by the Relevant Construction (Stage 1) and if Ground Movement is assessed at Stage 1 to be more than 1 mm, a Stage 2 assessment will be carried out;
 - 3.2.2 if the Building has been subject to a Stage 2 or Stage 3 assessment as described in the Appendix to this Deed:
 - 3.2.2.1 the assessed maximum tensile strain and an assessment of the predicted damage to the Building;
 - 3.2.2.2 the results of any structural inspection of the Building;
 - 3.2.2.3 any proposed monitoring specific to the Building;
 - 3.2.2.4 an assessment of whether Protective Works are required; and

- 3.2.2.5 details of any such Protective Works, including designs, method of working and programme of such Protective Works;
- 3.2.3 if the Building is a listed building, particulars of any such additional safeguarding measures as are referred to in paragraph 4.9 of the Appendix to this Deed.
- 3.3 The Report shall be sent to the Owner before commencement of the Relevant Construction, following which Project Co will (subject always to clauses 3.5 and 3.6 in a case where those clauses apply) be entitled to proceed with that construction without prejudice to any claim for compensation which the Owner may have (whether against TfL or Project Co).
- 3.4 Without prejudice to article 15 (Protective works to buildings) of the Order Project Co may, as often as it may reasonably require and upon giving not less than 14 days' notice in writing to the Owner, enter the Protected Property at any reasonable time for the purposes of carrying out the Assessment.

Cases in Risk Category 2 or below

3.5 Subject to clause 3.6 below, if the assessment of the predicted damage to the Building concerned contained in the Report is that the Building falls into Risk Category 2 (as set out in the Appendix to this Deed) or below, Project Co, if so requested by the Owner by notice given in writing to Project Co not later than 20 working days after the day on which the Report is Sent to the Owner, shall seek to consult with the Owner regarding the Report following which consultation (or if no request for consultation is made by the Owner within the time limit referred to above) Project Co shall be entitled to proceed with the Relevant Construction without prejudice to any claim for compensation which the Owner may have.

Cases in Risk Category 3 or above

3.6 If following the carrying out of the Assessment the assessment of the predicted damage to the Building concerned contained in the Report assesses that the Building falls into Risk Category 3 (as set out in the Appendix to this Deed) or above or if the Report recommends that Protective Works should be carried out wholly or partly from within the Building:-

- 3.6.1 the Owner shall be entitled to appoint an Engineer (at Project Co's cost in accordance with clause 3.6.9) save that:-
 - 3.6.1.1 in the event of Project Co entering into a Deed with any other person or persons (whether on, before or after the date of this Deed) in relation to the whole or part of the Building containing provisions similar to this clause 3:-
 - (a) the Owner's Engineer shall be appointed jointly by the Owner and the other person or persons with a legal interest in the Building (or, failing agreement between them, by the President of the Institution of Civil Engineers on the written application of either or, as the case may be, any of them), and
 - (b) as regards any disputes concerning the Report and the appropriate protective works for the Building (if any), for the purpose of obtaining a consistent outcome with respect to the Building capable of implementation by Project Co clause 9.12 has effect;
 - 3.6.1.2 where Project Co enters (or is to enter) into such a Deed as is mentioned in clause 3.6.1.1 above with any other person or persons, to enable the joint appointment to be made Project Co shall either send a notice in writing to the Owner specifying the other person or persons before the beginning of the period of 15 working days ending with the day on which the Report is Sent to the Owner under clause 3.3, or shall send that notice at some later time before the Report is Sent (in which case the Report shall be deemed to have been sent for the purposes of paragraphs 3.6.3 on the day 15 working days after the day on which that notice is Sent);
- 3.6.2 where the Owner decides not to appoint an engineer in accordance with clause 3.6.1 Project Co, if so requested by the Owner by Notice given in writing to Project Co not later than 20 working days after the day on which the Report is Sent to the Owner, shall seek to consult with the Owner regarding the Report following which Project Co shall be entitled to proceed with the Relevant Construction without prejudice to any claim for compensation which the Owner may have;

- 3.6.3 where an engineer is appointed in accordance with clause 3.6.1 the Owner's Engineer may, not later than 20 working days after the day on which the Report was sent to the Owner, give written Notice to Project Co stating whether or not the Report is agreed;
- 3.6.4 if the Owner's Engineer does not agree the Report the Notice shall contain detailed reasons for the objections to it;
- 3.6.5 if in Project Co's opinion (Project Co acting reasonably) the Notice contains reasonable objections:-
 - 3.6.5.1 Project Co and the Owner's Engineer shall seek to agree such amendments to the Report as are reasonably necessary to address those objections; and
 - 3.6.5.2 in the event that Project Co and the Owner's Engineer fail to agree the Report within 20 working days of the giving of the Notice the dispute shall be referred to dispute resolution in accordance with the provisions in clause 9.
- 3.6.6 if in Project Co's opinion (Project Co acting reasonably) the Notice contains unreasonable objections Project Co shall consult with the Owner's Engineer regarding the Report following which Project Co shall be entitled to proceed with the Relevant Construction without prejudice to any claim for compensation which the Owner may have and the right of the Owner to refer the dispute as to the reasonableness of the objections to dispute resolution in accordance with clause 9;
- 3.6.7 in the event that no Notice is served within the period specified in clause 3.6.3 the Report shall be deemed to have been agreed;
- 3.6.8 upon the Report being agreed or deemed to be agreed or upon determination of the dispute (as the case may be) Project Co shall be entitled to proceed with the Relevant Construction and the Relevant Protective Works in accordance with the Report as so agreed or determined; but if the Notice contains objections to the Report and the Report has not been agreed or finally determined within the period of 75 working days after the day on which the Report was Sent to the Owner then provided that Project Co has used its reasonable endeavours to agree the Report and facilitate the determination of the adjudicator Project Co shall be entitled to proceed with the Relevant Construction and the Relevant Protective Works specified in the Report
in accordance with its terms without prejudice to any claim for compensation which the Owner may have;

3.6.9 Project Co shall, within 20 working days of being called on to do so and of being provided with the appropriate documentation, repay to the Owner all reasonable costs, charges and expenses properly incurred by the Owner, including VAT thereon insofar as the same is not recoverable by the Owner (whether as a deduction against output tax or as a VAT credit or otherwise), in connection with:-

- 3.6.9.1 the services of the Owner's Engineer under clauses 3.6.2 to 3.6.4; and
- 3.6.9.2 the services of architects, surveyors, engineers and other technical advisers to whom the Owner's Engineer finds it reasonably necessary to refer in connection with clause 3.6.5(a).
- 3.7 Subject to the Conditions Precedent and clause 3.8 if Project Co carries out the Relevant Construction it shall also carry out the Relevant Protective Works (including the method of working and programme therefore).
- 3.8 The duty to carry out Protective Works under clause 3.7 has effect subject to the obtaining of any necessary consents, agreements or other approvals (which Project Co shall use reasonable endeavours to obtain) required for the purpose, whether required under any provision having effect under statute, or under any undertaking or agreement given or entered into by Project Co which governs the exercise of the powers conferred by the Order, or otherwise and the obligation to use reasonable endeavours to obtain such consents, agreements or other approvals includes an obligation:-
 - 3.8.1 where the Report in the form sent to the Owner identifies Protective Works as necessary, Project Co shall also serve notice under paragraph 5 of article 15 of the Order not later than 10 working days after the day on which the Report is so sent to the Owner on all persons who have not entered into a Deed in relation to the whole or part of the Building containing provisions similar to clause 3 and 4 of this Deed and who after reasonable inquiry by Project Co are (as at that time) found eligible to receive such notice in relation to the Building and the Protective Works so identified; and

- 3.8.2 after the sending of the Report to the Owner, to do what else is reasonable under article 18 of the Order to enable the Relevant Protective Works to be carried out at a time consistent with Project Co's construction timetable including, if a person gives notice of dispute under paragraph 6 of article 15 in relation to the Building (whether in consequence of a notice by Project Co given pursuant to paragraph 5 of article 18 of the Order or subsequently)-
 - 3.8.2.1 to pursue the arbitration with all due despatch; and
 - 3.8.2.2 to do all that is reasonable to argue the case in the arbitration in favour of the Relevant Protective Works in consultation with the Owner.
- 3.9 For the avoidance of doubt:
 - 3.9.1 for the purposes of clause 3.8 the circumstances in which Project Co shall be treated as not being able to carry out any protective works under article 18 to the Order includes a case where the decision of an arbitrator under paragraph 6 of article 15 of the Order precludes those works being carried out, or makes the carrying out of those works impractical; and
 - 3.9.2 if Project Co has done the things required by clause 3.8.2(a) and (b) but, on an arbitration under paragraphs of article 15 of the Order relating to the Building, a decision of the arbitrator has not been obtained by the time that would be required in order to maintain Project Co's construction timetable for the Relevant Construction (or for the Relevant Protective Works), Project Co may carry out the Relevant Construction in accordance with its construction timetable without carrying out such of the Relevant Protective Works as it could not or cannot lawfully or practicably carry out in accordance with that timetable because the decision has not been obtained.
- 3.10 Protective Works carried out in accordance with this Deed shall be treated as carried out under the powers conferred by article 15 of the Order, save that the Owner agrees that paragraph 6 of article 15 shall not apply to the Owner.

4. MONITORING OF PROTECTED PROPERTY

4.1 If the Report prepared under clause 3.2 above recommends that monitoring specific to the Building should be carried out, subject to clause 4.6 Project Co

shall, if it proceeds with the construction of the Authorised Works, undertake monitoring of the Building.

- 4.2 Except in so far as further or different types or methods of monitoring are recommended in that Report, monitoring shall consist of precise surveying of studs and targets on the outside of the Building, and shall begin prior to the commencement of the carrying out of the Relevant Construction and continue during the period of the Relevant Construction and shall end at the later of (a) six months from the Opening Date and (b) (if prior to the end date the monitoring has established any movement of the Building arising in consequence of the Relevant Construction) the date on which the monitoring indicates that such movement has reduced to a rate of 2mm per annum or less.
- 4.3 As soon as practicable after monitoring results have been obtained, whether interim or final, Project Co shall: -
 - 4.3.1 in the case of final results, make them available for inspection by theOwner for a period of three months commencing on the date on whichProject Co notifies the Owner that the monitoring has ceased; and
 - 4.3.2 in the case of both interim and final results, inform the Owner that Project Co will send to the Owner a copy of such of those results as the Owner may request.
- 4.4 Such monitoring will be recommenced at the written request of the Owner if: -
 - 4.4.1 the Owner can show reasonable grounds for concluding that building movement arising in consequence of the Relevant Construction of a rate greater than 2mm per annum has started again after the monitoring period specified in clause 4.2 or a monitoring period under this clause 4.4 has elapsed;
 - 4.4.2 that request is made within two years from the Opening Date; and
 - 4.4.3 any such recommenced monitoring shall continue until the monitoring indicates that building movement arising in consequence of the Relevant Construction has reduced to a rate of 2mm per annum or less.
- 4.5 The Owner hereby agrees that prior to and during the construction of the Authorised Works Project Co may as often as it may reasonably require, upon

giving not less than 7 days' notice in writing to the Owner, enter the Protected Property to monitor the effect of the Authorised Works (including entering to place and maintain studs and targets for the purposes of monitoring referred to above) Provided always that such inspection, entering, placing and maintaining shall be conducted with a duly authorised representative of the Owner and at reasonable hours and that before placing such studs and targets on any of the Protected Property Project Co shall consult with the Owner on their positioning.

4.6 The obligations of Project Co to carry out monitoring under this clause 4 are subject to the obtaining of any necessary consents, which Project Co shall use its reasonable endeavours to obtain.

5. SCHEDULE OF DEFECTS

- 5.1 Before it commences construction of so much of the Authorised Works as will or may affect the Building, Project Co shall at its own expense appoint in the joint names of Project Co and the Owner (and such other persons as Project Co may determine so that Project Co shall not be obliged to prepare more than one such schedule for the Protected Property) a reputable firm of chartered building surveyors or chartered engineers to prepare a schedule of defects existing in the Protected Property (including, so far as relevant and can be established from the visual inspection normally conducted in relation to the preparation of such schedules, a description of the apparent magnitude of any defect) and that firm shall submit a copy of the schedule of defects ("the original schedule") to Project Co and the Owner.
- 5.2 In a case where clause 5.1 applies, after Project Co has constructed the Authorised Works in the vicinity of the Building and either at the written request of the Owner or on the initiative of Project Co, a schedule shall be prepared at the expense of Project Co by the firm appointed under clause 5.1 (or a similar firm if the original firm is unable to act for whatever reason) similar to (and with similar inspection techniques as) the original schedule setting out what changes appear to have occurred in respect of the defects identified in the original schedule prepared pursuant to clause 5.1 (as amended pursuant to any procedure under clause 9), provided such request is made before the end of the period of 2 years from the Opening Date or if later (and building-specific monitoring of the Building is carried out under clause 4) the end of the period of three months from the day on which Project Co notifies the Owner that

monitoring has ceased, and provided further that Project Co will not be obliged to commission more than one such schedule for the Protected Property.

6. COMPENSATION FOR DAMAGE

- 6.1 In addition to any claim which the Owner may make under the provisions of any enactments incorporated with or applied by the Order with respect to compensation for lands taken or injuriously affected, the Owner may (subject to clause 6.2, 6.3 and 6.5) make a separate claim (a "clause 6.1 claim") upon Project Co for compensation for the reasonable and proper costs properly incurred by the Owner in remedying any material physical damage caused to the Protected Property by Ground Movement arising from the construction of the Authorised Works, provided that the claim is made before the end of the period of two years from the Opening Date, or if later (and where building-specific monitoring of the Building is carried out under clause 4) the end of the period of six months from the day on which Project Co notifies the Owner that monitoring has ceased.
- 6.2 For the avoidance of doubt, if the Building or the Protected Property suffers from a pre-existing defect or defects which are worsened by the construction of the Authorised Works, the compensation payable under clause 6.1 is limited to the additional costs of repair of the Protected Property which go beyond those that would be incurred upon remedying the pre-existing defect or defects (assuming the Authorised Works had not been constructed), provided always that Project Co may not assert any pre-existing defect that is not recorded in a schedule of defects prepared in accordance with clause 5 or otherwise specifically agreed in writing by Project Co and the Owner, in each case subject always to the outcome of any dispute determined in accordance with the procedure set out in clause 9.
- 6.3 Before carrying out any work in respect of which a clause 6.1 claim may be made, the Owner shall give not less than 28 days' notice in writing to Project Co, specifying the material physical damage concerned and the proposed remedial work; and if within the period of 28 days after the giving of such notice:-
 - 6.3.1 Project Co elects by notice in writing to the Owner to remedy all or part of that damage, then no claim may be made under that paragraph in respect of the damage or part, but Project Co shall be under a duty to remedy the damage or part to the reasonable satisfaction of the Owner

as soon as reasonably practicable thereafter in accordance with a specification agreed with the Owner or in default of such agreement, determined under clause 9 below; and/or

- 6.3.2 Project Co by notice in writing to the Owner requires the Owner to obtain competitive quotes for all or any of the remedial work, then before entering into a contract for or arranging for the carrying out of the work concerned the Owner shall (i) take reasonable steps to obtain not less than 3 competitive prices for the work and (ii) obtain the consent of Project Co to the quote to be accepted, such consent not to be unreasonably withheld.
- 6.4 If it appears to the Owner or Project Co that any damage in respect of which notice is given by the Owner under clause 6.3 is likely to be of a recurring nature by reason of the programme for or the nature of the Authorised Works, either of them may, in the relevant notice, require a timetable for the carrying out of the remedial work relating to the damage (including a timetable and specification of any interim repairs reasonably necessary in consequence of the damage) to be agreed or, in default of agreement, determined under clause 9; and for this purpose "the relevant notice" means:-
 - 6.4.1 where the requirement is made by the Owner, the notice given by him under clause 6.3;
 - 6.4.2 where the requirement is made by Project Co, and Project Co gives a notice of election under clause 6.3(a), the notice of election;
 - 6.4.3 where the requirement is made by Project Co, and Project Co gives notice requiring competitive quotes under clause 6.3(b), the notice requiring competitive quotes;
 - 6.4.4 where the requirement is made by Project Co and Project Co gives no such notice of election or notice requiring competitive quotes, a separate notice in writing stating the requirement, which is given within the period of 28 days after the giving of the notice by the Owner under clause 6.3; and
 - 6.4.5 The Owner hereby agrees that Project Co may as often as it may reasonably require and upon giving not less than 14 days' notice in writing to the Owner, enter the Protected Property at any reasonable time to carry out works in compliance with any duty under clause 6.3

or 6.4 but in doing so Project Co agrees that it will have due regard to any activities carried out by those with an interest in the Building.

- 6.5 For the avoidance of doubt, the Owner shall not be entitled to (and hereby accepts the fulfilment of the obligations of Project Co under this clause in satisfaction of any right to) compensation under any enactment as regards any damage or claim in respect of which the Owner is entitled to payment under clause 6.1 or which Project Co is under a duty to remedy under clause 6.3 or 6.4.
- 6.6 In clause 6.5 reference to "any enactment" includes reference to the Order.

7. COSTS

- 7.1 Project Co shall repay to the Owner all reasonable costs charges and expenses properly incurred by the Owner, including VAT thereon insofar as the same is not recoverable by the Owner (whether as a deduction against output tax or as a VAT credit or otherwise), in connection with: -
 - 7.1.1 the services of the Surveyor under clause 5.2 of this Deed;
 - 7.1.2 the services of the Surveyor in connection with the successful proving of a claim under clause 6.1 of this Deed; and
 - 7.1.3 the services of architects, surveyors, engineers and other technical advisers to whom the Surveyor finds it reasonably necessary to refer in connection with the successful proving of a claim under clause 7.1(b).
- 7.2 Before any services in respect of which repayment may be claimed under clause 7.1.2 and 7.1.3 above are undertaken, the Owner or the Surveyor shall give Project Co not less than 28 days' notice in writing of the services proposed to be undertaken, the basis on which any costs charges or expenses are to be calculated, and an estimate of the total amount of those costs charges and expenses.
- 7.3 Any amount payable under clause 7.1 shall be paid by Project Co within 20 working days of that amount being agreed between Project Co and the Owner or being determined in accordance with clause 9.

8. AS TO PROJECT CO'S LIABILITY IN CERTAIN CASES

8.1 The fact that any work or thing has been executed or done in accordance with a Report prepared or agreed under clause 3 above or in accordance with any decision of an adjudicator shall not relieve Project Co from any liability for damage caused to the Protected Property or affect any claim by the Owner in respect of such damage.

9. DISPUTES

- 9.1 Any dispute or difference arising between the parties hereto as to their respective rights duties and obligations under this Deed or as to any matters arising out of or in connection with the subject matter of this Deed shall be determined by adjudication in accordance with the provisions of this clause.
- 9.2 Either party may give notice in writing to the other referring the dispute to adjudication under this clause, and that notice shall briefly state the matter which is in dispute between them.
- 9.3 Unless the dispute in question is one that falls to be consolidated under clause 9.12 below with other disputes relating to the Building and an adjudicator has already been appointed for any of the disputes (under a provision in another Deed similar to this clause 9) which fall to be so consolidated (in which case that adjudicator shall act on the consolidated proceedings), the party giving notice under clause 9.2 shall upon giving that notice forthwith request the appointing body to nominate an independent person to act as adjudicator, who shall be a person professionally qualified for not less than 10 years and who is also a specialist in relation to the subject matter of the dispute, and the request shall ask the appointing body to nominate the adjudicator within 7 days of the notice being given with a view to the matter being referred to the adjudicator within that period.
- 9.4 The appointing body for the purposes of clause 9.3 shall be the President, Vice-President or other duly authorised officer of the Institution of Civil Engineers, except in the case of a dispute or difference with regard to the meaning or construction of this Deed, when the appointing body shall be the President, Vice-President or other duly authorised officer of the Law Society.
- 9.5 The terms of reference of the adjudicator shall be as follows: -

- 9.5.1 the adjudicator is to reach a decision within 28 days of the dispute being referred to him or within such longer period (if any) as may be agreed by the parties after the dispute has been referred to him;
- 9.5.2 the adjudicator may extend that period of 28 days by up to 14 days without the agreement of the parties to the dispute if the party referring the dispute consents;
- 9.5.3 the adjudicator must act impartially; and
- 9.5.4 the adjudicator may take the initiative in ascertaining the facts and the law.
- 9.6 In reaching his decision, the adjudicator shall act as an expert and not an arbitrator and he shall accordingly take into account his expert knowledge and judgement.
- 9.7 The parties hereto agree that the decision of the adjudicator shall be final and binding except in a case of manifest error.
- 9.8 In a case of manifest error the decision shall (so far as consistent with the terms of this deed) be binding until the matter is finally determined by legal proceedings or by agreement between the parties.
- 9.9 The adjudicator shall not be liable for anything done or omitted in the discharge or purported discharge of his functions as adjudicator unless the act or omission is in bad faith, and any employee or agent of the adjudicator shall be similarly protected from liability.
- 9.10 The incidence of the adjudicator's reasonable costs and fees in the adjudication shall lie (as between the parties to the dispute) at the award of the adjudicator.
- 9.11 This clause 9 shall apply to disputes falling both within and outside section 108 of the Housing Grants, Construction and Regeneration Act 1996, and (in the case of a dispute falling within that section) if there is any inconsistency between the provisions of this Deed and the requirements of subsections (1) to (4) of that section the inconsistency shall be resolved in favour of those subsections and those subsections shall to the extent of such inconsistency be deemed to be incorporated in this Deed and have effect accordingly (so that amongst other matters the Scheme for Construction Contracts is not intended to apply).

- 9.12 In the event of Project Co entering into a Deed with any person or persons other than the Owner (whether on, before or after the date of this Deed) in relation to the whole or part of the Building containing provisions similar to clause 3 above, then:-
 - 9.12.1 all disputes of the land mentioned in clause 3.6.6 relating to the Building shall be consolidated into a single proceeding with a single adjudicator for all of them and this clause 9 shall have effect accordingly; and
 - 9.12.2 the finding of the adjudicator shall have effect with respect to the Building for the purposes of clause 3 with respect to the Owner even if the Owner did not become or did not remain a party to the adjudication.

10. SERVICE OF NOTICES

- 10.1 Any notice in writing that is to be given by Project Co to the Owner shall be deemed effectively given if left at, or despatched by a postal service in which receipt is recorded addressed to, [specify address within United Kingdom] or such other address within the United Kingdom as the Owner notifies to Project Co in writing, and in the case of the documents referred to in the definition of "sent" in clause 2(1) above, the documents shall be deemed effectively given if posted by ordinary first class post to that address (whether or not received).
- 10.2 Any notice in writing that is to be given by the Owner to Project Co shall be deemed effectively given if left at, or sent by a postal service in which receipt is recorded addressed to, the address as set out in this Deed or at such other address within the United Kingdom as is notified in writing by Project Co to the Owner.

11. **RIGHTS OF THIRD PARTIES**

- 11.1 This Deed may be varied determined or supplemented without the consent of any third party.
- 11.2 This Deed does not and is not intended (save where this clause is in any other clause expressly (or by express reference) excluded) intended to confer any rights whatsoever on any person who is not a party to this Agreement pursuant to the Contracts (Rights of Third Parties) Act 1999.

IN WITNESS of which the parties have executed this document as a Deed on the day and year first before written

EXECUTED AS A DEED by the Parties on the day and year first before written

EXECUTED as a Deed by)
[PROJECT CO])
acting by two Directors or a)
Director and Secretary)

Director

Director/Secretary

EXECUTED as a Deed by)
[PARTY NAME])
acting by two Directors or a)
Director and Secretary)

Director

Director/Secretary

APPENDIX Ground Movement Assessment

1.0 Assessment and Mitigation Process

1.1 Project Co shall investigate the potential for ground movement associated with the design and possible construction methods using the process defined in paragraph 1.2 for the following purposes:

a) to assess risk of building damage by identifying those zones where the implementation of the design is likely to cause ground movements which will result in Risk Category 2 'Slight' being exceeded (see Table 1);

(b) to assess the risks of such degrees of damage occurring and consider alternative designs as necessary;

c) to undertake an assessment of the potential consequences where there is a significant likelihood that Risk Category 2 'Slight' will be exceeded and identify specifically what the risks are;

d) to design protective measures where necessary to mitigate against the risk of damage exceeding Risk Category 2;

e) to demonstrate that the environmental effects of excavation induced ground movements have been considered and taken account of in the design;

f) to assess the risk of damage to utilities and to design mitigation measures in agreement with the utility owner;

g) to assess the effects of excavation to existing above-ground and underground infrastructure and to design suitable mitigation measures;

h) to indicate where property may require demolition or structural modification; and

i) to enable the preparation of contingency plans to deal with residual risks.

1.2 A staged assessment process shall be followed to identify the risk and impact of construction activities on third parties. Five stages have been identified to assess the impact and risk from project inception to completion. The initial stages are aimed at maximising the benefit from assessments, whilst the latter are about managing and recording the impact of construction. The five stages are:

2.0 Stage 1 – Scoping

2.1 Schedules and plans shall be prepared to identify all third party assets assessed to experience ground movement exceeding 1mm using conservative parameters.

3.0 Stage 2 – Initial Assessment

3.1 Project Co shall carry out initial assessment calculations using simple empirically calibrated, methods and moderately conservative parameters to classify the risk of damage to assets. For building structures the risk should be classified in accordance with Table 1. For non-building infrastructure the level of risk should be determined by ensuring that deformations do not exceed tolerable values determined in consultation with the asset owner. These calculations shall be based on record drawings, where available and an inspection for assessment. Assets estimated to be a risk of damage greater than Risk Category 2 –Slight or where damage exceeds the agreed tolerable limits require further detailed assessment at Stage 3. A schedule and plans of predicted damage shall be prepared.

Building damage classification				
Risk Categ ory	Descripti on of degree of damage +	Description of typical and likely forms of repair for typical masonry buildings	Approx. crack width* (mm)	Max. tensile strain %
0	Negligible	Hairline cracks		< 0.05
1	Very slight	Fine cracks easily treated during normal redecoration. Perhaps isolated slight fracture in building. Cracks in exterior visible upon close inspection	0.1 to 1.0	0.05 to 0.075
2	Slight	Cracks easily filled. Redecoration probably required. Several slight fractures inside building. Exterior cracks visible; some repainting may be required for weathertightness. Doors and windows may stick slightly	1 to 5	0.075 to 0.15
3	Moderate	Cracks may require cutting out and patching. Recurrent cracks can be masked by suitable linings. Tuck pointing and possible replacement of a small amount of exterior brickwork may be required. Doors and windows sticking. Utility services may be interrupted. Weather tightness often impaired	5 to 15 or a number of cracks greater than 3	0.15 to 0.3
4	Severe	Extensive repair involving removal and replacement of walls especially over door and windows required. Window and door frames distorted. Floor slopes noticeably. Walls lean or bulge noticeably. Some	15 to 25 but also depends on number of cracks	> 0.3

		loss of bearing in beams. Utility services disrupted.		
5	Very Severe	Major repair required involving partial or complete reconstruction. Beams lose bearing, walls lean badly and required shoring. Windows broken by distortion. Danger of instability	Usually > 25 but depends on number of cracks	

+ In assessing the degree of damage, account must be taken of its location in the building or structure.

* Crack width is only one aspect of damage and should not be used on its own as a direct measure of it.

Burland, J.P. and Wroth, C.P., Settlement of Buildings and Associated Damage, Proceedings of a Conference on the Settlement of Structures, Cambridge, 1974, pp 611 – 54 and 764 – 810;

Table 1: Building Damage Classification

3.2 The heritage value of a Listed Building should be considered during the initial assessment by reviewing the sensitivity of the building structure and of any particular features together with the initial assessment calculations. The heritage assessment examines the following:

a) the sensitivity of the building / structure to ground movements and its ability to tolerate movement without significant distress. The potential for interaction with adjacent buildings / structures is also considered. A score within the range of 0- 2 should be allocated to the building/structure in accordance with the criteria set out in Table 2;

b) the sensitivity to movement of particular features within the building / structure and how they might respond to ground movements. A score within the range of 0-2 should be allocated to the building in accordance with the criteria set out in Table 2.

The scores for each of the two categories (a) and (b) should be added to the category determined in paragraph 3.1 to inform the decision making process. In general, Listed Buildings which score a total of 3 or higher should be subject to further assessment as part of the Stage 3 – Detailed Assessment. Buildings that score a total of 2 or less are predicted to suffer a degree of damage which may be easily repairable using standard conservation based techniques and hence no protective measures for the building's particular features should be required. However, ultimately the professional judgement of engineering and historic building specialists should be used to determine whether additional analysis is required.

Silvertown Tunnel Code of Construction Practice

Document Reference: 6.10

Score	Criteria		
	Sensitivity of the structure to ground movements and interaction with adjacent buildings	Sensitivity to movement of particular features within the building	
0	Masonry building with lime mortar not surrounded by other buildings. Uniform facades with no particular large openings.	No particular sensitive features	
1	Buildings of delicate structural form or buildings sandwiched between modern framed buildings which are much stiffer, perhaps with one or more significant openings.	Brittle finishes, e.g. tight-jointed masonry, which are susceptible to small movements and difficult to repair.	
2	Buildings which, by their structural form, will tend to concentrate all their movements in one location.	Finishes which if damaged will have a significant effect on the heritage of the building, e.g. cracks through frescos.	

 Table 2: Scoring for Sensitivity Assessment of Listed Buildings

4.0 Stage 3 - Detailed Assessment and Mitigation Design

- 4.1 Project Co shall carry out detailed assessments of structures that will be affected by the works so that any monitoring works and mitigation works can be designed and implemented.
- 4.2 For structures at risk of exceeding Risk Category 2 (Slight) Project Co's designer shall undertake a detailed assessment (more rigorous) to determine:

a) the influence of the structure and its foundations on the predicted ground movements;

b) the volume loss at which the risk of damage to the structure (or any sensitive finishes/features) is 'slight' or better;

c) whether this volume loss may be achieved by the proposed excavation design/control measures;

d) any special control measures required to reduce the predicted damage to acceptable levels (i.e. Risk Category 2 'slight' damage category or below) such as significantly higher face pressures with EPBM tunnelling and the practicality of these;

e) the amount of ground movement that the structure (and or any sensitive finishes/features) can accommodate without exceeding Risk Category 2 or any other agreed damage level;

f) the level of residual risk if intrusive mitigation measures are not implemented.

- 4.3 The detailed assessments should include a number of iterations to determine how the risk of damage to a building may be reduced. Asset-specific empirical models shall be prepared successively using moderately conservative and best estimate parameters. If after these iterations the use of empirical methods do not reduce the risk of building damage to acceptable levels (i.e. Damage Category 2 'slight' damage category and below), the damage assessment shall be refined by increasing the sophistication of the analysis with the aim of reducing the risk of asset damage to acceptable levels and to eliminate the asset from further assessment.
- 4.4 If the risk of damage cannot be shown to be reduced by detailed assessment to acceptable levels, then mitigation measures shall be designed.
- 4.5 The primary means of settlement mitigation shall be practical measures to control ground movement by good design and construction practice. This could include staged excavation sequences within sprayed concrete lining (SCL) works, ground treatment, face stabilisation, spiling / face dowels, increasing face pressure when using an tunnel boring machine (TBM), adopting stiffer walls/propping for rectangular shafts etc.
- 4.6 In the event that physical mitigation measures are still required (i.e. to control building damage to Damage Category 2 or less or to meet the third party asset Owner's requirements), Project Co shall seek to obtain the third party asset Owner's approval or may use TfL's powers under the Order as transferred to Project Co or as remain vested in TfL to undertake protective works.
- 4.7 Project Co shall also undertake a comparative risk assessment to demonstrate that the risks associated with installation/implementation of any intrusive mitigation measures (such as compensation grouting) are no worse than the risks associated with the base case.
- 4.8 The relevant Local Authority and English Heritage shall be consulted on the results of the Listed Building assessment reports and the proposals for protective measures, if any are required. English Heritage shall also be consulted in relation to Listed Buildings where they would normally be notified or consulted on planning applications or listed building consent applications.
- 4.9 When considering the need and type of protective measures for Listed Buildings, due regard should be given to the sensitivity of the particular features of the building which are of architectural or historic interest and the sensitivity of the structure of the building to ground movement. Where the assessment highlights potential damage to the features of the building which it will be difficult or impossible to repair and/or if that damage will have a significant effect on its heritage value, the assessment may recommend appropriate measures to safeguard those features either in-situ or by temporary removal and storage offsite if those with relevant interest(s) in the building consent.

- 4.10 The form of monitoring of Listed Buildings should be determined based on the results of the assessment process.
- 4.11 Where repair works are necessary they will require the consent of those with relevant interest(s) in the building.
- 4.12 For railway track and track support structures Project Co shall:

a) review the track surveys (including specifying additional surveys if required) and establish that ground movement can be accommodated without exceeding track standard operational tolerance in conjunction with the relevant Infrastructure Manager;

b) identify locations where fettling of the track is required pre construction and /or during construction to ensure the track geometry and clearances are acceptable.

- 4.13 Project Co shall prepare plans and sections showing the zone of influence of the works that is defined by ground movements exceeding 1mm.
- 4.14 Project Co shall develop an instrumentation and monitoring plan to validate that ground movements within the zone of influence are in accordance with design assumptions and that the infrastructure remains within acceptable limits. Project Co shall ensure that there is a clear distinction between parameters measured to confirm the change in any parameter is in accordance with the design and parameters measured to limit damage to the assets. This plan shall identify the minimum period of time required to obtain base line data for each monitoring point. Instrumentation adjacent to the railway, which will remain in place during traffic hours, shall conform to the Standards of the asset Owner.

Note: A competent engineer responsible for the works shall consider those factors which may influence the monitoring data and shall determine an appropriate period and frequency for baseline monitoring. This decision making process will include an element of engineering judgement to account for the possible effects of any underlying environmental trends in the assets under consideration.

4.15 Project Co shall demonstrate that the monitoring system complies with the British Tunnelling Society Monitoring Underground Construction best practice guide.

Note: A review of the monitoring system against the checklists provided in Appendix B of the BTS Monitoring Underground Construction best practice guide may be used as a tool to demonstrate compliance.

- 4.16 The detailed assessments shall define the control limits that need to be imposed on the TBM/SCL excavation in the zone of influence. Project Co shall state these control measures on drawings and specifications.
- 4.17 Project Co shall identify the critical parameters to be monitored and define the Asset Control Limits based on:

a) the ability of the asset or structure to withstand ground movement investigated during the assessments carried out in Stage 2 and 3.

b) the risk to third party operations

- 4.18 Project Co's designer shall link the Asset Control Limits to actions within the Emergency Preparedness Plan.
- 4.19 The Instrumentation and Monitoring Plan and Emergency preparedness Plan shall be agreed with the relevant third party asset Owner.

5.0 Stage 4 – Construction

- 5.1 Contingency plans shall be developed and agreed with third party asset Owners to cover the risks posed to the third parties before commencement of the Authorised Works.
- 5.2 Contingency plans shall be implemented where the results of monitoring or inspection so indicate.
- 5.3 Ground movement and construction progress records shall be maintained and reported in regular reviews when construction processes are taking place within the zone of influence.
- 5.4 Predictions and assumptions made during design in respect of both ground movement and the effects which such ground movement will have on adjacent assets shall be verified by measurement during construction.

6.0 Stage 5 – Completion and Close-out

6.1 After ground movement has stopped as confirmed by instrumentation Project Co shall prepare a "Completion Report". This shall include the following:

a) details of any modifications/mitigation measures to the existing structure;

b) graphs that show the ground movement and construction progress over time with at least 3 months duration of readings which show no change;

c) a schedule showing actual movement compared to predicted movement;

d) a schedule of defects recording only the exceptions (changes) identified during the post construction defects survey;

e) details of any remedial works undertaken; and

f) as-built records (including any temporary works remaining in situ).

Appendix C. CD&E Materials Commitments

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

SILVERTOWN TUNNEL

Construction, Demolition and Excavated Materials Commitments

April 2016



MAYOR OF LONDON

THIS PAGE IS INTENTIONALLY BEEN LEFT BLANK

Contents

Abbr	eviations4
GLOS	SSARY
1 li	ntroduction9
1.1	Scheme Description9
1.2	Vision10
1.3	Objectives
1.4	Structure11
2 0	Construction, Demolition and Excavated Waste Arisings
2.1	Overview
3 V	Vaste Hierarchy15
3.1	Waste Hierarchy Overview15
3.2	Definitions
4 C	Construction, Demolition and Excavated Materials Commitments
4.1	Overview17
4.2	CD&E Materials Commitment18
4.3	CD&E Environment and Communities Commitment21
4.4	Self Sufficiency Commitment
5 C	Conclusion25

List of Tables

Table 2-1 Projected CD&E Material Quantities for Greater London area	13
List of Figures	
Figure 3-1 Waste Hierarchy	15

ABBREVIATIONS

The abbreviations, descriptions and project terminology used within this document can be found in the table below.

BMW	Biodegradable Municipal Waste	
CL:AIRE	Contaminated Land: Applications In Real Environments	
CD&E	Construction, Demolition and Excavation	
CLP	The Classification, Labelling and Packaging	
CoCP	Code of Construction Practice	
DCO	Development Consent Order	
DoW CoP	Definition of Waste Development Industry Code of Practice	
EC	European Commission	
End-use	Final use of materials after being re-used, recycled or recovered (e.g. energy, aggregates, compost)	
EU	European Union	
EWC	European Waste Catalogue	
MMP	Materials Management Plan	
PPE	Personal Protective Equipment	
RSA	Receptor Site Assessment (for Excavated Materials only)	
SWMP	Site Waste Management Plan	
TfL	Transport for London	
TML	Transport Moisture Limit	
WEEE	Waste Electronic and Electrical Equipment	

WRAP	Waste And Resources Action Programme

GLOSSARY

A description of key terms and project terminology used within this document can be found in the table below.

Circular economy	A circular economy is based upon extracting the maximum value from a resource whilst in use. Followed by recovering and regenerating the product or material at each subsequent phase of its life
Proximity Principle	Requirement to manage, treat and/or dispose of waste close to the source in order to reduce environmental and costs impacts

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

SUMMARY

- S.1.1 The Silvertown Tunnel scheme (herein after referred to as 'the Scheme') is a new road tunnel under the River Thames connecting the Greenwich Peninsula and Silvertown. The Scheme involves the construction of a twin bore road tunnel providing a new connection between the A102 Blackwall Tunnel Approach on Greenwich Peninsula (Royal Borough of Greenwich) and the Tidal Basin roundabout junction on the A1020 Lower Lea Crossing/Silvertown Way (London Borough of Newham). The Silvertown Tunnel would be approximately 1.4km long and would be able to accommodate large vehicles including double-deck buses. The Boord Street footbridge over the A102 would be replaced with a pedestrian and cycle bridge.
- S.1.2 The Construction, Demolition and Excavation (CD&E) Materials Commitments document defines TfL's overall vision, aims and commitments for the management of CD&E materials on the Scheme and the proposed measures which will be taken to record and track progress towards achieving these commitments. The commitments included within this document have been prepared for a Development Consent Order (DCO) application.
- S.1.3 The aim of this document is to provide TfL's commitments with regard to the management of CD&E materials, and are the benchmarks by which the performance of the Scheme will be assessed. The commitments presented in this document are set to be challenging but achievable and in line with applicable regulations, similar infrastructure schemes and industry benchmarks.
- S.1.4 TfL is committed to achieve the following for the Scheme:
 - 80% (by weight) of CD&E materials to be re-used on site or removed from site for beneficial use with an aspiration to reach 95% (by weight).
 - Safely manage CD&E materials in order to minimise their impact on the environment and communities.
 - Follow the self-sufficiency and the proximity principles through the local management and end-use of CD&E materials.
- S.1.5 The commitments provide a clear message to stakeholders of what the Scheme sets out to achieve with regards to the management and end-use of CD&E materials.

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

1 INTRODUCTION

1.1 Scheme Description

- 1.1.1 The Silvertown Tunnel scheme (herein after referred to as 'the Scheme') is a new road tunnel under the River Thames connecting the Greenwich Peninsula and Silvertown. The Scheme involves the construction of a twin bore road tunnel providing a new connection between the A102 Blackwall Tunnel Approach on Greenwich Peninsula (Royal Borough of Greenwich) and the Tidal Basin roundabout junction on the A1020 Lower Lea Crossing/Silvertown Way (London Borough of Newham). The Silvertown Tunnel would be approximately 1.4km long and would be able to accommodate large vehicles including double-deck buses. The Boord Street footbridge over the A102 would be replaced with a pedestrian and cycle bridge.
- 1.1.2 New portal buildings would be located close to each tunnel portal to house the plant and equipment necessary to operate the tunnel.
- 1.1.3 The introduction of free-flow user charging on both the Blackwall and Silvertown Tunnels would play a fundamental part in managing traffic demand and would help to support the funding of the construction and operation of the Silvertown Tunnel.
- 1.1.4 The design of the Silvertown Tunnel would include a dedicated bus/coach and HGV lane, which would provide opportunities for TfL to provide additional cross-river bus routes.
- 1.1.5 Main construction works would likely commence in 2018 and would last approximately 4 years with the new tunnel opening in 2022/23. The main construction compound would be located at Silvertown, utilising the existing barge facilities at Thames Wharf along with a new temporary jetty for the removal of spoil and delivery of materials by river. A secondary site compound would be located adjacent to the alignment of the proposed cut and cover tunnel on the Greenwich Peninsula.
- 1.1.6 The Construction, Demolition and Excavation (CD&E) Materials Commitments defines the Scheme's overall objectives and commitments for the management of CD&E materials and the proposed measures which will be taken to record and track progress towards achieving these commitments. The commitments included within this document have been prepared for a Development Consent Order (DCO) application.

- 1.1.7 Throughout the Scheme's lifetime, CD&E materials will be generated and transported to material/waste receptor sites for their management, treatment and end-use.
- 1.1.8 CD&E materials (wastes) are defined under Article 3 (1) of the revised EU Waste Framework Directive² as 'any substance or object which the holder discards or intends or is required to discard'. It should be noted that a material being classed as a waste does not preclude it from becoming a 'non-waste', providing that it demonstrated that it has reached its 'end of waste' status (see Article 6 of the revised EU Waste Framework Directive). Within the United Kingdom, the process of demonstrating a waste has reached its end of waste status is typically demonstrated through the CL:AIRE Development of Waste Code of Practice.
- 1.1.9 The commitments set out within this document relate to the management and handling of waste and materials are based upon legislation and regulator guidance set out at the time of writing. Changes in regulation or its interpretation by regulating bodies may require an amendment to the commitments and aspirations. Should this be the case, then the impacts of the change shall be reviewed and the appropriate changes made.

1.2 Vision

1.2.1 The vision of The CD&E Materials Commitments is set out in TfL's Corporate Environmental Framework, which states:

"We aim to ensure that we use resources responsibly, by minimising our consumption of natural resources and encouraging the re-use and recycling of materials. The amount of waste produced increases or decreases in direct relation to the volume of maintenance or construction activities carried out as we expand our transport services. We intend to minimise waste as much as possible and reach a position where 'unwanted' materials are no longer referred to as 'waste', but are considered a potential resource."¹

1.3 **Objectives**

1.3.1 The objectives of The CD&E Materials Commitments document are the following:

¹ Mayor of London (2014) 'TfL Corporate Environment Framework.'

- To provide a clear statement of TfL's commitments in relation to maximising beneficial use and conforming to the principles set forth in the waste hierarchy².
- To provide an outline of the waste regulatory regime of how CD&E materials will be legally managed;

1.4 Structure

- 1.4.1 The CD&E Materials Commitment document is structured as follows:
 - Section 2: CD&E Materials Arisings. This section outlines the current quantities of CD&E materials in the Greater London area as well as the projected quantities for the Scheme.
 - Section 3: Waste Hierarchy. This section outlines the principles and definitions of the waste hierarchy which directly impact the CD&E materials commitments.
 - Section 4: CD&E Materials Commitments. This section outlines the commitments proposed for CD&E materials from the Scheme.

² The EU Waste Framework Directive 2008/98/EC

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

2 CONSTRUCTION, DEMOLITION AND EXCAVATED WASTE ARISINGS

2.1 Overview

- 2.1.1 The following section provides an overview on the quantity of materials generated from CD&E activities in the Greater London area and those projected for the Scheme.
- 2.1.2 These quantities are for information purposes and help provide perspective as to the contribution of materials from the Scheme compared to those projected for all schemes in the Greater London area.

Greater London Material Arisings

- 2.1.3 In 2012, the CD&E sector in the UK generated 100.23 million tonnes of material of which 85.24 million tonnes were generated in England³. In the Greater London area, the 2010 reported quantities generated by the CD&E sector were 20.5 million tonnes⁴.
- 2.1.4 The Scheme is expected to begin construction in 2018 with an expected opening date for the tunnel of 2022/ 2023. For illustrative purposes only during this time period, it is expected that the CD&E sector in the Greater London area will be generating between 10 and 11 million tonnes of material⁴ (per annum).
- 2.1.5 Regional estimated future CD&E waste arisings for Greater London are shown in the table below.

Projected Material Quantities from CD&E Sector (tonnes per annum) ⁴				
2010	2015	2020	2025	2030
9 753 000	10 203 000	10 512 000	10 736 000	11 029 000

 Table 2-1 Projected CD&E Material Quantities for Greater London area

³ Defra (2015), 'UK Statistics on Waste – 2010 to 2012'

⁴ Greater London Authority (2011) 'Future Waste Arisings in London 2010-2031: Summary Note.'

Silvertown Tunnel Scheme Projected Material Arisings

- 2.1.6 The Scheme is expected to generate approximately 1,194,000 tonnes of CD&E materials throughout the construction phase, more specifically 108,000 tonnes of non-excavated materials (arisings from construction and demolition activities) and 1,086,000 tonnes of excavated material, including material dredged from the River Thames to facilitate the construction and operation of the temporary jetty and NAABSA at the Silvertown site (if required). These material quantities have been extracted from the Scheme's up to date Cost Estimates and Site Waste Management Plan (SWMP), which has been based on the Scheme's reference design for DCO submission.
- 2.1.7 The SWMP is a live (working) document which requires the relevant sections to be completed at different stages throughout the lifespan of the Scheme. The plan is used to record the management of excavated materials and waste generated. Final completion of the SWMP should occur when the construction phase is complete. Therefore, at any point during Scheme construction, sections of the SWMP will be incomplete and material types and quantities will be updated.

3 WASTE HIERARCHY

3.1 Waste Hierarchy Overview

- 3.1.1 The internationally recognised waste hierarchy, shown in the figure below is set within the revised EU Waste Framework Directive 2008/98/EC. The waste hierarchy is a widely adopted benchmark for assessing a project or an activity's environmental performance, where different waste management options and processes are prioritised into the most and least environmentally favourable alternatives. The CD&E Materials Commitments for the Scheme are based around the principles of the waste hierarchy, which prioritises waste prevention, preparing for re-use, recycling and recovery over disposal to landfill.
- 3.1.2 Effective management of material usage and waste, actioned through following these principles, is crucial to improving resource management and minimising environmental impact.



Figure 3-1 Waste Hierarchy

3.2 Definitions

- 3.2.1 Each waste management option illustrated in the waste hierarchy is described below in the order of most to least preferred with regards to its environmental impact. The revised EU Waste Framework Directive defines the waste management options from the waste hierarchy as:
 - Prevention: "Measures taken before a substance, material or product has become waste that reduce: (a) the quantity of waste, including through the
re-use of products or the extension of the life span of products; (b) the adverse impacts of the generated waste on the environment and human health; or (c) the content of harmful substances in materials and products."

- Preparing for Re-use: "Checking, cleaning or repairing recovery operations, by which products or components of products that have become waste are prepared so that they can be re-used without any other pre-processing."
- Recycling: "Any recovery operation by which waste materials are reprocessed into products, materials or substances whether for the original or other purposes. It includes the reprocessing of organic material but does not include energy recovery and the reprocessing into materials that are to be used as fuels or for backfilling operations."
- Other Recovery: "Any operation the principal result of which is waste serving a useful purpose by replacing other materials which would otherwise have been used to fulfil a particular function, or waste being prepared to fulfil that function, in the plant or in the wider economy. Annex Il sets out a non-exhaustive list of recovery operations."
- Disposal: "Any operation which is not recovery even where the operation has as a secondary consequence the reclamation of substances or energy."
- 3.2.2 The waste hierarchy promotes the efficient use of material resources, reducing the amount of waste produced and reducing as far as reasonably practicable the amount of waste that falls under disposal.
- 3.2.3 In line with the waste hierarchy, disposal to landfill should only be a last resort, due to the range of potential adverse environmental effects and in order to conserve existing landfill capacity.

4 CONSTRUCTION, DEMOLITION AND EXCAVATED MATERIALS COMMITMENTS

4.1 Overview

- 4.1.1 Integral to this document are TfL's commitments, which demonstrate how waste materials are to be managed. These commitments will form part of the contract specification. The commitments also provide a means by which the performance of the contract will be assessed. The commitments presented in this document are set to be challenging, but achievable, and in line with similar infrastructure schemes and industry benchmarks. They aim to provide a clear message to stakeholders of what the Scheme sets out to achieve with regards to the management and end-use of CD&E materials.
- 4.1.2 The CD&E material commitments are based on a review of applicable regulations, policies, sustainability strategies and other similar sized infrastructure Schemes in London.
- 4.1.3 TfL is committed to achieve the following for the Scheme:
 - 80% (by weight) of CD&E materials to be re-used on site or removed from site for beneficial use with an aspiration to reach 95% (by weight).
 - Safely manage CD&E materials in order to minimise their impact on the environment and communities.
 - Follow the self-sufficiency and the proximity principles through the local management and end-use of CD&E materials.
- 4.1.4 Achievement of the objectives would be assessed, by TfL using specific reporting targets.
- 4.1.5 The commitments for CD&E materials are discussed in further detail below. Followed by further background and details of how the commitments can be achieved.

4.2 CD&E Materials Commitment

80% (by weight) of CD&E materials to be re-used on site or removed from site for beneficial use with an aspiration to reach 95% (by weight).

- 4.2.1 The commitment follows the principles set forth within the waste hierarchy. For the Scheme, beneficial use is an activity that meets one of the following criteria:
 - Ecological benefit or land reinstatement/landscaping: The activity will assist in ecological benefit and/or help to facilitate an approved change/alteration in land use or form.
 - Works (linked to a consented planning activity or permit) that aims to restore, enhance or be part of a land management scheme i.e. landfill or quarry.
 - Reduce the requirement for alternative material (waste or not) to be used for the purposes of any such Scheme.
- 4.2.2 In addition to the above, the beneficial use of material will pose no harm to the environment or human health.
- 4.2.3 In support of this commitment a Receptor Site Assessment (RSA) is currently underway and will be completed prior to the commencement of construction. The purpose of the RSA is to demonstrate that the commitment is feasible and achievable, and a transparent process and methodology in the evaluation of potential sites that may receive material / waste has been followed. The methodology allows both TfL and any third parties appointed by TfL to use this RSA process to determine whether a facility is suitable to receive materials / wastes generated by the Scheme in order that this commitment can be met. The methodology used in the Receptor Site Assessment has been included within the Environmental Statement (Document Reference 6.1).
- 4.2.4 The commitment is based on the anticipated quantities and composition of material. For materials that cannot be re-used, recycled or recovered by the Scheme, these would need to be transported to an externally managed and appropriate facility.

- 4.2.5 TfL's Corporate Environmental Framework states that "We will continue to minimise generation of waste as far as possible." ⁵ As per TfL's commitment, the prevention of waste generation from the Scheme will be prioritised through the development of an efficient design, sustainable procurement of materials and seeking opportunities through the waste hierarchy.
- 4.2.6 The waste hierarchy prioritises prevention as the preferred waste management option (see Section 3). Even though prevention of material generation will be prioritised, there will be unavoidable generation of CD&E material during the construction phase of the Scheme.
- 4.2.7 At a minimum, 80% (by weight) of the Scheme's CD&E materials equates to approximately 940,000 tonnes (as per Section 2).
- 4.2.8 In addition, there is a Scheme aspiration for 95% (by weight) of CD&E materials to be made available to other schemes where the material can be used for beneficial use (where possible and/ or technically feasible). This aspiration is in line with The London Plan⁶ which states that "95% recycling/ re-use of construction, demolition and excavation waste by 2020."
- 4.2.9 As the design of the Scheme progresses, materials will be identified for which a higher recycling and re-use rate may be achieved, if technically and economically feasible.
- 4.2.10 In line with TfL's Corporate Environmental Framework⁵ adherence to this commitment, and the aspiration of 95% may help to create innovative solutions for reducing wastes or its beneficial use.
- 4.2.11 As per the Duty of Care obligations (the (Environmental Protection (Duty of Care) Regulations 1991), all CD&E materials will be tracked from their origin to their final destination, including information on the quantities, classification and end destination.
- 4.2.12 Diversion from the disposal of waste will be achieved using one or more of the following industry best practices, as appropriate for the type of waste material produced:
 - Site Waste Management Plans (SWMPs): Adopting this best practice allows for the efficient management of materials and waste generated and transported. The SWMP helps to highlight materials and waste streams that could be more efficiently managed, for example redirecting recyclable/

⁵ Mayor of London (2014) 'TfL Corporate Environment Framework.'

recovered resources back to the manufacturing process, redirect usable materials to appropriate sites, and identifying materials and their adequate treatment to minimise impacts to the environment. The SWMPs is a live document which is regularly updated by the contractors of the Scheme. It helps to plan for the management of materials generated by CD&E activities as well as for recycling, re-use and recovery. Further details about the Scheme's SWMP can be found in the Site Waste Management (Appendix to Document Reference 6.10).

- 4.2.13 The SWMP will be the principal method of monitoring progress against the commitment. The details of which should be reported in line with the Code of Construction Practice (CoCP) (Document Reference 6.10).
 - Receptor Site Assessment (for Excavated Materials) (included within the Chapter 13 of the Environmental Statement (Document Reference 6.1): used to identify and assess the receptor sites which will be suitable for receiving and managing the excavated materials generated from the Scheme. The criteria used in the assessment have been developed to ensure that the sites conform to the vision and commitments detailed in this document. The RSA takes a range of sites (facilities) and provides a clear scoring system to the capabilities and functions of each site. Sites that are considered to recycle, re-use etc. score higher than those who dispose (landfill) material.

Monitoring

- 4.2.14 Progress in achieving the commitment will be monitored through population of the SWMP with the 'actuals' in terms of volume of waste produced and its end destination regularly recorded within the SWMP.
- 4.2.15 At defined periods the SWMP can be presented to TfL to evidence the Contractors progress. The issue of the SWMP will help to track progress in achieving (or exceeding) the commitment.
- 4.2.16 Should the information contained within the SWMP (and any other information the Contractor may present) indicate that the commitment would not be met then a 'Remedial Action Plan' may need to be prepared. This plan would be prepared by the Contractor following liaison with TfL.

4.3 CD&E Environment and Communities Commitment

Safely manage CD&E materials in order to minimise their impact on the environment and communities.

4.3.1 The safe management of CD&E materials from the Scheme will be implemented through a range of industry best practices in order to minimise the materials' impact on the environment and communities.

Measuring Safe Management of Materials

- 4.3.2 The safe management of CD&E materials will be achieved by using one or more of the following best industry practices along with liaison with stakeholders.
 - Code of Construction Practice (CoCP): The purpose of the CoCP is to set a framework to control possible impacts arising from the construction of a Scheme. The CoCP covers environmental, public health and safety aspects that may affect the interests of local residents, businesses, the general public and the surroundings in the vicinity of the Scheme. The control measures set out in the CoCP are based on the findings set out in the Environmental Statement.
 - The CL:AIRE Definition of Waste Development Industry Code of Practice (DoW CoP): demonstrating 'safe' in terms of limiting risks the environment and human health, the DoW CoP is used (where applicable) in order to maximise the re-use of excavated materials in a safe and efficient manner. The objective of the DoW CoP is to implement good industry practice (on a case by case basis) for classifying excavated materials which reached 'end of waste' status, as per the revised EU Waste Framework Directive 2008/98/EC. DoW CoP also provides a detailed step by step process to demonstrate that the requirements set within the DoW CoP has been followed and encourages the use of industry best practices, such as using a Materials Management Plan (MMP) for the materials generated on site, basing the MMP on a risk assessment, and demonstrating the use of the MMP in a Verification Report. Should the DoW CoP be used, the CoCP imposes a requirement upon the contractor to produce an MMP in liaison with relevant stakeholders.

• Transportable Moisture Limit (TML): defined in The Merchant Shipping (Carriage of Cargoes) Regulations 1999 as the 9/10^{ths} (or 90%) of the flow moisture point. This limit sets the standard for accepting cargoes which may liquefy for marine transport. Cargoes with a moisture content above the TML will not be transported by barge unless *"appropriate safety arrangements are made to the satisfaction of the Certifying Authority to ensure adequate stability in the case of cargo shifting, and the ship has adequate structural integrity."* Details of the commitments relating to the river transport of CD&E materials are set out within the CoCP.

4.4 Self Sufficiency Commitment

Follow the self-sufficiency and the proximity principles through the local management and end-use of CD&E materials.

- 4.4.1 The self-sufficiency and proximity principles are established in the revised EU Waste Framework Directive 2008/98/EC, which states that waste should be *"recovered in one of the nearest appropriate installations, by means of the most appropriate methods and technologies, in order to ensure a high level of protection of the environment and public health."*
- 4.4.2 These principles are also shared in the Mayor of London's 2015 London Plan which state that "London should manage as much of the capital's waste within its boundaries as practicable, enabling London and Londoners to receive environmental and economic benefits from its management."⁶

Measuring Self-Sufficiency and Proximity

4.4.3 Transport of CD&E materials to the nearest receptor sites will be a key aspect of meeting this commitment and forms part of the RSA methodology which the Contractor will be required to comply with as part of the CoCP (Document Reference 6.10). The nearest receptor sites which are the most suitable for the management and end-use of the excavated materials will be identified through the RSA. Based on the RSA methodology, receptor sites that are closer to the scheme would be attributed a higher score and would therefore rank more favourably than others at a greater distance.

⁶ Mayor of London (2015) 'The London Plan'

- 4.4.4 The RSA will be used to identify and assess the receptor sites which will be suitable for receiving and managing the excavated materials generated from the Scheme. The criteria used in the assessment have been developed to ensure that the sites conform to the vision and commitments detailed in this document
- 4.4.5 This criteria helps ensure the commitment to self-sufficiency and proximity for the management and end-use of CD&E materials is being met for the Scheme.
- 4.4.6 Permit or exemption details of all sites that have the potential to receive the excavated materials generated during the Scheme will be checked as part of the RSA methodology document to ensure all sites are legally complaint. It is recommended that such compliance is verified prior to any materials being sent to the receptor sites.

THIS PAGE IS INTENTIONALLY BEEN LEFT BLANK

5 CONCLUSION

- 5.1.1 The CD&E Materials Commitments document identifies the commitments for the Scheme in order to minimise impact on the environment and communities.
- 5.1.2 Through a review of applicable regulations, policies, sustainability strategies and other similar sized infrastructure projects in London, a list of commitments have been developed commensurate with TfL and the Mayor of London's sustainability visions.
- 5.1.3 Commitments for the Scheme are:
 - 80% (by weight) of CD&E materials to be re-used on site or removed from site for beneficial use with an aspiration to reach 95% (by weight).
 - Safely manage CD&E materials in order to minimise their impact on the environment and communities.
 - Follow the self-sufficiency and the proximity principles through the local management and end-use of CD&E materials.
- 5.1.4 The performance of the Scheme in line with the commitments will be measured as outlined in the CoCP. They will also prioritise maximising re-use, recycling and recovery in order to help drive innovative solutions to finding beneficial uses for the materials and reduce their impact on the environment and communities.

THIS PAGE IS INTENTIONALLY BEEN LEFT BLANK

Appendix D. Receptor Site Assessment

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK



Methodology

Receptor Site Assessment (for Excavated Materials)

TR010021

October 2016



MAYOR OF LONDON

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

Silvertown Tunnel

TR010021 Methodology Receptor Site Assessment (for Excavated Materials)

Planning Act 2008

The Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Document Reference: 6.10 Internal Code: ST150030-PLN-ZZZ-ZZ-DSD-0071

Regulation Number: 5(2)(Q) Author: Transport for London

Rev.	Date	Approved By	Signature	Description
0	29/04/16	David Rowe (TfL Lead Sponsor)		
1	01/11/2016	David Rowe (TfL Lead Sponsor)		

Page 3 of 54

Contents

List of	Abbreviations	6
Glossa	iry of Terms	7
1.		12
1.1	Overview	12
1.2	Aim	13
1.3	Principles	14
2.	METHODOLOGY	20
2.1	Overview	20
2.2	Assumptions	20
2.3	Methodology Overview	20
2.4	Step 1: Develop Long List	23
2.5	Step 2: Preliminary Viability Assessment	23
2.6	Step 3: Detailed Options Assessment	29
2.7	Preferred List	41
2.8	Reserve List	42
2.9	Use of the Lists	42
3.	CONCLUSION	44
Appen	dix A. Questionnaires	46

List of Tables

Table 1-1 The Principles used in the Receptor Site Assessment	. 15
Table 2-1 Permit/ Exemption Validity Sub-Criteria	. 24
Table 2-2 Accessibility Evaluation Sub-Criteria	. 26
Table 2-3 Scoring Indicators and Evaluation Criteria for the Detailed OptionsAssessment	. 31
Table 2-4 Scoring Guidance	. 39
Table 2-5 Carbon Values used to Calculate Carbon Footprint	. 40

List of Figures

Figure 2-1: Overview of Methodology	. 22
Figure 2-2: Overview of Preliminary Viability Assessment	. 28

List of Abbreviations

AQMA	Air Quality Management Areas
CCS	Compliance classification scheme
CD&E	Construction, Demolition and Excavation
CL:AIRE	Contaminated Land: Applications in Real Environments
DCO	Development Consent Order
Defra	Department of Environment, Food and Rural Affairs
EMOA	Excavated Materials Options Assessment
FOI	Freedom of Information
GHG	Greenhouse gases
HMRC	HM Revenue and Customs
HSE	Health, safety and environmental
km	Kilometres
m	Metres
N / A	Not applicable
Opra	Operational Risk Appraisal
РМ	Particulate matter
RSA	Receptor Site Assessment (for Excavated Materials)

Document Reference Number: 6.10

Glossary of Terms

Contractor	Anyone who directly employs or engages construction workers or manages construction work. Contractors include sub-contractors, any individual self-employed worker or a business that carries out, manages or controls construction work.
Detailed Options Assessment	A phase of the Receptor Site Assessment which assesses the receptor sites on the short list in more detail. The output is a preferred list of suitable receptor sites that meet the principles as outlined in the methodology. Each receptor site is scored against the criteria in each principle, with successful sites being included on the preferred list and remaining sites in the reserve list.
Excavated Material	Ground or other material removed during a construction process, usually by mechanical means.
Proximity Principle	Managing, treating and / or disposing of materials close to the material source in order to reduce environmental and costs impacts.
Receptor Site Assessment	A phased methodology and evaluation criteria to assess suitable sites to receive, treat and/or dispose of the excavated materials from the Scheme. The methodology provides a transparent process to stakeholders to ensure that the excavated material will be managed and treated with the least impact to the environment and communities.
Development Consent Order	The means of obtaining permission for developments categorised as Nationally Significant Infrastructure Projects. This includes energy, transport, water and waste projects.
List of Waste	A series of waste classification codes for all hazardous and non-hazardous wastes. List of Waste codes are often referred to as European Waste Catalogue codes.
Preliminary Viability Assessment	A phase of the Receptor Site Assessment which assesses the long list against high level criteria in order to identify a short list of potentially suitable receptor sites

	for the excavated materials from the Scheme
Reserve List	Receptor sites that fail to successfully pass the Detailed Options Assessment but may become viable/ suitable prior to construction starting.
The Scheme	The construction of a new bored tunnel under the River Thames between the Greenwich Peninsula and Silvertown, as well as necessary alterations to the connecting road network and the introduction of user charging at both Silvertown and Blackwall tunnels.
	A London government body responsible for most aspects of the transport system in Greater London. Its role is to implement transport strategy and to manage transport services across London.
Transport for London (TfL)	These services include: buses, the Underground network, Docklands Light Railway, Overground and Trams. TfL also runs Santander Cycles, London River Services, Victoria Coach Station and the Emirates Air Line.
	As well as controlling a 580km network of main roads and the city's 6,000 traffic lights, TfL regulates London's private hire vehicles and the Congestion Charge scheme.

SUMMARY

- S.1.1 The Silvertown Tunnel scheme (herein after referred to as 'the Scheme') is a new road tunnel under the River Thames connecting the Greenwich Peninsula and Silvertown. The Scheme involves the construction of a twin bore road tunnel providing a new connection between the A102 Blackwall Tunnel Approach on Greenwich Peninsula (Royal Borough of Greenwich) and the Tidal Basin roundabout junction on the A1020 Lower Lea Crossing/Silvertown Way (London Borough of Newham). The Silvertown Tunnel would be approximately 1.4km long and would be able to accommodate large vehicles including double-deck buses. The Boord Street footbridge over the A102 would be replaced with a pedestrian and cycle bridge.
- S.1.2 The Scheme aims to minimise disposal to landfill and promote safe management of materials at local facilities. In order to achieve this, a Receptor Site Assessment (for Excavated Materials) will be undertaken in order to identify the most suitable sites to receive, treat and/or dispose of the excavated material.
- S.1.3 The Receptor Site Assessment (for Excavated Materials) follows a standardised methodology and evaluation criteria which meets the vision and commitments set forth within the Scheme's Construction, Demolition and Excavated Materials Commitments document (Appendix to Document Reference 6.10). It also provides a transparent process to stakeholders in order to ensure that the excavated material will be managed and treated with the least impact to the environment and communities.
- S.1.4 A step by step process will be used in order to develop a preferred list of receptor sites which are the most suitable for receiving, treating and/or disposing of the incoming material from the Scheme.
- S.1.5 The Receptor Site Assessment consists of four key steps:
 - 1. The development of a long list of receptor sites which encompasses all possible receptor sites for the excavated materials from the Scheme;
 - 2. A Preliminary Viability Assessment to assess the long list against high level criteria in order to identify a short list of potentially suitable receptor sites for the excavated materials from the Scheme;
 - A Detailed Options Assessment to assess the short list against a more detailed set of criteria in order to identify a preferred list of

suitable receptor sites for the Scheme's excavated material. A reserve list is generated from sites that fail to successfully pass through the Detailed Options Assessment but may become viable/ suitable prior to construction; and

- 4. The preferred list is used by contractors tendering for the Scheme.
- S.1.6 Transport for London (TfL) aims to minimise disposal by only using receptor sites which meet or exceed the criteria presented in The Receptor Site Assessment (for Excavated Materials) for the Scheme.
- S.1.7 This document is intended to support the application for Development Consent Order for the Scheme. The Receptor Site Assessment (for Excavated Materials) methodology will be completed prior to the commencement of construction.

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

1. INTRODUCTION

1.1 Overview

- 1.1.1 The Silvertown Tunnel scheme (herein after referred to as 'the Scheme') is a new road tunnel under the River Thames connecting the Greenwich Peninsula and Silvertown. The Scheme involves the construction of a twin bore road tunnel providing a new connection between the A102 Blackwall Tunnel Approach on Greenwich Peninsula (Royal Borough of Greenwich) and the Tidal Basin roundabout junction on the A1020 Lower Lea Crossing/Silvertown Way (London Borough of Newham). The Silvertown Tunnel would be approximately 1.4km long and would be able to accommodate large vehicles including double-deck buses. The Boord Street footbridge over the A102 would be replaced with a pedestrian and cycle bridge.
- 1.1.2 New portal buildings would be located close to each tunnel portal to house the plant and equipment necessary to operate the tunnel.
- 1.1.3 The introduction of free-flow user charging on both the Blackwall and Silvertown Tunnels would play a fundamental part in managing traffic demand and would help to support the funding of the construction and operation of the Silvertown Tunnel.
- 1.1.4 The design of the Silvertown Tunnel would include a dedicated bus/coach and HGV lane, which would provide opportunities for TfL to provide additional cross-river bus routes.
- 1.1.5 Main construction works would likely commence in 2018 and would last approximately 4 years with the new tunnel opening in 2022/23. The main construction compound would be located at Silvertown, utilising the existing barge facilities at Thames Wharf along with a new temporary jetty for the removal of spoil and delivery of materials by river. A secondary site compound would be located adjacent to the alignment of the proposed cut and cover tunnel on the Greenwich Peninsula.

1.1.6 A large portion of the material that will be generated during the construction phase of the Scheme will be excavated material. The Scheme aims to minimise disposal to landfill and promote the safe management of materials at local facilities. As such a Receptor Site Assessment (for Excavated Materials) (herein after referred to as 'RSA') has been developed in order to identify the most suitable sites to receive, treat and/or dispose of the excavated material.

1.2 Aim

- 1.2.1 The aim of the RSA is to produce a list of preferred receptor sites for the excavated material produced by the Scheme.
- 1.2.2 The RSA provides a standard methodology and evaluation criteria which should be used in conjunction with the requirements and commitments set forth within the CD&E Materials Commitments document (Appendix to Document Reference 6.10). The RSA also provides a transparent process to stakeholders to ensure that the excavated material will be managed and treated commensurate with the least impact to the environment and communities.
- 1.2.3 The RSA uses a phased approach to develop a list of receptor sites which may be suitable to receive, treat and/or dispose of the CD&E materials from the Scheme. The approach is as follows:
- 1. The development of a long list of sites which encompasses all possible receptor sites for the excavated materials from the Scheme;
- 2. A Preliminary Viability Assessment to assess the long list against high level criteria in order to identify a short list of potentially suitable receptor sites for excavated materials;
- 3. A Detailed Options Assessment to assess the short list against a more detailed set of criteria in order to identify a preferred list of suitable receptor sites for the Scheme's excavated material. A reserve list is generated from sites that fail to successfully pass through the Detailed Options Assessment but may become viable / suitable prior to construction starting; and
- 1.2.4 The preferred list is taken forward and used by contractors tendering for the Scheme. Further to the Code of Construction

Practice, TfL will require the use of receptor sites which meet or exceed the criteria presented in this document.

- 1.2.5 In the instance that suitable sites on the preferred and reserve list are not available, or materials cannot be placed at a suitable facility either on site (subject to necessary consents and licences) or on these lists, or they do not allow for the Contractor to meet other commitments outlined by TfL, Contractors should assess receptor sites beyond the 100km radius criteria (including transfrontier sites) using the RSA methodology. For transfrontier sites, the effect of local legislation should be taken into account when applying the RSA methodology and evaluation criteria.
- 1.2.6 It is important to note that although this RSA relates to excavated material from the Scheme only, this methodology can also be applied (if necessary) to identify receptor sites for other materials, such as construction and demolition materials.
- 1.2.7 The RSA methodology identifies available receptor sites of potential suitability at the time of assessment and it is likely that other sites may become available before construction commences. Sites that fail to successfully pass through the Detailed Options Assessment but could potentially pass in the future (e.g. due to pending permit applications or land acquisition) will be placed onto the reserve list.
- 1.2.8 This document is intended to support the application for Development Consent Order (DCO) for the Scheme.

1.3 Principles

- 1.3.1 The seventeen principles presented below (see Table 1-1) provide the overarching criteria used to evaluate the receptor sites. They have been divided into four categories which address impacts to the environment and the public, promote sustainable transport options and evaluate the capability and operations of the receptor sites.
- 1.3.2 The methodology used to address these principles is described in more detail in Section 2.

Category	Principle	
	1. To follow the waste hierarchy.	
	2. To address and reduce contribution to climate change.	
	3. To reduce impact on air quality and noise.	
Environment	4. To protect the quality of water.	
	5. To protect and enhance biodiversity.	
	6. To ensure efficient use of land and resources.	
	7. To protect cultural heritage.	
	8. To protect local public amenities.	
Public	9. To reduce visual impact on surrounding areas.	
	10. To provide opportunities for a diverse workforce.	
Transport	11. To promote sustainable transportation.	
Transport	12. To manage materials locally, as per the proximity principle.	
	13. To identify receptor sites with suitable capacity and operational capability.	
	14. To use receptor sites with appropriate permits or exemptions.	
Operations	15. To ensure health and safety guidelines are followed.	
	16. To ensure legal compliance.	
	17. To develop solutions which minimise costs.	

Table 1-1 The Principles used in the Receptor Site Assessment

1.3.3 The principles are discussed in more detail below.

Environment

1.3.4 The principles included within the environment category address the impacts that the receipt (and treatment if required) of the incoming material to a waste facility may have on the environment and communities.

To follow the waste hierarchy

1.3.5 Receptor sites are assessed based on the waste hierarchy which prioritises prevention of waste followed by re-use, recycle, recover and disposal. In this case, waste prevention does not apply since the waste is already generated by the time it reaches the receptor site. Preference is given to sites which re-use, recycle and recover materials rather than those which dispose of the materials.

To address and reduce contribution to climate change

1.3.6 Receptor sites are assessed based on the impact the receipt/treatment/disposal of the incoming material has on greenhouse gas emissions and climate change. This criteria includes the greenhouse gas emissions from the site operations as well as the transport of the material from the Scheme to the receptor site.

To reduce impact on air quality and noise

1.3.7 Receptor sites are assessed based on the impact the receipt/treatment/disposal of the incoming material has on air emissions (e.g. dust, pollutants), noise and vibration levels which may cause harm and/ or nuisance to the surrounding areas. This includes a review of the site's air quality management plan.

To protect the quality of water

1.3.8 Receptor sites are assessed based on the impact the receipt/treatment/disposal of the incoming material has on water quality. This includes an evaluation on the site's proximity to

surface water bodies and groundwater, contamination pathways and risks to the receptors in the surrounding areas.

To protect and enhance biodiversity

1.3.9 Receptor sites are assessed based on the impact the receipt/treatment/disposal of the incoming material has on local habitats in the short and long term. This includes an evaluation on the site's proximity and impact on international, national and local nature conservation sites.

To ensure efficient use of land and resources

1.3.10 Receptor sites are assessed based on the impact the receipt/treatment/disposal of the incoming material has on the land uptake or footprint of the receptor site and surrounding areas.

To protect cultural heritage

1.3.11 Receptor sites are assessed based on the impact the receipt/treatment/disposal of the incoming material has on designated cultural heritage sites. This includes an evaluation on the site's proximity and impact on international, national and local cultural heritage sites.

Public

1.3.12 The principles included within the public category address the impacts that the receipt/treatment/disposal of the incoming material may have on public amenities, heritage, land and employment.

To protect local public amenities

1.3.13 Receptor sites are assessed based on the impact the receiving material and the site's operations will have on public amenities. Public amenities are defined as complimentary, benefits, resources and facilities provided to the general public (e.g. public toilets, recreational areas, parks, open spaces.)

To reduce visual impact of surrounding areas

1.3.14 Receptor sites are assessed based on the impact the receipt/treatment/disposal of the incoming material has on changes in the fabric, character and quality of the landscapes, as well the availability of views.

To provide opportunities for a diverse workforce

1.3.15 Receptor sites are assessed based on the impact the receipt/treatment/disposal of the incoming material has on the number of jobs at the receptor site. This includes an evaluation on the site's commitment to employing a diverse workforce.

Transport

1.3.16 The principles included within the transport category address the extent to which the transportation of the receipt/treatment/disposal of the incoming material promotes sustainability.

To promote sustainable transportation

1.3.17 Receptor sites are assessed based on their ability to receive material from various modes of transportation, including rail, river and road. Preference is given to receptor sites which have marine and rail access compared to road access only.

To manage materials locally, as per the proximity principle

1.3.18 Receptor sites are assessed based on their distance to the Scheme site. The proximity principle is the concept of treating and/ or disposing of materials at the closest point to the point of generation in order to minimise cost, carbon and impact on the environmental and communities.

Operations

1.3.19 The principles included within the operations category address the operational capability of the receptor sites to receive, treat and/or dispose of the incoming material.

To identify receptor sites with suitable capacity and operational capability

1.3.20 Receptor sites are assessed on the available capacity to receive, treat and/or dispose of the incoming material in the short and long term. This includes available storage space and equipment capability to handle the required volumes of incoming material.

To use receptor sites with appropriate permits or exemptions

1.3.21 Sites are assessed on having the necessary documentation to receive, treat and/or dispose of the incoming material. This includes (but is not limited to) having adequate permits or exemptions.

To ensure health and safety guidelines are followed

1.3.22 Receptor sites are assessed on meeting health and safety good practices when receiving/treating/disposing of the incoming material.

To ensure legal compliance

1.3.23 Receptor sites are assessed on their compliance with applicable regulations and an evaluation of their history of enforcement notices/cautions issued by the Environment Agency and/or decisions made by the Magistrates or Crown Court.

To develop solutions which minimise costs

1.3.24 Receptor sites are assessed on their consideration of cost minimisation during the receipt/treatment/disposal of incoming material at the receptor site.

2. METHODOLOGY

2.1 Overview

2.1.1 The Receptor Site Assessment (for Excavated Materials) uses a standardised methodology and evaluation criteria which meets the vision and commitments set forth within the Scheme's Construction, Demolition and Excavated Materials Commitments document (Appendix to Document Reference 6.10). It also provides a transparent process to stakeholders in order to ensure that the excavated material will be received, treated and/or disposed of with the least impact to the environment and communities.

2.2 Assumptions

- **2.2.1** The following assumptions are made throughout the Preliminary Viability Assessment:
- Materials arising from the Scheme will be separated at source;
- As per the proximity principle, only receptor sites which are within 100 km radius of the Scheme will be evaluated, unless such sites are unavailable and a receptor site is available at a greater distance beyond 100 km;
- New receptor sites, which may be available after the assessment is complete, are not included;
- Material quantities and composition may change following the development of the Scheme's detailed design completion and contract appointment; and
- Scheme timescales may change following the development of the Scheme's detailed design, completion and contract appointment.

2.3 Methodology Overview

- 2.3.1 Figure 2-1 provides an overview of the steps which will be followed in the RSA's methodology. The key steps which will be undertaken include:
- **Develop long list:** Develop a long list of sites which will encompass all possible receptor sites for the excavated materials from the Scheme;
- **Preliminary Viability Assessment:** This stage assesses the long list against high level criteria in order to identify a short list of potentially suitable receptor sites. The assessment consists of a series of basic requirements which may determine the operational suitability of the receptor site. Sites that pass through this assessment are included on the short list;
- Detailed Options Assessment: Receptor sites on the short list are taken through a more detailed assessment in order to identify a preferred list of suitable receptor sites. The assessment will use the principles as outlined in Section 1.3 as criteria to determine the receptor site's suitability to receive, treat and/or dispose of material and any impacts to the facilities environment and communities. Each receptor site will be scored against the criteria, with successful sites being included on the preferred list; and
- **Preferred list and reserve list:** The preferred list of receptor sites is used by contractors tendering for the Scheme. A reserve list of receptor sites will be generated from sites that fail to successfully pass the Detailed Options Assessment but may become viable/ suitable prior to construction starting.



2.4 Step 1: Develop Long List

- 2.4.1 The first step of the assessment aimed to develop a long list of sites which will encompass all possible receptor sites that may receive, treat and/or dispose of the excavated materials from the Scheme (i.e. sites that either hold an environmental permit or registered with an exemption).
- 2.4.2 In order to conform to the proximity principle, a criteria of 100 km radius from the Scheme was used to identify a list of potential receptor sites. This ensured that the search was focused on receptor sites within London, the east and the south east of England. No other pre-considerations relating to the suitability of the sites (cost, capacity, etc.) or environmental impacts were applied. For organisations which operate more than one potential receptor site, each site was recorded separately on the list.
- 2.4.3 The Environment Agency's list of permitted and exempt facilities¹ was used as the main source for the creation of the long list.
- 2.4.4 Sites which receive, treat and/or dispose of materials which were not relevant to the Scheme were also removed from the long list. Examples of receptor sites which were excluded from the long list include sites which process single waste streams not associated with 'soil and stones', such as electrical and electronic appliances, clinical wastes, and metals.

2.4.5

2.5 Step 2: Preliminary Viability Assessment

2.5.1 The second stage of the assessment aims to evaluate the operational capacity and high level viability of the receptor sites identified in the long list. This stage of the assessment aims to remove all sites that fail to comply with basic requirements of viable receptor sites.

¹ Environment Agency 'Waste Data Interrogatory'.
- 2.5.2 The criteria which will be used in the Preliminary Viability Assessment include:
- **Criteria 1: Receptor Site Capacity.** The capacity of the receptor site to receive, treat and/or dispose of volumes of excavated material in excess of 200,000 tonnes). Capacity refers to the total (originally permitted) capacity of the sites, without considering the available capacity (or void space) remaining at the sites. For permitted sites, capacity information is found in the environmental permit issued by the Environment Agency. For exempt sites, capacity information is dependent upon the exemption type.
- **Criteria 2: Validity of Permit/ Exemption**. The operator holds a valid permit or exemption, which allows them to receive, treat and/or dispose of the principal excavated material type(s), i.e. soil and stones. The sub-criteria below is used to assess a valid permit or exemption;

Permit/ Exemption	Sub-Criteria
	 An assessment of either the environmental permit or any recent variation will be undertaken to establish if the receptor site is licenced to receive, treat and/or dispose of the following: Inert and Non-hazardous (non-inert) soils and stones (List of Waste (LoW) Code: 17 05 04); Hazardous soils and stones (LoW Code: 17 05
Permitted sites	03*).
	These excavated materials streams are the two LoW codes identified in the Site Waste Management Plan (SWMP) (Appendix to Document Reference 6.10). Receptor sites which are permitted to receive, treat and/or dispose of either 17 05 04 only or 17 05 04 and 17 05 03* will be taken through to next stage of the assessment.
Exempt sites	A list of sites with exemptions can be obtained from the Environment Agency and filtered by the

Table 2-1 Permit/ Exemption Validity Sub-Criteria

Document Reference Number: 6.10

Permit/ Exemption	Sub-Criteria
	paragraph number, as outlined in Schedule 3 of The Environmental Permitting (England and Wales) Regulations 2010. Only sites with the following paragraph numbers are taken forward as they demonstrate receptor sites which could receive, treat and/or dispose of the materials likely to be produced from excavation (LoW codes 17 05 04 and 17 05 03*).
	 Disposal of Waste - D5 - Depositing waste samples for testing or analysis;
	 Treating Waste - T5 - Screening and blending waste;
	 Use of Waste - U1 - Use of waste in construction;
	 Use of Waste - U8 - Using waste for a specified purpose;
	 Use of Waste - U10 - Spreading waste to benefit agricultural land; and
	 Use of Waste - U11 - Spreading waste to benefit non-agricultural land.
	Storage of Waste exemptions were not considered as they apply to the temporary storage of waste only, and not at the point of production.

Note that any waste marked with an asterisk (*) is considered as a hazardous waste pursuant to Article 1(4), first indent, of Directive 91/689/EEC on hazardous waste, and subject to the provisions of that Directive unless Article 1(5) of that Directive applies.

• Criteria 3: Receptor Site Availability: The anticipated availability of the receptor site to receive, treat and/or dispose of the excavated material during the Scheme's construction (from year 2019 specifically).

If receptor sites state they plan to create additional space to remain operational in / beyond 2019, they remain on the short list.

- Criteria 4: Enforcement Notices, Court Cases and Cautions. Any enforcement notices, court cases (assumed to mean guilty by the Crown Prosecution Service in either a Magistrates Court or the Crown Court) and cautions will be gathered from the Environment Agency public register database². The following information will be gathered for each site: appropriate reference numbers, applicable date(s), a summary description and the act/action issued. This criteria would not preclude sites from being taken to the short list and the outcomes will be subject to further assessment during the Detailed Options Assessment.
- Criteria 5: Receptor Site Accessibility. The receptor site is accessible via road (i.e. lorry), water (i.e. barge) and / or rail (defined by distances to the nearest rail head/jetty defined by road travel). This criteria would not preclude a receptor site from being taken to the short list and the outcomes will be subject to further assessment during the Detailed Options Assessment.

Transportation	Criteria
Road	Accessible via road.
River	 Accessible if there is a jetty or the site has access to a jetty located within 3.2 km (2 miles) of the receptor site; Exclude brooks, culverts and streams that are less than 2 m wide or go underground to reach a river.
Rail	 Accessible if there is a rail head or the site has access to a rail head located within 3.2 km (2 miles) of the receptor site.

Table 2-2 Accessibili	y Evaluation	Sub-Criteria
-----------------------	--------------	--------------

² Environment Agency Public Register Database, Page 26 of 54

- 2.5.3 Information relating to each evaluation indicator will be obtained using one or more of the following:
- Web based sources and online literature;
- Direct contact (via telephone and email); and
- Information received directly from the Environment Agency (e.g. permit and licencing information).
- 2.5.4 Following completion of this stage all sites that pass the viability criteria listed above will be progressed onto the short list. Any sites that fail to comply with one or more of the criteria will not be progressed further unless they have the potential to become viable in the future. Future viability is measured by identifying sites which sites can meet the criteria in the future. Examples include sites which are expanding to increase capacity or sites which are in the process of applying for permits.
- 2.5.5 In addition, receptor sites for which evaluation information is unavailable will be progressed onto the short list, should the evaluation criteria not be available at the next stage the receptor sites will be removed. However, sites which refuse to provide information will not proceed to the next stage.



Figure 2-2: Overview of Preliminary Viability Assessment

Page 28 of 54

2.6 Step 3: Detailed Options Assessment

- 2.6.1 All sites on the short list are taken through the Detailed Options Assessment. This is a more detailed assessment of the receptor site's suitability in relation to the seventeen principles discussed previously.
- 2.6.2 At this stage, the principles are expanded further to include specific criteria used to score the receptor sites. The receptor sites on the short list are given a red, amber or green colour ('traffic light' grading system), along with an additional scoring grade for each evaluation criteria (see Table 2-1 for full list of criteria and scores). Those sites that perform best progress to the preferred list based on:
- The overall score awarded to the site;
- Number of red, amber and greens awarded to each site; and
- Professional judgement as to whether any individual red grade (or combination of grades) made a particular site unsuitable.
- 2.6.3 All of the receptor sites on the short list will be contacted again to obtain further details to inform the Detailed Options Assessment. More in depth telephone discussions and / or face to face interviews are held with the site operators using a standard questionnaire sheet as a guide (see Appendix A). The prompt sheet is used to ensure that all relevant topics are covered and that the sites can be effectively scored against each criteria with a detailed suite of supporting information.
- 2.6.4 To supplement and corroborate information received from the operators, a review of the site's planning consents and environmental permits/ exemptions is undertaken. Additionally, secondary sources are used to determine the environmental baseline of the site and obtain more detailed data on site operations. Secondary sources include (but are not limited to):

- Magic website³ (includes environmental and cultural designations, potential receptors, etc.);
- HM Revenue and Customs (HMRC) list of operators⁴;
- CL:AIRE Register of Materials list of donor and receiver sites⁵;
- Air Quality Management Areas (AQMA)⁶;
- 'What's in my Backyard' Environment Agency tool⁷ (includes flood zone areas, aquifers, permitted activities);
- Defra 'Greenhouse Gas Conversion Factor Repository⁸; and
- Other relevant documents, consents and plans.
- 2.6.5 Receptor sites that successfully pass through the Detailed Options Assessment by achieving the required score threshold will be placed on the preferred list. This list will contain sites that meet the Scheme's requirements with respect to the assessment principles and high level objectives detailed in the CD&E Materials Commitments document (Appendix to Document Reference 6.10). Sites that fail to meet the required score will not be included on the preferred list.

³ Magic, 'Map Application', http://www.magic.gov.uk/.

⁴ HM Revenue & Customs, 'Landfill Site Operators'.

⁵ CL:AIRE, 'Register of Materials'.

⁶ Defra, 'Air Quality Management Areas'.

⁷ Environment Agency, 'What's in my Backyard?'.

⁸ Defra, 'Greenhouse Gas Conversion Factor Repository'. Page 30 of 54

Document Reference Number: 6.10

Ŀ,	Dringinlag	Indicators	Evaluation Criteria	Evaluation Criteria				
Catego	Principles	Indicators	- 2	-1	0	+1	+ 2	
	1. To follow the waste hierarc hy	Extent to which end- use of the material follows the principles of the waste hierarchy.	More than 50% of the incoming material will be disposed in landfill.	The incoming material will be partly recovered and partly disposed in landfill.	The incoming material will be partly recovered and partly recycled at the receptor site.	The incoming material will be partly recycled and partly reused at the receptor site.	More than 50% of the incoming material will be beneficially reused at the receptor site, with no or minimal treatment.	
		Greenhouse gases (GHG) emitted through on site operations (including receipt, treatment and transportation of the incoming material).	The incoming material requires active treatment and onsite transportation and GHG emissions produced are expected to be high.	N / A	The incoming material requires minimal treatment or onsite transportation and GHG emissions produced are expected to be low.	N / A	N / A	
	2. To addres s and reduce contrib ution to climate change	Extent of additional vehicle movements required once incoming material has entered the receptor site.	A significant number of vehicle movements are required once the incoming material has entered the receptor site.	A low number of vehicle movements are required once the incoming material has entered the receptor site.	Minimal vehicle movements are required once the incoming material has entered the receptor site.	N / A	N / A	
Environment		Greenhouse gases (GHG) emitted during transportation of material to receptor site. This is calculated using standard carbon factors (see Section 2.6.11)	The carbon footprint of the material transport to the receptor site is in the highest 20% compared to other receptor sites.	N / A	The carbon footprint of the material transport to the receptor site is in the 20%-80% range compared to other receptor sites.	N/A	The carbon footprint of the material transport to the receptor site is in the lowest 20% compared to other receptor sites.	

Table 2-3 Scoring Indicators and Evaluation Criteria for the Detailed Options Assessment

Page 31 of 54

Document Reference: 6.10

cat ego ry	Prir	nciples	Indicators	Evaluation Criteria				
			Extent to which flood risk is altered by the receipt/treatment/dispo sal of incoming material at the receptor site, or in the local catchment.	The receipt/treatment/dispo sal of incoming material will significantly increase flood risk at the receptor site.	The receipt/treatment/dispo sal of incoming material will slightly increase flood risk at the receptor site.	The receipt/treatment/dispos al of incoming material will have minimal impact on the risk from flooding at the receptor site.	N / A	N / A
			Extent to which the receipt/treatment/dispo sal of the incoming material generates pollutants (within the permitted limits). This excludes transport to the site.	The receipt/treatment/dispo sal of the incoming material are expected to significantly increase pollutants within the permitted limits.	The receipt/treatment/dispo sal of the incoming material are expected to slightly increase pollutants within the permitted limits.	The receipt/treatment/dispos al of the incoming material are expected to have minimal impact on the generation of pollutants.	N / A	N / A
	3.	To reduce impact on air quality and noise	Extent to which air quality is impacted by the receipt/treatment/dispo sal of incoming material at the receptor site (excluding transport) in the short and long term.	The receipt/treatment/dispo sal of incoming material is expected to increase pollutants in the long term.	The receipt/treatment/dispo sal of incoming material is expected to increase pollutants in the short term.	The receipt/treatment/dispos al of the incoming material are expected to have minimal impact on the generation of pollutants.	N/A	N / A
			Extent to which the receipt/treatment/dispo sal of the incoming material generates noise (within the permitted limits). This excludes transport to the site.	The receipt/treatment/dispo sal of incoming material is expected to significantly increase noise levels at the receptor site (within permitted limits).	The receipt/treatment/dispo sal of incoming material is expected to slightly increase noise levels at the receptor site (within permitted limits).	The receipt/treatment/dispos al of the incoming material is expected to have minimal impact on noise levels at the receptor site.	N / A	N / A

Page 32 of 54

Document Reference Number: 6.10

ego ry	Prir	nciples	Indicators	Evaluation Criteria				
			Extent to which the receipt/treatment/dispo sal of the incoming material generates noise in the short and long term.	The receipt/treatment/dispo sal of incoming material is expected to increase noise levels in the long term.	The receipt/treatment/dispo sal of the incoming material is expected to increase noise levels in the short term.	The receipt/treatment/dispos al of the incoming material is expected to have minimal impacts on noise levels.	N / A	N / A
	4.	To protect the	Extent to which surface water quality is impacted by the material receipt/treatment/dispo sal at the receptor site.	The receipt/treatment/dispo sal of the incoming material at the receptor site will have adverse impacts on the local watercourses (within allowable limits).	The receipt/treatment/dispo sal of the incoming material at the receptor site will have slight adverse impacts on the local watercourses.	The receipt/treatment/dispos al of the incoming material at the receptor site will have minimal impact on the water quality of local watercourses.	N / A	N / A
		quality of water	Extent to which ground water quality is impacted by the material receipt/treatment/dispo sal at the receptor site.	The receipt/treatment/dispo sal of the incoming material at the receptor site will have adverse impacts on the underlying groundwater.	The receipt/treatment/dispo sal of the incoming material at the receptor site will have slight adverse impacts on the underlying groundwater.	The receipt/treatment/dispos al of the incoming material at the receptor site will have minimal impact on the water quality of the underlying groundwater.	N / A	N / A
	5.	To protect and enhanc e biodive rsity	Extent to which designated sites are impacted in the short and long term by the material receipt/treatment/dispo sal the receptor site.	The receipt/treatment/dispo sal of the incoming material is likely to have an adverse impact on biodiversity at the designated sites in the long term.	The receipt/treatment/dispo sal of the incoming material is likely to have an adverse impact on biodiversity at the designated sites in the short term.	The receipt/treatment/dispos al of the incoming material will have minimal impact on biodiversity at the designated sites.	The receipt/treatment/dispos al of the incoming material is likely to have a positive impact on biodiversity at the designated sites in the short term.	The receipt/treatment/dispos al of the incoming material is likely to have a positive impact on biodiversity at the designated sites in the long term.
	6.	To ensure efficien t use of	Extent to which incoming material replaces the need for virgin material use at	N / A	N / A	The incoming material will have minimal impact on the need for virgin material use at the	The incoming material will partly replace virgin material used at the receptor site.	The incoming material will directly replace virgin material used at the receptor site.

Page 33 of 54

Document Reference: 6.10

cat ego ry	Principles	Indicators	Evaluation Criteria				
	land	the receptor site.			receptor site.		
	es	Extent to which the efficiency of the receptor site's operations are impacted by the receipt/treatment/dispo sal of incoming material.	The receipt/treatment/dispo sal of the incoming material at the receptor site will significantly increase the use of water, electricity and/ or fuel.	The receipt/treatment/dispo sal of the incoming material at the receptor site will slightly increase the use of water, electricity and/ or fuel.	The receipt/treatment/dispos al of the incoming material at the receptor site will have a minimal impact on additional use of water, electricity and/ or fuel.	The receipt/treatment/dispos al of the incoming material at the receptor site will slightly reduce the use of water, electricity and/ or fuel.	The receipt/treatment/dispos al of the incoming material at the receptor site will significantly reduce the use of water, electricity and/ or fuel.
		Extent to which incoming material will affect land take (footprint) of the receptor site.	The receipt/treatment/dispo sal of the incoming material at the receptor site will significantly increase the overall footprint of the receptor site.	The receipt/treatment/dispo sal of the incoming material at the receptor site will slightly increase the overall footprint of the receptor site.	The receipt/treatment/dispos al of the incoming material at the receptor site will have minimal impact on the overall footprint of the receptor site	The receipt/treatment/dispos al of the incoming material at the receptor site will slightly decrease the overall footprint of the receptor site	The receipt/treatment/dispos al of the incoming material at the receptor site will significantly decrease the overall footprint of the receptor site
	7. To protec cultura heritag e	Extent to which designated sites are impacted in the short and long term by the material receipt/treatment/dispo sal the receptor site.	The receipt/treatment/dispo sal of the incoming material is likely to have an adverse impact on designated sites in the long term.	The receipt/treatment/dispo sal of the incoming material is likely to have an adverse impact on designated sites in the short term.	The receipt/treatment/dispos al of the incoming material is unlikely to have an impact on designated sites.	The receipt/treatment/dispos al of the incoming material is likely to have a positive impact on designated sites in the short term.	The receipt/treatment/dispos al of the incoming material is likely to have a positive impact on designated sites in the long term.
Public	8. To protec local public ameni es	Extent to which local public access are impacted by the material receipt/treatment/dispo sal at the receptor site.	The receipt/treatment/dispo sal of the incoming material at the receptor site will restrict public access in the long term.	The receipt/treatment/dispo sal of the incoming material at the receptor site will restrict public access in the short term.	The receipt/treatment/dispos al of the incoming material at the receptor site will not restrict local public access.	The receipt/treatment/dispos al of the incoming material at the receptor site will improve local public access in the short term.	The receipt/treatment/dispos al of the incoming material at the receptor site will improve local public access in the long term.

Page 34 of 54

Document Reference Number: 6.10

cat ego ry	Prin	nciples	Indicators	Evaluation Criteria				
	9.	To reduce visual impact of surroun ding areas	Extent to which the surrounding areas are visually impacted in the short and long term by the receipt/treatment/dispo sal of incoming material.	The receipt/treatment/dispo sal of incoming material at the receptor site will have adverse visual impacts on the surrounding area in the long term.	The receipt/treatment/dispo sal of incoming material at the receptor site will have adverse visual impacts on the surrounding area in the short term.	The receipt/treatment/dispos al of incoming material at the receptor site will have minimal visual impact on the surrounding area.	The receipt/treatment/dispos al of incoming material at the receptor site will have positive visual impacts on the surrounding area in the short term.	The receipt/treatment/dispos al of incoming material at the receptor site will have positive visual impacts on the surrounding area in the long term.
	10.	To provide opportu nities for a diverse workfor ce	Extent to which receipt/treatment/dispo sal of incoming material at the receptor site creates new job opportunities.	The receipt/treatment/dispo sal of incoming material contributes to a significant loss of jobs at the receptor site or local area.	The receipt/treatment/dispo sal of incoming material contributes to a slight loss of jobs at the receptor site or local area.	Receipt/treatment/dispo sal of incoming material will have a minimal impact on number of jobs at the receptor site or local area.	Receipt/treatment/dispo sal of incoming material will create a slight quantity of new job opportunities at the receptor site or local area.	Receipt/treatment/dispo sal of incoming material will create a significant quantity of new job opportunities at the receptor site or local area.
			Extent to which receipt/treatment/dispo sal of incoming material at the receptor site considers diversity of workforce.	N / A	N / A	N / A	N / A	A workforce diversity strategy is in place for the employment of a diverse work force at the receptor site.
	11.	To promot e sustain able transpo rtation	Extent to which sustainable forms of transport are available and considered for the receipt of incoming material at the receptor site.	The receptor site can only be accessed by road only.	N / A	The receptor site can be directly accessed by rail or marine transport but requires additional transport by road.	N / A	The receptor site can be directly accessed by rail or marine transport without the need for additional transport by road.
Transport	12.	To manag e materia I locally.	Average distance from the excavation site to the receptor site.	N / A	The receptor site is 75 km to 100 km from the excavation site.	The receptor site is 50 km to 75 km from excavation site.	The receptor site is 25 km to 50 km from the excavation site.	The receptor site is less than 25 km from the excavation site

Page 35 of 54

Silvertown Tunnel

Receptor Site Assessment (for Excavated Materials)

Document Reference: 6.10

cat ego ry	Princi	iples	Indicators	Evaluation Criteria				
	ati ptype	as per he proximi y principl e						
	13. T ic	To dentify	Likelihood of the receptor site to conform to the project's timescales.	The receptor site will be available to receive/treat/dispose of material during less than 25% of the timescale proposed.	The receptor site will be available to receive/treat/dispose of material during 25% to 50% of the timescale proposed.	The receptor site will be available to receive/treat/dispose of material during 50% to 75% of the timescales proposed.	The receptor site will be available to receive/treat/dispose of material during 75% to 100% of the timescale proposed.	The receptor site will be available to receive/treat/dispose of material during 100% of the timescale proposed.
	r s c y o	r sites with suitabl e capacit y and operati onal capabili ty	The receptor site's ability to receive/treat/dispose of various types of incoming material.	N / A	N/A	The receptor site is only able to receive/treat/dispose of one type of incoming material (based on geotechnical and chemical characteristics).	The receptor site is able to receive/treat/dispose of two to four types of incoming material (based on geotechnical and chemical characteristics).	The receptor site is able to receive/treat/dispose of five or more types of incoming material (based on geotechnical and chemical characteristics).
	ty		The receptor site's capacity to receive/treat/dispose of the volume of incoming material.	The receptor site has the capacity to receive/treat/dispose of less than 25% of the incoming material.	The receptor site has the capacity to receive/treat/dispose of 25% to 50% of the incoming material.	The receptor site has the capacity to receive/treat/dispose of 50% to 75% of the incoming material.	The receptor site has the capacity to receive/treat/dispose of 75% to 100% of the incoming material.	The receptor site has the capacity to receive/treat/dispose of 100% of the incoming material.
Operations	14. T r v a ri p	To use recepto sites with approp riate permits	The receptor site has the appropriate permits/ exemptions to receive/treat/dispose of and treat the incoming materials.	The receptor site does not have the appropriate permits/ exemptions to receive/treat/dispose of and treat the incoming materials. There are no plans to	The receptor site does not have the appropriate permits/ exemptions to receive/treat/dispose of and treat the incoming materials. The permit/ exemption	The receptor site has the appropriate permits/ exemptions to receive/treat/dispose of and treat the incoming materials.	N/A	N / A

Page 36 of 54

Document Reference Number: 6.10

cat ego ry	Prir	nciples	Indicators	Evaluation Criteria				
		exempt ions		apply for the appropriate permits/ exemptions.	application is currently being processed.			
	15.	To ensure health and safety guideli nes are followe d	Extent to which health, safety and environmental (HSE) best practices are followed.	The receptor site has had at least one recorded HSE accident resulting in injury or fatality in the last year.	The receptor site has had at least one recorded HSE accident resulting in injury or fatality in the last two to five years.	The receptor site has not had any recorded accidents in more than five years.	N/A	N / A
	16.	. To ensure legal compli ance	Extent to which the site operator has been issued with a ruling by UK court of law.	The site operator has been formally charged guilty by a UK court of law.	N / A	The site operator has never been found guilty by a UK court of law.	N / A	N / A.
			ensure legal Extent to which the operator is in risk of breaching environmental regulations.	The site operator has been issued with enforcement notice/caution from the Environment Agency.	N / A	The site operator or receptor site has not been issued with enforcement notices from the Environment Agency.	N / A.	N / A.
	17.	To develo p solutio ns which minimis e costs	Extent to which cost minimisation is considered for the receipt/treatment/dispo sal of incoming material at the receptor site.	Cost minimisation options appraisal for the receipt/treatment/dispo sal of incoming material are not typically undertaken at the receptor site.	Cost minimisation options appraisal for the receipt/treatment/dispo sal of incoming material are sometimes undertaken at the receptor site.	Cost minimisation options appraisal for the receipt/treatment/dispos al of incoming material are always undertaken at the receptor site and can be supported by appropriate reporting.	N/A	N / A

Silvertown Tunnel

Receptor Site Assessment (for Excavated Materials)

Document Reference: 6.10

cat ego ry	Principles	Indicators	Evaluation Criteria				
		Extent to which the receipt/treatment/dispo sal of the incoming material at the receptor site will increase operational costs.	The receptor site's gate fee is in the highest 20% compared to other receptor sites.	N / A	The receptor site's gate fee is in the 20%-80% range compared to other receptor sites.	N / A	The receptor site's gate fee is in the lowest 20% compared to other receptor sites.

- 2.6.6 Details of the scoring procedure are provided below.
- 2.6.7 Each receptor site is scored against the seventeen principles and their criteria. Scores range from -2 to +2 with negative impacts assigned negative scores (red colour), neutral impacts assigned a 0 score (amber colour) and positive impacts assigned positive scores (green colour).
- 2.6.8 The majority of the criteria requires a qualitative assessment, while only a select few require a quantitative assessment. Qualitative criteria provide flexibility during the evaluation in order to allow for professional judgment to be made on the receptor site's information. Quantitative criteria is not always readily available and, thus, it can add complexity in the evaluation process of a receptor site.
- 2.6.9 Table 2-4 below provides guidance on how the scores should be interpreted when assessing the receptor sites.

	Scores				
	-2	-1	0	+1	+2
Level of Impact	Major Adverse	Minor Adverse	Neutral	Minor Benefit	Major Benefit
	Significant Increase/ Decrease	Slight Increase/ Decrease	Neutral	Slight Increase/ Decrease	Significant Increase/ Decrease
	Long term Effect	Short term Effect	Neutral	Short term Effect	Long term Effect

Table 2-4 Scoring Guidance

2.6.10 During the assessment of each principle, human health and ecological receptors will be considered to determine the level of impact the receptor site's activities will have on the environment and communities. The sites are then scored depending on their level of impact against the receptors. As an example, close proximity to more sensitive human health and ecological receptors (e.g. internationally or nationally designated locations, high

exposure areas) pose a higher risk and thus are scored more negatively.

Carbon Calculations

- 2.6.11 In order to assess the principle 'to address and reduce contribution to climate change' under the Environment category, a carbon footprint of transport modes to receptor sites will be calculated as follows:
- The carbon footprint is first calculated using the formula below. Sum of: Total tonnes by mode × distance travelled by mode × CO2 eq conversion factor = Total kg CO2 eq
- Where more than one transport mode is used, the kg CO2 eq per tonne for each mode is calculated and then these are added together. The value is then converted to a kg CO2 eq per tonne accepted by the receptor site.
- The carbon values used to calculate the carbon footprint are based on publicly available datasets, see table below. The Defra 2015 conversion factors for transport are used and are as follows:

kg CO₂ eq per tonne per km	HGV (all diesel) all articulates (50% laden)	Cargo ship, general cargo, average	Rail, freight train
Scope 3	0.93742	0.013155	0.02601

Table 2-5 Carbon Values used to Calculate Carbon Footprint⁹

⁹ Defra, 'Greenhouse Gas Conversion Factor Repository', http://www.ukconversionfactorscarbonsmart.co.uk/ Page 40 of 54

- The Heavy Goods Vehicle (HGV) carbon value has been assumed to be 50% laden and the Rail and Cargo ship options have been assumed to be 100% laden as specified in the Defra conversion factors.
- Scope 3 emissions have been used as these are used for downstream transportation of material/goods.
 - 2.6.12 After calculating the carbon footprint of the transport mode to each site, the values are compared against each other in order to identify site's which rank highest, middle and lowest carbon footprint. High carbon footprint values are scored lower than lower carbon footprint values.

Scoring

- 2.6.13 After assessing and scoring each site, the scores for the individual principles are added to obtain a total score. Each site is ranked from highest to lowest score in order to determine whether it is included in the preferred list, the reserve list, or discounted entirely.
- 2.6.14 Below is a guide to the scoring thresholds for progression:
- Progress to preferred list: Score of 0 to +34; and
- Option not progressed (but with potential to be included on the reserve *list):* Score of -62 to -1.

2.7 Preferred List

- 2.7.1 The preferred list includes receptor sites which scored 0 to +34 and therefore will have a neutral or positive impact on the environment and the communities. The receptor sites which do not progress are those sites which scored -62 to -1 and, therefore, will have a negative impact on the environment and local communities.
- 2.7.2 The receptor sites should be ranked, focussing on sites with the most positive impact. Both lists are taken forward and receptor sites from the reserve list can be used in case any problems exist with using a site in the preferred list.

- 2.7.3 Special attention should be given to the negative scores for each site. It may be that a receptor site scores positive overall, but negatively in one or two areas that may be of specific concern.
- 2.7.4 Professional judgment will be used in order to determine whether those negative scores should be prioritised over the positive scores in other criteria. For example, a receptor site might score positive in all criteria except having the right permit/ exemption to receive, treat and/or dispose of the incoming material. In this case, professional judgment should be made to determine whether this site can be included in the preferred list given that it cannot receive/treat/dispose of the material.
- 2.7.5 During this further assessment of the preferred list, it may be concluded that some receptor sites may no longer be viable.

2.8 Reserve List

- 2.8.1 It must be noted that the RSA represents available receptor sites at the time of assessment and it is likely that other sites may become available and suitable before construction commences. Sites that fail to successfully pass through the Detailed Options Assessment but could potentially pass in the future (due to pending permit applications or land acquisition) will be placed onto the reserve list.
- 2.8.2 Receptor sites may move on or off the preferred list if their performance alters, for example:
- If a receptor site obtains planning consent, it may move on to the preferred list from the reserve list
- If a receptor site finds alternative materials and no longer requires the Scheme's excavated material, it may move off the preferred list.

2.9 Use of the Lists

2.9.1 The full RSA will be completed prior to the commencement of construction.

- 2.9.2 The preferred list and reserve list are used by contractors tendering for the Scheme. The inclusion of a receptor site on the preferred list or reserve list does not ensure that the site would be ultimately utilised as a receptor site for the excavated material arising from the Scheme. Should additional receptor sites be identified in the future, they would go through the same assessment process, as detailed in this document.
- 2.9.3 In the instance that suitable sites on the preferred and reserve list are not available, or materials cannot be placed at a suitable facility either on site (subject to necessary consents and licences) or on these lists, or they do not allow for the Contractor to meet other commitments outlined by TfL, Contractors should assess receptor sites beyond the 100km radius criteria (including transfrontier sites) using the RSA methodology. For transfrontier sites, the effect of local legislation should be taken into account when applying the RSA methodology and evaluation criteria.
- 2.9.4 TfL aims to minimise disposal by only using receptor sites which meet or exceed the criteria presented in this RSA methodology document.

3. CONCLUSION

- 3.1.1 The methodology used throughout the RSA provides a standard approach to assess the suitability of receptor sites to receive, treat and/or dispose of the incoming material from the Scheme.
- 3.1.2 The criteria used throughout the RSA are aligned with the commitments within the scheme's CD&E Materials Commitments document (Appendix to Document Reference 6.10).
- 3.1.3 The RSA provides a transparent process to stakeholders in order to ensure that the excavated material will be managed and treated with the least impact to the environment and communities.
- 3.1.4 The RSA will be completed prior to the commencement of construction. This will ensure that suitable sites will be in place to receive, treat and/or dispose of the excavated material from the Scheme.

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

Appendix A. Questionnaires

Preliminary Viability Assessment

- A.1.1 The following list of questions can be used as guidance for research or communications with the receptor sites in the long list. These questions may not necessary if the information is available through other sources.
- A.1.2 In addition to these questions, general information about the receptor site (e.g. contact name, address, date contacted, etc.) should also be recorded.
- A.1.3 Guidance questions for the Preliminary Viability Assessment include (but are not limited to) the following:

Preliminary Viability Assessment Criteria	Guidance Questions
The capacity of the receptor site to receive/treat/dispose of volumes of excavated material in excess of 200,000 tonnes)	 What is the total volume of material that the site can receive/treat/dispose of (tonnes per annum or total tonnes)? What is the end-use of the materials? Are there any potential future barriers / improvements to capacity (e.g. future contracts or expansions)?
The operator holds a valid permit or exemption, which allows them to receive/treat/dispose of the principal excavated material type(s), i.e. soil and stones	 Does the site have the appropriate permits / exemptions to receive/treat/dispose of the material?
The anticipated availability of the receptor site to receive, treat and/or dispose of the excavated material during the Scheme's construction (from the year 2019)	 What is the period of time which the site can receive/treat/dispose of the required capacity? Are there any future timescale constraints? Does the site have a future strategy or plan (e.g. expansion)?
Any enforcement notices, court cases and cautions from the	 Has the operator or receptor site been issued with any enforcement notices, court

Preliminary Viability Assessment Criteria	Guidance Questions
Environment Agency issued to the operator and / or receptor site	 cases or cautions in the past? If so, what was the type of offense, what date did the offense occur and what was the action taken?
The receptor site is accessible via road (i.e. lorry), water (i.e. barge) and / or rail	 Is the site accessible via road, water or rail? For water and rail access does the site have access to existing infrastructure for receiving material, e.g. jetty / rail head?

Detailed Options Assessment

- A.1.4 The following list of questions can be used as guidance for interviews (via telephone, email, face to face interviews) with the receptor sites in the short list. These questions may not necessary if the information is available through other sources.
- A.1.5 These questions may be used to prompt the sites to provide more details which may be helpful in the detailed assessment.
- A.1.6 In addition to these questions, general information about the receptor site (e.g. contact name, address, date contacted, etc.) should also be recorded. Guidance questions for the Detailed Options Assessment include (but are not limited to) the following:

Detailed Options Assessment Criteria			Guidanco Questions
Category	Principles	Indicators	Guidance Questions
t	1. To follow the waste hierarchy	Extent to which end- use of the material follows the principles of the waste hierarchy.	 What type of treatment is used? What is the end-use of the incoming material? What portion of the incoming material is disposed, recovered, recycled or re-used?
Environmer	2. To address and reduce contribution to climate change	Greenhouse gases (GHG) emitted through on site operations (including receipt, treatment and	 Is a carbon management in place at the site? Will the incoming material have to be transported once on site

Page 47 of 54

Detailed C	Detailed Options Assessment Criteria		Guidanco Questiene
Category	Principles	Indicators	Guidance Questions
		transportation of the incoming material).	 If so, how will it be transported? Will the incoming material have to be treated once on site – If so, how will it be treated?
		Extent of additional vehicle movements required once incoming material has entered the receptor site.	 Will the incoming material require movement via a vehicle once on site? What is the total distance material will travel on site, before it is placed?
		Greenhouse gases (GHG) emitted during transportation of material to receptor site. This is calculated using standard carbon factors (see Section 2.6.11)	 What is the transport mode used to transport material to the site? (rail/river/road)
		Extent to which flood risk is altered by the receipt/treatment/dispo sal of incoming material at the receptor site, or in the local catchment.	 Is the site in a flood risk area? Are there any water abstraction points on or near to the site? Has an assessment been made on the impact on local flood risk due to receipt/ treatment of the incoming materials? Is there any information in your permit that covers flood risk?
	3. To reduce impact on air quality and noise	Extent to which the receipt/treatment/dispo sal of the incoming material generates pollutants (within the permitted limits). This	 Please provide details of any air quality abatement measures in place. Do you believe the receipt/ treatment of incoming material will

Detailed Options Assessment Criteria		Guidance Questions	
Category	Principles	Indicators	Guidance Questions
		excludes transport to the site.	have an impact on local air quality?
		Extent to which air quality is impacted by the receipt/ treatment/ disposal of incoming material at the receptor site (excluding transport) in the short and long term.	 Will local air quality be affected in the short term due to the receipt/ treatment of incoming material? Will local air quality be affected in the long term due to receipt/ treatment of incoming material?
		Extent to which the receipt/treatment/ disposal of the incoming material generates noise (within the permitted limits). This excludes transport to the site.	 Please provide any details about any local noise complaints. Are there any noise abatement measures in place?
		Extent to which the receipt/treatment/dispo sal of the incoming material generates noise in the short and long term.	 Will local noise levels be affected in the short term due to receipt/ treatment of incoming material? Will local noise levels be affected in the long term due to receipt/ treatment of incoming material?
	 To protect the quality of water 	Extent to which surface water quality is impacted by the material receipt/ treatment/ disposal at the receptor site.	 Are there any surface water bodies within the site boundary or close to the site? What measures are in place to prevent contamination to the water bodies from the receipt/ treatment of the incoming material?
		Extent to which ground water quality is impacted by the material receipt/ treatment/disposal at	 Are there any aquifers within the site boundary or close to the site? What measures are in place to prevent

Page 49 of 54

Detailed Options Assessment Criteria		Cuidanas Quastiana	
Category	Principles	Indicators	Guidance Questions
		the receptor site.	 contamination entering the groundwater on site? Have there been any historical issues with groundwater contamination?
	5. To protect and enhance biodiversity	Extent to which designated sites are impacted in the short and long term by the material receipt/ treatment/ disposal the receptor site.	 Are there any designated areas on site or in close proximity? What measures are in place to prevent designated areas being detrimentally affected? Would local biodiversity be impacted in the short term by the receipt/ treatment of the incoming material at the receptor site? Would local biodiversity be impacted in the long term by the receipt/ treatment of the incoming material at the receptor site?
	6. To ensure	Extent to which incoming material replaces the need for virgin material use at the receptor site.	 What percentage of the virgin material used on site will be replaced by the incoming material? Where is the virgin material sourced from?
	efficient use of land and resources	Extent to which the efficiency of the receptor site's operations are impacted by the receipt/treatment/dispo sal of incoming material.	 Will the efficiency of the receipt/ treatment of incoming material increase/ decrease?

Detailed Options Assessment Criteria		Guidance Questions	
Category	Principles	Indicators	
		Extent to which incoming material will affect land take (footprint) of the receptor site.	• Will the site footprint be affected by the receipt/ treatment of the incoming material?
	7. To protect cultural heritage	Extent to which designated sites are impacted in the short and long term by the material receipt/ treatment/disposal the receptor site.	 Are there any scheduled ancient monuments or archaeological features on or in close proximity to the site which were assessed/ mitigated prior to the site becoming operational? Would the receipt/ treatment of the incoming material have any effects on local scheduled ancient monuments or archaeological sites in the short term? Would the receipt/ treatment of the incoming material have any effects on local scheduled ancient monuments or archaeological sites in the short term? Would the receipt/ treatment of the incoming material have any effects on local scheduled ancient monuments or archaeological sites in the short term?
	8. To protect local public amenities	Extent to which local public access are impacted by the material receipt/ treatment/disposal at the receptor site.	 Is there odour abatement measures in place at the site? Are there currently any public rights of way (PRoW) on the site? Will the receipt of material have an impact on PRoWs?
Public	9. To reduce visual impact of surrounding areas	Extent to which the surrounding areas are visually impacted in the short and long term by the receipt/ treatment/disposal of	 What screening is there currently in place at the site to reduce visual impact? Is the proposed restoration for the site

Page 51 of 54

Detailed Options Assessment Criteria		Guidanaa Quaatiana	
Category	Principles	Indicators	Guidance Questions
		incoming material.	 different to surrounding landscape? How will the receipt/ treatment of incoming material affect the local landscape in the short term? How will the receipt/ treatment of incoming material affect the local landscape in the long term?
	10. To provide	Extent to which receipt/treatment/dispo sal of incoming material at the receptor site creates new job opportunities.	• Would the receipt/ treatment of incoming material have any impact on employment at the site or local area?
	for a diverse workforce	Extent to which receipt/treatment/dispo sal of incoming material at the receptor site considers diversity of workforce.	 Does the site have a workforce diversity strategy? Does the site currently employ a diverse workforce?
	11. To promote sustainable transportatio n	Extent to which sustainable forms of transport are available and considered for the receipt of incoming material at the receptor site.	 How can the site be accessed (road/rail/barge)? Will any major upgrades need to be completed in order to receive material by rail/barge?
Transport	12. To manage material locally, as per the proximity principle	Average distance from the excavation site to the receptor site.	• What is the distance (miles/ kilometres) from the source site to the receptor site, by rail/ road/ barge?
Operatio ns	13. To identify receptor sites with suitable	Likelihood of the receptor site to conform to the project's timescales.	• Over what timeframe could you receive/treat/dispose of the incoming material?

Detailed C	ptions Assessme	ent Criteria	Guidanaa Quaatiana
Category	Principles	Indicators	
	capacity and operational capability	The receptor site's ability to receive/ treat/dispose of various types of incoming material.	 What material types are listed in your permit/ exemption? Are there any limitations to the materials you can receive/treat/dispose of?
		The receptor site's capacity to receive/treat/dispose of the volume of incoming material.	 How much material are you permitted to receive/ treat/dispose of? How much material could you receive/treat/dispose of on a daily/ monthly/ yearly basis based on your expected capacity?
	14. To use receptor sites with appropriate permits/ exemptions	The receptor site has the appropriate permits/ exemptions to receive/treat/dispose of and treat the incoming materials.	 Please can you provide a copy of permit or exemption issued by the Environment Agency Information relating to any permit constraints If permits/exemptions have not been granted yet, when do you anticipate they will be in place? Can you receive/treat/dispose of material prior to the pending permits being issued?
	15. To ensure health and safety guidelines are followed	Extent to which health, safety and environmental (HSE) best practices are followed.	 How many health and safety incidents have you had in the past year? Does the site have a HSE advisor? Does the site have a HSE plan in place? Does the site have ISO 18001 certification?

Detailed Options Assessment Criteria		Guidance Questions	
Category	Principles	Indicators	
	16.To ensure	Extent to which the site operator has been issued with a ruling by UK court of law.	 Has the site operator been formally charged guilty by a UK court of law?
	legal compliance	Extent to which the operator is in risk of breaching environmental regulations.	Has the site operator has been issued with enforcement notice/ caution from the Environment Agency?
	17.To develop solutions	Extent to which cost minimisation is considered for the receipt/treatment/dispo sal of incoming material at the receptor site.	 Has the site undertaken a cost minimisation options appraisal for the receipt/ treatment of incoming material?
	wnich minimise costs	 site. Extent to which the receipt/treatment/dispo sal of the incoming material at the receptor site will increase operational costs. Will ope the site the site the receipt of the incoming material at the receptor site will increase operational costs. 	 Will operational costs at the site be effected by the receipt/ treatment of the incoming material? What are the site's gate fees for the incoming material?

Appendix E. Site Waste Management Plan

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

SILVERTOWN TUNNEL

Site Waste Management Plan

April 2016



MAYOR OF LONDON

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

Contents

List of	Abbreviations	6	
Gloss	ary of Terms		
1.	INTRODUCTION	13	
1.1	Purpose		
1.2	Responsibilities		
1.3	Purpose of the SWMP	15	
1.4	Site Description		
2.	TYPES OF WASTE		
2.1	Overview		
2.2	Construction and Demolition wastes (C&D)		
2.3	Excavated Materials (E)		
2.4	Tunnel Materials	21	
2.5	Contaminated Waste	21	
2.6	Hazardous Waste (including asbestos containing waste)		
2.7	Vegetation		
2.8	Fit Out Material		
3.	WASTE FORECAST		
3.1	Overview	25	
4.	WASTE MANAGEMENT STRATEGY		
4.1	Overview	27	
4.2	Reduction and Reuse		
4.3	Segregation of Non-excavation Waste		
4.4	Colour-coded Skips		
4.5	Contaminated Land and Hazardous Waste		
4.6	Disposal and Treatment Options		
5.	TRAINING		
5.1	Overview		
6.	MONITORING		
6.1	Overvie	W	39
-------------	---------	---	----
7.	ACTUA	L QUANTITIES	41
7.1	Overvie	ew	41
8.	REVIEV	V	43
8.1	Overvie	W	43
9.	POST S	CHEME COMPLETION	45
Appendix A.		Waste Forecast Tables (WRAP SWMP Extract)	47
Appendix B.		Waste Carriers (WRAP SWMP Extract)	49
Appendix C.		Actual Waste Movements	51

List of Tables

Table 3-1 Overall Material / Waste Arisings	26
Table 4-1 Waste Minimisation Measures	35
Table 8-1 Review Checklist	43

List of Figures

Figure 1-1 Location of Scheme	
Figure 4-1: Waste Hierarchy	27

List of Abbreviations

CD&E	Construction, Demolition and Excavation
CD&EMC	Construction, Demolition and Excavation Materials Commitments
CL:AIRE DoW CoP	Contaminated Land: Applications in Real Environments Definition of Waste: Code of Practice
CoCP	Code of Construction Practice
DCO	Development Consent Order
DEFRA	Department for Environment Food & Rural Affairs
DLR	Docklands Light Railway
DoC	Duty of Care
Edoc	Electronic Duty of Care
EA	Environment Agency
EAL	Emirates Air Line
EMS	Environmental Management System (for the Scheme)
ELoW	European List of Waste
ES	Environmental Statement
LOW	List of Waste
NSIP	Nationally Significant Infrastructure Project
PEMP	Project Environmental Management Plan

Silvertown Tunnel Site Waste Management Plan Document Reference: 6.10

RSA	Receptor Site Assessment (for Excavated Materials)
SWMP	Site Waste Management Plan
SPV	Special Purpose Vehicle
ТВМ	Tunnel Boring Machine
TfL	Transport for London
WAC	Waste Acceptance Criteria
WEEE	Waste Electrical and Electronic Equipment
WRAP	Waste and Resources Action Programme

Glossary of Terms

Contractor	Anyone who directly employs or engages construction workers or manages construction work. Contractors include sub-contractors, any individual self-employed worker or business that carries out, manages or controls construction work.
Development Consent Order (DCO)	This is a statutory order which provides consent for the project and means that a range of other consents, such as planning permission and listed building consent, will not be required. A DCO can also include provisions authorising the compulsory acquisition of land or of interests in or rights over land which is the subject of an application.
	http://infrastructure.planninginspectorate.gov.uk/help/glossary-of- terms/
ELoW	The European List of Waste provide definition and code assignment for hazardous and non-hazardous wastes.
Contractor	A term defined under the Construction Design and Management Regulations (2015). Contractors are appointed by the client to coordinate the construction phase of a project where it involves more than one contractor.
Envisaged Design	Design proposals that the consultation and DCO application will refer to.
The Scheme	The construction of a new bored tunnel with cut and cover sections at either end under the River Thames (the Silvertown Tunnel) between the Greenwich peninsula and Silvertown, as well as necessary alterations to the connecting road network and the introduction of user charging at both Silvertown and Blackwall tunnels.
Transport for	A London government body responsible for most aspects of the transport system in Greater London. Its role is to implement transport strategy and to manage transport services across London.
London (ITL)	These services include: buses, the Underground network, Docklands Light Railway, Overground and Trams. TfL also runs Santander Cycles, London River Services, Victoria Coach Station and the Emirates Air Line.

	As well as controlling a 580km network of main roads and the city's 6,000 traffic lights, TfL regulates London's private hire vehicles and the Congestion Charge scheme.
Waste and Resources Action Programme (WRAP)	WRAP is a registered charity that works with businesses, individuals and communities to achieve a circular economy by helping them reduce waste, develop sustainable products and use resources in an efficient way.

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

SUMMARY

- S.1.1 This Site Waste Management Plan (SWMP) details the anticipated waste streams to be generated during the construction phase of the Silvertown Tunnel project and the means that could be implemented to minimise and manage the waste. This document should be read in conjunction with the associated Waste and Resources Action Programme (WRAP) excel template which provides estimates for the type and volume of anticipated waste arising from the construction phase of the Scheme.
- S.1.2 This SWMP has been developed to provide a consistent framework for managing and documenting waste generated during the construction phase. The plan will be updated throughout the construction phase of the Scheme. This SWMP submitted with the DCO application presents the framework in which the SWMP will subsequently be developed by the contractor.

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

1. INTRODUCTION

1.1 Purpose

- 1.1.1 This Site Waste Management Plan (SWMP) has been developed to reflect the Silvertown Tunnel envisaged design and the associated quantities of waste that are anticipated to be generated during the construction phase of the Silvertown Tunnel project (herein referred to as the 'Scheme').
- 1.1.2 SWMPs are developed and maintained to facilitate good practice on construction sites, recording duty of care information and by that preventing waste crime. They are also an important tool for improving environmental performance and reducing waste disposal costs for schemes.
- 1.1.3 This document summarises estimates of how much waste is anticipated to be generated during the construction phase of works (covering construction, demolition and excavation (CD&E) waste) and to provide an initial indication as to whether materials and waste streams have the potential to be reused, recycled, recovered or disposed of based on information available at this stage.
- 1.1.4 The main phases for a SWMP include:
 - The preparation stage which is the responsibility of TfL prior to construction; and
 - The construction stage which is the responsibility of the Contractor from when construction begins.
- 1.1.5 The SWMP is a live document that will be updated throughout the construction phase by the Contractor.
- 1.1.6 The Waste and Resource Action Programme (WRAP) SWMP template has been / will be (as the Scheme progresses) utilised to record the following details:
 - Record the Contractor's responsible personnel, where known, who will be involved in the project;
 - Forecast waste that is anticipated to be produced during the project, recorded using the appropriate European List of Waste (ELoW) code and waste description;

- Supply details on waste minimisation actions, by both TfL and the Contractor; and
- Contain a register of waste carriers and their waste carrier registration number and details of the site the waste and/or material will be taken to.
- 1.1.7 For information on TfL's overall vision, objectives and commitments for the management of CD&E materials the Construction, Demolition and Excavated Materials Commitments (CD&EMC) report (Appendix to Document Reference 6.10) should be consulted.
- 1.1.8 The SWMP will be used to review the progress of the Scheme towards the targets contained within the CD&EMC, which are also detailed within this document.
- 1.1.9 SWMPs set a framework to facilitate good practice on site and are an important tool for improving environmental performance, meeting regulatory control and reducing waste disposal costs.

1.2 Responsibilities

- 1.2.1 TfL and the appointed contractor shall take all reasonable steps to ensure that:
 - All waste from the site will be dealt with in accordance with the waste Duty of Care as set out in Section 34 of the Environmental Protection Act 1990¹ (as amended) and the Waste (England and Wales) Regulations 2011²
 - Emphasis is put on the waste hierarchy to ensure that waste is dealt with in the priority order of: prevention; preparing for re-use; recycling; other recovery (for example, energy recovery); disposal as per the Waste Regulations 2011²; and
 - Materials will be handled efficiently and waste managed appropriately.

² The Waste (England & Wales) Regulations 2011 (SI 2011/988) (as amended 2012 (SI 2012/1889) and 2014 (SI 2014/656)).

¹ Environmental Protection Act 1990 (c. 43) Part II, section 34 (as amended in 1996 and 1999).

- 1.2.2 TfL are responsible for preparing the SWMP prior to construction.
- 1.2.3 This plan has been developed at this early stage to ensure a guidance document is in place. On award of Scheme's contract, the appointed Contractor will be responsible for developing this plan and ensuring it relates specifically to the works associated with the solution adopted for the completion of the works.
- 1.2.4 The Contractor and appointed site manager are responsible for instructing workers, overseeing and documenting results in the SWMP. Copies of this plan shall be distributed to the design team, TfL project manager, project director and each subcontractor involved in the Scheme. This will be undertaken every time the plan is updated.

1.3 Purpose of the SWMP

- 1.3.1 This SWMP has been developed to provide a consistent framework for managing and documenting waste generated during the construction phase and to highlight areas in which waste could be reduced, reused, recycled and those that require disposal. This SWMP also records design decisions that demonstrate good and best practice in waste minimisation and management. By recording these project details contractors are able to:
 - Estimate waste and identify actions to reduce waste and cost;
 - Record actual waste movements; and
 - Review the Scheme's performance.
- 1.3.2 At detailed design stage, actions will be taken to design out waste and specify that materials and components with a recycled content are used in the construction of the Scheme where practicable. These considerations are in line with the designing out waste principles and adopting best practice approaches advocated by WRAP.
- 1.3.3 The SWMP is designed to follow the key stage process of:
 - Plan;
 - Implementation;
 - Measure; and
 - Review.
- 1.3.4 The creation of a SWMP allows the Contractor to demonstrate how their approach to excavated materials and waste has taken into account; the waste hierarchy, Contaminated Land: Applications in Real Environments

Definition of Waste: Code of Practice (CL:AIRE DoW CoP) code of practice and Duty of Care (DoC).

- 1.3.5 Best practice suggests that the SWMP process is to be implemented at the earliest opportunity within the Scheme to encourage the process of developing design and construction solutions that contribute towards the minimisation of waste.
- 1.3.6 The SWMP will outline the methods required to minimise waste, manage waste produced responsibly, measure the quantities and costs of waste produced effectively and on review, provide lessons learned to enhance future schemes.
- 1.3.7 The SWMP WRAP Excel template is a live working document that requires the relevant sections to be completed at different stages throughout the development of the Scheme. Final completion of the spreadsheet will not happen until the construction phase of the Scheme has finished. Therefore, at any point during the construction phase, sections of the spreadsheet will be incomplete, pending completion.
- 1.3.8 The Contractor is expected to take ownership of, and update the SWMP document, completing the various sections until it has reached the point of being finalised and signed off at the point of the overall Scheme completion. The SWMP will identify the types and quantities of waste that are produced throughout the construction phase of the Scheme and will identify management options for each type of waste, paying attention to the waste hierarchy (see Section 4).
- 1.3.9 It shall also state stringent requirements for the control and disposal of hazardous wastes (see Section 7.6).
- 1.3.10 The adoption of the SWMP will help to ensure the Scheme fulfils its legal obligations towards waste management and Duty of Care. The SWMP shall be communicated to selected employees and sub-contractors on the Scheme.
- 1.3.11 Adopting an extended Duty of Care i.e. ensuring that sub-contractors and facility operators with a good and proven record of environmental performance and compliance are appointed to treat and/or receive materials from the site, is also recommended.
- 1.3.12 The measurement and control of fly-tipping will not be included in the SWMP as this will be controlled by on-site measures.

1.4 Site Description

- 1.4.1 The Silvertown Tunnel scheme (herein after referred to as 'the Scheme') is a new road tunnel under the River Thames connecting the Greenwich Peninsula and Silvertown. The Scheme involves the construction of a twin bore road tunnel providing a new connection between the A102 Blackwall Tunnel Approach on Greenwich Peninsula (Royal Borough of Greenwich) and the Tidal Basin roundabout junction on the A1020 Lower Lea Crossing/Silvertown Way (London Borough of Newham). The Silvertown Tunnel would be approximately 1.4km long and would be able to accommodate large vehicles including double-deck buses. The Boord Street footbridge over the A102 would be replaced with a pedestrian and cycle bridge.
- 1.4.2 New portal buildings would be located close to each tunnel portal to house the plant and equipment necessary to operate the tunnel.
- 1.4.3 The introduction of free-flow user charging on both the Blackwall and Silvertown Tunnels would play a fundamental part in managing traffic demand and would help to support the funding of the construction and operation of the Silvertown Tunnel.
- 1.4.4 The design of the Silvertown Tunnel would include a dedicated bus/coach and HGV lane, which would provide opportunities for TfL to provide additional cross-river bus routes.
- 1.4.5 Main construction works would likely commence in 2018 and would last approximately 4 years with the new tunnel opening in 2022/23. The main construction compound would be located at Silvertown, utilising the existing barge facilities at Thames Wharf along with a new temporary jetty for the removal of spoil and delivery of materials by river. A secondary site compound would be located adjacent to the alignment of the proposed cut and cover tunnel on the Greenwich Peninsula.
- 1.4.6 Figure 1-1 below, shows the sites location and surrounding features.





2. TYPES OF WASTE

2.1 Overview

- 2.1.1 Waste streams categorised as either excavation (E) or demolition (D) are those which exist within the Scheme footprint already, such as topsoil, subsoil, vegetation and demolition materials etc.
- 2.1.2 Imported materials are those which are imported to site for inclusion into the temporary and permanent construction works (such as concrete, construction aggregates, asphalt and cabling etc.) which also produce a waste stream. Included within this waste stream is an estimated quantity of product packaging. This waste stream is produced from a range of potentially avoidable activities such as damaged materials and the over ordering of materials. This waste stream is described as construction (C) waste within the SWMP.
- 2.1.3 In order to assist the management and segregation of waste, estimations have been made of the types and quantities that will be generated during the construction phase of the Scheme. For this the WRAP SWMP Excel template has been utilised (Appendix A).
- 2.1.4 For all CD&E waste there are a number of considerations to be implemented in terms of management; such as waste reduction, segregation of waste, disposal of waste, financial impacts of waste disposal and recording, monitoring, education and reviewing data (see Section 7).
- 2.1.5 The following sections discuss the main waste streams anticipated as a result of the Scheme construction phase (covering CD&E). This list is not exhaustive, however it forms a basis for the SWMP based on the current envisaged design. Any additional streams will be included in the plan as part of the updates.

2.2 Construction and Demolition wastes (C&D)

- 2.2.1 Construction and demolition wastes will largely consist of inert material; concrete (structural and pavement), bricks and glass.
- 2.2.2 Construction waste figures are included in the SWMP (Appendix A). These approximate figures have been generated by calculating the percentage of imported material that will become a waste. These are calculated by using the WRAP industry standard wastage rates for imported material.

- 2.2.3 Currently the SWMP does not account for workforce waste streams, for example Waste Electrical and Electronic Equipment (WEEE) and organic waste or any operational waste. Once details of workforce are known such waste streams are to be incorporated into the SWMP.
- 2.2.4 Any waste produced through the importation of materials needs to be monitored and included in the SWMP as the Scheme progresses. Where possible, consideration will be given to the use of recycled imported material such as concrete or those with a higher recycled content. However, due to the high level of specification expected of the material required for the tunnel structure, this may not be considered a viable option.
- 2.2.5 Demolition waste figures are also included within the SWMP (Appendix A). Demolition figures are calculated based on the assumption that all materials generated as a result of the demolition works will be waste.
- 2.2.6 Demolition works are anticipated to consist of the removal of above ground buildings and structures.

2.3 Excavated Materials (E)

- 2.3.1 Excavated materials will be generated through tunnel excavation, the construction of portal launch chamber and surrounding highways.
- 2.3.2 In relation to materials removed from site it is anticipated the majority of the material would be excavated material, for which beneficial uses would be sought. Beneficial use, in terms of this Scheme is defined within the CD&EMC report (Appendix to Document Reference 6.10).
- 2.3.3 Appropriately experienced staff, familiar with working on brownfield sites and with the contaminant groups anticipated will supervise the excavation works to manage the segregation of spoil materials. Site-derived materials of a similar nature will be stockpiled together and any changes in the physical and/or chemical properties will prompt further segregation.
- 2.3.4 As part of the construction phase works, excess material from tunnelling and other excavation activities will be removed from the construction sites at Greenwich Peninsula and Silvertown by road or river transport.
- 2.3.5 Where possible alternatives have been exhausted there will be a requirement to dispose of excavated material, by licensed waste carriers to licensed landfill sites. The material will be handled in accordance with the Waste (England and Wales) Regulations 2011².

2.4 Tunnel Materials

- 2.4.1 The bored tunnel section is likely to be constructed using a segmental concrete lining, excavated through the use of a TBM, which will form the structural (or primary) lining. The type of TBM is envisaged to be a closed face earth pressure balance configuration. This machine works to balance the pressure of the excavation by controlling the rate of material excavation as opposed to the use of pressurised slurry. The condition of the excavated material will vary however it is anticipated that this method will not result in the production of slurry.
- 2.4.2 Final TBM selection will be determined by the contractor based on the assessment of the construction risk with consideration for tunnel alignment depth, associated ground pressure, ground cover, anticipated geology and depth of the water table. The envisaged design has therefore ensured that sufficient temporary land is available for a slurry separation plant if this option is chosen.
- 2.4.3 Due to the size of the tunnel bores the anticipated excavated spoil at any one time is likely to consist of more than one type of material (i.e. different strata).
- 2.4.4 If the water content of the excavated material is considered too high for transport by barge it may be necessary to dry it out prior to transportation.
- 2.4.5 The TBM will be maintained underground and by-products of this process will typically be oils and greases, and will therefore require subsequent classification for inclusion within the SWMP.

2.5 Contaminated Waste

- 2.5.1 Given the nature of the works and site history there is the potential for excavation works to give rise to potentially contaminated material that will require remediation and/or appropriate disposal.
- 2.5.2 Site wide remediation was undertaken at Greenwich during the late 1990s. It is understood that notable sources of contamination, such as tar tanks and known contamination hot spots were removed, groundwater remediation was undertaken and near surface soils were removed or cleaned prior to landscaping. However, it is understood that contaminated materials remain at depth beneath much of the site. Allowance will need to be made in the forecasted waste for the removal of impacted material that may be encountered.

2.6 Hazardous Waste (including asbestos containing waste)

- 2.6.1 It is anticipated that a small amount of hazardous material will be produced from CD&E activities.
- 2.6.2 Hazardous waste including any contaminated excavated spoil will be kept separate and secure in receptacles in line with the Waste (England) Regulations 2011². Doing so will also act to reduce cross contamination.
- 2.6.3 The waste will then be removed from site and treated in accordance with all applicable legislation.
- 2.6.4 Asbestos based materials may arise during the excavation of the ground for tunnels and portals especially in areas of previously high industrial use and the historic gas works. Asbestos fragments in soil and low levels of loose fibres in soils have been identified within some of the samples taken during the ground investigations. Therefore, waste streams containing asbestos have been estimated within the SWMP.
- 2.6.5 The edge of one of the historic gas work buildings on Greenwich Peninsula is located above the proposed alignment. Therefore, there is the possibility that the foundations or items of infrastructure (including asbestos sheeting) remain underground. No records have been found detailing the demolition works associated with these features.
- 2.6.6 Currently, the full extent of below ground structures (i.e. piles) and their interaction with the proposed tunnel alignment is not fully known.
- 2.6.7 During demolition and construction activities additional streams of hazardous waste are anticipated, including: oils and grease from equipment maintenance, batteries, waste paint and solvents and fluorescent tubes. Quantities of this waste stream have not been estimated at this stage, once plant set up is established this waste stream should be accounted for.

2.7 Vegetation

2.7.1 In order for construction to take place, areas of vegetation, comprising mainly of grass and shrubs will require clearance in advance of general excavation works. If Japanese Knotweed, or any other invasive species, is located then special measures will be required to deal with this vegetation (such as classification and disposal of the waste as a 'controlled waste' under the Environmental Protection Act 1990 (c. 43) (as amended in 1996 and 1999).

2.7.2 As a minimum all vegetation waste should be diverted from landfill, unless it has been identified as an invasive species and no other options are available. The greatest opportunity for the sustainable management of vegetation waste is through recycling into compost.

2.8 Fit Out Material

- 2.8.1 Imported materials for the final fit out of the tunnels and associated infrastructure will meet a pre-designed specification, which will have taken into consideration when designing out waste measures.
- 2.8.2 Any WEEE waste generated should be reused or recycled. Where practicable this should be managed through a take-back or other local recycling schemes. A full audit trail should be kept for such WEEE compliant schemes (in accordance with regulations).

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

3. WASTE FORECAST

3.1 Overview

- 3.1.1 One of the initial stages of completing the SWMP is to forecast anticipated waste arisings. This allows for early stages of designing out waste to be incorporated into the Scheme.
- 3.1.2 The waste estimates produced are formulated from available data for the Scheme including current envisaged design and cost estimates. Details used are consistent with other application documents.
- 3.1.3 Appendix A contains the forecast waste tables from the WRAP spreadsheet produced for the construction phase and covers CD&E waste from both the temporary and permanent works for the Scheme. The waste forecast tables set out the current estimates of waste types and quantities that are anticipated. This list is not exhaustive and additional waste streams shall be added when they occur.
- 3.1.4 The forecast waste table provides important information such as identifying the waste activity (either C,D or E), the waste stream, the material classification, the ELoW code, quantity and the management methods.
- 3.1.5 This information is duplicated in subsequent tabs within the spreadsheet. Actual waste movements are completed by the Contactor during construction, which feed into the key performance indicators (KPIs) tab.
- 3.1.6 Demolition waste and excavation materials quantities are calculated as a whole for an activity, i.e. assuming all of the material is classified as a waste and generally based on estimated volumes.
- 3.1.7 Construction waste quantities are calculated using WRAP and industry standard wastage rates. The rates assume a certain percentage of the imported material will become a waste.
- 3.1.8 As well as determining where waste has been generated there are other considerations to waste management such as waste reduction, segregation of waste, disposal of waste and the financial impacts of waste disposal which are completed further into the Scheme and discussed in subsequent sections of the SWMP.

Table 3-1 Overall Material / Waste Arisings

Activity	Total Material / Waste (tonnes)	
Construction Phase: Excavation		
Excavated material	1,086,000	
Construction Phase: Non-excavation		
Demolition waste	66,000	
Construction waste	23,000	
Total	1,175,000	

4. WASTE MANAGEMENT STRATEGY

4.1 Overview

4.1.1 The waste management strategy for the Scheme will follow the accepted waste hierarchy (Figure 4-1) which is taken from the Waste Framework Directive (2008/98/EC). This shows the priority that must be applied when making decisions on reuse, recycling, recovery or disposal of each type of waste generated during the construction phase.



Figure 4-1: Waste Hierarchy

- 4.1.2 Contractors, design teams and suppliers are encouraged to minimise the amount of waste produced at the work sites. Waste arisings have been considered in terms of their suitability for:
 - Reuse on site: as part of the development or future maintenance works;
 - Reuse off site: materials of reusable value but for which a need cannot be demonstrated on site; and
 - Recycling off site: where materials are suitable for recycling/treatment at an appropriately permitted facility.
- 4.1.3 The SWMP helps to ensure best practice and sustainable waste management is considered throughout. The SWMP will be updated with

every design feature that will potentially have an impact on waste prevention, reduction or re-use.

- 4.1.4 The prevention of waste is the preferential option of waste management in terms of the hierarchy. Contractors are therefore required, through the SWMP to identify actions relating to waste prevention/minimisation.
- 4.1.5 In line with the Schemes commitment of the diversion of 80% (by weight) of CD&E materials to be re-used on site or removed from site for beneficial use with an aspiration to reach 95% (by weight), the following sections aim to highlight potential areas of waste minimisation and management that will be considered as part of the Scheme's waste management strategy.

4.2 Reduction and Reuse

- 4.2.1 By implementing a SWMP from design stage it is possible to reduce the amount of waste produced as part of the Scheme.
- 4.2.2 The contractor will maximise opportunities for the potential for reuse and recycling of all waste streams on site.
- 4.2.3 Construction waste, or waste arising from imported material may be minimised through careful product specification and use.
- 4.2.4 Over-purchasing can lead to significant wastage and will be avoided in the first place. Ensuring materials are ordered for delivery shortly before they are used on the Scheme will also avoid possible damage and therefore wastage.
- 4.2.5 A continual review of the type of surplus materials being generated on site will aid in identifying approaches to reduce the amount produced.
- 4.2.6 Site set-up should maximise reuse or recycling and minimise disposal to landfill.
- 4.2.7 Materials delivered to the Scheme will be received and controlled by the Contractor. Materials will be stored to minimise the potential of damage or wastage. Measures will include off-ground storage e.g. on pallets, remaining in original packaging, protection from rain damage or collision by plant or vehicles.
- 4.2.8 The materials storage area will be secured during out of hours to prevent unauthorised access.

- 4.2.9 The use of precast concrete segments for the tunnel and other components of the Scheme will act to reduce the amount of concrete waste on site.
- 4.2.10 For any surplus material that may arise on site other options will be explored. These may include reuse on another site or take back options with the manufacturer.
- 4.2.11 If applicable, surplus inert excavated materials with some engineering strength (e.g. stone, bricks, clay, rubble, rock) may be suitable for reuse in land reclamation schemes. The material could be reused in other schemes in the surrounding area, if such a scheme was proceeding at the same time, to avoid disposal at landfill and the associated impacts and costs. Such a reuse scheme will need to be in accordance with all applicable legislation. This will require compliance with the criteria and thresholds for an Environment Agency issued exemption (U1 or U11 may be applicable) or it may require a permit under the Environmental Permitting Regulations 2010³ as amended or compliance with CL:AIRE DoW CoP by production a materials management plan.
- 4.2.12 Although onsite soil reuse options will be used where applicable, it is still likely that there will be a requirement for importation of additional fill materials with specific properties such as structural backfill and topsoil.
- 4.2.13 If any materials, which are deemed acceptable, are produced from the enabling works (e.g. good quality topsoil), this will be stored and re-laid, within the Scheme or if this is not possible it will be sent to a topsoil recycling facility.

4.3 Segregation of Non-excavation Waste

- 4.3.1 Successful recycling relies upon early planning, clear responsibility and space within a compound for segregation and storage. Shelter may be needed to prevent some materials such as cardboard and paper from deteriorating while being sorted or awaiting collection.
- 4.3.2 A specific area shall be laid out and labelled to facilitate the segregation of materials, where possible, for potential recycling, reuse and return.

³ The Environmental Permitting Regulations, as amended in 2011 (SI 2011/2043), 2012 (SI 2012/630) and 2014 (2014/255)

- 4.3.3 Enclosed and lockable skips prevent deterioration of waste and also stops unauthorised access to the skips. Recycling and waste receptacles (e.g. skips) are to be kept clean and clearly marked in order to avoid contamination of materials. The following waste streams are currently identified for segregation:
 - Plastics;
 - Mixed inert (e.g. concrete and rubble);
 - Hazardous (e.g. asbestos, polychlorinated biphenyls);
 - Mixed non-hazardous (welfare waste and general waste);
 - Metal (e.g. copper and iron);
 - Wood (e.g. fencing/hoarding);
 - Food (canteen waste);
 - Paper and cardboard (office waste);
 - WEEE (e.g. cables, disused electrical appliances and equipment); and
 - Oils and oily rags.
- 4.3.4 Space requirements within the compound will need to be identified to accommodate skips and storage of reusable materials. Individual waste streams will be kept separate including the use of an excavated material only area.
- 4.3.5 For all waste management options within the red line boundary of the Scheme, consideration will need to be given to identifying whether waste exemptions or permits are required to enable the storage and treatment of waste materials.
- 4.3.6 Waste management options will be supported by the identification of appropriately permitted waste management and recycling facilities within an acceptable proximity to the Scheme. Further details are contained within the Receptor Site Assessment (RSA) document (Appendix to Document Reference 6.10) within the Environment Statement (Document Reference 6.1)

4.4 Colour-coded Skips

- 4.4.1 Different coloured skips (or sufficiently clear labelling) will be used to ensure that construction workers are clear about where to put each type of waste. This will reduce the levels of contamination in the skips and increases the likelihood that a load will not subsequently be rejected once the waste stream has been sent off-site for reprocessing. In cases where the load is rejected, the likely destination will be landfill (which will increase the costs of the Scheme).
- 4.4.2 Typical segregated skip categories and management methods include:
- 4.4.3 WRAP recommends the following materials and associated colour-code to aid material segregation on site:
 - Wood (red/yellow);
 - Plastic (orange);
 - Plasterboard (blue);
 - Bricks and rubble (green);
 - Cardboard/paper (no colour coding specified); and
 - General waste (no colour coding specified).
- 4.4.4 The colour schemes adopted on site must be made clean to operative and simple labelling illustrating suitable material types for collection may be used in addition to colour coding.
- 4.4.5 Skips will be monitored by the Contractor to ensure that contamination of segregated skips does not occur.

4.5 Contaminated Land and Hazardous Waste

- 4.5.1 Samples should be taken at predetermined time frames appropriate to the type and quantity of soil.
- 4.5.2 Soils will be placed in clearly identified stockpiles and chemical testing undertaken to confirm the potential for reuse on site, or, if considered inappropriate for reuse (due to geotechnical or chemical properties or being surplus), to inform off site treatment and/or disposal routes. Where soil materials meet the geotechnical and chemical criteria for reuse given the proposed end use scenario, such materials may be reused on site, if

required. Any surplus materials will be removed from site for either direct beneficial use elsewhere (such as land remediation schemes) or for recycling or recovery at an appropriately permitted off-site facility. Where excavated materials are affected by contamination, such materials will be separated and sent for either treatment, where appropriate, or disposal at appropriately permitted facilities.

- 4.5.3 The cost of hazardous waste treatment and disposal is significantly higher than treatment or disposal of non-hazardous or inert waste. Through identifying areas of contamination early on, the scheme layout and construction methods to be adopted could be amended to minimise the handling of such materials, potentially reducing the scheme costs. Any soils removed from site during construction will be subject to appropriate chemical characterisation and Waste Acceptance Criteria (WAC) testing to determine their destination facility i.e. type of landfill etc⁴.
- 4.5.4 Hazardous waste cannot be re-used on site nor can it be mixed with nonhazardous or inert waste. There is a statutory requirement under the Landfill Directive⁴ to pre-treat any waste (including hazardous waste) at the point of origin or at an alternative suitable site prior to disposal. Pre-treatment may reduce the cost of disposal by rendering the waste non-hazardous.
- 4.5.5 Where potentially hazardous materials are encountered, appropriate mitigation measures will be employed on site. The following list details best practice approaches:
 - Prepare a 'quarantine' receiver location with a bunded perimeter to a suitable size compared to the assumed quantity of materials (a maximum height to be agreed);
 - The storage area will have an impermeable base to prevent leachate escape to groundwater and will be protected against flood damage or inundation with any accumulated rainwater regularly emptied and managed appropriately. The area will be regularly checked and kept in a good condition, as well as being protected from vandalism; and
 - Appropriate spill kits will be available, located near to the hazardous waste storage area and checks carried out regularly to ensure they are adequately stocked.

⁴ Landfill Directive 99/31/EC and Council Decision 2003/33/EC. Page 32 of 52

- 4.5.6 Once appropriate analysis has been completed, treatment and/or disposal of materials/waste can be carried out. Where this waste needs to be removed from site, a suitable disposal facility will be sourced.
- 4.5.7 Suitable facilities will be assessed as part of the RSA process.
- 4.5.8 The suitable facility must have the relevant licenses and permissions to receive the waste. Completed waste transfer notes (or consignment notes for hazardous waste) detailing the relevant waste codes must be raised and the waste moved from site with a licensed waste carrier. Wastes shall be moved under the requirements of the Duty of Care. Responsibility for the basic characterisation of waste rests with the producer.
- 4.5.9 All site staff will be made aware of their responsibilities and liabilities in terms of hazardous waste handling and management at site inductions and repeated during toolbox talks.
- 4.5.10 When hazardous waste is generated all efforts will be made to ensure it is not stored on site for any longer than is necessary.

4.6 Disposal and Treatment Options

- 4.6.1 It is anticipated that the majority of excavated material from the Scheme will be taken as "natural" material, the volume of which exceeds that for reuse on site. This material will therefore be transported to a suitable location such as a remediation scheme or another scheme. This approach works towards the commitments detailed within CoCP (Document Reference 6.10)
- 4.6.2 For excavated materials, suitable treatment, recycling and disposal facilities within a reasonable proximity of the Silvertown Tunnel site will be identified by the Contractor using the RSA. The RSA uses pre-determined criteria to assess the suitability of receptor sites to receive and treat the excavated materials. Sites which are closer to the Scheme will score higher than those which are further away.
- 4.6.3 For construction and demolition waste, suitable treatment, recycling and disposal facilities within a reasonable proximity of the Silvertown Tunnel site will be identified by the Contractor.
- 4.6.4 The Landfill Directive⁴ requires that disposal sites are classified into one of three categories dependent on the chemical composition of the material; these are hazardous, non-hazardous or inert.

- 4.6.5 The ability for waste to be deposited at these sites will be dependent on the available space and the conditions imposed on the sites through the relevant licence/permit. Such facility details have been assessed as part of the RSA process.
- 4.6.6 For excavated materials that are confirmed to be non-hazardous or inert, by chemical screening and are below hazardous waste thresholds, there are a number of reuse and recycling options that could be explored, both on and off site.
- 4.6.7 The excavated materials (excluding hazardous materials) could potentially be used as infill, bunding and/or landscaping for the Scheme. Further uses could include the construction or maintenance of pavements, footings for fencing etc. Materials produced could also be used in the laying of roads around the site or stored for later use, providing there are adequate storage areas and the materials are adequately managed to minimise dust and runoff. The use of these materials would need to be undertaken under the relevant regulatory controls and appropriate guidance.
- 4.6.8 For any vegetation removed consideration will be given to mulching and/or composting. Reuse of such materials will be considered where possible i.e. mulch or compost to be reused back in the Scheme for landscaping purposes.
- 4.6.9 Table 4-1 details waste minimisation measure that may be implemented during the construction phase.

Table 4-1 Waste Minimisation Measures

Waste Minimisation Measures		
	• Excavation will be for highways, tunnels and portals and foundations. It is anticipated that any waste produced through the construction of the tunnels will be cut and fill, it is anticipated that a small amount maybe reused elsewhere on site.	
Excavation	• Surplus excavated materials including soils, gravels and man-made fill can potentially generate the largest quantities of all the waste streams that will be removed off site. Preference for disposal should be given to off-site reuse following the waste hierarchy.	
	• Excavated material suitable for reuse, where appropriate, will be stored for reuse as landscaping material or infilling.	
Minimisation of vegetation clearance at the design	 The small amount of vegetation on site is predominantly grass with some shrubs; clearance of vegetation has the potential to be insignificant due to the nature of the area as former commercial and industrial/gasworks. 	
phase	 Identify, during the design phase, ways to minimise the loss of vegetation on site. Where minimisation is not possible, composting or mulching the vegetation will be considered for reuse in landscaping within the Scheme. 	
Minimisation of impacted materials arisings	 Where possible impacted material arisings will be treated and reused on site. This can act to minimise potential transport and disposal costs. This approach will be standard practice among designers and contractors. 	
Contractor targets	 The Contractor will consider setting off-cut/surplus targets for sub-contractors with a positive incentive scheme for on-site waste champions. 	
	On site good practice construction wastage rates for the total amount of construction material handled on site should be achieved.	
Imported material	• Enabling the purchase of materials in shape/dimension and form that minimises the creation of off-cuts waste. Avoiding over-purchasing as this can lead to significant wastage and will be avoided in the first place. Ensuring materials are ordered for delivery shortly before they are used on the Scheme will also avoid possible damage and therefore wastage.	
	 Secure storage to prevent damaged materials/theft. Keeping deliveries packaged until they are ready to be used and the inspection of deliveries on arrival helps to reduce damage and wastage. 	

Silvertown Tunnel

Site Waste Management Plan

Document Reference: 6.10

Waste Minimisation Measures		
Use of take back schemes	 Some suppliers offer a take back scheme, which will be utilised where practicable, particularly for packaging and pallets. 	
Monitoring and review	 Data from waste removal and the periodic review process (required as part of the SWMP) used to assess whether the waste objectives are being met, and if not to review procedures to steer the Scheme towards achieving them. This will require clear responsibilities to be identified, supported with authority and incentives to act on any deviations from the SWMP. 	
Education and awareness	 Site inductions are to include waste minimisation education and will be mandatory. Regular toolbox talks which all contractors and site workers will be expected to attend will also cover waste management education. 	
Consideration of end of life materials	 Consideration will be given to what will happen to the materials specified when they reach the end of their useful life. Where possible, elements will be designed for repair, modular repair, recycling at the end of life or safe disposal. The use of hazardous materials, in particular, will be minimised. 	

5. TRAINING

5.1 Overview

- 5.1.1 The Contractor will provide training to employees on the Scheme on the SWMP, roles and responsibilities and DoC.
- 5.1.2 On-site instruction and training covering appropriate waste separation, handling, storage, recycling, and reuse and return methods to be used by all parties (e.g. sub-contractors), at all appropriate stages of the Scheme.
- 5.1.3 Training will consist of an initial toolbox talk that will form part of the site induction process that introduces and explains the requirements of the SWMP and the concept of the waste hierarchy, as well as regular toolbox talks that provide updates on wider environmental issues. Workshops and other forms of training may be utilised where appropriate. Appropriate staff will be identified to attend these.
- 5.1.4 Where possible the Scheme will appoint a waste champion to act as a point of contact to deal with any waste queries from staff and promoting best waste management practice and be responsible for the upkeep of the SWMP.

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

6. MONITORING

6.1 Overview

- 6.1.1 Under the Waste (England and Wales) Regulations 2011² and in line with Duty of Care, the Contractor has a number of responsibilities which it will execute as follows:
 - To ensure that suitable storage is made available, including correct signage;
 - To check the waste carrier has the appropriate waste carrier licenses; and
 - To retain Waste Transfer Notes / Consignment Notes for 2/3 years as appropriate.
- 6.1.2 Responsibility for waste management lies with the Contractor unless a contractual agreement with sub-contractors to manage their own waste arisings exists the contractor would still ensure that the Duty of Care process is in place when this situation occurs.
- 6.1.3 Duty of Care details are to be logged within the WRAP SWMP template under the appropriate tabs. Details are to include the waste management licenses, waste carrier licenses and exempt site licenses for waste management contractors employed on the Scheme. All waste management contractors licences must be checked and verified before any waste movement occurs.
- 6.1.4 Appendix B contains an extract from the WRAP SWMP template which will be populated with such detail, including waste management facility information once the Scheme commences.
- 6.1.5 All Waste Transfer Notes will be safely stored for two years. Consignment notes for the transport of hazardous waste will be held for three years. The Scheme could also consider using electronic transfer notes rather than paper based. An example of such a system is Department for Environment Food & Rural Affair's Electronic Duty of Care (Defra Edoc).
- 6.1.6 In addition to monitoring and recording the performance of contractors, waste management facilities could be monitored periodically through the lifespan of the contract. This will ensure that high standards of compliance
and environmental performance are maintained throughout the supply chain, including accurate recording of waste types and the origin/destinations of wastes.

- 6.1.7 Waste monitoring will be included as an agenda item at construction progress meetings. In addition, this SWMP will be communicated to the whole project team (including TfL) at regular management meetings.
- 6.1.8 Throughout the project the Contractor will be responsible for reviewing the performance of all parties which are involved in the management of waste at the sites.
- 6.1.9 Progress towards the commitments contained in the CD&EMC (Appendix to Document Reference 6.10) will be reported regularly and performance will be reviewed and monitored in line with the CoCP (Document Reference 6.10)
- 6.1.10 A comparison of forecast and actual waste streams with key associated notes should be documented. This information could consist of explanation for diversion, lessons learnt and revision of plans.

7. ACTUAL QUANTITIES

7.1 Overview

- 7.1.1 Within the SWMP a record of the type and quantities of waste actually being produced and what has happened to this waste will be kept.
- 7.1.2 Appendix C contains an extract from the 'Actual Waste Movements' tab of the WRAP SWMP spreadsheet, to be completed. The table provides an example of how the waste movements will be logged, which ultimately feeds into actual waste figures for each waste activity (on subsequent WRAP template tabs).
- 7.1.3 Under this section the Contractor would be responsible for recording the: waste activity, ELoW code, method of management and where required the off-site carrier and destination and waste totals (amongst others).
- 7.1.4 Maintaining these records will also help to identify which waste streams are not achieving anticipated recycling potentials so that alternative methods to handle that waste stream can be explored for the remainder of the Scheme.
- 7.1.5 The actual quantities table within the WRAP SWMP template will be completed during construction phase and data taken from the amounts recorded on waste transfer notes and hazardous waste consignment notes.
- 7.1.6 This information then feeds into the summary of KPIs tab, which details key commitment information such as the amount of recycled, reused/beneficially reused and diverted from landfill quantities.

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

8. REVIEW

8.1 Overview

- 8.1.1 The SWMP shall be reviewed on a regular basis, with waste data being entered frequently i.e. fortnightly or monthly. Further reviews shall take place where any significant changes occur. A log will be kept of when the plan has been reviewed and the outcomes.
- 8.1.2 An appropriate monitoring regime of the waste objective and targets shall be put in place.
- 8.1.3 Table 8-1 provides an example of a SWMP review checklist. This checklist should be completed at the start of the project, where possible and reviewed throughout the duration of project.

CHECKS – ENTER YES OR NO	YES	NO
Has a Waste management (WM) Contractor(s) been appointed with terms and conditions agreed?		
Have all WM Contractor(s) Carriers & Disposal Licences been checked & verified?		
Has a data reporting procedure been agreed with the WM Contractor(s)?		
Has a waste compound and segregation area been adequately set up & resourced?		
Has a SWMP implementation planning meeting been set up?		
Has the waste management document control process been set up?		
Have all necessary staff read & signed the SWMP?		
Have all site and sub-contractor staff been trained / briefed on SWMP requirements?		
Have waste management objective and targets been set?		
Has the SWMP received approval from the Project Lead?		
List comments and further actions if necessary.		

Table 8-1 Review Checklist

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

9. POST SCHEME COMPLETION

- 9.1.1 At the end of the Scheme, both TfL and the Contractor are responsible for reviewing, revising and refining the SWMP as necessary, to ensure best practice and to identify if lessons could be learned for the next time a similar scheme is undertaken. This review will aim to identify and conclude the following:
 - Confirmation that the SWMP has been monitored and updated within the defined timescales;
 - An explanation of any deviation from the original plan;
 - A comparison of the estimated quantities of each waste type against the actual quantities generated; and
 - An action plan to address the lessons that have been learnt from the Scheme that could be implemented for the next scheme.
- 9.1.2 An estimation of the cost savings (if any) that have been achieved through the measures undertaken to minimise, reuse, recycle or recover waste arisings rather than disposal to landfill will be recorded.
- 9.1.3 The "reporting" tab of the SWMP summarises key performance indicators such as: diversion from landfill, cost of waste disposal and recovery of materials.

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

Appendix A. Waste Forecast Tables (WRAP SWMP Extract)

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK







Forecast Waste						Forecast	Quantities	Calculated (Converting be	Quantities	
C, D or E Activity	Waste Stream	Material Type	Further description of waste - optional	Suggested LOW Code	Waste or Re-Use	(m³)	(tonnes)	(m³)	(tonnes)	Forecast provided by
Excavation	Packaging	plastic packaging	plastic packaging	15 02 02	Off-site destination	###	###	###	###	A.N Other
Excavation	Inert - Soil & stones	soil and stones (inert) other than those mentioned in 17 05 03	material from surface works.	17 05 04	Off-site segregated		56,319	45055.20	56319.00	S. Panesar
Excavation	Non Haz (Non Inert) - Soil & stones	Topsoil (non inert)	material from surface works and tunnel boring.	17 05 04	Off-site segregated		740,005	592004.00	740005.00	S. Panesar
Excavation	Segregated Haz - Soil & stones	soil and stones containing dangerous substances	material from surface works and tunnel boring.	17 05 03*	Off-site segregated		194,965	155972.00	194965.00	S. Panesar
Construction	Inert - mixture of concrete, bricks, tiles etc.	concrete	grout (tunnel) - miss pours and surplus - industry waste rate 7.5%. Values from Cost Estimate (Feb 2016).	17 01 01	Off-site segregated	1,003		1003.00	1273.81	S. Panesar
Construction	Inert - mixture of concrete, bricks, tiles etc.	concrete	STL (secondary tunnel lining) in situ concrete - miss pours and surplus concrete, industry waste rate 2.5%. Values from Cost Estimate (Feb 2016).	17 01 01	Off-site segregated	375		375.00	476.25	S. Panesar
Construction	Inert - mixture of concrete, bricks, tiles etc.	concrete	PTL (primary tunnel lining) - segmental concrete - damage to components - based on industry average waste rate of 5%. Values from Cost Estimate (Feb 2016).	17 01 01	Off-site segregated		1,662	1308.66	1662.00	S. Panesar
Construction	Inert - mixture of concrete, bricks, tiles etc.	concrete	structural concrete - miss pours and surplus - based on industry average waste rate of 2.5%. Values from Cost Estimate (Feb 2016)	17 01 01	Off-site segregated	3,074		3074.00	3903.98	S. Panesar
Construction	Inert - mixture of concrete, bricks, tiles etc.	concrete	mass concrete - miss pours and surplus - based on industry average waste rate of 2.5%. Values from Cost Estimate (Feb 2016).	17 01 01	Off-site segregated	374		374.00	474.98	S. Panesar
Construction	Inert - mixture of concrete, bricks, tiles etc.	concrete	structural concrete - utility building - miss pours and surplus - based on industry average waste rate of 2.5%. Values from Cost Estimate (Feb 2016).	17 01 01	Off-site segregated		78	61.42	78.00	S. Panesar

Transport for London (TfL) Atkins Silvertown Reference Design

0



Forecast Waste						Forecast	Quantities	Calculated (Converting be	l Quantities tween m ³ and t)	
C, D or E Activity	Waste Stream	Material Type	Further description of waste - optional	Suggested LOW Code	Waste or Re-Use	(m ³)	(tonnes)	(m ³)	(tonnes)	Forecast provided by
Construction	Inert - mixture of concrete, bricks, tiles etc.	mixtures of concrete, bricks, tiles and ceramics other than those mentioned in 17 01 06	Ruble and hardcore-utility building - damage to components, miss pours and surplus concrete / screed / blinding - based on industry average waste rate of 2.5%. Values from Cost Estimate (Feb 2016).	17 01 07	Off-site mixed		18	14.52	18.00	S. Panesar
Construction	Metals	iron and steel	structural steel - off cuts based on industry average waste rate of 7.55%. Values from Cost Estimate (Feb 2016).	17 04 05	Off-site segregated		110	268.29	110.00	S. Panesar
Construction	Metals	iron and steel	structural steel (reinforcing bar) - off cuts - based on industry average waste rate of 7.55%. Values from Cost Estimate (Feb 2016).	17 04 05	Off-site segregated	1,438		1438.00	589.58	S. Panesar
Construction	Inert - Glass	glass	Structural glass- based on industry average waste rate of 5%. Values from Cost Estimate (Feb 2016).	17 02 02	Off-site segregated		1	1.64	1.00	S. Panesar
Construction	Metals	mixed metals	VRS offcuts or surplus material. Based on 29kg per m. Value taken from Cost Estimate (Feb 2016)	17 04 07	Off-site segregated		1	2.38	1.00	S. Panesar
Construction	Metals	mixed metals	non ferrous - offcuts - calculated on 1% of total length of service buildings in metres. Values from tunnel service building quantities (Feb 2016)	17 04 07	Off-site segregated		4	9.52	4.00	S. Panesar
Construction	Metals	iron and steel	structural steel / reinforcements (utilities buildings) - offcuts - based on industry average waste rate of 7.55%. Values from tunnel service building qauntities (Feb 2016)	17 04 05	Off-site segregated		136	331.71	136.00	S. Panesar
Construction	Inert - mixture of concrete, bricks, tiles etc.	bricks	bricks -off cuts, damaged and surplus material - based on industry average waste rate of 5%. Values from tunnel service building quantities (Feb 2016)	17 01 02	Off-site segregated		218	181.67	218.00	S. Panesar

Transport for London (TfL) Atkins Silvertown Reference Design

0





Forecast Waste						Forecast	Quantities	Calculated (Converting be	l Quantities tween m ³ and t)	
C, D or E Activity	Waste Stream	Material Type	Further description of waste - optional	Suggested LOW Code	Waste or Re-Use	(m³)	(tonnes)	(m ³)	(tonnes)	Forecast provided by
Construction	Inert - mixture of concrete, bricks, tiles etc.	tiles and ceramics	tiles - offcuts, damaged and surplus material - based on industry average waste rate of 5%. Values from tunnel service building quantities (Feb 2016)	17 01 03	Off-site segregated		2	3.39	2.00	S. Panesar
Construction	Gypsum (17 08 02)	gypsum-based construction materials other than those mentioned in 17 08 01	Gypsum from utilities buildings. 5% wasteage rate. Values from Cost estimate (Feb 2016).	17 08 02	Off-site segregated		12	36.36	12.00	S. Panesar
Construction	Packaging	mixed packaging	packaging from all components delivered to the site for works. No data available so assumed 1% of total waste as a result of imported material from the service buildings	15 01 06	Off-site mixed		126	600.00	126.00	S. Panesar
Construction	Other C&D segregated waste	track ballast other than those mentioned in 17 05 07	type 1 sub base - surplus. based on industry average waste rate of 13%. Values from Cost Estimate (Feb 2016)	17 05 08	Off-site segregated	3,405		3405.00	3701.92	S. Panesar
Excavation	Inert - Soil & stones	soil and stones (inert) other than those mentioned in 17 05 03	from cosntruction work- imported fill (soil); surplus material. based on industry average waste rate of 13%. Values from Cost Estimate (Feb 2016).	17 05 04	Off-site segregated	7,906		7906.00	9882.50	S. Panesar
Construction	Mixed C&D waste (17 09 04)	mixed construction and demolition wastes other than those mentioned in 17 09 01, 17 09 02 and 17 09 03	lighting - offcuts, damaged and surplus material. Based on 1% of total length of tunnels in metres. Values taken from Cost Estimate (July 2015) and (Feb 2016).	17 09 04	Off-site mixed	28		28.00	24.36	S. Panesar
Construction	Other C&D segregated waste	bituminous mixtures other than those mentioned in 17 03 01	bitumen surfacing - including roads and tunnels - miss pours and surplus material. Assumed to include kerbs, pathways, cycleways). Waste rate of 4.9%. Values taken from Cost Estimate (Eeb 2016)	17 03 02	Off-site segregated	619		619.00	507.58	S. Panesar
Construction	Wood	wood	Timber from utilities buildings. based on industry average waste rate of 3.8%. Values from Cost Estimate (Feb 2016).	17 02 01	Off-site segregated		17	50.00	17.00	S. Panesar

WIGP

Transport for London (TfL) Atkins Silvertown Reference Design

0



Forecast Waste						Forecast	Quantities	Calculated (Converting bet	Quantities ween m ³ and t)	
C, D or E Activity	Waste Stream	Material Type	Further description of waste - optional	Suggested LOW Code	Waste or Re-Use	(m³)	(tonnes)	(m³)	(tonnes)	Forecast provided by
Construction	Other C&D segregated waste	household plastics	drainage pipes, cables and ducts: offcuts - damaged and surplus material. Based on 1% of total length in metres. Value taken from Cost Estimate (Feb 2016)	20 01 39	Off-site segregated	368		368.00	51.52	S. Panesar
Demolition	Wood	wood	timber from building demolition	17 02 01	Off-site segregated	566		566.00	192.44	S. Panesar
Demolition	Gypsum (17 08 02)	gypsum-based construction materials other than those mentioned in 17 08 01	gypsum from building demolition	17 08 02	Off-site segregated	963		963.00	317.79	S. Panesar
Demolition	Metals	mixed metals	nonferrous metal from building demolition	17 04 07	Off-site segregated	482		482.00	202.44	S. Panesar
Demolition	Metals	iron and steel	ferrous metal from building demolition.	17 04 05	Off-site segregated	28,825		28825.00	11818.25	S. Panesar
Demolition	Inert - Glass	glass	glass from building demolition.	17 02 02	Off-site segregated	48		48.00	29.28	S. Panesar
Demolition	Inert - mixture of concrete, bricks, tiles etc.	tiles and ceramics	tiles form building demoliton.	17 01 03	Off-site segregated	157		157.00	92.63	S. Panesar

Transport for London (TfL) Atkins Silvertown Reference Design

0

25/04/2016

Appendix B. Waste Carriers (WRAP SWMP Extract)

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK







peeny waste oe					Specify wa	ste Manage	ement Facilities				
Name	Contact Details	Date checked with Environment Agency (dd/mm/yyyy)	Registration Number	Expiry Date (dd/mm/yyyy)	Name		Type of facility	% reused if known	% recycled if known	% energy recovery if known	tota of
											1
											_
											_
											_
											_
											_
											_
											_
											_
											_
											+
											+
											+
											+
											+
											+
											+
											+
											+
											+

Transport for London (TfL) Atkins Silvertown Reference Design 0

ms y	Overall diverted from landfill / recovery
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%
	0%

Appendix C. Actual Waste Movements

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK





? Tell me about this sheet

Waste Totals

Display summary as:	Waste Stream	Total waste arising (Tonnes)	Total material retained on site (Tonnes)	Total waste sent offsite (Tonnes)	Total waste to landfill (Tonnes)	Total waste recovered offsite (Tonnes)	Cost of waste disposal
Tonnes	Inert - Soil & stones						£0.00
	Hazardous - Soil & stones						£0.00
	Non Haz (Non Inert) - Dredgings						£0.00
	Segregated Haz - Soil & stones						£0.00
	Gypsum						£0.00
	Metals						£0.00
	Wood						£0.00
	Packaging						£0.00
	Inert - Building rubble						£0.00
	Inert - Glass						£0.00
	Mixed Hazardous - C&D waste						£0.00
	Mixed C&D waste						£0.00
	Segregated Haz Waste						£0.00
	Other C&D segregated waste						£0.00
	Total						£0.00

Actual W	aste Mov	rements											١	Naste Totals		
Movement Number	C, D or E Activity	Waste Stream	Material Type	Further description of waste - optional	LOW Code used	On or off-site destination	Off-site carrier	Off- site destination	Overide facility recovery rate for individual skip	Overall diversion from landfill / recovery (further detail on Sheet 4)	Date of Movement(s) (dd/mm/yyyy)	(m³)	(tonnes)	Actual Cost	£/m³	£/t
1										100%						
2										100%						
3										100%						
4										100%						
5										100%						
6										100%		<u> </u>				
7										100%		<u> </u>				
8										100%						
9										100%						
10										100%						
11										100%						
12										100%						
13										100%						
14										100%						
15										100%						
16										100%						
17										100%						
18										100%						
19										100%						
20										100%						
21										100%						
22										100%						
23										100%						
24										100%						
25										100%						
26										100%						
27										100%						
28										100%						
29										100%						
30										100%						
31										100%						
32										100%						
33										100%						
34										100%						
35										100%						
36										100%						
37										100%						
38										100%						

Transport for London (TfL) Atkins Silvertown Reference Design 0

Appendix F. Groundwater Monitoring Strategy

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

NTKINS

Silvertown Tunnel

Groundwater Monitoring Strategy Transport for London

September 2016

Notice

This document and its contents have been prepared and are intended solely for Transport for London's information and use in relation to the baseline groundwater monitoring for Silvertown Tunnel.

Atkins Limited assumes no responsibility to any other party in respect of or arising out of or in connection with this document and/or its contents.

Document history

Job numb	er: 5123288		Document ref: STWTM-ATK-EWE-XXXX-RP-EN- 0002					
Revision	Purpose description	Originated	Checked	Reviewed	Authorised	Date		
1.0	Draft - internal	КВ	ESW	RS		09 March 2016		
2.0	Working draft	KB	ESW	RS		18 March 2016		
P03	Formal issue	KB	ESW	RS / JS	MRM	30 June 2016		
P04	Incorporating Pinsent and TfL's comments	КВ	ESW	RS / JS	MRM	15 August 2016		
P05	Incorporating TfL and the Environment Agency's comments	KB	RS	RS	MRM	08 September 2016		

Client signoff

Client	Transport for London
Project	Silvertown Tunnel
Document title	Silvertown Tunnel Groundwater Monitoring Strategy
Job no.	5123288
Copy no.	Rev P05
Document reference	STWTM-ATK-EWE-XXXX-RP-EN-0002

Table of contents

Chapter Pages 1. Introduction 1 1.1. Objectives 1 1.2. **Project Context** 2 2. Scope of Groundwater Monitoring 3 3. Site Setting 4 3.1. 4 Geology 3.2. Hydrogeology 5 3.3. Licensed Abstractions 5 3.4. **Historical Site Uses** 6 3.5. 6 Silvertown Tunnel Construction Methodology 3.6. Potential Pollutant Linkages 7 9 4. Groundwater Monitoring Strategy 4.1. Groundwater Monitoring Strategy 9 9 4.2. Baseline Assessment and Compliance Criteria for Groundwater 4.3. 10 Alert and Trigger Levels 4.4. Sentinel Monitoring Boreholes 10 4.5. **Deliverables** 11 4.6. Assessment of Risk 11 4.7. **Contingency Action Plan** 11 4.8. Groundwater Monitoring and Verification Plan 12 5. **Requirements for Groundwater Monitoring and Verification Plan** 13 5.1. Proposed Scope of Groundwater Monitoring and Verification Plan 13 5.2. Monitoring 13 5.3. **Monitoring Schedule** 15 5.4. 16 Dewatering 6. **Contingency Planning** 18 6.1. Assessment Criteria (Alert Levels) 18 6.2. Compliance Criteria (Trigger Levels) 19 6.3. 20 **Emergency Measures** 21 7. Reporting 7.1. Notification report 21 7.2 21 **Factual Report** 7.3. Interpretative Report 22

Appendices

Appendix A. Proposed Baseline Monitoring Programme

- A.1. Proposed Baseline Monitoring Programme
- A.2. Access Arrangements and Health and Safety
- A.3. Baseline Monitoring Network
- A.4. Baseline Monitoring Programme

Appendix B. Proposed Sample Collection Methodology

- B.1. Groundwater Elevation Loggers (Divers) and Barologgers
- B.2. Interface Meter
- B.3. Monitoring Groundwater Chemistry
- B.4. Baseline Monitoring Reporting

Appendix C.	Analytical Suite
Appendix D.	Greenwich Peninsula Borehole logs

Appendix E. Silvertown Site Borehole logs

Tables

Table 3-1	Regional Geological Sequence	4
Table 3-2	Typical Strata depth and thicknesses for Silvertown and Greenwich	4
Table A-1	Groundwater Monitoring Location Specifications - Greenwich Peninsula	28
Table A-2	Groundwater Monitoring Location Specification – Silvertown	29
Table A-3	Greenwich Peninsula Monitoring Requirements	30
Table A-4	Silvertown Monitoring Requirements	30
Table C-1	Analytical Suite -A	35
Table C-2	Analytical Suite - B	35
Table C-3	Additional Analytical Suite - C	36

List of Abbreviations

AGS	The Association of Geotechnical and Geoenvironmental Specialists
AOD	Above Ordinance Datum
BGS	British Geological Society
BTEX & MTBE	Benzene Toluene Ethylbenzene Xylene and Methyl tert-butyl ether
CEMP	Construction Environmental Management Plan
CLEA	Contaminated Land Exposure Assessment
CQA	Construction Quality Assurance
COC	Contaminants of Concern
COCP	Code of Construction Practice
DCO	Development Consent Order
DNAPL	dense non aqueous phase liquid
DQRA	Detailed Quantitative Risk Assessment
ECC	Buro Happold Engineering
EH	Oxidation/Reduction Potential
EOH	end of hole
EPA	Environmental Protection Agency.
EPR	Environmental Permitting Regulations
GAC	granular activated carbon
GLA	Greater London Authority
GP-LTGWM	Greenwich Peninsula Long Term Groundwater Monitoring
m	metre
MCERTS	Environment Agency Monitoring Certification Scheme
MS	Microsoft
NAPL	non aqueous phase liquid
QA/QC	Quality Assurance / Quality Control
Р	Proven
PAH	polycyclic aromatic hydrocarbons
PINS	Planning Inspectorate
RTD	River Terrace Deposits
STWTN-GI	Atkins Silvertown Tunnel Ground Investigation Project
SVOC	Semi-Volatile Organic Compounds
ТВМ	Tunnel Boring Machine
TfL	Transport for London
TOC	Total Organic Carbon
TPH CWG	Total Petroleum Hydrocarbons Criteria Working Group
UKAS	United Kingdom Accreditation Service
VOC	Volatile Organic Compounds

1. Introduction

The Silvertown Tunnel, a Transport for London (TfL) managed project, will provide an additional highway transport link between the Greenwich Peninsula and the Royal Docks at Silvertown. The Silvertown Tunnel project is currently in the process of obtaining a Development Consent Order (DCO), with the DCO application accepted for examination on 31 May 2016 by the Planning Inspectorate (PINS). Detailed design of the scheme is to be completed following engagement with a Contractor with procurement to be undertaken concurrently with the DCO process. At detailed design stage, further details regarding the exact tunnel alignment and dewatering requirements will be established.

This document outlines the Groundwater Monitoring Strategy which has been developed to inform the Environmental Statement and Code of Construction Practice, part of the DCO application. It is intended to provide the necessary overarching framework to allow for production of a detailed Groundwater Monitoring and Verification Plan, as part of detailed design and as required by the Code of Construction Practice, and to assess the potential impact to groundwater resources as a result of tunnel construction.

1.1. Objectives

This Groundwater Monitoring Strategy is intended to outline the requirements for the monitoring of groundwater during the baseline, construction and post construction phases of the Silvertown Tunnel project.

The overall aim of this document is to set out a strategy to monitor and manage the effects of the Silvertown Tunnel development, during and post construction, on groundwater quality and quantity within both the Upper and Lower Aquifers across Greenwich and Silvertown as a result of construction (including dewatering if required). This includes measures to monitor the following:

- baseline groundwater quality and quantity conditions prior to construction works being undertaken;
- changes in groundwater quality and levels as a result of dewatering and the effectiveness of mitigation measures to counter these effects;
- changes to the local hydrogeological regime as a result of construction and the effects on neighbouring structures and associated potential contaminant transport;
- the effects of long-term changes in groundwater pressures and the design of the tunnels and structures to accommodate these changes;
- potential for mobilisation of contaminants already present in groundwater by the creation of alternative pathways, or significantly altering existing pathways using sentinel monitoring boreholes to assess the impacts;
- early identification of trends towards derogation of pumping rights. Such monitoring would also inform mitigation actions that would need to be taken in the event of (or trends towards) derogation; and
- potential increased turbidity in groundwater due to the physical action of tunnelling construction within the Lower Aquifer and subsequent migration with the prevailing groundwater flow.

This document sets out a strategy to monitor groundwater during the life cycle of the Silvertown Tunnel project. The strategy is set out as follows:

- the development and implementation of baseline groundwater monitoring, where the data will be used to inform the following:
 - detailed tunnel design and subsequent dewatering requirements (if required);
 - site specific preliminary groundwater risk assessments and detailed groundwater risk assessments if required;
 - development of alert and trigger levels for construction;
 - a Groundwater Monitoring and Verification Plan; and

- a Dewatering Strategy (if required).

The content of the various report sections is as follows:

- Sections 3 provide site specific information with regards to the Silvertown Tunnel alignment including geology and hydrogeology, details of known sources of groundwater contamination and an outline of potential source-pathway-receptor linkages present along the alignment;
- Section 4 provides details and objectives of the Groundwater Monitoring Strategy;
- Section 5 outlines the requirements to be included within the Groundwater Monitoring and Verification Plan, including the overarching management protocols for the construction phase (and post construction phase) and an outline content for a monitoring plan to be produced by the appointed Contractor or their nominated representatives; and
- Section Error! Reference source not found. outlines the requirements to be included in the dewatering strategy and dewatering plan and summarises the project's approach to dewatering.

1.2. Project Context

1.2.1. Site Details

The proposed Silvertown Tunnel will link the Greenwich Peninsula at the A102 Blackwall Tunnel approach to the Royal Docks at Silvertown Way. The proposed tunnel alignment is shown on the General Arrangement Drawings (document reference 2.2). The tunnel will pass under the River Thames and will follow the general alignment of the London (Emirates) Cable Car.

For the purposes of groundwater monitoring the site is divided into two separate areas:

- **Silvertown** located north of the River Thames within the London Borough of Newham. This area includes the Thames and Clyde Wharves with the Canning Town London Underground and Docklands Light Rail (DLR) station located to the north. The Royal Victoria Docks are located to the east approximately 100 m away from the proposed northern tunnel portal entrance; and
- **Greenwich Peninsula** located to the south of the River Thames within the Royal Borough of Greenwich. This development area includes land around Edmund Halley/Millennium Way and Cutter Lane, with the O2 arena located on the northern edge. The area extends south within the confines of the Blackwall Tunnel approach and West Parkside.

Numerous assessments have been completed for the Silvertown Tunnel project which include in depth descriptions of the site. A concise background description is included within this strategy document (Section 3.4). For further information regarding the site and hydrogeological setting reference should be made to the Silvertown Tunnel 6.10 Environmental Statement, referenced below.

• Transport for London, 2016. Silvertown Tunnel 6.1 Environmental Statement TR010021 (ST150030-PLN-ZZZ-DSD-ZZ-0066), April 2016.

2. Scope of Groundwater Monitoring

The groundwater monitoring programme will be initiated ahead of construction to enable collection of baseline data. The data obtained would then be used to quantify any potential impacts produced by the Scheme. The groundwater monitoring programme would be designed to allow for ongoing construction monitoring. This would then be used to compare against pre-construction predictions and to initiate trigger and control measures, in the event of an identified risk starting to materialise. The Groundwater Monitoring and Verification Plan will contain measures for the monitoring of baseline, construction and post construction effects.

The monitoring regimes will be described in a Groundwater Monitoring and Verification Plan to be implemented for the life of the project. This is to ensure that TfL and the Contractor can identify and respond to changes in groundwater levels or groundwater quality as a result of any changes in the project design and/or the site conceptual model.

2.1.1. Baseline Groundwater Monitoring

Pre-construction baseline monitoring should commence as soon as practicable and continue until the commencement of construction or the implementation of the construction phase of the Groundwater Monitoring and Verification Plan.

The pre-construction monitoring will be used to establish a baseline which will inform the setting of alert and trigger levels, for both water quality and groundwater elevations, against which the construction and post construction phases can be compared. These alert and trigger levels will be proposed by TfL or their nominated representatives for discussion and agreement with the Environment Agency, as the principal controlled waters regulator. Any changes in the alert and trigger levels proposed by the appointed Contractor must be approved by TfL and agreed with the Environment Agency before being adopted.

2.1.2. Construction Monitoring

Construction groundwater monitoring will be undertaken throughout the project construction programme. The results shall be reported to TfL and the Environment Agency.

The monitoring requirements during construction will be outlined in the Groundwater Monitoring and Verification Plan (Section 5) to be completed following detailed design by the Contractor.

2.1.3. Post Construction Monitoring

Post construction monitoring will be required as part of the Groundwater Monitoring and Verification Plan. The extent of this monitoring will depend on the extent of impact the development has on the groundwater environment which will be identified through assessment of the baseline and construction monitoring data and the associated detailed design. TfL or its nominated representative will be responsible for ensuring that post construction monitoring is undertaken and satisfactorily reported.

Post construction monitoring will be undertaken until groundwater elevations and groundwater quality stabilise (details of what constitutes 'stabilised' conditions are to be confirmed). It is envisioned that a minimum duration of six months of monitoring will be required following the completion of construction, however the length of post construction monitoring will need to be assessed and agreed with the Environment Agency following detailed design stage.

3. Site Setting

3.1. Geology

The regional geology of the project area includes extensive made ground located to the northeast and southeast of the proposed routes of the Silvertown Tunnel. Table 3-1 below presents the regional geological sequence and typical strata boundaries and associated thicknesses are presented in Table 3-2 below.

Table 3-1	Regional	Geological	Sequence

System	Series	Formation		
Quaternary	Holocene	Alluvium		
	Pleistocene	River Terrace Deposits (RTD)		
Palaeogene	Eocene	Bagshot Formation		
		Thames Group (London Clay)		
	Palaeocene	Lambeth Group		
		 Woolwich Formation: Shelley clay, laminated beds with sand channels, 2 to 4 m thick in central London 		
		 Reading Formation: Highly variable; mottled clay and silt. Fine to medium grained sand in layers and channels may constitute up to 60% 		
		 Upnor Formation: Glauconitic sand with thin clay seams 		
		Thanet Sands		
Cretaceous	Upper	Chalk Group		

Table 3-2 Typical Strata depth and thicknesses for Silvertown and Greenwich

Formation	Greenwich			Silvertown		
	Top (m AOD) (Min – Max)	Bottom (m AOD) (Min – Max)	Thickness (m) (Min – Max)	Top (m AOD) (Min – Max)	Bottom (m AOD) (Min – Max)	Thickness (m) (Min – Max)
Made Ground	2.13 – 5.72	-0.91 – 2.57	0.91 – 6.2	1.35 – 5.28	-9.22 – 1.76	1.0 - 14.50
Alluvium	-0.91 – 2.57	-3.950.48	1.22 – 4.5	-3.23 – 1.76	-5.95 (EOH) - -1.1	1.45 – 7.7 (EOH)
River Terrace Deposits	-3.95 – -0.48	-10.966.88	5.95 - 8.38	-5.84 – -1.1	-8.74 - 4.43	1.6 – 4.4
London Clay Formation	-10.966.88	-16.93 – -11.86	0.9 - 6.8	-9.224.43	-22.3 – 16.54	9–17.9 (P)
Harwich Formation	-16.9314.48	-22.7615.39	1.02 – 5.83	-20.7619.48	-25.4820.54	0.52 - 5.17
Lambeth Group	-22.766.88	-35.2618.8	8.9 - 14.8	-25.4820.15	-40.08 (EOH) - -27.83	5.0 – 15.83
Upnor Formation	-35.26	-37.41	2.15	-38.9736.37	-40.4739.33	1.5 – 2.96
Thanet Formation	-37.41 – -18.8	-45.9 – 29.5	10.7 – 12.5 (P)	-40.4739.33	-52.52 - 50.44	10.02 – 13.19
Chalk	-45.9	N/A	N/A	-52.5250.44	N/A	N/A

P – Proven EOH – End of Hole Scour features have been identified during multiple investigations (BGS boreholes, Blackwall Tunnel assessments, Greenwich Peninsula gasworks assessments and Crossrail assessments) within the vicinity of the proposed Silvertown tunnel alignment. Three known scour features located within the vicinity of the assessment area are the Blackwall scour feature, Lea scour feature and the Greenwich Peninsula scour feature, the presence of further scour features in the vicinity of the site cannot be ruled out.

At the current design stage (reference design), geological cross sections produced during the Silvertown Tunnel Ground Investigation, indicate the Tunnel will be constructed through the Made Ground, Alluvium River Terrace Deposits, London Clay, Harwich and Lambeth Group.

3.2. Hydrogeology

There are two distinct aquifers identified in the area; an Upper Aquifer present within the RTD which is located above the low permeability London Clay Formation; and a Lower Aquifer within the Lambeth Group, Thanet Sand and Chalk. The Chalk is generally considered to be in hydraulic continuity with the overlying Thanet Sand and Lambeth Group though this is constrained to some degree by the elevated fines and subsequent relatively low vertical permeability continuity of the lower part of the Lambeth Group and the Thanet Sand. Also localised perched aquifers have been identified within the Alluvium and Made Ground.

The Environment Agency has classified the different geological units based on the water-bearing permeability of the deposit from which groundwater can be extracted. The Environment Agency has classified the Chalk as a Principal Aquifer based on its high fracture permeability meaning it has the ability to provide a high level of water storage and may support water supply and/or river base flow on a strategic scale. The Lambeth Group and Thanet Sands are classified as bedrock Secondary A Aquifers by the Environment Agency meaning they are capable of supporting water supplies at a local rather than strategic scale, and in some cases forming an important source of base flow to rivers. The RTD is also classified as a Secondary A Aquifer by the Environment Agency while the Alluvium is classified as the Secondary Undifferentiated Aquifer meaning the deposit could not be classified as either a Secondary A or a Secondary B (predominantly lower permeability layers which may store and yield limited amounts of groundwater due to localised features such as fissures, thin permeable horizons and weathering).

The London Clay is considered relatively impermeable and where present effectively hydraulically separates the Upper and Lower Aquifers. The London Clay is thickest on the northern side of the River Thames (Silvertown assessment area) and is thin (and maybe locally absent) on the southern side of the River Thames (Greenwich assessment area). Scour features also have the potential to increase the hydraulic connectivity between the Upper and Lower Aquifers. The Lambeth Group has been described as being predominantly granular in east London, and as such there may be some connectivity between the Upper and Lower Aquifers in the area where the London Clay is locally thin or absent. Where present, the laminated beds at the base of the Lambeth Group decreases this connectivity. The Upper Aquifer is likely to be subject to tidal influence due to its proximity to the River Thames. Perched water may be present within the Made Ground.

According to the Envirocheck report (2013) the proposed tunnel crossing is over 3 km away from the closest (to the north and southwest) groundwater source protection zones.

Long term groundwater monitoring has been completed in the vicinity of the Site for previous projects. These include the Greenwich Peninsula Long-Term Groundwater Monitoring programme and groundwater monitoring associated with the Crossrail project.

3.3. Licensed Abstractions

At the time of DCO submission there are 30 Environment Agency licensed abstractions within 5 km of the Silvertown Tunnel from both the Upper and Lower Aquifers as well as surface water. Of the 30, four licensed abstractions are within close proximity to the Silvertown Tunnel alignment on the Greenwich Peninsula and two are in close proximity to the tunnel alignment on the Silvertown side. Full details of these abstractions are presented in the Environmental Statement, Chapter 12 Geology and Soils. Further assessment into active licensed abstractions and the potential risk to this receptor will be assessed within the Assessment of

Risk described in Section 4.6 to be completed following both detailed design and baseline groundwater monitoring.

3.4. Historical Site Uses

Both Silvertown and the Greenwich Peninsula have historically been used for heavy industrial activity.

3.4.1. Silvertown

The principal contamination sources in the Silvertown assessment area are related to the former land uses including railway land (including coal and goods depots), manure works, chemical works, oil storage facilities, dockland warehouses, garages, and engineering works, as well as those associated with continued light industrial use. Present day activities include an aggregate manufacturing facility, waste recycling facilities and a scrap metal yard.

In addition to the potential on-site sources above, a number of potential off-site sources have been identified including the paint works to the south of Clyde Wharf and surrounding warehouses. Historical off-site land uses such as a former manure works, iron works and soap works are also considered potential contamination sources.

A landfill is situated at the location of the proposed northern tunnel portal and is associated with the infilling of the former Western Entrance lock to the Royal Victoria Dock.

3.4.1.1. Potential Groundwater Sources

An extensive investigation into identifying potential sources of groundwater contamination within the Silvertown site has not been completed to date. The main potential source of groundwater contamination is the former western entrance to Victoria Dock which has been backfilled and is designated as a landfill, the Silvertown Tunnel alignment crosses this dock at an angle. The various other land uses listed above, either along the alignment or in the vicinity, also have the potential to be sources of groundwater contamination.

3.4.2. Greenwich Peninsula

Historically the Greenwich Peninsula was dominated by the South Metropolitan Gas Works; considered to be a major source of ground and groundwater contamination. A single remaining gas holder and ancillary equipment currently operated by SGN Ltd. are the sole above ground remnants of the facility. The key sources of contamination identified during phases of ground investigation and remediation works in the 1990s on the Greenwich Peninsula, such as tar tanks and contamination hotspots, were removed; however residual soil and groundwater contamination is known to remain beneath much of the assessment area. This is particularly the case within the RTD, where residual dense non aqueous phase liquids (DNAPLs) have pooled in depressions in the London Clay surface.

To the southwest of the remaining gasholder, neighbouring the Blackwall Tunnel Approach, is a registered waste treatment / recovery site. It is understood that the site has since surrendered its licence and been issued with a completion certificate by the Environment Agency.

3.4.2.1. Potential Contaminant Groundwater Sources

Groundwater monitoring at Greenwich Peninsula has shown that the Upper Aquifer is predominantly contaminated, and is therefore considered a source of contamination. The north of the Greenwich Peninsula has specifically been identified as a zone of increased contamination within the Upper Aquifer as well as localised points within the Lower Aquifer. This area is understood to have held the main tar works including underground tar storage tanks, the London Clay is also locally thin or absent in this area and the Blackwall scour feature is located to the north northwest. In addition, inorganic contamination (specifically ammoniacal nitrogen) has been identified within the Lower Aquifer in other parts of the peninsula.

3.5. Silvertown Tunnel Construction Methodology

It is envisioned that two main proposed construction methods of cut-and-cover and bored tunnel will be used, Appendix B of the Construction Method Statement (ES Appendix 4A) (document reference 6.3) indicates the respective sections. It should be noted that the stated envisioned construction could be adopted for the scheme though the precise methods adopted by the contractor may vary.

The approaches to the tunnel at Silvertown and Greenwich are envisioned to be formed via cut-and-cover techniques as these sections are considered to be too shallow to be completed as bored tunnels. The walls of the cut-and-cover section are intended to be constructed using a combination of secant or diaphragm piling techniques. Once piled walls have been constructed it is probable that one of two methods of cut-and-cover construction will be implemented either top-down construction or bottom-up construction.

3.6. Potential Pollutant Linkages

A brief discussion of the potential pollutant linkages, as discussed within Transport for London Silvertown Tunnel Environmental Statement Chapter 12 (document reference 6.2) is presented below. Further assessment into the pollutant linkages present at the site and the risk to receptors will be assessed within the Assessment of Risk described in Section 4.6 to be completed following both detailed design and baseline groundwater monitoring.

3.6.1. Sources

- **Silvertown** ground investigations and monitoring have identified contamination both within soil and groundwater at Silvertown. A number of potential sources of contamination within the assessment area and surrounding area which may be mobilised by dewatering activities have also been identified. The primary potential source of contamination is considered to be within the RTD groundwater itself.
- **Greenwich Peninsula** sources of contamination within the assessment area and surrounding area, predominantly associated with the historical gasworks site use, have the potential to be mobilised by dewatering activities. Contamination has previously been identified within the Upper Aquifer and in localised areas of the Lower Aquifer.

A comprehensive groundwater contamination assessment has not been completed within close proximity of the Silvertown Tunnel alignment.

3.6.2. Pathways

The potential pathways considered relevant are discussed below:

- dewatering from the shallow aquifer (RTD) and or Lower Aquifer (Thanet Sand / Chalk);
- creation of preferential pathways: The introduction of the proposed tunnel between the Silvertown and Greenwich sites or the removal of underground structures prior to using the tunnel boring machine (TBM) may create preferential pathways for contamination within the shallow aquifer;
- direct discharge of abstracted groundwater to the surrounding environment a quantitative risk
 assessment should be undertaken at detailed design stage to provide assessment criteria against which
 the discharged water can be monitored.
- migration of contaminants from off-site (e.g. wider Greenwich Peninsula etc.) towards the area of dewatering;
- migration of contaminants towards cleaner areas of the Greenwich Peninsula;
- vertical migration of contaminants from RTD following increased hydraulic gradient between the two aquifers, this may be of particular relevance where London Clay is thin or absent;
- lowering of piezometric heads and interference with other nearby abstractions a quantitative assessment may be required to assess the potential impact of local abstractions; and
- hydraulic connectivity between the upper and Lower Aquifers: local thinning of the London Clay has been documented on the Greenwich Peninsula, there is also the possibility for scour features to be present which have the potential to increase the hydraulic connectivity. Piezometric pressures also suggest that in areas of the peninsula there may be a hydraulic connection between the Upper and Lower Aquifers.

3.6.3. Receptors

The following receptors have been identified with the potential to be impacted by changes in quality, quantity or both:

- potential discharge points. Previous dewatering in the area, undertaken by Crossrail, was discharged into the River Thames, nearby docks, or reinjected into the aquifer. Discharge options will be assessed following detailed design but it is likely that similar options for discharge as employed by Crossrail will be considered;
- third party groundwater users. Multiple third party users have been identified in the vicinity of the sites;
- groundwater within the Upper Aquifer; and
- groundwater within the Lower Aquifer.

3.6.4. Monitoring of linkages

The identified potential linkages discussed above are based on current information. The potential linkages will be further assessed through investigation and monitoring as the design develops to help inform the Groundwater Monitoring and Verification Plan.

4. Groundwater Monitoring Strategy

4.1. Groundwater Monitoring Strategy

4.1.1. Objectives and Rationale

Based on the source-pathway-receptor linkage model, monitoring of groundwater is proposed to enable the following:

Baseline:

- collection of a groundwater level dataset to be used as a baseline such that an assessment can be completed as to whether or not the tunnel has significantly impacted groundwater flow during construction and operation;
- collection of a groundwater quality dataset to establish baseline groundwater quality, identify trends and determine trigger levels, where possible;
- assessment of risk of mobilisation and migration of pre-existing contaminants affecting the groundwater (i.e. if source-pathway-receptor linkages exist);
- direct the placement of sentinel monitoring boreholes to be utilised during construction to ensure that mobilisation of contaminants is not adversely affecting identified receptors; and
- enable the development of site specific alert and trigger levels and the development of a contingency plan.

Construction and Post Construction Phases:

- Collection of groundwater quality samples during both phases, to establish whether:
 - mobilisation and migration of contaminants has taken place; and
 - significant changes in water quality as a result of dewatering and tunnel seepages have occurred.
- To ensure alert and trigger levels are not exceeded and if they are that the contingency plan is activated.

4.2. Baseline Assessment and Compliance Criteria for Groundwater

The Code of Construction Practice (document reference 6.10) sets out that the Contractor will be required to undertake pre-construction baseline monitoring to help inform the Groundwater Monitoring and Verification Plan. To facilitate this, a proposed baseline groundwater level and quality monitoring plan has been included in Appendix A. A proposed list of monitoring wells is included within the proposed baseline monitoring plan. Following an assessment into the viability of these wells the Baseline Monitoring Plan will be submitted to the Environmental Agency for information. Baseline monitoring should be implemented as soon as practicable and continue until the commencement of construction or the implementation of the monitoring plan outlined in Section 4.8. It is suggested that at least 12 months of monitoring is completed prior to the commencement of construction of the Silvertown Tunnel, but this shall be agreed with the regulators in advance of monitoring.

A Baseline Monitoring Report will be produced incorporating the collected baseline groundwater level and water quality data and will outline a typical range for both groundwater levels and selected contaminants of concern (COC) for the site. The report will also include initial assessment and compliance criteria (developed in conjunction with the Groundwater Risk Assessment detailed in Section 4.5) against which project impacts can be assessed as part of the Groundwater Monitoring and Verification Plan. Baseline

data, in conjunction with the detailed design, will also be used to select sentinel monitoring boreholes which will be used to monitor the potential movement of contamination throughout the construction phase of the project. The initial criteria will be derived in consultation with the Environment Agency.

A two-tier system of screening criteria shall be adopted:

- assessment criteria (alert levels) will be used to provide an early warning of potential issues; and
- compliance criteria (trigger levels) will be used to identify potentially significant risks to receptors.

4.3. Alert and Trigger Levels

Alert levels will be used on a day-to-day basis by the Contractor. Alert levels aim to draw the attention of the Contractor's site management team to the development of adverse trends in groundwater conditions. The alert levels should be treated primarily as an early warning system to enable appropriate investigative or corrective measures to be implemented so that the risk of breaching a compliance limit can be reduced and before impacts can cause harm to the environment or human health.

Comparison of measured water quality with trigger levels will be used to demonstrate compliance with agreed water quality standards to stakeholders. Trigger levels are standards to be agreed with the Environment Agency through the approval of the Groundwater Monitoring and Verification Plan. Breaches of the trigger levels should be prevented. Should a breach of a trigger level occur a Contingency Action plan will be implemented, this is outlined in Section 4.7.

Alert and trigger levels will not be derived for every water quality monitoring determinand. Instead, key indicator parameters will be identified using a risk-based approach that reflects the baseline water quality dataset, extent of dewatering required, discharge points and identified receptors. The indicator parameters will be agreed with the Environment Agency through the approval of the Groundwater Monitoring and Verification Plan. The choice of indicator parameters will be reviewed by the Contractor at least annually for adequacy; any proposed changes must be approved by TfL and the Environment Agency prior to implementation.

Alert and trigger levels may be derived on a borehole-specific basis. The levels will be based on the baseline dataset and will take into account existing variability and trends. Statistical methods will be used, where possible, to derive appropriate alert and trigger levels. The alert and trigger levels will be reviewed at least annually by the Contractor for adequacy; any proposed changes must be approved by TfL and the Environment Agency prior to implementation.

The alert and trigger levels, and associated procedures, will determine whether contingency plans need to be activated. The assessment process will involve evaluation of the significance of a departure from baseline conditions.

4.4. Sentinel Monitoring Boreholes

Sentinel monitoring boreholes, which will provide an early warning to the mobilisation of contamination, will be selected following analysis of the baseline dataset and completion of the detailed design and subsequent scope of dewatering required. These monitoring boreholes could be a combination of already established monitoring locations and newly installed boreholes, depending on the specific scope of dewatering and availability of existing monitoring locations. The selected sentinel monitoring borehole details, justifications and associated alert and trigger levels will be submitted to the Environment Agency by the contractor for review and approval in the Groundwater Monitoring and Verification Plan.

In the event that new monitoring boreholes / sentinel monitoring boreholes are required, a dataset will need to be established prior to the commencement of construction. A dataset of 12 months of monitoring for each sentinel monitoring boreholes is desirable. However in the event that new sentinel monitoring boreholes are required a minimum four rounds (over a minimum of four months) of monitoring data should be collected (groundwater level and quality) to provide a reasonable baseline, however, this will depend on the scope of dewatering and will be subject to Environment Agency agreement.
4.5. Deliverables

A one-off pre-construction groundwater Baseline Monitoring Report will be provided to the Environment Agency. The Baseline Monitoring Report which will be completed prior to the start of construction, will contain the following:

- relevant historical and current groundwater level and quality data;
- interpretation of data;
- trends in key water quality parameters;
- appropriate alert and trigger levels defined at this stage; and
- an outline of sentinel boreholes selected for the project to monitor the movement / migration of contamination.

4.6. Assessment of Risk

If during detailed design it is identified there is a potential risk to receptors then a Groundwater Risk Assessment will be required to assess all source-pathway-receptors linkages, as explained in the Code of Construction Practice.. If dewatering is required as part of the construction process a detailed qualitative risk assessment (DQRA) maybe required to assess the risks to controlled waters from dewatering. This/these document(s) will be submitted to the Environment Agency for information. The Groundwater Risk Assessment / DQRA will inform the following:

- the requirement for further assessment of identified potential impacts to receptors through detailed risk assessments;
- further development / verification of appropriateness of alert and trigger levels;
- the development of the Groundwater Monitoring and Verification Plan including the requirement for additional and/or more frequent monitoring relative to the baseline programme (including long term monitoring) based on the risk identified;
- inform additional groundwater monitoring requirement to be implemented during periods of dewatering; and
- remedial groundwater treatment as mitigation shall be proposed where a detailed quantitative risk assessment identifies an unacceptable risk to receptors. The proposal shall include an options appraisal and full details of the remedial works required and how they are to be undertaken.

4.6.1. Deliverables

The Groundwater Risk Assessment and the DQRA if required will be completed and submitted to the Environment Agency, for information, following the completion of detailed design, the establishment of dewatering requirements and baseline monitoring.

4.7. Contingency Action Plan

A Contingency Action Plan will be developed following baseline monitoring in conjunction with the Baseline Monitoring Report, Groundwater Risk Assessment and Groundwater Monitoring and Verification Plan. If during construction or operation, trigger levels are exceeded, the Contingency Action Plan will be implemented. The Contingency Action Plan will include the following:

- notification of the exceedances to the Environment Agency, as soon as possible;
- determining the cause of any exceedances;
- evaluation of location, likely scale, duration and effect; and
- identification of appropriate mitigation or remediation measures.

In all cases, the need for remediation should be balanced against the risk posed to groundwater receptors and the benefits gained by remediation. Remedial actions and their objectives will be agreed in advance with the Environment Agency.

Any exceedances of alert levels will be reported to the Environment Agency using the agreed reporting process outlined in Section 5. Only in cases where trigger levels are subsequently exceeded will the Environment Agency be contacted specifically.

4.8. Groundwater Monitoring and Verification Plan

A Groundwater Monitoring and Verification Plan is to be developed to implement the Groundwater Monitoring Strategy (Section 4.1) during construction and post-construction and will be developed by the Contractor following detailed design. The Groundwater Monitoring and Verification Plan will include the following aspects as well as include measures identified within the following assessments:

- provide a methodology and schedule for groundwater level and quality monitoring;
- baseline monitoring data including updated alert and trigger levels;
- detailed design dewatering requirements; and
- measures identified through any risk assessment / detailed risk assessment completed for the project.
- provide methodology and schedule for additional groundwater and quality monitoring to be carried out during dewatering activities (if required)
- outline post-construction monitoring requirements both in frequency and duration. The post construction monitoring requirements will depend on the extent of dewatering required for construction.

The objectives of the Groundwater Monitoring and Verification Plan are to:

- enable routine monitoring of groundwater levels and quality during construction and post-construction to discern breaches of alert and trigger levels;
- review combined monitoring results across the project area as a whole, during construction, to assess impacts; and
- carry out routine monitoring of groundwater levels and quality during construction and post-construction to discern breaches of alert and trigger levels.

5. Requirements for Groundwater Monitoring and Verification Plan

The Groundwater Monitoring and Verification Plan document aims to aid the implementation of the Groundwater Monitoring Strategy outlined in Section 4.1.

The objectives of the Groundwater Monitoring and Verification Plan are:

- to ensure groundwater level and quality monitoring is appropriate with respect to detailed design using a risk-based groundwater monitoring network;
- enable routine monitoring of groundwater levels and quality during construction is undertaken by Contractors to discern breaches of alert and trigger levels;
- to outline post-construction monitoring requirements and provide a mechanism for cessation of monitoring. The duration of post-construction monitoring will depend on the effects construction activities have on the groundwater environment;
- provide a methodology to ensure consistency of monitoring techniques and outline quality assurance and quality control requirements; and
- the Contractor shall be responsible for preparing the Groundwater Monitoring and Verification Plan. The Groundwater Monitoring and Verification Plan must be approved by TfL and the Environment Agency prior to construction works which may affect groundwater on site.

5.1. Proposed Scope of Groundwater Monitoring and Verification Plan

The Groundwater Monitoring and Verification Plan will include the following elements:

- management structure, including roles and responsibilities within Contractor team;
- monitoring network already in existence;
- monitoring measurements and laboratory analysis to be obtained and the methodology;
- monitoring and analysis schedules;
- data management procedures, data review procedures, including comparison with assessment and compliance criteria;
- contingency actions;
- reporting procedures; and
- details of the quality assurance / quality control (QA/QC) protocols that will be implemented to provide confidence in the groundwater monitoring results.

The Groundwater Monitoring and Verification Plan shall be a live document and subject to regular review by TfL for adequacy (at least annually). The Contractor will not deviate from the approved Groundwater Monitoring and Verification Plan without written permission from TfL and the Environment Agency.

5.2. Monitoring

The Groundwater Monitoring and Verification Plan will present details regarding the groundwater monitoring network that will be monitored during the construction and post-construction phases.

As a minimum, groundwater monitoring will be undertaken in the monitoring boreholes summarised in Appendix A. Additional monitoring locations may be required following the completion of detailed design and subsequent risk assessments (if required).

The Contractor will be responsible for maintaining the groundwater monitoring network during the construction phase and ensuring critical monitoring wells are retained throughout works and post-construction.

The Groundwater Monitoring and Verification Plan will include a protocol and schedule for regular monitoring location inspection and maintenance. This will include procedures for the repair and replacement of monitoring locations as required.

Each new or repaired monitoring point will be designed, constructed, supervised and certified in accordance with normal engineering construction quality assurance (CQA) procedures. Detailed borehole logs for each monitoring point will be provided.

An up-to-date register of all monitoring points will be incorporated within the Groundwater Monitoring and Verification Plan and Annual Interpretative Report.

5.2.1. Monitoring Methodology

The Contractor will be responsible for undertaking groundwater monitoring, a proposed methodology is included in Appendix B.

The Groundwater Monitoring and Verification Plan will present details regarding the groundwater monitoring methodology that will be implemented during the construction and post-construction phases for approval by the Environment Agency.

The Groundwater Monitoring and Verification Plan will specify a process for managing any required changes in monitoring methods, procedures or protocols over the life of the project taking into account the need for reliable comparison of future monitoring results against the baseline dataset.

The Groundwater Monitoring and Verification Plan will specify the following for each methodology used:

- the measurement method;
- detailed protocols for sampling and/or measurement and record keeping; and
- QA/QC protocols.

5.2.2. Sampling and field analysis

The Groundwater Monitoring and Verification Plan will specify field protocols with respect to:

- monitoring network condition inspections;
- in situ groundwater measurement protocols:
 - manual measurements; and
 - data loggers.
- water sampling method(s) and protocols, including purging (where appropriate);
- field analysis protocols;
- sample handling protocols;
- sample containers and preservation;
 - sample labelling;
 - sample storage and transportation; and
- QA/QC protocols.

The samples collected and measurements made should not be affected by contamination associated with surface run-off, contact with the sampling equipment or extraneous matter that may have entered the monitoring structure. Nor should they be affected by the products of reaction with materials used in the construction of the monitoring point, sampling equipment or containers.

The Groundwater Monitoring and Verification Plan shall reference appropriate protocols for managing any waste waters generated during sampling procedures (e.g. purging).

The Groundwater Monitoring and Verification Plan shall reference appropriate protocols for managing HSE hazards and risks during field activities.

5.2.3. Laboratory analysis

The Groundwater Monitoring and Verification Plan shall specify laboratory analysis protocols with respect to:

- laboratory selection, including accreditation requirements (UKAS and MCERTS accredited);
- analytical suite (as per the project list at the time);
- analytical methods;
- detection limits; and
- QA/QC protocols.

The analytical methods adopted should not be affected by cross contamination, poor recovery and interference or instrument errors.

5.3. Monitoring Schedule

The Groundwater Monitoring and Verification Plan shall present a detailed schedule for groundwater monitoring during construction, dewatering and post-construction phases of work. The monitoring schedule will ensure that identified risks to groundwater from construction activities are monitored such that potential moderate and large adverse effects are identified such that actions can be implemented to ensure the security of receptors.

5.3.1. Data Management and Review

The Contractor shall be responsible for managing the groundwater monitoring data collected.

The Groundwater Monitoring and Verification Plan will specify protocols for data management, including:

- data collection;
- data collation;
- data verification; and
- data storage.

As a minimum, monitoring data will be checked for laboratory and scheduling inconsistencies within five working days of receipt.

The Contractor will be responsible for keeping original monitoring records and submitting these to TfL on completion of construction.

Water level and quality data will be stored in Excel-compatible formats.

Initial alert and trigger levels will be provided to the Contractor by TfL at the start of the construction period.

The Contractor or nominated representative will be responsible for reviewing the groundwater monitoring data collected and for interpreting the results.

The data review process will be undertaken in a timely manner. The Groundwater Monitoring and Verification Plan will specify the required timescales for data reviews and turnaround times at analytical laboratories.

The Contractor's review of the verified data will include:

- comparison of actual against specified monitoring;
- evaluation of significance of QA/QC data;
- application of assessment and compliance tests;
- a review of the conceptual site model (i.e. current understanding of the hydrology and hydrogeology of the site); and
- a review of the significance of risks and impacts including comparison of data collected against alert and trigger levels.

The Contractor will be responsible for ongoing review of the alert and trigger levels for adequacy during the construction period.

When risks are re-evaluated or monitoring data reveal unexpected variations or trends, it may be necessary to review and occasionally change assessment or compliance criteria. However, any proposed changes to the alert and trigger levels used in the Groundwater Monitoring and Verification Plan need to be justified technically and may be implemented only after consultation and agreement between the Contractor, TfL and the Environment Agency.

5.4. Dewatering

Dewatering may be required to complete the construction of the Silvertown Tunnel, if dewatering is required the following approach will be taken and appropriate measures will be included within the Groundwater Monitoring and Verification Plan.

5.4.1. Approach to Dewatering

When developing the final construction methodology, the Contractor will consider suitable control methods to manage groundwater ingress and dewatering. The Contractor will liaise with the Environment Agency to determine the need for detailed assessment of dewatering impacts.

The project will implement measures to ensure that the following objectives are met:

- the project is compliant with Environmental Permitting Regulations (EPR) Schedule 22 (England and Wales) 2010 with respect to dewatering;
- adequate protection of surface water and groundwater resources, in line with principles as set out under the Water Resources Act 2003 (required for abstraction licence applications);
- construction effects arising directly or indirectly from dewatering systems and from discharges to the water environment are mitigated; and
- movements of poor quality groundwater as a result of dewatering activities will be minimised.

The Environment Agency propose to implement changes to the licensing policy which will mean that licences with damaging effects on the environment will not be issued by the Environment Agency. A number of previously exempt activities, including dewatering, are likely to require authorisation from the Environment Agency (licences or permits) from 2016 onwards. As set out in the COCP the Contractor will be responsible for obtaining all necessary licences and permits from the Environment Agency prior to dewatering commencing (TfL has applied through the DCO to disapply the need for abstraction licences and/or permits in which case the Contractor will need to meet the requirements stipulated within the DCO). The Contractor will be responsible for ensuring compliance with the conditions of all licences and permits granted by the Environment Agency.

On the Greenwich Peninsula and, potentially at Silvertown, it is likely that water pumped from the ground via dewatering systems will require some form of treatment prior to discharge. An appropriate discharge point/s

will be assessed prior to the commencement of dewatering and approval from appropriate statutory authorities will be obtained (i.e. the Environment Agency and/or Thames Water).

5.4.2. Dewatering Plan

A Dewatering Plan will be developed the objectives of which are to ensure that any negative environmental impacts arising directly or indirectly from dewatering systems and discharges to the water environment are mitigated.

The plan will be developed by the Contractor and submitted to the Environment Agency for approval as a requirement of the Groundwater Monitoring and Verification Plan. The structure and content of the plan will include the following:

- an overview of expected dewatering;
- preliminary and detailed (if required) dewatering risk assessment;
- summarise all the licences and permits to abstract and discharge from dewatering systems issued by the Environment Agency; and
- A Method Statement.

Prior authorisation will be obtained from Thames Water for groundwater discharge to sewer and an outline of pre-treatment measures will be provided to the Environment Agency and/or Thames Water for approval.

5.4.3. Dewatering Method Statement

Method statements shall include the following:

- main discharge points;
- details of bore installation;
- details of monitoring network (surface and groundwater);
- details of equipment used;
- proposed construction sequence;
- licences and permits from the Environment Agency;
- prior authorisation from local sewerage provider; and
- outline of any pre-treatment required prior to discharge, approved by the Environment Agency/Thames Water.

5.4.4. Dewatering Overarching Management Protocols

TfL or their nominated representative will review and approve the site-specific dewatering plans prior to submission to the Environment Agency for approval.

TfL will review the monitoring reports, as required under the various permits and provide comments to Contractor within four weeks of receipt of all the reports.

6. Contingency Planning

The Groundwater Monitoring and Verification Plan will make reference to the Contingency Action Plan with any necessary updates required. The Contingency Action Plan will describe a phased contingency response process that will be implemented so that risks to the project and nearby receptors can be assessed and managed in the event of any breach of approved alert or trigger levels.

Where risks are unacceptable, corrective or remediation measures will be initiated and a strategy to monitor their effectiveness will be determined within one week of the trigger level exceedance being confirmed by the laboratory and in consultation with the Environment Agency.

In all cases, the need for remediation should be balanced against the risk posed to groundwater receptors and the benefits gained by remediation. In complex cases, specialist advice should be taken and remedial actions and their objectives agreed in consultation between the Contractor and the Environment Agency.

Recommended contingency strategies are outlined below; the Groundwater Monitoring and Verification Plan may include alternative or additional contingency plans as appropriate. The Contingency Action Plan shall specify the required timescales and parties responsible for implementing each action.

6.1. Assessment Criteria (Alert Levels)

Assessment criteria are intended to draw the attention of site management to the development of adverse trends in monitoring data or the breach of a specified alert level. The assessment process will involve the evaluation of the significance of a departure from baseline conditions. The assessment criteria should be treated primarily as an early warning system to enable appropriate investigative or corrective measures to be implemented, particularly where there is the potential for a compliance limit to be breached.

The Groundwater Monitoring and Verification Plan shall specify the assessment criteria being applied, including the alert levels and the protocols for identifying adverse trends.

If an alert level is breached on a single occasion the following actions will be taken by the Contractor:

- notify Site Management Team;
- repeat sampling round if no routine sampling is planned within seven days; and
- report in next quarterly factual monitoring report.

If an alert level is breached on three consecutive monitoring rounds or if the annual review process identifies significant adverse trends in four monitoring rounds, further assessment will be required as follows:

- characterise observed issues (desk-based data review), including:
 - parameter(s) involved;
 - magnitude of exceedance(s);
 - frequency of exceedance(s);
 - spatial pattern; and
 - temporal trends.
- review relevant site activities;
- review potential cause(s) for observed results;
- review monitoring plan for adequacy;
- undertake additional investigation or monitoring (if required) to better characterise issue; and
- identify likely cause(s) for observed results.

If the observed issue is considered to relate to the project the following actions will be taken by the Contractor:

- consult with regulator(s)/stakeholders (as appropriate);
- review significance of potential risks to receptors;
- implement mitigation measures (as appropriate) to try and halt or reverse adverse trends and/or manage risks for receptors; and
- carry out ongoing monitoring at an increased frequency and review to confirm effectiveness of measures implemented.

The assessment findings and actions will be reported in the next monitoring report.

6.2. Compliance Criteria (Trigger Levels)

If a trigger level is breached on a single occasion:

- notify Site Management Team in writing;
- undertake Assessment of Significance (desk-based data review):
 - parameter(s) involved;
 - magnitude of exceedance(s);
 - frequency of exceedance(s);
 - spatial pattern;
 - temporal trends; and
 - risks to nearby receptors.
- repeat sampling round if no routine sampling is planned within seven days; and
- prepare notification report and inform key stakeholders.

If a trigger level breach:

- occurs on two consecutive monitoring rounds;
- is part of a significant adverse trend;
- is considered to indicate an immediate and significant risk to nearby receptors.

The Contractor will undertake the following additional actions:

- prepare a notification report and inform key stakeholders (e.g., Environment Agency, TfL), as soon as possible;
- notify other relevant stakeholders within one week as appropriate, e.g., other abstractors;
- review relevant activities occurring on site;
- review potential cause(s) for observed results;
- review existing monitoring plan for adequacy;
- undertake additional targeted monitoring or investigation (if required) to characterise the issue better; and
- identify likely cause(s) for observed results.

If the observed issue is considered to relate to the project:

• review significance of risks to nearby receptors;

- develop strategy within one week of the Trigger Level exceedance being confirmed by the laboratory, to
 mitigate and/or remediate issue(s) in consultation with key stakeholders;
- implement mitigation and/or remedial measures (as appropriate) to reduce risk of future breaches affecting project or nearby receptors; and
- ongoing monitoring and review to confirm effectiveness of measures implemented.

The assessment findings and actions will be reported in Notification Report to be submitted to TfL and the Environment Agency within a time frame approved by TfL and the Environment Agency.

6.3. Emergency Measures

Additional groundwater monitoring may be required in the event of a pollution incident accidentally occurring at or near the site. The scope of any such additional monitoring will be proposed by the Contractor for the approval of TfL and the Environment Agency. The scope should be targeted to reflect the nature of the pollution incident and potential risks. This cannot be readily defined in advance of any incident occurring. Instead, the Contingency Action Plan will include a phased process by which the necessary scope of additional monitoring is proposed by the Contractor in consultation with TfL and the Environment Agency following a pollution incident.

In the event of a pollution incident, the Contractor shall inform the Environment Agency as soon as practicably possible and the Contractor will comply in full with all of their requirements as well as requirements outlined within the Silvertown Construction Environmental Management Plan (CEMP).

In addition the Contractor will complete the following actions:

- review significance of risks to nearby groundwater receptors;
- notify other relevant stakeholders within one week as appropriate, e.g., other nearby abstractors;
- review the existing Groundwater Monitoring and Verification Plan for adequacy;
- design additional targeted monitoring or investigation (if required) to better characterise issue; consult with the Environment Agency and other stakeholders as appropriate;
- undertake additional targeted monitoring or investigation (if required) and review results;
- develop strategy to mitigate and/or remediate issue(s) in consultation with key stakeholders;
- implement mitigation and/or remedial measures (as appropriate) to reduce risk of future breaches affecting project or nearby receptors; and
- ongoing monitoring and review to confirm effectiveness of measures implemented.

The assessment findings and actions will be reported within the notification report to be produced and submitted to TfL and the Environment Agency within a time frame approved by TfL and the Environment Agency as part of the Contingency Action Plan.

7. Reporting

The Contractor will be responsible for reporting the groundwater monitoring results during construction as part of the Groundwater Monitoring and Verification Plan. The Groundwater Monitoring and Verification Plan will set out that monitoring reporting will include the following types of reports, which will be presented in the formats set out in the rest of this chapter:

- notification reports (if required);
- factual reports for each monitoring round; and
- interpretative reports (frequency to be agreed with the Environment Agency via the Groundwater Monitoring and Verification Plan).

The monitoring reports shall be:

- submitted on time;
- quality assured; and
- collated and presented in a consistent format.

The templates for the monitoring reports will be approved in advance by TfL in consultation with the Environment Agency. The reporting templates will not be altered without approval from TfL.

The reports will be submitted electronically in PDF format. The factual data will be submitted in an Excelcompatible format.

7.1. Notification report

Notification reports will be used to disseminate information regarding breaches of Trigger Levels or pollution incidents. These reports should provide clear, concise information and carry a recommendation for action (or describe action taken). Notification reports should be issued within a time frame approved by TfL and the Environment Agency.

Notification reports will include:

- date and time of issue of report;
- name, position and contact information for person issuing report;
- date and time of monitoring surveys or observations that confirm the breach of a compliance limit, or an
 actual pollution incident;
- pollution incident recorded or compliance limit breached;
- · details of any emergency contingency actions implemented; and
- an indication of the urgency of response needed by TfL and/or the Environment Agency.

In instances where assessment criteria or compliance limits are breached regularly and action is being implemented by the site operator (e.g., where remedial measures are underway or where the source of contamination to groundwater is being investigated), alternative ongoing reporting procedures may be agreed between the Contractor, TfL and the Environment Agency to avoid unnecessary duplication of notification reports.

7.2. Factual Report

Factual monitoring reports shall be prepared by the Contractor quarterly. Minimum factual monitoring report requirements are as follows:

- self-audit of compliance with monitoring plan requirements;
- groundwater level data in graphical form;
- key groundwater quality data to be displayed in graphical form;
- breaches of alert levels in tabular form;
- breaches of trigger levels in tabular forms;
- urgent actions;
- mitigation and/or remedial measures (if required) to manage groundwater-related risk to project or nearby receptors; and
- factual datasets (electronic).

The quarterly factual reports will be submitted to TfL within four weeks of the end of each quarter. The quarterly reports will be made available to the Environment Agency.

7.3. Interpretative Report

Interpretative monitoring reports shall be prepared by the Contractor annually. Minimum Annual Interpretative report requirements are as follows:

- self-audit of compliance with monitoring plan requirements;
- updated register of monitoring network;
- groundwater level data in graphical form;
- spatial contour plots of groundwater levels in Upper and Lower Aquifers;
- spatial plots of key COC in upper and Lower Aquifer to an appropriate scale;
- groundwater quality data in graphical form for agreed indicator species (outlined in the Groundwater Monitoring and Verification Plan);
- breaches of Alert Levels in tabular form;
- breaches of Tigger Levels in tabular form;
- summary of QA/QC checks e.g. ionic balance calculations;
- review of Key groundwater issues and patterns (in context of existing conceptual model and baseline dataset);
- update or refinement of conceptual model (if appropriate);
- likely causes of significant groundwater issues;
- influence of site activities on groundwater;
- influence of off-site/third party activities on local and regional groundwater;
- assessment of risk to nearby receptors;
- identification of any data gap at the site;
- proposed modification to monitoring plan (if required);
- changes to monitoring network;
- changes to monitoring frequency;
- changes to sampling methodology;
- changes to analytical suite;
- proposed revision to alert and trigger levels (if appropriate);

- mitigation and/or remedial measures (if required) to manage groundwater-related risk to project or nearby receptors; and
- other proposed modifications to site activities to better manage risk to/from groundwater (if appropriate).

The annual reports will be submitted by the Contractor in draft to TfL within four weeks of monitoring year end. The approved draft final annual report shall be submitted to the Environment Agency within two months of monitoring year end. Following receipt of corrections and comments from the Environment Agency, the final report will be produced within one month of receipt of these comments.

Appendices

Appendix A. Proposed Baseline Monitoring Programme

A.1. Proposed Baseline Monitoring Programme

The objectives of this Baseline Monitoring Programme are as follows:

- to establish a data set such to direct monitoring requirements for construction and post construction phases of works;
- to allow the development of Alert and Trigger levels;
- to allow for the verification of predictions, and to confirm that actual performance is as expected; and
- to assess what further monitoring bores (new installations) are required to implement the Groundwater Monitoring Strategy during the construction of the Silvertown Tunnel.

Baseline monitoring comprises the following tasks:

- continuous groundwater elevation monitoring using groundwater elevation data loggers;
- manual groundwater level dip measurement to validate and convert data into levels and provide information on levels in monitoring bores not installed with data loggers; and
- collection of groundwater chemistry samples and subsequent chemical analysis for an appropriate suite of determinants.

As the detailed design and construction programme is developed this monitoring strategy will be periodically reviewed and amended, where necessary.

A.2. Access Arrangements and Health and Safety

Many of the proposed monitoring locations belong to a third party, permission to use these boreholes will need to be obtained prior to monitoring being undertaken. Additionally the proposed monitoring scope requires accessing areas of the Greenwich Peninsula and Silvertown Site which are managed by a variety of different parties. To facilitate access permission from these parties must be obtained and they should be informed of the works and provided the relevant works and health and safety information in advance of each proposed monitoring visit and/or be compliant with specific requirements of the affected parties. For certain areas within the sites a permit to dig / work system will need to be completed prior to entry to undertake the monitoring.

In addition to access arrangements, a Health and Safety risk assessment, method statement and appropriate contaminated land exposure risk assessment (CLERA) will need to be completed in advance of the works to outline the methodology for the monitoring and sampling and to assess any relevant hazards and proposed appropriate mitigation / control measures.

A.3. Baseline Monitoring Network

An initial groundwater level monitoring network, utilising existing boreholes, is proposed and the boreholes are presented below in Table A-1 and Table A-2. The monitoring network has been selected based on the following:

- to ensure that baseline water level data from both the Upper (RTD) and Lower Aquifers (Thanet Sand and Chalk);
- assess the connectivity of the Lambeth Group with the underlying Lower Aquifer at both Silvertown and Greenwich Peninsula via groundwater level monitoring;
- to obtain baseline groundwater quality data from both the Upper (RTD) and Lower Aquifers (Thanet Sand and Chalk); and
- to facility the assessment of contamination source zones and the selection and or installation of sentinel monitoring boreholes.

The monitoring network includes monitoring boreholes from several projects completed in the vicinity of the Silvertown Tunnel Scheme, the projects include the following:

- Greenwich Peninsula long term groundwater monitoring (GP-LTGWM) locations which form part of the Housing and Land Directorate of Greater London Authority's (GLA) monitoring network.
- London Cable Car Project (Emirates Cable Car) completed by Buro Happold (ECC).
- Crossrail Limmo Peninsula Project (Crossrail).
- Atkins Silvertown Tunnel Ground Investigation Project (STWTN-GI).
- A borehole condition survey of all selected monitoring locations should be completed as soon as possible to establish the current status of the proposed monitoring boreholes / piezometers. Pending the results of the condition survey, installation of further monitoring boreholes maybe required. Assuming all monitoring locations selected and defined in Table A-1 and Table A-2 are useable / suitable to be sampled, it is considered that the following monitoring boreholes will be required:

Greenwich Peninsula

- Chalk monitoring borehole installed adjacent to BHN1(A) and BHN1(B);
- Chalk monitoring borehole adjacent to BHN2(A);
- Lambeth Group monitoring borehole adjacent to BHN5(A);
- Chalk monitoring borehole adjacent to BHN5(A) see comments under Table A-1 below.

Silvertown

• Chalk monitoring borehole installed adjacent to NT BH02;

Borehole construction will be in accordance with BS ISO 5667-22: 2009, Water Quality – Sampling – Part 22: Guidance on the design and installation of groundwater monitoring points.

Monitoring boreholes have been separated into three groups; primary, secondary and auxiliary monitoring boreholes, the groups represent levels of decreasing monitoring requirements. Primary monitoring boreholes (and piezometers) represent monitoring positions located in strategic locations to obtain critical data for future assessments. Secondary monitoring boreholes (and piezometers) and auxiliary monitoring boreholes are aimed at obtaining water levels and chemistry from the greater assessment areas.

Location	Project	Easting	Northing	Ground Level (mAOD)	Depth to Base (m bgl)	Indicative Depth to Groundwater (m bgl)	Installation
			Primary M	onitoring E	Boreholes		
River Terr	ace Deposit N	Ionitoring	Boreholes				
BHN1(A)	GP-LTGWM	539164	179925	5.67	12.3	6.7	Monitoring Bore
BHN5(A)*	GP-LTGWM	539488	179955	5.68	15.0	6.6	Monitoring Bore
BHN6(A)	GP-LTGWM	539331	179808	4.92	15.0	5.8	Monitoring Bore
BHN7(A)	GP-LTGWM	539179	179560	2.43	13.0	3.1	Monitoring Bore
BHN9(A)	GP-LTGWM	539727	179631	7.27	15.0	7.1	Monitoring Bore
Lambeth (Group Monito	ring Boreh	oles				
BHN1(B)	GP-LTGWM	539167	179927	5.62	50.0	12.5	Monitoring Bore
BHN5(B)			To be	installed			Monitoring Bore
Thanet Sa	nd Formation	and Chalk	Monitorin	g Borehol	es		
BHN1(C)			To be insta	alled - Chal	k		Monitoring Bore
BHN2(B)			To be insta	alled - Chal	k		Monitoring Bore
BHN5(C)			To be inst	alled -Chall	k		Monitoring Bore
		S	econdary l	Monitoring	Boreholes		
River Terr	ace Deposit N	Ionitoring	Boreholes				
BHN2(A)	GP-LTGWM	539332	180158	5.80	14.0	6.3	Monitoring Bore
BHN8(A)	GP-LTGWM	539570	179480	4.56	14.0	5.4	Monitoring Bore
BNH12(A)	GP-LTGWM	539414	179299	2.77	15.0	3.5	Monitoring Bore
EB3A	STWTN-GI	539241	179657	2.92	15.0	4.3	Monitoring Bore
Thanet Sa	nd Formation	and Chalk	Monitorin	g Borehol	es		
BHN7(B)	GP-LTGWM	539178	179557	2.51	45.0	3.8	Monitoring Bore
BHN9(B)	GP-LTGWM	539722	179637	7.27	45.0	8.1	Monitoring Bore
BH(AB)	GP-LTGWM	539405	179301	-	97	1.3	Monitoring Bore
			Auxiliary M	lonitoring	Boreholes		
River Terr	ace Deposit N	Ionitoring	Boreholes				
BHN3(A)	GP-LTGWM	539047	1803023	6.06	14.0	6.7	Monitoring Bore
BHN4(A)	GP-LTGWM	538828	180155	6.23	14.5	7.0	Monitoring Bore
BHN10(A)	GP-LTGWM	539917	179424	6.05	15.0	6.8	Monitoring Bore
BNH11(A)	GP-LTGWM	539783	179279	5.32	14.5	6.1	Monitoring Bore
BHN13(A)	GP-LTGWM	539540	179000	1.21	9.0	2.3	Monitoring Bore
Thanet Sa	nd Formation	and Chalk	Monitorin	g Borehol	es		
BHN4(B)	GP-LTGWM	538829	180157	6.22	60.0	11.9	Monitoring Bore
BH106	GP-LTGWM	538849	180119	5.56	60.0	12.2	Monitoring Bore

Table A-1	Groundwater Monitoring Location Specifications - Greenwich Peninsula
-----------	--

 BH106
 GP-LTGWM
 538849
 180119
 5.56
 60.0
 12.2
 Monitoring Bore

 *
 the Environment Agency has indicated that there is an application to decommission BHN5(A), if this application is approved an assessment should be undertaken to assess options to obtain/install a like for like, replacement monitoring borehole as this monitoring well is deemed necessary.

Table A-2	Groundwater Monitoring	Location	Specification – Silvertown
	0.04.14.14.0	,	000000000000000000000000000000000000000

Location	Project	Easting	Northing	Ground Level (mAOD)	Depth to Base (m bgl)	Indicative Depth to Groundwater (m bgl)	Installation
	I	Primary Mo	onitoring B	oreholes a	and Piezome	ters	
River Terrace	e Deposit Mor	nitoring Bo	res/Piezon	neters			
G16A	STWTN-GI	540009	180357	2.29	10.5	3.0	Monitoring Bore
G15A	STWTN-GI	539891	180266	2.68	10.3	2.7	Piezometer
G22B	STWTN-GI	539814	180622	3.54	8.5	4.0	Piezometer
Lambeth Gro	up Monitoring	g Borehole	s/Piezome	ters			
G15B	STWTN-GI	539891	180266	2.68	26.9	6.9	Piezometer
G16B	STWTN-GI	540009	180359	2.29	26.0	3.0	Piezometer
Thanet Sand	Formation an	d Chalk M	onitoring E	Boreholes			
NIT BH02	ECC	539973	180437	5.28	60.0	6.2	Monitoring Bore
NT BH02	ECC	539850	180286	5.16	40	7.2	Piezometer
NT BH02 (B)			To be inst	alled -Chal	k		Monitoring Bore
		Au	ixiliary Mor	nitoring Bo	oreholes		
River Terrace	e Deposit Mor	nitoring Bo	reholes/Pie	ezometers		-	
NS BH01A	ECC	540153	180702	4.87	11.4	6.67	Monitoring Bore
CH81(RTD)	Crossrail	539617	180734	5.40	30.0	6.3	Monitoring Bore
CH82(RTD)	Crossrail	539580	180624	5.22	13.0	5.9	Monitoring Bore
CH82(RTD)	Crossrail	539580	180624	5.22	30.0	5.4	Monitoring Bore
CH97(RTD)	Crossrail	540077	180851	1.39	6.65	4.6	Monitoring Bore
CH96(RTD)	Crossrail	539938	180808	2.61	7.7	3.0	Monitoring Bore
CH30(RTD)	Crossrail	541030	180978	1.44	10.20	1.91	Monitoring Bore
Harwich Forr	nation and La	mbeth Gro	oup Monito	ring Boreh	oles		
CH96(SND)	Crossrail	539938	180808	2.61	28.7	6.20	Monitoring Bore
NT BH01	ECC	539869	180301	2.76	50.0	3.96	Monitoring Bore
CH3R(LMB)	Crossrail	538305	180534	3.40	25.60	8.26	Monitoring Bore
CH4R(HF)	Crossrail	538396	180586	2.44	15.60	7.46	Monitoring Bore
CH27(HF)	Crossrail	540706	180869	3.10	23.50	4.07	Monitoring Bore
CH30(UF)	Crossrail	541030	180978	1.44	33.20	8.61	Monitoring Bore
Thanet Sand	Formation an	d Chalk M	onitoring E	Boreholes		-	
SN6RA(CK)	Crossrail	539274	180803	2.7	78.22	15.4	Monitoring Bore
CH91R(CK)	Crossrail	539425	181013	10.94	114.0	18.1	Monitoring Bore
CH3R(CK)	Crossrail			3.40	49.60	15.95	Monitoring Bore
CH7R(TS)	Crossrail	538723	180810	5.92	45.00	18.33	Monitoring Bore
CH4R(TS)	Crossrail	538396	180586	2.44	42.41	21.22	Monitoring Bore
CH27(TS)	Crossrail	540706	180869	3.10	44.70	8.61	Monitoring Bore

A.4. Baseline Monitoring Programme

The groundwater monitoring strategy / programme comprises of the elements presented below, a proposed methodology is presented in Appendix B:

1. Collection of continuous groundwater elevation data from primary monitoring bores (and primary piezometers, Silvertown assessment area) using divers, corrected for atmospheric pressure using barologgers, over as long as possible a period prior to commencement of construction works.

- 2. Manual groundwater elevation monitoring (dip readings) and groundwater samples for chemical analysis should be collected as per the schedules outlined in Table A-3 for Greenwich Peninsula Table A-4 for Silvertown. It is suggested that these schedules be completed for the first year of monitoring at which time an assessment should be undertaken to reassess further data requirements.
- 3. For the first month, weekly groundwater monitoring should be completed on monitoring boreholes/piezometers containing continuous loggers to provide re-assurance that the data recorded by the loggers are not drifting.

Groundwater samples should be scheduled for analysis of the analytical suites presented in Appendix C, refer to Groundwater monitoring suites have been developed based on historical site uses identified during the Mott Macdonald (2013) Phase 1 Contamination Assessment and the Atkins (2016) Ground Investigation.

4. Table C-1 for Suite A, Table C-2 for Suite B and Table C-3 for Suite C.

Table A-3 and Table A-4 below present the initial and pre-construction monitoring schedules for Greenwich Peninsula and Silvertown respectively.

Table A-3 **Greenwich Peninsula Monitoring Requirements**

			Mon	itoring	
Location	Manual Dip	Chemistry Suite A	Additional Chemistry Suite C	Continuous Elevation (divers)	Barometric Pressure
Primary Monitoring Boreholes	Monthly*	Monthly*	Quarterly	Hourly	BHN1(A)
Secondary Monitoring Boreholes	Monthly*	Quarterly		-	-
Auxiliary Monitoring Boreholes	Quarterly	Quarterly		-	-

* following 6 months of monitoring reduce frequency to every two months.

Table A-4 Silvertown Monitoring Requirements

			Monito	pring	
Location	Manual Dip	Chemistry Suite A and B**	Additional Chemistry Suite C **	Continuous Elevation (divers)	Barometric Pressure
Primary Monitoring Boreholes and Piezometers	Monthly*	Monthly*	Quarterly	Hourly-	NT BH02
Auxiliary Monitoring Boreholes and Piezometers	Quarterly	Quarterly		-	-

* following 6 months of monitoring reduce frequency to every two months.

** groundwater chemistry sampling from accessible monitoring locations.

Appendix B. Proposed Sample Collection Methodology

B.1. Groundwater Elevation Loggers (Divers) and Barologgers

B.1.1. Equipment Specification

The rental/procurement cost is to be covered by the Contractor. The divers must have:

- a minimum tolerance of a 20 m water column;
- adequate storage capacity;
- sufficient battery to monitor throughout the entire period;
- be compatible with the barologgers;
- resistant to the potential presence of non aqueous phase product (NAPL); and
- for piezometer sites the data logger needs to fit a 19 mm piezometer.

The barologgers must have:

- adequate storage capacity;
- a sufficient battery to monitor throughout the entire period; and
- be compatible with the divers.

The level divers and barologgers will need to be suspended within the borehole. The material used to suspend the divers must be:

- resistant to corrosion and chemical attack; and
- non-stretch (maximum of 0.01% of length).

B.1.2. Methodology

The groundwater diver installation methodology is summarised below:

- 1. The Contractor shall undertake an equipment test prior to commencing placing the logging equipment in the specified boreholes, this shall include ensuring that the equipment is functioning correctly;
- 2. The divers and barologgers shall be programmed to monitor groundwater elevation data at 30 minute intervals. The divers and barologgers shall be programmed to monitor at the same time;
- 3. Measure the depth to groundwater using a dipmeter (as discussed in Section B.2);
- 4. The divers will be placed such that they are within the water table at all times;
- 5. The divers shall be securely attached;
- 6. The barologgers shall be installed at a depth immediately below ground level, to ensure that they are at no point submersed below groundwater;
- 7. The groundwater level shall be manually dipped (as discussed in Section B.2) at a time corresponding to a diver measurement to ensure accurate conversion to levels; and
- 8. The frequency the data is required to be downloaded will be dependent upon the minimum diver or barologgers memory capacity. At a minimum it shall be at every chemical monitoring interval.

B.2. Interface Meter

Manual readings of groundwater elevation data shall be obtained during each monitoring round using an interface meter at every borehole location specified in Table A-1 and A-2 (in Appendix A.3). The boreholes shall be dipped at the precise time the divers and barologgers are taking a reading (this can be at different times for different locations) and will occur prior to any monitoring.

The monitored parameters shall include:

- 1. Depth at which any free phase product (NAPL) is encountered.
- 2. Depth at which groundwater is encountered.
- 3. Depth to the base of each borehole.
- 4. Date and time that the groundwater level reading is obtained.

B.3. Monitoring Groundwater Chemistry

Groundwater samples shall be obtained from boreholes as specified within Table A.-3 and Table A-4.

Prior to sampling each of the selected monitoring boreholes, the depth to water and the depth to the base of each borehole will be collected relative to ground level and top of borehole casing (as detailed above). Water levels will be monitored using an oil/water interface meter, capable of measuring the depth to water and also capable of detecting and measuring the thickness of any free phase hydrocarbons which may be present on the surface of the groundwater or in the base of the borehole. The water level will be recorded prior to any sampling at a point in time that coincides with a diver reading. The diver should be removed prior to sampling.

Prior to water quality sampling, the divers (if installed) must first be removed from the boreholes. To obtain a representative sample of the in situ groundwater each borehole will be purged prior to sampling for a minimum of three bore volumes using either a submersible or surface pump which must be variable speed. In situ measurement of groundwater physico-chemical properties (as a minimum: temperature, redox (E_H), pH and electrical conductivity) will be undertaken using an inline flow cell and probes and samples will be collected once hydrochemical parameters have stabilised (+/- 10%) and the sample considered representative of aquifer conditions. Should hydrochemical parameters not stabilise within a 25 minute period, this should be noted and a sample should be collected as normal. Where a limited volume of groundwater is present within a bore, a grab sample may be taken, whereby the sample is taken directly after measuring the depth to water and the depth to the base of the borehole, without measuring hydrochemical parameters. The amount of water purged will be measured using an in-line gauge and a description of the purge water and the volume purged will be recorded on the appropriate record sheet.

The Contractor is to allow for suitable storage and subsequent disposal of the purge water. Previous monitoring on the Greenwich Peninsula has used a granular activated carbon (GAC) filter before temporary storage in a water bowser and subsequent disposal to foul sewer under suitable discharge consent. Alternatively the Contractor may explore options to tanker off site, by an approved waste Contractor, for treatment / discharge at a suitable waste water treatment facility. Temporary storage of purge water on site must be in suitable, sealed containers at the agreement of the site operative.

All monitoring equipment will be thoroughly cleaned between monitoring borehole locations to prevent cross contamination. All meters will be stored in accordance with the manufacturer's instructions. New tubing associated with the pump will be used for each new location.

After purging, the sample tubing will be used to collect the pre-arranged volume of sample within the dedicated laboratory supplied sample containers. All bottles and caps will be rinsed in the sample water prior to being completely filled to exclude air. All samples will be stored out of direct sunlight in a cool location such as a cool box with frozen ice packs to keep samples below 4°C. Care will be taken to ensure that the brim or inside of the sampling containers is not touched. When filling any volatile organic compounds (VOC) vials the Contractor shall ensure there is no headspace by filling the inverted lid with sample as well, and set the peristaltic/bladder pump to low flow to minimise the presence of bubbles.

All sample vessels will be labelled with appropriate information for the particular borehole, as follows (minimum information; company name, site details, borehole number and date).

Samples will be clearly labelled as follows:

- site name and job number;
- date and time of sampling;
- sample location (borehole no.);
- sample number; and
- sample depth.

Water samples will be dispatched to the laboratory on the same day of sampling under full chain of custody documentation

Subsequent to monitoring and sampling, the Contractor will ensure that each borehole location and the general work environment, is left in a clean and tidy manner or as originally found.

The Contractor shall inform the relevant parties when the work has been completed and the appropriate permit / documentation will be signed off. Any equipment to be left on site overnight or following the monitoring will be agreed in advance with the site and collection arranged as soon as practicable.

B.4. Baseline Monitoring Reporting

- As a minimum, the following information will need to be obtained after each monitoring visit / event for collation and onward reporting to interested parties:
- Factual report outlining the details of the monitoring visit / event including the date and time of monitoring / sampling, atmospheric conditions pre and during the visit, the condition of the monitoring locations at the time of the visit, tabulated manual groundwater dip data, groundwater monitoring sheets and associated tabulated physico-chemical properties.
- AGS-compatible files containing the raw diver and logger data.
- MS Excel file(s) containing a database of the corrected groundwater level data (using manual dip and logger data).
- A database containing all chemical and field test results gathered during the monitoring event in MS Excel format or similar.
- Electronic copies of all laboratory reports (AGS or MS Excel format).

Appendix C. Analytical Suite

Groundwater monitoring suites have been developed based on historical site uses identified during the Mott Macdonald (2013) Phase 1 Contamination Assessment and the Atkins (2016) Ground Investigation.

Table C-1 Analytical Suite -A

Determinand	Maximum MDL	Unit
рН	+/-0.1	pH units
Total Organic Carbon (TOC)	<0.1	mg/l
Arsenic (dissolved)	< 0.15	µg/l
Cadmium (dissolved)	< 0.02	µg/l
Chromium (dissolved)	< 0.2	µg/l
Lead (dissolved)	< 0.2	µg/l
Mercury (dissolved)	< 0.05	µg/l
Selenium (dissolved)	< 0.6	µg/l
Boron (dissolved)	<10	µg/l
Copper (dissolved)	< 0.5	µg/l
Nickel (dissolved)	< 0.5	µg/l
Zinc (dissolved)	< 0.5	µg/l
Vanadium (dissolved)	< 0.2	µg/l
Chromium - Hexavalent (dissolved)	<5	µg/l
Cyanide - Total	<10	µg/l
Cyanide - Free	<10	µg/l
Cyanide - Complex	<10	µg/l
Ammonium (NH4 ⁺)	<15	µg/l
Sulphate as SO ₄	<0.03	mg/l
Sulphide	<5	µg/l
PAH - Speciated (EPA 16)	<0.01	µg/l
TPH CWG	<10	µg/l
BTEX & MTBE	<1	µg/l

Table C-2 Analytical Suite - B

Determinand	Maximum MDL	Unit
VOC	<1	µg/l
	<0.01 - <0.1	
SVOC	(Depending	ua/l
	on	P9/1
	compound	

Table C-3 Additional Analytical Suite - C

Determinand	Maximum MDL	Unit
Phenols - Speciated	<0.05	µg/l

Appendix D. Greenwich Peninsula Borehole logs

			Nor BO	west I	-lols .E LO	t Sc G - (oil Er Cabl	ngii E P	neeri PERCU	ng L [.] Issio	td. N		Hole ID. N1A Header
Contra	ct No.	F12	080		Meth	nod	Cable	Percu	ission (Coordinate	s 5	39163.6 E	
Client	L	Eng	lish Partne	erships	Drilli Drille	ng Rig er	Pilcon GW	1500	(Ground Lev Drientation	/el 5. V	79925.3 N .67m AOE ertical)
Consu	ltant	WS	Atkins		Logg	ged by	AH		[[Date Starte Date Comp	d 0 leted 1	7/09/2001 0/09/2001	
				PROGRESS						DR	ILLING DE	TAILS	
Date	Ti	me	Hole depth	Casing depth	Water depth		Remarks		Hardbore from depth	Hardbore to depth	Chiselling hours	J F	Remarks
07/09/20 07/09/20 10/09/20 10/09/20	001 08 001 13 001 08 001 16	800 300 800 630	0.00 2.00 2.00 15.00	0.00 1.70 1.70 14.50	Dry Dry Dry 9.00	SOH EOS SOS EOH							
Hole	Max dep	CAS	ING Casing I	Max depth of	Date	Time	Strike at	Rise	WATER S		w Cas	ing denth	Casing dep
diam.	hole at	dia.	diameter o	casing of dia.			depth	dept	th to rise		at s	trike time	to seal flov
-													
			GENER	AL NOTES				T	1	SPT D	ETAILS		
2. D 3. S 4. W	tandpipe Vater strik	seai in installi kes not	stalled and ed upon bi ted 1.70m	a rearilied 3.50 orehole compl and 8.00m.	-5.50m. etion.								
		-							* Seating b	lows only.			
NB AII	depths in	n metre	es, all dian	neters in millin	netres,				* Seating b	lows only.	F	orm Al	RIAL CP HEADER

	Norwest H	olst So LOG - (DILE CAB	ngi LE F	ne Per	er C	ing Lt	d. N	Hole II N1A Sheet 1 o	D. 1 f 2
Contract No. Project Client Consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cab Pilco GW AH	le Percu	ission		Coordinates Ground Level Orientation Date Started Date Complete	53916 17992 5.67m Vertic 07/09/ ed 10/09/	3.6 E 25.3 N AOD al 2001 2001	
	Description of Strata		Legend	Depth Below	O.D.		Sampling	SPT N &	SPT type	Inst
MADE GRO sub-angular sized concre	UND: Brown slightly clayey very s to subrounded brick, flint and med ete gravel	andy lium		<u>G.L.</u>		D	0.50	(0 blows)		
MADE GRO sand of lime.	UND: Pale green gravelly fine-coa . Gravel is fine-coarse brick and as	nrse sh.		1.70	3.97	D	2.80			
MADE GRO sand. Gravel	UND: Black very gravelly fine-coal I fine-coarse brick and ash. locally mottled black CLAY. Slight ly contaminated (ALLUVIUM).	rse		4.50	- 1.17	D	4.50			
Dark brown a remains. (AL	amorphous PEAT with occasional LUVIUM)	plant	stle stle stle e stle stle stle stle stle stle e stle stle e stle stle stle stle stle stle stle stle stle stle stle stle	6.00	-0.33	D	6.00			
Dark brown r plant remains	mottled grey organic CLAY with at s. (ALLUVIUM).	pundant	2 shis shis s shis shis shis shis shis sh	7.20	-1.53	D	7.20			
Grey brown TERRACE D	very silty fine-medium SAND. (RIV DEPOSIT).	/ER		8.80	-3.13	D	8.80			0000
√B All depths i	in metres, all diameters in millimet	ires.	X 0,000	9.90	-4.23	D	9.90			
See heade strikes. Se	er sheet for details of drilling, progr e legend sheet for key to symbols	ess and water						Version	ARIAL CP 1.00	LOG
								Revised	01/10/199	В

Norwest Holst Soil Engineering Ltd. BOREHOLE LOG - CABLE PERCUSSION

Hole ID. N1A

Sheet 2 of 2

1000						***
Contract No.	F12080	Method	Cable Percussion	Coordinates	539163.6 E	
Project	Greenwich Village				179925.3 N	
		Drilling Rig	Pilcon 1500	Ground Level	5.67m AOD	
Client	English Partnerships	Driller	GW	Orientation	Vertical	
		Logged by	AH	Date Started	07/09/2001	
Consultant	WS Atkins			Date Completed	10/09/2001	

Description of Strata	Legend	Below	O.D. Level	Sampling	SPT N a	SF	PT type	nstall- ation
Grey fine-coarse sandy flint GRAVEL. Sand is fine-coarse. (RIVER TERRACE DEPOSIT).		14.30 15.00	-9.33	D 14.30				
NB All depths in metres, all diameters in millimetres.			F		For		- ARIAL CP LC	G
see neader sneet for details of drilling, progress and water strikes. See legend sheet for key to symbols.					Ver	sion	1.00	
					Rev	ised	01/10/1998	

R			Nor BC	wes	st F HO	Holst	Sc SG	oil - C	Er CAE	ngi BLE	ine E 8	eeri & RO	ng L TARY	td.		Hole ID. N1B Header
Contra Projec	act No st	. F12 Gre	080 enwich V	illage		Metho	d a Ria		Cable	& Rot	tary	(Coordinates	5 1	539166.8 179926.9	E N
Client Consu	ıltant	Eng WS	llish Partn Atkins	nerships		Drillers Logge Core b	d by arrel		GW & AH	TF			Drientation Date Started Date Comple	ted 2	/ertical 1/09/200 21/09/200	1
						Core b	oit		Rock F	Roller						
Date	•	Time	Hole depth	PROG Cas de	RESS	Water depth		Rem	narks		Har	rd strata m depth	CABLE PEI Hard strata to depth	RCUSS Hours		AILS Remarks
11/09/2 11/09/2 19/09/2 20/09/2 20/09/2 21/09/2 21/09/2	001 001 001 001 001 001 001 001	0800 1730 0730 1730 0730 1730 0730 1730	0.00 15.00 33.00 33.00 42.30 42.30 50.00	0 14 15 15 15 15	00 .80 .80 .00 .00 .00 .00	Dry 4.00 4.00 0.00 0.90 0.00 6.70 6.70	SOH EOH C SOH R EOS SOS EOS SOS EOH	able otary								
		CAS	SING								<u> </u>					
Hole diam.	Max hole	depth of at dia.	Casing diameter	Max de casing d	pth of of dia.	Date	Time	Stri de	ike at epth	Rise dept	to 1	Time take to rise	en Flow	Cas at s	sing deptr strike time	Casing dept to seal flow
200 150	2	5.50 12.30 50.00	250 200 200	5.5 15.(15.(0											
From	1	RO To	TARY DF	RILLING	DETA	ILS Flush	Cor		Denth	Type		cromonte	SPT DET	AILS	otion in m	
depti 15.00	n D	depth 50.00	ty Water	pe Mud		Good	diame 0*	ter		1360						
			GENEF	RAL NO	TES											
1. F 2. E 3. V 4. F 5. S	Hand d Benton Vater : Rotary Standp	lug servic ite seal in strikes no drilling cc ipe install	e pit to 1. Istalled ar ted at 3.8 Intinued to led upon I	20m. nd redrill 0m and o 50.00n borehole	ed 3.5 9.90m n depti	0-5.50m. h. letion.					* 9	Seating h	ows only			
NB AI	l depti ater st	ns in metr rike rise ti	es, all dia ime in mir	imeters i nutes, ha	n millir ard stra	metres, ata time in l	hours		L		<u>*</u>	Seating b	lows only.	Form /	ARIAL COM	BINED HEADER
				-			τ. Γ							Version	1.00	0

	BOREH	DLE LOG	SOII G - C	ABL	LE &	ROTAF	Lta. RY		N1I Sheet 1	3 of 5
Contract No.	F12080	Method	Ca	ble & Ro	otary	Coordina	ates	5391	66.8 E	
Project	Greenwich village	Drilling Rig	Pilo	:on & H	ansEng	Ground	evel	1799: 5.62r	26.9 N	
Client	English Partnerships	Drillers	GV	V & TF	unseng	Orientati	on	Vertic	cal	
o 11 1	110 111	Logged by	AH			Date Sta	rted	11/09	9/2001	
Consultant	WS Atkins	Core Barrel	Ro		r	Date Co	mpleted	21/09	9/2001	
	A Martin Andrew Andr				1 			441 T		
	Description of Strata		Legend	Below	O.D.	Sampling	Blow c	ount to	SPT N	Ir
MADE GRO	OUND: Brown slightly clayey ve	ery sandy		<u> </u>	-			0100	d doput	
fine-coarse	sub-angular to subrounded bri	ick, flint								
	le graver				Ē					
					E					3
					-					-
				1 50	4 10					-
MADE GRO	OUND: Pale green gravelly fine	e-coarse		1.50	- 4.12					-188
sand of lime	e. Gravel is fine-coarse brick a	nd ash.								
					-					-
					Ē					-188
	8				-					
		Station of the		2.80	2.82					-
sand. Grave	DUND: Black very gravelly fine el fine-coarse brick and ash.	-coarse			E I				3	
										-88
					E I	×.				-18
										-188
					E					-188
										-188
01				4.50	1.12					-18
organic odo	our. Slightly contaminated (ALL	light UVIUM).								
•	,				- 1					-18
										18
					-					-
										1111
			atte atte atte	6.00	0.38		3			1
Dark brown remains (A	amorphous PEAT with occasi LLUVIUM)	onal plant	e sites sites o sites sites sites							11111
	······		a shka shka s shka shka shka ta shka shka		-					1 1
			able able ables a able ables a			20				1 1
			attle attle attle a attle attle a attle attle attle		_					1 -
Deels	mattlad group and a bit to a		a sulla sulla s	7.20	-1.58					- I - I
abundant pl	lant remains. (ALLUVIUM).	in			L I					
			====							111111
				3	-					
	2									
					-					1 1 -
				8.80	-3.18					1 1
Very soft br	own very sandy CLAY. (ALLU	VIUM).			E I					1 1
					-					1 1
										112111
	12									1111111
			0000	9.90	4.28					
IB All depths	s in metres, all diameters in mil	limetres.					Fo	orm Af	RIAL COMBINE	
See head	er sneet for details of drilling, i	progress and wate	er				Ve	ersion 1	00	
ounce. O	se logend sheet for key to syll						R	evised 01	/10/1998	

	Norwest H BOREHOL	olst S E LOG	Soil G - C	Enç ABI	gine E &	ering L	.td. Y		Hole II N1E Sheet 2 o	D. 3 of 5
Contract No. Project	F12080 Greenwich Village	Method	Cat	ole & Ro	otary	Coordinat	ies	5391(1799)	36.8 E 26.9 N	
Client	English Partnerships	Drilling Rig Drillers	Pilo GW AH	on & Ha / & TF	ansEng	Ground L Orientatic Date Star	evel in ted	5.62n Vertic	n AOD al	
Consultant	WS Atkins	Core Barrel	Roo	ck Rolle	r	Date Con	pleted	21/09	/2001	
	Description of Strata		Legend	Depth Below G.L.	O.D. Level	Sampling & coring	Blow cou drive U1	nt to	SPT N & depth	ln a
is fine-coars	brown extremley closely fissured C LONDON CLAY).	CLAY.		14.30						
IB All depths	in metres, all diameters in millime	tres.					Fom	Al	RIAL COMBINED	111111111111111111

ľ

Contract No. F12880 Greenwich Village Method Cable & Rotary Coordinates S39166.8 E 179220 N Client English Partnerships Drilling Rig Drillers Piloon & HansEng GW & TF Ground Level Ground Level S20100.00 Consultant WS Akins Core Bit Rock Roller Date Starshot Vertical Date Completed 2100/2001 Description of Strata Logged by AH Logged by Core Bit Core Bit Rock Roller Secription SPT N & depth drive U100 \$PT N & depth drive U100 \$PT N & depth Firm to stiff brown extremely closely fissured CLAY. Secription of Strata Lagend Bethot Bethot Con Sampling Blow count to & dopth SPT N & depth Firm to stiff brown extremely closely fissured CLAY. Secription of Strata Secription drive U100	N ;	Norwest H BOREHOL	lolst S	Soil G - C	Enę ABL	gine E &	ering l ROTAF	_td. ₹Y		Hole I N1E Sheet 3	D. 3 of 5
Description of Strata Legend Description Sampling Boltw Blow count to drive U100 SPT N F/m to stiff brow extremely closely fissured CLAY. (Weathered LONDON CLAY). Sampling Blow count to drive U100 SPT N Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata Image: Strata <th>Contract No. Project Client Consultant</th> <th>F12080 Greenwich Village English Partnerships WS Atkins</th> <th>Method Drilling Rig Drillers Logged by Core Barrel Core Bit</th> <th>Cal Pilo GW AH Roo</th> <th>ole & Ro con & Hi / & TF ck Rolle</th> <th>otary ansEng r</th> <th>Coordina Ground Orientati Date Sta Date Co</th> <th>ates Level ion arted mpleted</th> <th>5391 1799 5.62 Verti 11/0 21/0</th> <th>166.8 E 926.9 N m AOD ical 9/2001 9/2001</th> <th></th>	Contract No. Project Client Consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Drillers Logged by Core Barrel Core Bit	Cal Pilo GW AH Roo	ole & Ro con & Hi / & TF ck Rolle	otary ansEng r	Coordina Ground Orientati Date Sta Date Co	ates Level ion arted mpleted	5391 1799 5.62 Verti 11/0 21/0	166.8 E 926.9 N m AOD ical 9/2001 9/2001	
Firm to stiff brown extremiley closely fissured CLAY. (Weathened LONDON CLAY).		Description of Strata		Legend	Depth Below	O.D. Level	Sampling & coring	Blow co	ount to	SPT N & depth	lins
	Firm to stiff t (Weathered	prown extremley closely fissured LONDON CLAY).	CLAY.								

	Norwest F BOREHOL	LE LOG	011 6 - C	Eng ABL	gine _E &	ering L ROTAR	td. Y		N1E Sheet 4 o	, f 5
Contract No. Project Client Consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Drillers Logged by Core Barrel Core Bit	Cat Pilc GW AH Roc	ole & Ro on & Ha / & TF ck Rolle	otary ansEng r	Coordina Ground L Orientatio Date Sta Date Cor	tes .evel on rted mpleted	53910 17992 5.62n Vertic 11/09 21/09	66.8 E 26.9 N n AOD cal 1/2001 1/2001	
	Description of Strata		Legend	Depth Below	O.D.	Sampling & coring	Blow co	ount to	SPT N & depth	In
rım to stiff (Weathered	prown extremiey closely fissured LONDON CLAY).	ULAY.								
Green sand	y CLAY	s)								
NB All depths See head	s in metres, all diameters in millim ler sheet for details of drilling, pro	etres. gress and wat	er				Fc	orm A	RIAL COMBINED	
	, , , , , , , , , , , , , , , , , , ,		-1				Ve	ersion 1.	00	

·

Construct No. F12000 Greenwich Village Method Cable & Rotary Coordinates 539166.6 E 179282.8 N Dient English Patherships Drillers GW & TF Ground Level 582m.00D Jonsultant WS Atkins Core Barel Core Barel Core Barel Core Barel Core Barel Date Startadon 1109/2001 Description of Strata Leggerd Date Scong Blow count to drive U100 SPT N It Green sandy CLAY Solu Long Date Sampling Blow count to drive U100 SPT N It Green sandy CLAY Solu Long Date Secong	N i	Norwest BOREHC	Holst S	Soil Er G - CAE	ngine BLE &	ering I ROTAF	_td. ≀Y	Hole ID. N1B Sheet 5 of 5						
Description of Strata Legend Biolog Legend Biolog Legend Biolog Level Creen sandy CLAY Green sandy CL	Contract No. Project Client Consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Drillers Logged by Core Barrel Core Bit	Cable & Pilcon & GW & Ti AH Rock Ro	Rotary HansEng =	Coordin Ground Oriental Date St Date Co	ates 539 179 Level 5.62 ion Ver arted 11/0 ompleted 21/0	166.8 E 926.9 N 2m AOD tical 09/2001 09/2001						
Green sandy CLAY		Description of Strata	9999 (*** 9919 - 1240) 1	Legend Belo	w O.D.	Sampling & coring	Blow count to drive U100	SPT N In & depth a						
NB All depths in metres, all diameters in millimetres.	Green sand	omplete at 50.00 m.	χ.											
See beader speet of belais of drilling, brodless and water	NB All depth	s in metres, all diameters in mil	limetres.	ter	15		Form	ARIAL COMBINED LO						
							Revised	01/10/1998						
R			Nor BO	west REHOL	H <mark>ols</mark> .E LO	t Sc G - 0	D <mark>il E</mark> r Cabl	ngi E F	neer PERC	i <mark>ng L</mark> USSIC	td. N			Hole ID. N2A Header
----------------------	-------------------------------	--	--	---	-----------------------------	------------------------	-------------------------------	-------------	-----------------------	---	-----------------	---	------------------------------------	----------------------------------
Contra	ict No.	F12	080		Meth	nod	Cable	Percu	ission	Coordinate	es	53933	2.3 E	
Client	Itant	Eng	lish Partne Atkins	age erships	Drilli Drille Logg	ng Rig er ged by	Pilcon GW JFK	1500		Ground Le Orientatior Date Starte	evel n ed	18015 5.80m Vertica 17/09/ 17/09/	8.3 N AOD al 2001 2001	I
Date		Time	Hole	PROGRESS Casing	Water	1	Remarks		Hardbore	DF Hardbore	RILLING		.S R	emarks
17/09/20	001	0800	depth 0.00	depth	depth	801			from dept	th to depth	hou	rs		
17/09/20	001	1730	14.00	14.00	0.00	EOH								
		CAS	SING	1					WATER					
Hole diam.	Max of hole	depth of at dia.	Casing diameter	Max depth of casing of dia.	Date	Time	Strike at depth	Rise dep	to Time t th to ri	aken Flo	w (Casing d at strike	epth time	Casing de to seal flo
250	1	4.00	250 150	4.70 14.00										
1 4	land d		GENER	AL NOTES						SPT D	DETAILS	3		
2. B 3. S 4. V	ientoni itandpi Vater s	ite seal in ipe install strikes no	istalled an led upon b ted at 3.40	d redrilled 3.50 orehole comp)m and 10.40n	D-5.50m. letion. n.									
						-			* Seating	blows only				
NB All wa	i depth ater str	is in metr rike rise ti	es, all diar ime in min	neters in millir utes, chisellind	netres, g time in h	ours.						Form	AF	RIAL CP HEAD
					-	-						Version	1.0	00

F

	BOREHOLE	LOG - (ngi LE F	PER	er C	USSIC	(a.)N		N2A Sheet 1 c	D. \ of 2
Contract No. Project Client Consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cabl Pilco GW JFK	e Percu on 1500	ission	State 1 State 1	Coordinates Ground Level Orientation Date Started Date Complet	ed	53933 18015 5.80m Vertica 17/09/ 17/09/	2.3 E 8.3 N AOD al 2001 2001	
	Description of Strata		Legend	Depth Below	O.D.		Sampling	SP	TN&	SPT type	ins
MADE GRO MADE GRO subangular-r brick, quartz MADE GRO subangular-r and concrete	UND: Tarmac UND: Black / Brown fine-coarse rounded gravel. Gravel is of red & and flint. UND: Black / brown fine-coarse rounded gravel. Gravel is of ash, t a.	yellow prick		<u>G.L.</u> 0.20 0.50	5.30 5.30	D	0.70		NOWS)	& depth	
Very soft gre Clay has con	ey CLAY with occasional yellow m itaminated odour. (ALLUVIUM)	ottling.		4.50	1.30	D	3.60				
Discis				6.80	-1.00	D	6.80				
Black spong wood fragme	y fibrous PEAT containing rootlets ents. (ALLUVIUM)	s and	a stille stille a stille stille a e stille stille a stille stille a stille stille stille e stille stille a stille stille stille a stille stille stille a stille stille a stille stille a stille				45	والمتحد والمتحد والمتحد والمتحد			
			ables ables ables to ables ables a ables ables ables to ables ables a	9.80	-4.00	D	9.80				11111
B All depths i	in metres, all diameters in millime	tres.	<u></u>]			1		[Form	ARIAL CE	

Contract No. Project F12080 Greenwich Village Method Dilling Rig Dilling Rig Dill		BOREHOLE	LOG - (DILE CABI	ngi LE F	PER	er C	ing Lt USSIO	d. N	N2 Sheet 2	D. A of 2
Description of Strata Legend Balow Level Sampling SPT N.8. SPT byee In Very soft purple grey CLAY occasional shell fragments noted. (ALLUVIUM). Image: CLAY occasional shell fragments. noted. (ALLUVIUM). Image: CLAY occasional shell fragments. noted. (ALLUVIUM). Image: CLAY occasional shell fragments. (RIVER TERRACE DEPOSITS). Image: CLAY occasional shell f	Contract No. Project Client Consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cabl Pilcc GW JFK	e Percu on 1500	ission		Coordinates Ground Level Drientation Date Started Date Complete	53933 18015 5.80m Vertic 17/09 d 17/09	2.3 E 38.3 N AOD al /2001 /2001	
Very soft purple grey CLAY occasional shell fragments noted. (ALLUVIUM). Grey slightly sandy GRAVEL Gravel is fine-coarse angular-submodel chert, fint, quart and shell fragments. (RIVER TERRACE DEPOSITS). Cable Percussion boring complete at 14.00 m. Cable Percussion boring complete at 14.00 m.		Description of Strata		Legend	Depth Below	O.D. Level		Sampling	SPT N &	SPT type & depth	Ins
	Very soft pur shell fragmer Grey slightly angular-subn fragments. (F	ple grey CLAY occasional hts noted. (ALLUVIUM). sandy GRAVEL. Gravel is fine-co ounded chert, flint, quart and shel RIVER TERRACE DEPOSITS).	parse 		10.70	-4.90	D	10.80			
NB All depths in metres, all diameters in millimetres. See header sheet for details of drilling, progress and water		in motoo, all diamotors in millima	trac								

		BC	DREHOL	E LC)G - (igi E P	PERCI	ING L JSSIO	ia. N		N3A Header	
Contra	ct No. F	12080 Greenwich V	illage	Met	hod	Cable	Percu	ission	Coordinates	s 5	39046.9 E		
Client	E tant V	nglish Partr	nerships	Drill Drill Logg	ing Rig er ged by	Pilcon GW AH	1500		Ground Lev Orientation Date Starte	vel 6. V d 1:	6.06m AOD Vertical 12/09/2001 13/09/2001		
Date	Time	Hole	PROGRESS Casing	Water		Remarks		Hardbore	DR	LLING DE	TAILS	Pemarke	
40/00/00		depth	depth	depth				from dept	to depth	hours			
12/09/20	01 1730	14.00	0.00	0.00	EOH			1.70 2.20	2.00 2.40	0.50 0.50			
		ASING							CTDWC0				
Hole diam.	Max depth of hole at dia	of Casing	Max depth of	Date	Time	Strike at	Rise	to Time ta	ken Flo	w Cas	ing depth	Casing der	
250 150	4.50 14.00	250 150	4.50 14.00										
		GENER	RAL NOTES									I	
1. H 2. B 3. S 4. W	and dug ser entonite sea andpipe ins 'ater strike n	vice pit to 1. I installed ar talled upon I oted at 11.8	20m. nd redrilled 3.5(borehole compl 0m.	0-5.50m. etion.		Dept	Туре	Increme	ntal blow cou	unt/penetra	ation in mr	n	
								* Seating	blows only.				
NB All	depths in m	etres, all dia	meters in millin	netres,						F			
wa	ter strike ris	e time in mir	nutes chicolling	time in h	oure							VAL OF TIEADE	

N :	Norwest H BOREHOLE	olst So	DILE CAB	ngi LE F	nee PER	er	ing Lt	d. N	Hole II N3A Sheet 1 o). of 2
Contract No. Project Client Consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cabl Pilcc GW AH	e Percu on 1500	ission		Coordinates Ground Level Orientation Date Started Date Complete	53904 18030 6.06m Vertic: 12/09/ ed 13/09/	6.9 E 2.5 N AOD al 2001 2001	
	Description of Strata		Legend	Depth Below	O.D.		Sampling	SPT N &	SPT type	In
MADE GRO sandy grave subrounded	UND: Brown very clayey fine-coa I. Gravel fine-coarse angular to brick, flint and ash.	arse		<u> G.L. </u>	-	D	0.50	(0 blows)	a depin	
MADE GRO fine-coarse, subangular a	UND: Black clayey gravelly sand gravel fine-medium angular to ash. Strong odour noted.	. Sand is		2.60	3.46	D	2.60			
Brown mottle odour. (Cont	ed black CLAY. Slightly organic a taminated ALLUVIUM).	and tar		4.50	-1.64	D	4.50			
Dark brown remains. (AL	amorphous PEAT with abundant LUVIUM).	plant	when some and the some and t							
NB All depths See heade	in metres, all diameters in millimer sheet for details of drilling, pro	etres. gress and water	fa shka shka s					Form	ARIAL CF	>LO
otrikon Sc	a legend cheat for key to symbol								4.00	1.1

	Norwest H BOREHOLE	olst So LOG - (DILE CABI	ngi LE F	nee PER	er C	ing Lt USSIO	d. N	Hole ID N3A Sheet 2 o). 1 f 2
Contract No. Project Client Consultant	F12080 Greenwich Peninsula English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cabl Pilco GW AH	e Percu	ission	() () () []	Coordinates Ground Level Drientation Date Started Date Complete	53904 18030 6.06m Vertic 12/09 d 13/09	46.9 E 02.5 N 0 AOD al /2001 /2001	
- ALIA	Description of Strata	*	Legend	Depth Below	O.D.		Sampling	SPT N &	SPT type	Ins
Dark brown a remains. (AL Grey CLAY v organic odou	amorphous PEAT with abundant LUVIUM). with occasional plant remains, slig ur. (ALLUVIUM)	plant ght	Aller alle	10.50	-4.44	D	10.50			
Grey silty ve coarse flint C (RIVER TER	ry sandy angular to rounded fine GRAVEL. Sand is predominantly o RACE DEPOSITS).	to coarse.		12.00		D	12.00			
Cable Percu	ssion boring complete at 14.00 m	1.								
	· ·									
					-					
NB All depths See heade	in metres, all diameters in millime er sheet for details of drilling, prog	etres. gress and water						Form	ARIAL CP	LOC
strikes. Se	e leaend sheet for key to symbol	c						Version	1.00	



Norwest Holst Soil Engineering Ltd.

Hole ID. N4A

Header

BOREHOLE LOG - CABLE PERCUSSION

Contract No.	F12080	Method	Cable Percussion	Coordinates	538827.7 E
Project	Greenwich Village				180154.6 N
		Drilling Rig	Pilcon 1500	Ground Level	6.23m AOD
Client	English Partnerships	Driller	GW	Orientation	Vertical
		Logged by	JFK	Date Started	13/09/2001
Consultant	WS Atkins			Date Completed	15/09/2001

			PROGRESS	6			DR	ILLING DET	AILS
Date	Time	Hole depth	Casing depth	Water depth	Remarks	Hardbore from depth	Hardbore to depth	Chiselling hours	Remarks
13/09/2001 13/09/2001 14/09/2001 14/09/2001	0800 1730 0800 1730	0.00 3.70 3.70 14.70	0.00 3.70 3.70 14.70	Dry 3.50 3.70 0.00	SOH EOS SOS EOH	1.40 2.80 3.50 3.70	1.60 3.10 3.70 4.20	0.75 0.75 1.25 3.25	

	CAS	SING					1	VATER STR	RIKES		
Hole diam.	Max depth of hole at dia.	Casing diameter	Max depth of casing of dia.	Date	Time	Strike at depth	Rise to depth	Time taken to rise	Flow	Casing depth at strike time	Casing depth to seal flow
250 150	4.50 14.70	250 150	4.50 14.70			deput	deptit	10 1136			to seal now

GENERAL NOTES			SPT DETAILS		
 Hand dug service pit to 1.20m. Bentonite seal installed and redrilled 3.50-5.50m. 	Depth	Туре	Incremental blow count/pene	etration in	mm
 Standpipe installed upon borehole completion. Water strike noted at 3.50m (contaminated) and 10.00m. 					
			* Seating blows only.		
NB All depths in metres, all diameters in millimetres, water strike rise time in minutes, chiselling time in hours				Form	ARIAL CP HEADER
				Version	1.00
			· · · · · · · · · · · · · · · · · · ·	Revised	01/10/1998

	Norwest H BOREHOLE	olst So LOG - (DILE CAB	ngi LE F	PER	er C	ing Lt	d. N	Hole II N4A Sheet 1 o). of 2
Contract No. Project Client Consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cab Pilco GW JFK	le Percu on 1500	ission		Coordinates Ground Level Orientation Date Started Date Complete	53882 18015 6.23m Vertica 13/09/ ed 15/09/	7.7 E 4.6 N AOD al 2001 2001	
	Description of Strata		Legend	Depth Below	O.D.		Sampling	SPT N &	SPT type	In
MADE GRO	UND: Tarmac			0.20	6.03	-		-		
MADE GRO fine cohesive medium-coa MADE GRO gravel. Sand	UND: Brown grey sandy gravel. S e material. Gravel is subrounded, rse concrete, brick and flint. UND: Brown green subrounded s I is coarse subangular quartz. Gra	and is andy average as a second se		0.50	[_ 5.73 [_	D	0.50			0. 0. 0. 0. 0.
MADE GRO	medium quartz, flint and lime. UND: Grey clayey gravel. Clay is	soft.		1.70	- 4.53	D	1.70			
flint.	oangular-subrounded, medium bri	ck and				D	2.60			
3.50m Grave	el is coarse. Slightly contaminated	L			-	D	3.50			
Very soft gre petroleum oc	ey and black mottled CLAY, conta Jour. (ALLUVIUM)	ining		4.20	2.03	D	4.20			1111111111
				7.00		D	7.00			
Black spongy petroleum oc	y fibrous PEAT. Contains strong dour. (ALLUVIUM)		avec solle, solle, a solle, sol					المتعدية المتحدية		
Grey and bla material. Slig	ick organic CLAY. Some fibrous p ht organic odour. (ALLUVIUM)	eaty	to solve	9.10	-2.87	D	9.10		-	
				10.00	-3 77		10.00	Ē		1
NB All depths	in metres, all diameters in millime	tres.	-1	10,00	-9.0			Form	ARIAL CP	1988
See heade strikes. Se	er sneet for details of drilling, prog e legend sheet for key to symbols	ress and water					(+):	Version	1.00	200
2	Serve shoet for hey to symbolis									

Contract No. F12080 Method Cable Parcussion Coordinates 53827.7 E Project Greenwich Peninsula Drilling Rig Pilcon 1500 Ground Level 623m ADD Consultant WS Atkins Drilling Rig Pilcon 1500 Ground Level 623m ADD Consultant WS Atkins Depting Rig Depting Rig Depting Rig Sampling SPT N & SPT type In Grey and black organic CLAY. Some Bbrous peaty 0 00 3.77 D 10.00 Sampling SPT N & SPT type In Grey and black organic CLAY. Some Bbrous peaty 0 00 3.77 D 10.00 3.77 D 10.00 Sampling SPT N & SPT type In Grey and black organic CLAY. Some Bbrous peaty 0 00 0.00 Sampling SPT N & SPT type In Grey & white same arganic addar. (ALLUX/UM) Grey & Swite same arganic model. Sampling Sum arganic addar. Sampling Sum arganic addar. Sampling Sum arganic addar. Sum arganic adda	N	Norwest He BOREHOLE	olst Sc LOG - (оіl E САВ	ngi LE F	ne PER	ering L	.td. DN	Hole II N4A Sheet 2 o	D. M of 2
Description of Strata Legen Description Balow Sampling SPT type (Ublows) Set Type 8 depth Gray and black organic CLAY. Some fibrous peaty material. Sliph regarding dots r. (ALLUVIM). 10.00 3.77 D 10.00 3.77 D 10.00 10	Contract No. Project Client Consultant	F12080 Greenwich Peninsula English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cabl Pilco GW JFK	e Percu on 1500	ission	Coordinates Ground Leve Orientation Date Started Date Comple	5388 1801 6.23r Vertio 13/09 eted 15/09	27.7 E 54.6 N n AOD xal 1/2001 1/2001	
Grey and black organic CLAY. Some fibrous peaty material. Silver TERRACE DEPOSITS) D 10.00 -3.77 D 10.00 Grey & white sandy angular fine-coarse filth CRAVEL. Sand is coarse. (RIVER TERRACE DEPOSITS) D 0.00 -4.77 D 10.00 Cable Percussion boring complete at 14.70 m. D 14.70 -8.47 D 0.00 4B< All depths in metres, all diameters in millimetres. See beader sheet for details of drilling, progress and water strikes. See beader sheet for details of drilling, progress and water strikes. See beader sheet for details of drilling, progress and water strikes. See beader sheet for details of drilling, progress and water Term ARAUL CP LOC		Description of Strata		Legend	Depth Below	O.D. Level	Sampling	SPT N & (U biows)	SPT type & depth	Ins
NB All depths in metres, all diameters in millimetres. See header sheet for details of drilling, progress and water strikes. See legend sheet for key to symbols. Version	Cable Percus	ack organic CLAY. Some fibrous p ght organic odour. (ALLUVIUM) a sandy angular fine-coarse flint G ise. (RIVER TERRACE DEPOSITS ssion boring complete at 14.70 m.	eaty RAVEL. S)		10.00	8.47				
vision to vision hour of hour of hour hour hour hour hour hour hour hour	NB All depths i See heade	in metres, all diameters in millimet er sheet for details of drilling, progr e legend sheet for key to symbols	res. ess and water	J			L	Form	ARIAL CP	LOG

N		Norv BC	West I	Holst	: Soil DG - (Er CAE	ngi BLE	neer & RC	ing L	td.		Hole ID. N4B Header
Contrac Project	ct No. F12 Gre	2080 eenwich Vil	llage	Metho	d g Rig	Cable Pilcon	& Rot	ary eco T6	Coordinates Ground Leve	538 180 el 6.22	829.2 E 157.3 N 2m AOD	
Client Consult	Eng tant WS	glish Partne S Atkins	erships	Driller Logge Core b Core b	s d by parrel pit	GW & JFK Rock F	HJ Roller		Orientation Date Started Date Comple	Ver 18/ eted 27/	tical 09/2001 09/2001	
Date	Time	Hole	PROGRESS Casing depth	S Water depth	Rer	narks		Hard strata	CABLE PE Hard strata	RCUSSIOI Hours	N DETAI R	LS emarks
18/09/20 18/09/20 19/09/20 20/09/20 20/09/20 24/09/20 24/09/20 25/09/20 25/09/20 26/09/20	01 0800 01 1730 01 0800 01 1730 01 0800 01 0800 01 0800 01 1800 01 1820 01 0800 01 1820 01 0800 01 1730 01 0800	0.00 5.50 5.50 16.50 18.20 38.00 18.20 38.00 56.00 56.00	0.00 5.50 5.50 16.50 18.20 38.00 18.20 38.00 56.00 56.00	Dry Dry Dry Dry Dry Dry Dry Dry Dry Dry	SOH EOS SOS EOH Cable EOS SOH Rotan SOS EOS	1		3.50 3.80 5.50 17.40	3.60 4.00 6.10 18.20	0.50 0.75 1.25 2.00		
	CAS	SING						WATER S	STRIKES			
diam.	Max depth of hole at dia.	diameter	Max depth of casing of dia.	Date	Time Str	ike at epth	Rise dept	to Time tak	en Flow	Casin at stri	g depth ke time	Casing dep to seal flo
	000	130	56.00								-	
From	RC	TARY DRI	ILLING DETA sh	ILS Flush	Core	Depth	Type	Increment	SPT DET	AILS	on in mn	 1
18.20	60.00	Wate	e r	Good	0*				•			
1. Ha 2. Be 3. Ro de 4. St 5. W 10	and dug servic entonite seal ir otary drilling ca opth. andpipe instal ater strike not 0.00m.	GENER be pit to 1.2 Installed an ontinued fro lled upon b ed at 3.50r	AL NOTES 20m. d redrilled 3.5 om 18.20m to orehole comp n (contamina	0-5.50m. 60.00m oletion. led) and								
								* Section				

	Norwest F BOREHO	HOIST S	i - C	En(ABl	gine _E &	ering L ROTAR	_td. XY	N Shee	4B t 1 of 6
Contract No. Project	F12080 Greenwich Peninsula	Method	Cal	ole & R	otary	Coordina	ites (538829.2 E 180157.3 N	
Client	English Partnerships	Drilling Rig Drillers	GW	:on & E / & HJ <	deco 16	Ground I Orientation	Level (on \	6.22m AOD Vertical	
Consultant	WS Atkins	Core Barrel Core Bit	Ro	ck Rolle	ər	Date Cor	mpleted 2	27/09/2001	
	Description of Strata		Legend	Depth Below	O.D.	Sampling	Blow coun	t to SPT	N Ir
MADE GRO	UND: Tarmac		****	<u> </u>	6.02				- ::
MADE GRO is fine. Grav concrete, bri MADE GRO sandy grave subrounded	DUND: Brown grey sandy gravel. el is subrounded, medium-coarse ick and flint. DUND: Brown and green subroun el. Sand is coarse quartz. Gravel , medium quartz flint and lime.	Sand e ded is		0.50	5.72				
MADE GRO subangular-	UND: Grey clayey gravel. Grave subrounded, medium brick and fl	l is int.		2.50	3.72				
3.50m Grave	el is coarse. Slightly contaminate	d.		4.20	2.02				
petroleum o	dour. (ALLUVIUM)	ng -							
Black spong petroleum oo	y fibrous PEAT. Contains strong dour. (ALLUVIUM)		We alke a she she alke a she she alke a she she alke alke alke alke alke alke alke alke alke alke alke alke alke alke alke alke alke alke	7.00					
Blue grey Cl	AY with gravel. (ALLUVIUM)		able, able, able, able, able, a salle, able, a able, able, able of the able	9.40	-3.18				l
B All depths	in metres, all diameters in millim	etres.		10.00					188
See heade	er sheet for details of drilling, pro	gress and wate	r				Form	ARIAL COMBI	NED LOG
strikes Se	e legend sheet for key to symbo	S					Version	1.00	

I

Contract No. P12080 Project Greenwich Peninsula Dilling Rig Client English Partnerships Differs Loggade by Core Bit Rock Roller Description of Strata Description of Strata De	Contract No. F120 Project Gree Client Engli Consultant WS / Blue grey CLAY wit Brown sandy GRAN DEPOSITS)	080 enwich Peninsula ish Partnerships Atkins Description of Strata ith gravel. (ALLUVIUM) VEL. (RIVER TERRACE	Method Drilling Rig Drillers Logged by Core Barrel Core Bit	Cab Pilc GW JFK Roc	ole & Ro con & Eo / & HJ (ck Rolle	otary deco T6	Coordina Ground I Orientati	ates Level	53882 18015 6.22m	29.2 E 57.3 N	
Description of Strata Legent Depth Biolog D.D. Biolog Sampling & coring Biow rount to drive U100 SPT N & depth Blue grey CLAY with gravel. (ALLUV/UM) DEPOSITS) 0.0 -3.78	Blue grey CLAY with Brown sandy GRAN DEPOSITS)	Description of Strata ith gravel. (ALLUVIUM) VEL. (RIVER TERRACE		Legend		r	Date Sta Date Co	irted mpleted	Vertic 18/09 27/09	al /2001 /2001	
Blue grey CLAY with gravel. (ALLUVIUM) 0 0 0 0 0 0 Brown sandy GRAVEL. (RIVER TERRACE 0 0 0 0 0 0 DEPOSITS) 0 0 0 0 0 0 0 Stiff grey CLAY (driller's description) - (LONDON 0 0 0 0 0 0 Stiff grey CLAY (driller's description) - (LONDON 0 0 0 0 0 0 16.50-17.60m Brown grey sandy GRAVEL (Scour feature?) 0 0 0 0 0 0	Blue grey CLAY with Brown sandy GRAN DEPOSITS)	ith gravel. (ALLUVIUM) VEL. (RIVER TERRACE		Logena	Depth Below	O.D.	Sampling	Blow cou	unt to	SPT N	In
Stiff grey CLAY (driller's description) - (LONDON CLAY/LAMBETH GROUP)											
	Stiff grey CLAY (dri CLAY/LAMBETH G 16.50-17.60m Brow feature?)	iller's description) - (LONDO GROUP) vn grey sandy GRAVEL (Sco	bur		13.10						

	Norwest H BOREHO	Holst S LE LOG	Soil 5 - C	Enę ABL	gine _E &	ering l	_td. RY		Hole II N4E Sheet 3 c	D. 3 of 6
Contract No. Project Client Consultant	F12080 Greenwich Peninsula English Partnerships WS Atkins	Method Drilling Rig Drillers Logged by Core Barrel Core Bit	Cat Pilc GW JFK Roc	con & Ed (& HJ (& Rolle	otary deco T6 er	Coordina Ground Orientati Date Sta Date Co	ates Level on irted mpleted	53882 18015 6.22m Vertic 18/09 27/09	29.2 E 57.3 N AOD al /2001 /2001	
	Description of Strata		Legend	Depth Below	O.D.	Sampling	Blow co	ount to	SPT N	In
Stiff grey CL CLAY/LAMB	AY (driller's description) - (LONE SETH GROUP)	DON		28.00	21.78					
(GLAUCONI	TÌC/THANET SAND)									
IB All depths i See heade	in metres, all diameters in millime er sheet for details of drilling, pro-	etres.	r <u>aansial</u> r			-3 	For	n ARI	AL COMBINED L	
oou neade	a one of details of drining, prog	jiess and wate					1 20			

Contract No. Project Client Consultant Green SAND (GLAUCONI [*]	F12080 Greenwich Peninsula English Partnerships	Method Drilling Rig	Ca		the second second		57#N 3			
Green SAND (GLAUCONI	WS Atkins	Drillers Logged by Core Barrel Core Bit	Pilo GV JFF	con & Er V & HJ K	otary deco T6 er	Coordina Ground Orientati Date Sta Date Co	ates Level on arted mpleted	5388 1801 6.22n Vertic 18/09 27/09	29.2 E 57.3 N n AOD cal v/2001 v/2001	
Green SAND (GLAUCONI	Description of Strata		Legend	Depth Below	O.D.	Sampling & coring	Blow co	ount to	SPT N	lr
	(driller's description). TIC/THANET SAND)			<u>G.L.</u>					& depin	

Dontract No. Project P12080 Greenwich Peninsula Method Cable & Rotary Dilling Rig Logged by Core Barrol Core Bit Coordinates Rock Roller 53829.2 E 180157.3 ADD Orientation 53829.2 E 180157.3 ADD Orientation Consultant WS Atkins Drilling Rig Logged by Core Barrol Core Bit Pilcon & Edeo 16 Rock Roller Ground Level Date Starka Sampling # 40092001 Date Starka Description of Strata Legend Barrol Core Bit Doub Rock Roller Sampling # 40092001 Blow count to # 40092001 SPT N & \$ 4.6eph In Barrol Core Bit Sampling # 4.6eph Blow count to # 4.0eph SPT N & \$ 4.6eph In Barrol Core Bit In Barrol Barrol Core Bit In Barrol Barrol Barrol Core Bit In Barrol Barr	Contract No. F Project C Client E Consultant V	E12080 Greenwich Peninsula English Partnerships WS Atkins	Method Drilling Rig Drillers Logged by Core Barrel Core Bit	Cal Pilo GW	ble & Ro	otary	Coordina	ates	53882	9.2 E	-
Description of Struta Legend Both Green SAND (dillers description). (GLAUCONITIONTHANET SAND) (GLAUCONITIONTHANET SAND)	Green SAND (Description of Strate		Ro	V & HJ < ck Rolle	deco T6 r	Ground Orientat Date Sta Date Co	Level ion arted mpleted	18015 6.22m Vertic 18/09 27/09	i7.3 N i AOD al /2001 /2001	
Green SAND (driller's description). (GLAUCONITIC/THANET SAND) C	Green SAND (Description of Strata		Legend	Depth Below	O.D. Level	Sampling & coring	Blow cou	unt to	SPT N & depth	In
	GLAUCONITIO	driller's description). C/THANET SAND)									

Î

	BOREHOL	E LOG	- C	Eng ABL	gine .E &	ering L ROTAR	.td. Y		N4E Sheet 6 c). } of 6
Contract No. Project Client Consultant	F12080 Greenwich Peninsula English Partnerships WS Atkins	Method Drilling Rig Drillers Logged by Core Barrel Core Bit	Cab Pilco GW JFK Roc	on & Eo (& HJ (& Rolle	otary deco T6 r	Coordina Ground L Orientatio Date Sta Date Cor	tes .evel on rted mpleted	53882 18015 6.22m Vertic 18/09 27/09	29.2°E 57.3 N 1 AOD :al /2001 /2001	
	Description of Strata	L	.egend	Depth Below	O.D.	Sampling & coring	Blow co	unt to	SPT N & depth	In
Green SAN (GLAUCON	D (driller's description). ITTIC/THANET SAND)									
White CHAL CHALK) 52.50-53.10 description)	LK (driller's description) (UPPER Om Hard LIMESTONE (drillers			52.50						
NB All depths	in metres, all diameters in millime	etres.		00.00	-00.701		For	m AF	RIAL COMBINED	
See head	ler sheet for details of drilling, prog	gress and water					For	m AF	MAL COMBINED	LUG

I

		No	rwest	Hols	t So	oil Er	ngi	neeri	ng Lt	td.		Hole ID. N5A
		BC	REHO	E LO	G - (CABL	ΕP	ERCL	JSSIO	N	Sing depth strike time ration in mm	Header
Contra Projec Client Consu	act No. F t G E Iltant V	12080 Greenwich V Inglish Partr VS Atkins	illage nerships	Meth Drilli Drille Logg	nod ng Rig er led by	Cable Pilcon ID AH	Percu	ssion	Coordinates Ground Lev Orientation Date Starte Date Comp	s 5: 11 vel 5. V d 04 leted 05	39487.8 E 79954.5 N 68m AOD ertical 4/09/2001 5/09/2001)
Doto	Timo	Liele	PROGRESS	3	1				DR	ILLING DE	TAILS	
Dale		depth	depth	depth		Remarks		from depth	to depth	hours) F	Remarks
04/09/2 04/09/2	001 0800 001 1730	0.00	0.00 15.00	Dry 0.00	SOH						539487.8 E 179954.5 N 5.68m AOD Vertical 04/09/2001 05/09/2001 DETAILS lling Ring casing depth at strike time etration in mm Revised 01/	
	A No. F12080 Greenwich Village Method Cable Percussion Coordinates 539437.3 E 179984.5 N English Partnerships Drilling Rig Pilon Ground Level 58m.AOD Logged by AH Dotter Stando Ordentation Vertical 1ant WS Askins Driller D Detection Ground Level 58m.AOD 1ant WS Askins Driller D Detection Ground Level 04002001 1ant WS Askins Date Completed 05/09/2001 Date Completed 05/09/2001 1ant WS Askins Date Time depth depth depth Gepth Remarks 1 1730 0.00 0.00 0.00 Date Date Completed 05/09/2001 1 0.00 0.00 0.00 Date Time Strike at 10.50 m. SPT DETAILS and dug son (Casing Max depth of Logne Date Time Strike at 10.50 m. SPT DETAILS and dug son (Casing Max depth of 1.20 m. SPT DETAILS SPT DETAILS and dug son (Casing dug time to 1.20 m. SPT DETAILS SPT DETAILS and dug son (Casing dug time to 1.20 m. SPT Detai											
Hole diam.	Max depth of hole at dia	of Casing diameter	Max depth of casing of dia.	Date	Time	Strike at depth	Rise dept	to Time ta	ken Flor	w Cas	ing depth trike time	Casing depth to seal flow
150	15.00	150 -	15.00									
1 1		GENE	RAL NOTES					1	SPT D	ETAILS		
2. E 3. S 4. V	Bentonite sea Standpipe ins Vater Strike a	l installed a talled upon at 10.50m.	nd redrilled 3.5	0-5.50m. oletion.								
								* Seating t	plows only.			
NB AI	I depths in mater strike ris	etres, all dia e time in mi	ameters in milli nutes, chisellir	metres, ig time in h	ours.			* Seating I	blows only.	F	orm A	RIAL CP HEADER

I

l

	BOREHOLE	LOG - (DII E CABI	ngi LE F	PER	er C	ing Lt USSIO	d. N	N5A Sheet 1	D. 4 of 2
Contract No. Project Client	F12080 Greenwich Village English Partnerships	Method Drilling Rig Driller	Cabl Pilco ID	e Percu on	ission	 	Coordinates Ground Level Orientation	53948 17995 5.68m Vertic	37.8 E 54.5 N 5 AOD al	
Consultant	WS Atkins	Logged by	AH			1	Date Started Date Complete	04/09 ed 05/09	/2001 /2001	
	Description of Strata		Legend	Depth Below	O.D. Level		Sampling	SPT N &	SPT type & depth	lr á
MADE GROU	UND: Tarmac			0.10	- 5.58			5		-
MADE GROU MADE GROU Gravel subar	<u>UND: Concrete</u> UND: Brown slightly gravelly san ngular-rounded brick and flint. Sa	dy clay. nd /		0.30 0.50	- 5.38 - 5.18 -	D J	0.50 0.50			
\fine-coarse. MADE GROU subangular to	UND: Brown slightly clayey sand	y/		1.30	- 4.38		č.		-	
MADE GROU	UND: Grey mottled black clay.					J	1.50 1.50			
MADE GROU Gravel subar fine-coarse.	UND: Dark grey sandy gravelly c ngular-rounded flint and brick. Sa	lay. nd		2.50	- - - - - - -	D J	2.50 2.50			
Light brown r odour.	nottled dark grey CLAY. Slight o	rganic		3.50	2.18	DJ	3.50 3.50			
Grey slighty s Sand is fine.	sandy CLAY with occasional roof Slight organic odour.	ilets.		4.50	 	D	4.50 4.50		Ĕ	
Dark brown c rootlets.	organic CLAY, with occasional re	lic		5.70	-0.02	D	6.00 6.00		*	
			······································	8 30	- 260	C.	8 30			
Very dark gre	ey very sandy CLAY.			в.3U		- D -	9.80 9.80			<u>entrestrestres</u>
NB All depths i	in metres, all diameters in millime	etres.	Him Ith		L	J	5.00			
See heade	r sheet for details of drilling, prog	gress and water						Form	ARIAL CI	PLO
etrikes So	e legend sheet for key to symbol	s						Versio	1.00	

-

I

Ì

	Norwest Ho BOREHOLE	olst Sc LOG - (DILE CABI	ngi LE F	nee PER	er C	ing Lt USSIO	d. N		Hole ID N5A Sheet 2 of). f 2
Contract No. Project Client Consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cabl Pilco ID AH	e Percu	ssion		Coordinates Ground Level Orientation Date Started Date Complete	53 17 5.6 Ve 04 04	9487.8 9954.9 88m A rtical /09/20 /09/20	3 E 5 N OD 01 01	
	Description of Strata		Legend	Depth Below	O.D.		Sampling	SPT N	&	SPT type	Ins
Very dark gre	ey very sandy CLAY.			G.L.	-						-
Grey sandy G	SRAVEL (TERRACE GRAVEL).			10.70	-9.32	DJ	10.70				
					- - -			-			
B All depths in See heade	n metres, all diameters in millime r sheet for details of drilling, prog	tres. ress and water						Fo	rm	ARIAL CP	LOG
	a logand about for log to such als							110	-	1 00	

4

Į,

Contract No. F12080 Greenwich Village Method Cable Percussion Coordinates 539331.4 E 1779808.0 N Client English Partnerships Drilling Rig Diller Pilcon 1500 Ground Level 4.99m ACD Consultant WS Atkins Doller ID Orientation Vertical 3108200 Date Time Holg Casing Water Remarks Date Stated 3108200 1708 0.00 Date State 3108200 1708 F208 DRILLING DETAILS Date Time Holg Casing Water Remarks Herdbore DRILLING DETAILS Stream 7.30 7.30 7.30 5.30 SCS SCH Driller taken Flow Casing depth Gasing depth </th <th>ſ</th> <th></th> <th>Nor BO</th> <th>west REHOL</th> <th>Hols .e lo</th> <th>t So G - 0</th> <th>o<mark>il E</mark>r Cabl</th> <th>ngi E F</th> <th>ne PER</th> <th>erir CU:</th> <th>n<mark>g L</mark>i ssio</th> <th>td. N</th> <th></th> <th></th> <th>Hole ID. N6C Header</th>	ſ		Nor BO	west REHOL	Hols .e lo	t So G - 0	o <mark>il E</mark> r Cabl	ngi E F	ne PER	erir CU:	n <mark>g L</mark> i ssio	td. N			Hole ID. N6C Header
Project Greenwich Wilage Dilling Rig Pilcon 1500 Ground Level 4.38m AOD Orientation VWS Atkins Date Started 3100/2001 Date Completed Office Remarks Date Time Hole Gepth Gept	tract No	lo. F12	080		Meth	nod	Cable	Percu	ussion	Co	ordinates	S	5393	31.4 E	
Client English Partnerships Driller ID Correntation Vertical Logged by AH Date Started 3108/2001 Consultant WS Atkins	ect	Gre	enwich Vill	lage	Drilli	ng Rig	Pilcon	1500		Gi	round Lev	vel	1798 4.98r	08.0 N n AOD)
Consultant WS Atkins Description Date Completed Diodxio Date Time Holp Casing Water Remarks Trade Trade Cosing Cosing Remarks Trade Trade Cosing Cosing Remarks Trade Trade Cosing Remarks Holport Hardbore Hole Max depth of Casing (Max depth of Casing of dia) Date Time Strike at Rese to Hole Max depth of Casing of dia) Date Time Strike at Rese to Time file Hole Max depth of Casing of dia) Date Time Strike at Remarks Hole Max depth of Casing of dia) Date Time Strike at Remarks Hole Max depth of Casing of dia) Date Time Strike at Remarks Hole Max depth of Casing of dia) Date Time Strike at <td< td=""><td>nt</td><td>Eng</td><td>llish Partne</td><td>erships</td><td>Drille</td><td>er aed by</td><td>ID AH</td><td></td><td></td><td></td><td>rientation</td><td>d</td><td>Vertio</td><td></td><td></td></td<>	nt	Eng	llish Partne	erships	Drille	er aed by	ID AH				rientation	d	Vertio		
PROGRESS Date Time Holg depth Casing depth Water Remarks Remarks DRILLING DETAILS 31082001 1300 0.00 0.00 Dry Sold Ecos Hardbore Hardbore Chiselling Remarks 31082001 1300 0.00 5.00 Ecos Sold Ecos	sultant	t WS	Atkins		52	,,				Da	ate Comp	leted	01/09	9/2001 9/2001	
Gent Hole of Casing Market Index Joint Stress Water Strikes noted at 5.00m, 6.20m and 5.10m. Date Time Strike at 10m mm SPT DETAILS	ate	Time	Hole	PROGRESS	Water		Remarks		Hard	hore	DR	ILLING	DETA	ILS	lomarko
31082001 1400 0.00 Dry SOH 31082001 1730 7.50 7.50 6.20 SOS 01092001 1730 7.50 7.50 6.20 SOS 01092001 1730 7.50 15.00 15.00 SOS 01092001 1730 7.50 15.00 SOS SOS 01092001 1730 7.50 15.00 SOS SOS 01092001 1730 7.50 15.00 SOS SOS 01092001 1730 15.00 15.00 SOS SOS 01092001 1730 15.00 IS.00 SOS SOS 01092001 1730 15.00 IS.00 SOS SOS 01092001 1730 IS.00 SOS SOS SOS 01092001 Incremental Statistics Incremental blow count/penetration in mm Standpipe installed upon borehole completion. SOm and 9.10m. Incremental blow count/penetration in mm 4. Water Stirkes noted at 5.00m, 6.20m and 9.10m. Incremental blow count/penetration in mm Incremental blow count/penetration in mm	2/00.04		depth	depth	depth				from	depth 1	to depth	hou	rs		emarks
CASING WATER STRIKES Hard diameter casing of dia. Date Time Strike at Rise to Time taken Flow Casing depth Case at strike time to rise Image: strike time Image: strike time <t< td=""><td>8/2001 8/2001 9/2001 9/2001</td><td>1400 1730 0800 1730</td><td>0.00 7.50 7.50 15.00</td><td>0.00 7.50 7.50 15.00</td><td>Dry 5.00 6.20 6.20</td><td>SOH EOS SOS EOH</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>	8/2001 8/2001 9/2001 9/2001	1400 1730 0800 1730	0.00 7.50 7.50 15.00	0.00 7.50 7.50 15.00	Dry 5.00 6.20 6.20	SOH EOS SOS EOH									
WATER STRIKES Hole Max depth of hole at dia. Casing Max depth of diameter casing of dia. Date Time Strike at depth Res to Hole Time taken to rise Flow Casing depth at strike time Casing depth to rise dam diameter casing of dia. GENERAL NOTES .															
Hole Max depth of Casing Max depth of dameter casing of da. Date Time Strike at dameter Rise to Time taken to rise Flow Casing depth Casing depth Casing of dameter casing dameter casing dameter casing of dameter casing damet		CAS							10/0						
GENERAL NOTES SPT DETAILS 1. Hand dug service pit to 1.20m. Depth Type 2. Bentonite seal installed and redrilled 2.50-4.50m. Depth Type 3. Standpipe installed upon borehole completion. Peth Type 4. Water Strikes noted at 5.00m, 6.20m and 9.10m. Incremental blow count/penetration in mm	e Max	ax depth of ole at dia.	Casing I diameter	Max depth of casing of dia.	Date	Time	Strike at	Rise	to Tir	ne take	n Flo	w	Casing	depth	Casing de
1. Hand dug service pit to 1.20m. Depth Type Incremental blow count/penetration in mm 2. Bentonite seal installed and redrilled 2.50-4.50m. Standpipe installed upon borehole completion. Depth Type Incremental blow count/penetration in mm 4. Water Strikes noted at 5.00m, 6.20m and 9.10m. Incremental blow count/penetration in mm Incremental blow count/penetration in mm			GENER	AL NOTES	-						SPT D				
 2. Bentonite seal installed and redrilled 2.50-4.50m. 3. Standpipe installed upon borehole completion. 4. Water Strikes noted at 5.00m, 6.20m and 9.10m. 	Hand	dug servic	e pit to 1.2	20m.			Dept		Incre	ementa		unt/pen	etration	in mr	n
	Stand Water	dpipe install er Strikes no	led upon b oted at 5.00	orehole comp 0m, 6.20m and	letion. d 9.10m.										
NB All depths in metres, all diameters in millimetres, water strike rise time in minutes, chiselling time in hours.	All dept	oths in metri	es, all dian	neters in millir	netres,	ours.			* Sea	ating blo	ws only.		Form	AF	RIAL CP HEADE

N	Norwest Ho BOREHOLE	olst So LOG - (D <mark>il</mark> E	ngi _E F	nee PER	er C	ing Li USSIC	td. N	Hole II N6C Sheet 1 c	D. Cof 2
Contract No. Project	F12080 Greenwich Village	Method	Cabl	e Percu	ission	(Coordinates	53933 17980	1.4 E 8.0 N	
Client	English Partnerships	Driller Logged by	ID AH	11 1000		([Drientation Date Started	Vertic 31/08/	al /2001	
Consultant	WS Atkins					0	Date Complet	ed 01/09/	/2001	
	Description of Strata		Legend	Depth Below	O.D. Level		Sampling	SPT N &	SPT type & depth	Ins
TOPSOIL: B	ark chippings and sandy soil			0.30	4.68			-		1.0
MADE GROU fine-coarse s	UND: Brown slightly clayey very andy subangular - subrounded brick flipt and gravel	/		0.50 0.70	4.48 4.28	D	0.40			
MADE GROU	UND: Orange brown mottled sand o rounded concrete and flint grave	y /	/				1.20	-	á	-
MADE GROU sandy, slight	UD: Dark brown mottled orange billy gravelly clay with rare concrete	rown					1.20		6	
subangular-r	ounded flint and brick.					D	2.00			
MADE GRO	UND: Black very sandy ash, brick	gravel.		2.30	- 2.68 -			-		
						D	3.00			
				3.90	1.08	D	4.00	-		
Brown mottle	ed black CLAY. Slight organic odo	ur.						2		
					1.1.1.1		5.00			
							5.00			
					-					
					ليريين	D	6.00			
					-	-	7.00	-		
Brown, grey	mottled CLAY with spaced lamina	ations of		7.00	-2.02		7.00			
TIDFOUS PEA	1			0.00			8.00			TIII.
Dark brown a remains.	amorphous PEAT with occasional	plant	aster aster aster to aster aster aster aster aster a aster aster aster aster aster	0.00	-3.02		0.00			
Grey slightly fine-coarse f	sandy subrounded to rounded lint GRAVEL. Sand is coarse.	114	0.000	0.00			0.50			1.1.1.1
						D	9.50			
NB All depths	in metres, all diameters in millime	tres.	0,0,0		F			F		-100
See head	er sheet for details of drilling, prog	ress and water						Form	ARIAL C	PLO
strikes. Se	ee legend sheet for key to symbols	S.						Versio	n 1.00	

ł

ß	Norwest Ho BOREHOLE	blst Sc LOG - (DILE CABI	ngi _E F	nee PER	ering Lt	d. N	Hole II N6C Sheet 2 c).)) f 2
Contract No. Project Client Consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cabl Pilco ID AH	e Percu	ssion	Coordinates Ground Level Orientation Date Started Date Complete	53933 17980 4.98m Vertica 31/08/ ed 01/09/	1.4 E 8.0 N AOD al 2001 2001	
	Description of Strata		Legend	Depth Below	O.D.	Sampling	SPT N &	SPT type	Ins
Grey slightly fine-coarse fli Cable Percus	sandy subrounded to rounded int GRAVEL. Sand is coarse.			15.00					
NB All depths i See heade	in metres, all diameters in millimeter r sheet for details of drilling, progra	res. ess and water					Form	ARIAL C	PLO

l



Norwest Holst Soil Engineering Ltd.

Hole ID. N7A

Header

BOREHOLE LOG - CABLE PERCUSSION

		· · · · · · · · · · · · · · · · · · ·			
Contract No.	F12080	Method	Cable Percussion	Coordinates	539179.4 E
Project	Greenwich Village				179560.0 N
		Drilling Rig	Pilcon 1500	Ground Level	2.43m AOD
Client	English Partnerships	Driller	ID	Orientation	Vertical
		Logged by	AH	Date Started	02/09/2001
Consultant	WS Atkins			Date Completed	03/09/2001

			PROGRESS	3			DR	ILLING DET	AILS
Date	Time	Hole depth	Casing depth	Water depth	Remarks	Hardbore from depth	Hardbore to depth	Chiselling hours	Remarks
02/09/2001 02/09/2001 03/09/2001 03/09/2001	0800 1730 0800 1730	0.00 5.00 5.00 13.20	0.00 5.00 5.00 13.20	Dry 5.00 6.00 6.00	SOH EOS SOS EOH				

	CAS	SING					V	WATER STR	RIKES		
Hole diam.	Max depth of hole at dia.	Casing diameter	Max depth of casing of dia.	Date	Time	Strike at depth	Rise to depth	Time taken to rise	Flow	Casing depth at strike time	Casing depth to seal flow
300 150	4.00 13.20	300 150	4.00 13.00			uepar	deptit	10 1150		at strike time	to seal now
		-									

GENERAL NOTES			SPT DETAILS		
 Hand dug service pit to 1.20m. Bentonite seal installed and redrilled 2.00-4.00m. 	Depth	Туре	Incremental blow count/pene	etration in	mm
 Standpipe installed upon borehole completion. Water Strike noted at 5.00m. 					
			-		
			* Seating blows only.		
NB All depths in metres, all diameters in millimetres,				Form	ARIAL CP HEADER
water strike rise time in minutes, chiselling time in hours.				Version	1.00
				Revised	01/10/1998

	Norwest H BOREHOLE	olst Sc LOG - (DILE CABI	ngi LE F	nee PER	er C	ing Lt USSIO	d. N		Hole I N7/ Sheet 2	D. A of 2
Contract No. Project Client Consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cabl Pilco ID AH	e Percu	ission		Coordinates Ground Level Drientation Date Started Date Complete	ed (539179 539179 2.43m /ertica 02/09/2 03/09/2	9.4 E 0.0 N AOD al 2001 2001	
	Description of Strata		Legend	Depth Below	O.D.		Sampling	SPT	N &	SPT type	Ins
Dark brown w with rare sub	very clayey fine-coarse organic S rounded flint gravel.	AND ed flint h.		13.00		D	13.00				
NB All depths i See heade	in metres, all diameters in millime er sheet for details of drilling, prog	etres. gress and water	1		1	I	I	r F	Form	ARIAL C	PLOC

R		Nor BC	west DREHC	Holst	So SG -	- CAE	ngi BLE	ineeri E & RC	ng L	td.		Hole ID. N7B Header
Contra	t Gr	2080	illage	Metho	d	Cable	& Ro	tary (Coordinates	Ę	539177.7 E	
Client	Er	iglish Partr	nerships	Drilling Drillers Logged	ı Rig S d by	Dando IP & H AH	o 2000 IJ	& Edeco T60 ([Ground Leve Drientation Date Started		179557.4 N 2.51m AOE Vertical 19/09/2001)
Consu		5 Atkins		Core b Core b	arrel it	Rock	Roller		Date Comple	eted (01/10/2001	
Data	Time	Liele	PROGRESS	3					CABLE PE	RCUSS	ION DETA	ILS
Date		depth	depth	depth		Remarks		from depth	Hard strata to depth	Hours	s F	Remarks
19/09/2 19/09/2 20/09/2 20/09/2 21/09/2 21/09/2 24/09/2 24/09/2 28/09/2 28/09/2	001 0800 001 1730 001 0800 001 1730 001 0800 001 1730 001 0800 001 1730 001 0800 001 1730 001 0800 001 1730 001 0800 001 1730	0.00 5.20 5.20 17.50 26.30 26.30 26.50 26.50 28.00	0.00 5.20 13.50 24.00 26.30 26.50 26.50	Dry Dry 13.50 4.30 6.10 6.10 6.10 0.00	SOH EOS SOS EOS EOS SOS EOH Ca SOS RO EOS	able otary		2.30 23.70 23.80 26.30 26.50	4.00 23.80 24.30 26.50 26.50	2.00 1.00 2.50 4.00 1.00		
29/09/2	001 0800	28.00	26.50	0.00	SOS							
Hole diam.	Max depth of hole at dia	Casing	Max depth of	Date	Time	Strike at	Rise	to Time take	TRIKES en Flow	Ca	sing depth	Casing depth
300 200	4.50 13.50	300 200	4.50 13.50	19/09/2001 20/09/2001	0000	1.50 6.00	1.50	0 20 0 20	NO RIS	E	1.50 6.00	3.00 13.50
											<u>_</u>	
From		DTARY DF	ILLING DETA	ILS Flush	Core	e Dept	n Type	e Incrementa	SPT DET al blow cour	TAILS	ration in mr	n
26.50) 45.00	Wa	ter		0*							
		GENER	RAL NOTES									
1. F 2. E 3. F 4. S	land dug servi Bentonite seal nishing cable Rotary open ho Standpipe insta	ce pit to 1. installed al percussior ole follow o alled upon	20m. nd redrilled 2.5 n at 26.50m. n from 26.50 - borehole comp	0-4.50m 45.00m. oletion.				* Seating b	lows only.			
NB AI	I depths in me	tres, all dia	ameters in milli	metres,		1		, Jeaniy D	iono only.	Form	ARIAL COMB	INED HEADER
W	ater strike rise	time in mi	nutes, hard str	ata time in l	hours					Version	1.00	
										Revised	01/10/1998	1

N	Norwest H BOREHOL	olst S E LOG	ioil 6 - C	Eng ABL	gine .E &	ering l ROTAF	_td. XY	Hole ID. N7B Sheet 1 of 5
Contract No. Project Client	F12080 Greenwich Peninsula	Method Drilling Rig Drillers	Cab Dar	ndo 200	otary 0 & Edec	Coordina o T6 Ground I Orientati	ates 5391 1795 _evel 2.51n	77.7 E 57.4 N n AOD
Consultant	WS Atkins	Logged by Core Barrel Core Bit	AH	k Rolle	r	Date Sta Date Co	rted 19/09 mpleted 01/10	/2001 /2001
	Description of Strata		Legend	Depth Below	O.D.	Sampling & coring	Blow count to	SPT N
MADE GRO gravelly fine subangular-	OUND: Dark brown clayey, slightly -coarse sand. Gravel subrounded fine-coarse flint.			0.50	_ 2.01			
MADE GRO clay. Sand is subrounded	OUND: Brown slightly gravelly sand s fine-coarse. Gravel subangular to flint and brick.	ly D			-			1
1.30m Conc Soft grey CL	rete boulder. (driller's description) _AY (driller's description)			1.40	1.11			
					-			
				3.00	0.49			
Soft brown o	organic CLAY. (driller's description)	- shes shes - shes shes - shes shes - shes shes - shes shes	3.50	-0.99			
Dark brown plant remair	amorphous PEAT with occastiona ns.	L	مالاد مالاد مالاد مالاد مالاد م الاد مالاد مالاد م مالاد مالاد م مالاد مالاد مالاد مالاد مالاد ما					
Grey sandy GRAVEL. S	subangular-rounded fine-coarse fil and is coarse.	int		4.20	-1.69			1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
					-			
8.00m Grey subrounded	and brown sandy subangular to I fine-coarse GRAVEL. Sand is coa	arse.						
9.00m Occa	asional flint COBBLE							
NB All depths See head	s in metres, all diameters in millime der sheet for details of drilling, prog	etres. Iress and wat	er			<u></u>	Form A	
See nead	ier sneet for details of unining, prog	i ess anu wat	61				Varian 1	00

I

	Norwest F BOREHOI	Holst S	Soil G - C	Eng ABL	gine _E &	ering l ROTAF	_td. ≀Y	Hole N7 Sheet 2	ID. B of 5
Contract No. Project Client Consultant	F12080 Greenwich Peninsula English Partnerships WS Atkins	Method Drilling Rig Drillers Logged by Core Barre Core Bit	Ca Da IP AH	ble & Ro ndo 200 & HJ ck Rolle	otary 00 & Edec er	Coordina co T6 Ground Orientati Date Sta Date Co	ates 5 1 Level 2 on V Irted 1 mpleted 0	539177.7 E 79557.4 N 2.51m AOD /ertical 9/09/2001 01/10/2001	
	Description of Strata	i oisas	Legend	Depth Below	O.D. Level	Sampling & coring	Blow count	t to SPT N	Ins
Grey sandy GRAVEL. S Very stiff ex Fissures rar smooth occ	subangular-rounded fine-coarse and is coarse. tremely fractured grey brown CL/ ndomly orientated undulose to pla asionally slightly polished.	flint AY. mar,		13.20	-10.69				וואסוואסוואסוואסוואסוואסוואסוואסוואסווא
Stiff grey br occasional s	own slightly sandy CLAY with shell fragments. Sand is fine.	ents.		17.50					
NB All depths See head strikes. S	s in metres, all diameters in millim der sheet for details of drilling, pro see legend sheet for key to symbo	netres. ogress and wa	ter				Form Versio	ARIAL COMBINE	

I

	BOREHO	Holst S LE LOG	oll - C	Enę ABL	gine .E &	ering L ROTAR	td. Y		Hole II N7E Sheet 3 c	D. 3 of 5
Contract No. Project Client Consultant	F12080 Greenwich Peninsula English Partnerships WS Atkins	Method Drilling Rig Drillers Logged by Core Barrel	Cat Dar IP & AH	ndo 200 MJ HJ	otary 0 & Edeco	Coordina T6 Ground L Orientatio Date Sta Date Cor	tes evel on rted npleted	53917 17955 2.51m Vertic 19/09/ 01/10/	77.7 E 77.4 N AOD al /2001 /2001	
		Core Bit	Roc	ck Rolle	r					
	Description of Strata		Legend	Below	O.D. Level	Sampling & coring	Blow cou	nt to	SPT N & depth	lr.
Angular coar with abundar	rse gravel size LIMESTONE frag nt shells.	iments		23.70	-21.19					1 1 1 1 1 1
Very stiff ext CLAY with o regularly orie occasionally	remely closely fissured grey broc ccasional shell fragments. Fissu entated, undulose to irregular, sn slightly polished.	wn res are nooth		24.30						
Hard LIMES	TONE (Driller's description)			26.30	-23.79					
Grey brown	and green fine SAND			28.00						
25.00m Abu	ndant shells noted.				-					1 1 1 1 1
the second second second			and the second							12
IB All depths	in metres, all diameters in millim	ietres.					Form	n AF	NAL COMBINE	DLO

I

Ĩ

I

Contract No. P12080 Method Cable & Rotary Coordinates 538177.7 E Project Greenwich Peninsula Drilling Rig Dando 2000 & Edeco TB Ground Level 25.1m ADD Client English Partnerships Drilling Rig Dando 2000 & Edeco TB Ground Level 1906/2001 Core Barrel Core Barrel Core Barrel Core Barrel Date Started 1906/2001 Core Barrel Rock Roller Date Completed 01/10/2001 Service Service 38.0mg Grey brown and green fine SAND Logged Bow Level Service Service <td< th=""><th></th><th>Norwest F BOREHOI</th><th>loist S _E LOG</th><th>ioil 6 - C</th><th>Eng ABL</th><th>gine E &</th><th>ering L ROTAR</th><th>.td. Y</th><th>N7E Sheet 4 d</th><th>3 of 5</th></td<>		Norwest F BOREHOI	loist S _E LOG	ioil 6 - C	Eng ABL	gine E &	ering L ROTAR	.td. Y	N7E Sheet 4 d	3 of 5
Defining Rig English Partnerships Defining Rig Cores Bit Usged by AH Description of Strata Core Bit Core Bit Co	Contract No. F Project G	12080 Greenwich Peninsula	Method	Cat	ole & Ro	otary	Coordina	tes 53 17	9177.7 E 9557.4 N	
Logged by Consultant AH Date Started 1999/201 Description of Strata Logged by Core Bit Rock Roller Date Completed 91/10/201 Grey brown and green fine SAND Logged by Core number of Strata Logged by Logged by Core Bit Core Strate Logged by Logged by Core Strate Sampling Blow count to Strate Strate Image: Strate	Client E	nglish Partnerships	Drilling Rig Drillers	Dar IP 8	ndo 200 & HJ	0 & Edeco	o T6 Ground L Orientatio	evel 2. on Ve	51m AOD ertical	
Description of Strata Loopend Description Description of Strata Loopend Below D.D. Sampling Blow count to SPT N n Grey brown and green fine SAND MB All depths in metres, all diameters in millimetres. See header sheet for defails of drilling, progress and water	Consultant V	VS Atkins	Logged by Core Barrel	AH		r	Date Star Date Cor	ted 19 npleted 01	9/09/2001 1/10/2001	
Open of Line Level & coring drive U100 & depth r Grey brown and green fine SAND Image: Same of Line Same of Li		Description of Strata		Logond	Depth	O.D.	Sampling	Blow count	to SPT N	Ins
Green silty fine SAND. (THANET SAND) 38.00 56.49 Image: solution of the solution o	Grey brown and	d green fine SAND		Legend	G.L.	Level	& coring	drive U100	& depth	a
NB All depths in metres, all diameters in millimetres. Form ARIAL COMBINED LO See header sheet for details of drilling, progress and water Version 1.00	Green silty fine	SAND. (THANET SAND)			38.00	-35.49				
See header sheet for details of drilling, progress and water strikes. See legend sheet for key to symbols.	NB All depths in	metres, all diameters in millin	netres.	هامن استامین می			•	Form	ARIAL COMBINE	DLO
	See header	sheet for details of drilling, pro	ogress and wat	ter				Version	1.00	

I

Dantract No. F12080 Greenwich Peninaula Method Drillers Cable & Rotary Danto 2000 & Edeco T6 Greund Level Coordinates 179654 A N Orientation 539177.7 E 179657 A N Orientation 2llant English Partneships Drillers Drillers IP & HJ Loggod by Core Barrel Core Bit Dando 2000 & Edeco T6 Ground Level Ground Level Drientation Vertical Using Rig Data Started Using Rig Core Barrel Core Bit Data Started Data Complete Using Rig Data Complete 10092001 Description of Strata Legend Bore Dow Level Sampling & coring Blow count to drive U100 SPT N & core Bit & core Bit A Green sity fine SAND. (THANET SAND) In 45.00 -42.49 In 45.00	Description of Strata Logend Pactor Data Coordinates (1984) SSI17.7 E (1984) Drilling Rig Usaged by Ansultant Data Data Edge Tilling Rig Usaged by AH Data Started (1982) SSI17.7 E (1984) Donsultant English Partnerships Drilling Rig Usaged by Core Barrel Data Started (1982) StanADD (1982) Description of Strata Logend Barrel Dethy Core Barrel O.D. Sampling Blow count to drive U100 SPT N If Creen silty fine SAND. (THANET SAND) Image A Image A Image		Norwest H BOREHO	Holst S	ioil 6 - C	Eng ABL	gine _E &	ering L ROTAR	.td. Y	Hole ID. N7B Sheet 5 of 5
Description of Strata Legend Depty Book O.D. Sampling Biow count to SPT N In Green silty fine SAND. (THANET SAND) Image: Comparison of Strata Image	Description of Strata Logend Bow O.D. Sampling Biow count to SPT N In Creen silty line SAND. (THANET SAND) Image: Council of Strata Image: Council of Strata	Contract No. Project Client Consultant	F12080 Greenwich Peninsula English Partnerships WS Atkins	Method Drilling Rig Drillers Logged by Core Barrel Core Bit	Cal Dar IP & AH Roo	ole & Ro ndo 200 & HJ ck Rolle	otary 10 & Edec r	Coordinat co T6 Ground L Orientatio Date Star Date Com	tes 5 1 evel 2 on V ted 1 ppleted 0	39177.7 E 79557.4 N .51m AOD /ertical 9/09/2001 1/10/2001
Oreen silly fine SAND. (THANET SAND) OL OL <thol< th=""> OL OL</thol<>	Oreen silly fine SAND. (THANET SAND) 0.0 0.0 0.0 0.0 0.0 0.0 Borehole complete at 45.00 m. 45.00 42.49 1.0 1.0 0.0 B Al depths in metres, all diameters in millimeters. See header sheet for details of drillino. procress and water Fm ARU.COMMEND.COM		Description of Strata		Legend	Depth Below	O.D.	Sampling	Blow count	to SPT N
	IB All depths in metres, all diameters in millimetres. See header sheet for details of drilling, progress and water	Green silty fi	mplete at 45.00 m.			45.00				



Norwest Holst Soil Engineering Ltd. BOREHOLE LOG - CABLE PERCUSSION

Hole ID. N8A

Header

Contract No.	F12080	Method	Cable Percussion	Coordinates	539570.1 E
Project	Greenwich Village				179480.0 N
		Drilling Rig	Dando 2000	Ground Level	4.56m AOD
Client	English Partnerships	Driller	IP	Orientation	Vertical
		Logged by	JFK	Date Started	23/08/2001
Consultant	WS Atkins			Date Completed	23/08/2001

PROGRESS							DRILLING DETAILS					
Date	Time	Hole depth	Casing depth	Water depth	Remarks	Hardbore from depth	Hardbore to depth	Chiselling hours	Remarks			
23/08/2001 23/08/2001	0800 1730	0.00 14.00	0.00 14.00	Dry Dry	SOH EOH	2.30	2.60	0.50				

	CAS	SING					\	WATER STR	RIKES		
Ho dia	e Max depth of hole at dia.	Casing diameter	Max depth of casing of dia.	Date	Time	Strike at depth	Rise to depth	Time taken to rise	Flow	Casing depth at strike time	Casing depth to seal flow
30 15	0 5.00 0 14.00	300 150	5.00 14.00	23/08/2001	0000	8.30	7.20	20.0		8.30	13.50
					-						

GENERAL NOTES			SPT DETAILS		
 Hand dug service pit to 1.20m. Bentonite seal installed and redrilled 4.0-6.0m. 	Depth	Туре	Incremental blow count/pene	stration in	mm
3. Standpipe installed upon borehole completion.					
	1		Seating blows only.		
NB All depths in metres, all diameters in millimetres, water strike rise time in minutes, chiselling time in hours				Form	ARIAL CP HEADER
				Version	1.00
				Revised	01/10/1998

N	Norwest H BOREHOLE	lolst Sc E LOG - (DILE CABI	ngi _E F	nee PER	er C	ing Lt USSIO	d. N	Hole N8 Sheet	ID. A 1 of 2
Contract No. Project	F12080 Greenwich Village	Method Drilling Rig	Cabl	e Percu lo 2000	ssion	(Coordinates Ground Level	53957 17948 4.56m	70.1 E 80.0 N n AOD	
Client Consultant	English Partnerships WS Atkins	Driller Logged by	IP JFK			C [[Drientation Date Started Date Complete	Vertic 23/08 ed 23/08	al /2001 /2001	
	Description of Strata		Legend	Depth Below	O.D. Level		Sampling	SPT N & (U blows)	SPT type & depth	e Ins
MADE GRO fine-coarse f	UND: Brown sand and subround lint and angular ash.	led		0.50	4.06	D	0.50	-		
MADE GRO medium grav	UND: Dark grey-black fine sand vel of ash, wood and occasional	, round roots.		1.00	3.56	D	1.00			
MADE GRO gravelly clay and ash.	UND: Dark grey mottled white s v. Sand is fine. Gravel is fine-coa	andy rse brick			-					
Black fibrous	s PEAT.		alter alter alter alter alter alter alter alter alter alter alter alter alter alter alter alter alter alter	2.00	2.56	D	2.00			rterri
			ashka ashka ashka ashka ashka ashka ashka a	2.80	1.76	D	2.80			-
Grey brown Soft grey bro viens of mot	fine sandy mottled black CLAY own CLAY with occasional rootle tled black contamination.	ets and		3.00	1.56	D	3.00			
Stiff grey bro	own CLAY			4.00		D	4.00			
Soft black fit	brous PEAT containing wood fra	gments.	عالد عالد عالد عالد عالد عالد عالد عالد عالد عالد عالد عالد عالد عالد عالد عالد عالد د عالد عالد د عالد عالد عالد د عالد عالد عالد	5.00	0.44 - - - - - - -	D	5.00			ليبيبليب
			م مالو مالو م مالو مالو مالو مالو ب مالو مالو مالو مالو مالو مالو مالو مالو مالو مالو مالو مالو			D	6.00	- 		. l
Grey brown fine-coarse	and white sandy subrounded-ro flint GRAVEL. Sand is coarse.	unded,	ملاد مالد مالد مالد مالد مالد مالد مالد	7.00	-2.44	D	7.00			
						D	8.00			
						D	9.00	-		
NB All depths	in metres, all diameters in millir	netres.	0.00		-	D	10.00			
See head	ler sheet for details of drilling, pr	ogress and water	e e					Form	ARIAI	CP LOO
strikes. Se	ee legend sheet for key to symb	ols.						Versie	1.00	

	Norwest H BOREHOLE	lolst Sc ELOG - (DIIE CABI	ngi _E F	nee PER	er C	ing Lt	d. N	Hole I N8 Sheet 2	D. A of 2
Contract No. Project Client Consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cabl Dano IP JFK	e Percu do 2000	ssion		Coordinates Ground Level Drientation Date Started Date Complete	53957 17948 4.56m Vertic: 23/08/ d 23/08/	0.1 E 0.0 N AOD al /2001 /2001	
	Description of Strata	TRACK	Legend	Depth Below	O.D. Level		Sampling	SPT N &	SPT type & depth	Inst
Grey brown a fine-coarse fl	and white sandy subrounded-rou int GRAVEL. Sand is coarse.	inded,				D	10.00 11.00 12.00			
Stiff grey CL	AY			13.00		D	13.00 13.20			
NB All depths	in metres, all diameters in millin er sheet for details of drilling, pro	netres. ogress and water			<u> </u>			Form	ARIAL	
See neau										

Ń		Nor BOR	west H	<mark>-lols</mark> t E LO	t So G - C	I En	gir E P	neeri ERCL	ng Lt JSSIO	d. N		Hole ID. N9B Header
Contract Project	t No. F12 Gre	080 enwich Villa	age	Meth	od	Cable I	Percu	ssion	Coordinates		539727.4 E 179630.9 N 7.27m AOE	
Client Consult	Eng ant WS	llish Partner Atkins	rships	Drille Logg	r ed by	GW AH	1500		Orientation Date Started Date Compl	d (eted (Vertical 04/09/2001 05/09/2001	
			PROGRESS						DRI	LLING D	ETAILS	
04/09/200 04/09/200 05/09/200 05/09/200	11/11/20 11 1400 11 1730 11 0800 11 1730	0.00 7.00 7.00 15.00	0.00 6.00 6.00 15.00	Dry Dry Dry Dry 9.90	SOH EOS SOS EOH			1.20 from depth 3.30 4.20	3.60 4.70	0.50 0.75		
Hole	CAS Max depth of	SING Casing	Aax depth of	Date	Time	Strike at	Rise	to Time ta	STRIKES	w Ca	asing depth	Casing dep
250 150	6.00 15.00	250 150	6.00 15.00									
		GENER	AL NOTES						SPT D	ETAILS		
2. Be 3. St 4. W	antonite seal i andpipe insta ater Strikes n	nstalled and lled upon b oted at 5.90	d redrilled 4.70 orehole comp)m and 9.90m	0-6.70m. letion.			i yρe		Intel DIOW CO	а то репе		
								* Seating	blows only.	264 Figure	0-2-07-7	
NB All	depths in me	tres, all diar	meters in milli	metres,	ooure			* Seating	blows only.		Form	ARIAL CP HEADE

	Norwest H	olst Sc LOG - (DII E CABI	ngi _E F	nee PER	ering L CUSSIC	ta. DN	N9E Sheet 1 c	5. 3 of 2
Contract No. Project Client Consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cable Pilco GW AH	e Percu n 1500	ssion	Coordinates Ground Leve Orientation Date Started Date Comple	53972 17963 I 7.27m Vertic 04/09, ted 05/09,	7.4 E i0.9 N i AOD al /2001 /2001	
	Description of Strata	- 446	Legend	Depth Below	O.D.	Sampling	SPT N &	SPT type	lr
MADE GRO Gravel is su Sand is fine-	UND: Brown sandy slightly grave bangular-subrounded flint and brid -coarse.	lly clay. ck.		4.80	2.47				
MADE GRO sand. Grave brick and lir	DUND: Brown very gravelly fine-m el is fine-coarse, subangular-round nestone.	edium ded flint,		5.70	-	D 5.70			1 1 1 1 1 1 1 1
MADE GRC slightly grav subrounded	DUND: Brown-grey mottled dark g velly, slightly sandy CLAY. Gravel d-round brick and flint. Slight odou	rey r noted.		7.50		J 5.70 D 6.60 J 6.60			
Black amor	phous PEAT.		مالاه مالاه ما الم مالاه مالاه مالاه مالاه مالاه مالاه مالاه ما	4 9 10			-		
Grey slightl remains. S	ly sandy CLAY with occasional pla and is fine.	ant		9.90	-2.63				
NB All depth	s in metres, all diameters in millim	netres.		9		1	Form	ARIAL	CPL
Soo boo	der sheet for details of drilling hro	mess and water						the second se	

ł.

I

Norwest Holst Soil Engineering Ltd. BOREHOLE LOG - CABLE PERCUSSION

Hole ID. N9B

Sheet 2 of 2

Contract No.	F12080	Method	Cable Percussion	Coordinates	539727.4 E	
Project	Greenwich Village				179630.9 N	
		Drilling Rig	Pilcon 1500	Ground Level	7.27m AOD	
Client	English Partnerships	Driller	GW	Orientation	Vertical	
		Logged by	AH	Date Started	04/09/2001	
Consultant	WS Atkins			Date Completed	05/09/2001	

Description of Strata	Legend	Depth Below	O.D. Level	Sampling	SPT N & (U blows)	SPT type & depth	Install- ation					
Grey slightly sandy subangular-rounded, fine-coarse flint GRAVEL. Sand is coarse.		G.L. 14.80 15.00										
See header sheet for details of drilling, progress and water					For	n ARIAL C	PLOG					
strikes. See legend sheet for key to symbols.					Reve	ised 01/10/10	98					
					Rev	isea 01/10/19	90					
N		Nor BO	west l	Hols E LO	t So G - 0	I En	giı E P	neeri ERCU	ng Lt Issio	:d. N		Hole ID. N9C Header
---	--	---	---	--	--	-------------------------------	----------------------	--------------------------------	--	--	---	---------------------------
Contrac Project Client Consult	Eng	080 enwich Vil lish Partne Atkins	lage erships	Meth Drilli Drille Logg	nod ng Rig er ged by	Cable Pilcon GW & AH	& Rot & Har TF	ary (nsEngland (([Coordinates Ground Lev Drientation Date Started Date Compl	s 539 179 el 6.5 Ve d 05/ leted 03/	9722.4 E 9636.9 N 54m AOD rtical /09/2001 /10/2001	
Date	Time	Hole	PROGRESS	Water		Romarks		Hardbore	DR	LLING DE	TAILS	Remarks
05/09/200 05/09/200 06/09/200 06/09/200 06/09/200 24/09/200 24/09/200	01 0800 01 1630 01 0800 01 1630 01 0800 01 0800 01 1730	depth 0.00 2.00 15.50 15.50 15.50	depth 0.00 2.00 15.50 15.50 15.50 15.50	depth Dry Dry 10.00 10.00 10.00 7.40	SOH EOS SOS EOH Ca SOH Ro EOS	ble		from depth 1.80 2.00	<u>to depth</u> 2.00 2.30	1.25 0.75		
25/09/200 25/09/200 26/09/200 26/09/200 27/09/200	01 0800 01 1730 01 0800 01 1730 01 0800 CAS	15.50 19.30 19.30 23.00 23.00	19.50 19.50 19.50 23.00 23.00	7.40 0.00 2.50 0.00 5.25	EOS SOS EOS SOS	*		WATER	STRIKES			
Hole Jiam.	Max depth of hole at dia.	Casing diameter	Max depth of casing of dia.	Date	Time	Strike at depth	Rise dep	to Time ta	ken Flo	w Casi at st	ng depth rike time	Casing de to seal fle
200 150	26.00 45.00	200 150	26.00 28.00									
1 1	land dug serviv	GENER	RAL NOTES				Tun	Incromor	SPT D	ETAILS	tion in m	m
2. fi fi 3. R 4. S 5. V	entonite seal i nishing cable p Rotary open ho Standpipe insta Vater strike at	nstalled ar bercussior le follow o lled upon 10.00m	nd redrilled 4.0 n at 15.50m. n from 15.50 - borehole comp	0-6.00m 45.00m. bletion.								
NR A	Il denths in me	tres all di	ameters in mill	imetres				* Seating	blows only		Form	
W DIN	ater strike rise	time in m	inutes, chiselli	ng time in	hours.						Version	1.00
										<u> </u>	Davida and	

I

Contract No. Project Client Consultant MADE GROU (Driller's desc MADE GROU brick obstruct MADE GROU obstructions (F12080 Greenwich Peninsula English Partnerships WS Atkins Description of Strata JND: Soil brick, wood and glass cription)	Method Drilling Rig Drillers Logged by Core Barrel Core Bit	Cab Pilc GW AH Roo	ole & Ro on & Ha / & TF ck Rolle	tary ansEnglan	Coordina d Ground L Orientatio Date Sta Date Cor	tes 53 17 .evel 6. on Ve rted 05	39722.4 E 79636.9 N 54m AOD ertical 5/09/2001	
MADE GROU (Driller's desc MADE GROU brick obstruct MADE GROU obstructions (Description of Strata JND: Soil brick, wood and glass cription)	fill.	Legend	Denth			npieteu 03	3/10/2001	
MADE GROU (Driller's desc MADE GROU brick obstruct MADE GROU obstructions (JND: Soil brick, wood and glass pription)	fill.		Below	O.D.	Sampling	Blow count	to SPT N	Inst
MADE GROU brick obstruct	- C			<u> </u>	5.24	& coning			
obstructions (JND: Clay fill with large concrete tions (Driller's desccription)	e and		2.00	4.54				
MADE GROU	UND: Brown grey mottled dark g	ILGÅ		5.90	- 0.64				
slightly grave subrounded f	elly, slightly sandy clay. Gravel to rounded of brick and flint.		مربع مادر مادر	7.50	-0.96				
Black amorp	hous PEAT		ته علام عائلام مالام علام علام تم علام مالام مالام عالام عالام مالام عالام عالام مالام عالام عالام	8.20	-1.66				
Grey slightly remains. Sar	sandy CLAY with occasional pland is fine.	ant		9.00					
NB All depths	in metres, all diameters in millim	netres.	0.0.0	<u>ק</u> 0.00	F 0.00		Form	ARIAL COMBIN	
See heade strikes. Se	er sheet for details of drilling, pro ee legend sheet for key to symbo	ogress and wa ols.	ter	5 4 (Versio	n 1.00	

Contract No.	F12080	LOG Method	Cab		E&F	Coordina	tes	53972	Sheet 2 o	of 5
Project Client Consultant	Greenwich Peninsula English Partnerships WS Atkins	Drilling Rig Drillers Logged by Core Barrel Core Bit	Pilc GW AH	on & Ha / & TF ck Rollei	ansEngland	d Ground I Orientati Date Sta Date Cor	-evel on rted mpleted	17963 6.54n Vertic 05/09 03/10	36.9 N n AOD al /2001 /2001	
*	Description of Strata		Legend	Depth Below	O.D.	Sampling	Blow cou	unt to	SPT N & depth	Ins
Grey and br subangular flint GRAVE coarse.	own slightly sandy to rounded fine to coarse L. Sand is predominantly			15.00	-8.46					
Stiff grey fis Green grey	very silty fine SAND.)		16.00	9.46					
	а 2			19.50						
Grey brown	n fine to coarse SAND (Scour feat	ure?)			E					

Contract No. Project	F12080 Greenwich Peninsula	Method	Cat	ole & Ro	otary	Coordina	ates (539722.4 E 179636.9 N	
Client	English Partnerships	Drillers Logged by	GW AH	0 & TF	ans⊏ngian	Orientati Date Sta	on N Inted (Vertical 05/09/2001	
Consultant	WS Atkins	Core Barrel Core Bit	Roo	ck Rolle	r	Date Co	mpleted (03/10/2001	
	Description of Strata		Legend	Depth Below G.L.	O.D. Level	Sampling & coring	Blow cour drive U1	t to SPT N 00 & depth	Inst
Grey brown	fine to coarse SAND (Scour featu	ire?)		24.20			07		
Grey CLAY	with abundant shell fragments.			25.80					
Grey very s fragments.	ilty fine SAND with abundant she	11		28.10					
Green very	silty fine SAND. (Glauconitic San	ld)							

I

	Norwest H BOREHOL	olst S E LOG	oil I G - C	Enç ABL	gine .E &	ei R	ring L OTAR	td.		Hole I N90 Sheet 4	D. C of 5
Contract No. Project Client Consultant	F12080 Greenwich Peninsula English Partnerships WS Atkins	Method Drilling Rig Drillers Logged by Core Barrel Core Bit	Cab Pilc GW AH Roc	on & Ha ' & TF k Rolle	otary ansEngla r	and	Coordina Ground L Orientati Date Sta Date Cor	tes evel on rted mpleted	5397 1796 6.54r Vertic 05/09 03/10	22.4 E 36.9 N n AOD cal 9/2001 0/2001	
	Description of Strata		Legend	Depth Below	O.D. Level	2	Sampling & coring	Blow of drive	ount to U100	SPT N & depth	Ins ati
Grey brown	silty fine SAND. (Glauconitic Sand)			36.20							00000
NB All depths	s in metres, all diameters in millim	etres. gress and wa	iter		87 - 1946	94			Form	ARIAL COMBIN	ED LOC
Joo nodu	as ones for details of arming, proj							1			

Dortract No. F12080 Method Cable & Rotary Coordinates 539722.4 E Ty7836.00 English Partnerships Drilling Rig Drillers Pilcon & HansEngland Core Barrel Ground Level 634m AOD 20nsultant WS Atkins Core Barrel Core Barrel Date Started Odf/02001 20nsultant WS Atkins Core Barrel Core Barrel Date Started Odf/02001 20nsultant WS Atkins Core Barrel Core Barrel Sampling Blow count to SPT N Core barrel Description of Strata Legend Core Barrel Core Barrel Sampling Blow count to SPT N In Grey brown silly fine SAND (Thanet Sand) Image Sampling Blow count to SPT N In Sampling Blow count to SPT N Sampling Blow count to SPT N In Borehole complete at 45.00 m. Image Sampling Blow count to SPT N In Sampling Blow count to SPT N Sampling Sampling Blow count to SPT N Sampling Sampling Sampling Sampling Sampling Sampling S	K	Norwest H BOREHOI	lolst S	Soil G - C	Enç ABL	ginee _E & F	ring L ROTAR	.td. Y	H N She	ole ID. 19C et 5 of 5
Description of Strata Legend Depth O.O. Sampling Blow court to SPT N In Grey brown sity fine SAND (Thanet Sand) 4 <t< th=""><th>Contract No. Project Client Consultant</th><th>F12080 Greenwich Peninsula English Partnerships WS Atkins</th><th>Method Drilling Rig Drillers Logged by Core Barrel Core Bit</th><th>Cal Pilo GV AH Ro</th><th>ole & Ro con & H / & TF ck Rolle</th><th>otary ansEngland er</th><th>Coordinat Ground L Orientatic Date Star Date Con</th><th>tes evel on ted npleted</th><th>539722.4 E 179636.9 N 6.54m AOD Vertical 05/09/2001 03/10/2001</th><th></th></t<>	Contract No. Project Client Consultant	F12080 Greenwich Peninsula English Partnerships WS Atkins	Method Drilling Rig Drillers Logged by Core Barrel Core Bit	Cal Pilo GV AH Ro	ole & Ro con & H / & TF ck Rolle	otary ansEngland er	Coordinat Ground L Orientatic Date Star Date Con	tes evel on ted npleted	539722.4 E 179636.9 N 6.54m AOD Vertical 05/09/2001 03/10/2001	
Strey brown silly fine SAND (Thanet Sand) 45.00 -38.46 Borehole complete at 45.00 m. 45.00 -38.46 Borehole complete at 45.00 m. 45.00 -38.46		Description of Strata		Legend	Depth Below G.L.	O.D. Level	Sampling & coring	Blow cour drive U1	nt to SP1 00 & de	N Insta pth atio
NB All depths in metres, all diameters in millimetres. Form ARIAL COMBINED LOG See header sheet for details of drilling, progress and water Version 1.00	Grey brown	mplete at 45.00 m.			45.00					
	NB All depths See head strikes. Se	in metres, all diameters in millim ler sheet for details of drilling, pro	etres. gress and wate	er		<u>ــــــــــــــــــــــــــــــــــــ</u>	J	Form	ARIAL COM	IBINED LOG

m	ere	Merebrook Consulting Limited Tel: 01773 829988 Fax: 01773 829993 email: consulting@merebrook.co					Equipment and Methods	Borehole No
110		consulting		Fax: 01773 8293	93 Marakasikasi			BHN10C
A	N Idom G	ROUP COMPANY		email: consulting	@merebrook.co.	.uk		Sheet 1 of 2
Proj	ect N	ame			Project N	No.	Co-ords	Hole Type
Plot	MO1	01				62	539917E - 179424N	
LOC	ation:	Greenv	VICN P	reninsula Rivers	ide South			5cale
							0.091	
Clie	nt:	Bellway	/ Hom	nes (Thames Ga	ateway) Ltd		Dates: 09/07/2013	OTF
Well	Water Strike	Sample Depth (m)	es & In Type	Situ Testing Results	Depth in metres (thickness)	Legend	Stratum Des	scription
					(0.90)		MADE GROUND. Medium dense dark bi medium SAND. Gravel is fine to medium and rare ash.	rown clayey gravelly fine to of flint, brick, clinker
					0.90 (0.30) 1.20		MADE GROUND. Orange meshed geote slightly gravelly sandy CLAY. Gravel is fi flint, brick and rare clinker	extile membrane over a dark brown -1 ne to medium of
							MADE GROUND: Stiff to firm black sand fine to medium angular to sub rounded c and ash. Sand is medium to coarse	ly gravelly CLAY. Gravel is of flint, brick, clinker
					(2.60)			-3
					3.80		Soft to firm bluish grey CLAY	-4
					(3.90)			-6
					7.70 (1.30)	Alter stole stole atter stole atter atter stole atter sto	Soft to stiff dark brown fibrous PEAT	-8
					9.00	nike sile Ne sile ske S S S S S S S S S S S S S S S S S S S	Dense greyish brown silty SAND and GF coarse rounded to sub rounded of flint. S coarse. Increasing silt from 14.5m	3AVEL. Gravel is medium to Sand is medium to
-			Туре	Results		an anns	Continued next s	sheet
Rem	arks:	Hand dug se drilling, bent with a respo Upstanding	ervice in tonite se onse zon cover us	spection pit to 1.0mbg eal at 4m. Monitoring ir le between 9.0 -15.0m sed	II. Reduced diamet Istallation - 15m I. Bentonite seal.	ter	IVN - in-situ hand vane IPP - in-situ pocket penetrometer SPT - in-situ standard penetration test (spoon) CPT - in-situ standard penetration test (cone) PID - in-situ photoionization detector	 D - small disturbed sample (tub) J - amber glass jar (250ml) V - amber glass jar (60ml) B - bulk disturbed sample U - U100

\cap	ere	brook		Merebrook Const Tel: 01773 82998	ulting Limited		Equipme	ent and Methods	Borehole No BHN10C	
	u ulana a	Consulting		email: consulting	@merebrook.co	.uk			Sheet 2 of 2	
Pro	iect N	ame			Project I	No.	Co-orde			
Plot		01			MER005	562	539917E - 1	79424N	Cable	
	ation	Greenv	vich P	Peninsula Rivers	ide South				Scale	
	u	choom					6.691		1:50	
									Logged By	
Clie	ent:	Bellway	y Horr	nes (Thames Ga	ateway) Ltd		Dates:	09/07/2013	OTE	
Wall	Water	Sample	es & In	Situ Testing	Depth in metres					
	Water	Sample Depth (m)	Type	Results	Depth in metres (thickness) (6.00)	Legend	Dense greyish coarse rounde coarse. Increa	Stratum Descri	ption /EL. Gravel is medium to id is medium to	- 11
										- - 19 - - - - - -
			Туре	Results					1	-
Rem	narks:	Hand dug s drilling, ben with a respo Upstanding	ervice in tonite se onse zor cover u	Ispection pit to 1.0mbg eal at 4m. Monitoring in the between 9.0 -15.0m sed	yl. Reduced diame nstallation - 15m n. Bentonite seal.	ter	IVN - in-situ IPP - in-situ SPT - in-situ sta CPT - in-situ sta PID - in-situ	hand vane pocket penetrometer andard penetration test (spoon) andard penetration test (cone) photoionization detector	D - small disturbed sample (J - amber glass jar (250ml) V - amber glass jar (60ml) B - bulk disturbed sample U - U100	tub)



Norwest Holst Soil Engineering Ltd. BOREHOLE LOG - CABLE PERCUSSION

Cable Percussion Contract No. F12080 Method Greenwich Village Dando 2000 Drilling Rig RK **English Partnerships** Driller

AH Logged by

Coordinates Orientation

539783.6 E 179279.0 N Ground Level 5.32m AOD Vertical Date Started 30/08/2001 Date Completed 31/08/2001

N11B Header

Hole ID.

Client

Project

Consultant

WS Atkins

		F	PROGRESS	;			DR	LLING DET	AILS
Date	Time	Hole depth	Casing depth	Water depth	Remarks	Hardbore from depth	Hardbore to depth	Chiselling hours	Remarks
30/08/2001 30/08/2001 31/08/2001 31/08/2001	1430 1630 0800 1500	0.00 1.80 1.80 15.00	0.00 1.80 1.80 15.00	Dry Dry Dry 7.45	SOH EOS SOS EOH	1.80	2.00	1.00	

	CAS	SING					V	VATER STR	IKES		
Hole diam.	Max depth of hole at dia.	Casing diameter	Max depth of casing of dia.	Date	Time	Strike at depth	Rise to depth	Time taken to rise	Flow	Casing depth at strike time	Casing depth to seal flow
300 150	7.00 15.00	300 150	7.00 15.00	31/08/2001	1230	7.80	7.45	20.0	SLOW	7.80	15.00

GENERAL NOTES			SPT DETAILS		
1. Hand dug service pit to 1.20m.	Depth	Туре	Incremental blow count/pene	tration in n	nm
 Standpipe installed upon borehole completion. 					
		ļ			
•		ļ			
			* Seating blows only.		
NB All depths in metres, all diameters in millimetres,				Form	ARIAL CP HEADER
water strike rise time in minutes, chiselling time in hours.				Version	1.00
				Revised	01/10/1998

N	BOREHOLE	LOG - (CABL	EF	PER	CL	JSSIO	N		Sheet 1 o	of 2
Contract No. Project	F12080 Greenwich Village	Method Drilling Rig	Cable	e Percu lo 2000	ssion	G	oordinates round Level	5 1 5	539783.6 179279.0 5.32m AC	E N DD	
	English Partnerships	Driller Logged by	RK AH			0 D	rientation ate Started ate Complete	/ 5 be	/ertical 30/08/200 31/08/200	01	
Consultant		an a		Death	[]						
- 262-2=	Description of Strata		Legend	Below G.L.	O.D. Level		Sampling	SPT (U bl	N & S ows)	PT type & depth	ln a
TOPSOIL: E subrounded	Brown gravelly sand. Gravel is -round flint.					J	0.00 0.00	<u></u>			
MADE GRC gravelly clay	DUND: Orange brown sandy, sligh y. Gravel is subrounded-round flin	tly t. Sand		1.00	4.32	D J	1.00 1.00	لمتعليه			11111111
1.80m Cond MADE GRC	crete obstruction DUND: Dark brown sandy, slightly	gravelly		2.00	3.32	D J	1.80 2.00 2.00				
ciay. Sand fine-coarse	brick, flint, concrete, metal, wire a	ind tiles.			و و و و و و و و و و و و و و و و و و و	D	3.00	ي بريا يهي يا ي			
				4 50		DJ	4.00 4.00				Trifici
MADE GRO Sand is coa	OUND: Olive green slightly sandy arse brick.	clay.		4.00	- 0.02	J	4.50	1.1.1.1.1.1			111111111
Green grey pockets. O ALLUVIUM	mottled brown CLAY with occasio ccasional relic rootlets. (Contamin	onal white ated		5.40	-0.08	J	5.40 5.40				
Dark browr remains	amorphous PEAT with abundant	t plant	مالذ مالذ مالذ د مالد مالد مالد مالد د مالد مالد مالد مالد د مالد مالد مالد مالد مالد مالد	5.00 5 5 6 6 6 6 6 6 6 6 6 6 6 6 6			0.00	ليتناب			Turili
Dark brown	n gravelly CLAY	- 147 - 2483	1 344 344 344 344 344 344 344 344 344 34	7.00	- 1.68	DJ	7.00 7.00	لعبيها			1.1.1.1.1
Brown very flint GRAV	y sandy subangular-rounded, fine- EL. Sand is coarse.	-coarse	00000	7.80	-2.48	D	7.80 8.00				111111111
	æ			0.0.0.0	با به به ایر به برا ب	D	9.00				
			0,0	0	F	D	10.00				
NB All depth See hea	ns in metres, all diameters in millin ader sheet for details of drilling, pro	netres. ogress and wate	r						Form	ARIAL	CPL
strikes.	See legend sheet for key to symb	ols.							Version	1.00	177

N	Norwest H BOREHOLE	olst Sc LOG - (DILE CABI	ngi _E F	nee PER	er	ING LI USSIO	d. N	N1 Sheet	1B 2 of 2
Contract No. Project Client Consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cable Danc RK AH	e Percu lo 2000	ssion		Coordinates Ground Level Drientation Date Started Date Complete	53978 1792 5.32n Vertic 30/08 ed 31/08	33.6 E 79.0 N n AOD al /2001 /2001	
	Description of Strata		Legend	Depth Below	O.D.		Sampling	SPT N &	SPT typ	e Inst
Brown very sa flint GRAVEL	andy subangular-rounded, fine-o . Sand is coarse.	poarse		G.L. 15.00			10.00 11.00 12.00 13.00 14.00 15.00	(U blows)	& deptr	
								trend constructions after the		
					ſ					_1_
NB All depths in See beado	n metres, all diameters in millime	etres.						Form	ARIAL	CP LOG



Norwest Holst Soil Engineering Ltd. BOREHOLE LOG - CABLE PERCUSSION

Hole ID.

Header

Contract No. F12080 Method Cable Percussion Coordinates 539413.7 E Project Greenwich Village 179298.6 N Drilling Rig Dando 2000 Ground Level 2.77m AOD Client **English Partnerships** Driller IP Orientation Vertical Logged by AH Date Started 03/09/2001 WS Atkins Consultant Date Completed 05/09/2001

			PROGRESS	3			DR	ILLING DET	AILS
Date	Time	Hole depth	Casing depth	Water depth	Remarks	Hardbore from depth	Hardbore to depth	Chiselling hours	Remarks
03/09/2001 03/09/2001 04/09/2001 04/09/2001 05/09/2001 05/09/2001	0800 1630 0800 1630 0800 1630	0.00 1.20 1.20 3.00 3.00 15.00	0.00 0.00 3.00 3.00 15.00	Dry Dry 2.50 3.50 7.30	SOH EOS SOS EOS SOS EOH	1.30 2.30	1.30 2.50	0.75 0.50	

	CAS	SING					\ \	VATER STR	RIKES		
Hole diam.	Max depth of hole at dia.	Casing diameter	Max depth of casing of dia.	Date	Time	Strike at depth	Rise to depth	Time taken to rise	Flow	Casing depth at strike time	Casing depth to seal flow
300 150	4.50 15.00	300 150	4.50 15.00	04/09/2001 05/09/2001	0000	2.50 7.30	2.50 3.30	20.0 20.0		0.00 6.00	3.00 0.00

					A	
	GENERAL NOTES			SPT DETAILS		
1. 2.	Hand dug service pit to 1.20m. Bentonite seal installed and redrilled 4.00-6.00m.	Depth	Туре	Incremental blow count/pene	etration in	mm
3.	Standpipe installed upon borehole completion.					
				* Seating blows only.		
NB	All depths in metres, all diameters in millimetres,				Form	ARIAL CP HEADER
	water strike rise time in minutes, chiselling time in hours.				Version	1.00
					Revised	01/10/1998

A:	Norwest F BOREHOLI	lolst Sc E LOG - (DILE CABL	ngi _E F	nee PER	ering Lt	d. N	Hole ID N12/ Sheet 1 o). A of 2
Contract No. Project Client Consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cable Danc IP AH	e Percu	ssion	Coordinates Ground Level Orientation Date Started Date Complete	53941 17929 2.77m Vertica 03/09/ d 05/09/	3.7 E 8.6 N AOD al 2001 2001	
	Description of Strata		Legend	Depth Below	O.D. Level	Sampling	SPT N &	SPT type & depth	in
MADE GRO subangular-n	UND: Light grey slightly clayey rounded concrete and flint grave DUND: Black sandy clinker and a	sandy el ash gravel.		1.30	1.47				
Grey mottle	d dark brown organic CLAY.			2.50	0.27				
				5.00	-2.23		<u>. </u>		
Dark brown plant remair	and black amorphous PEAT. C ns.	Occasional	cs stille stole attes stole				· · · · · · · · · · · · · · · · · · ·		
Light green fine-mediun	grey very sandy CLAY. Sand is n.	5	a sike sike	6.70	-3.93				
Brown sanc GRAVEL. S	dy subangular-rounded, fine-coa Sand is coarse.	arse flint			4.93 - - - - - - - - - - - - -				
		-	0.000						1111
NB All depth See head	s in metres, all diameters in mill der sheet for details of drilling, p	limetres. progress and water	•				Form	ARIAL	CP L
strikes. S	See leaend sheet for key to sym	bols					Versi	on 1,00	

I

ľ

	Norwest He BOREHOLE	olst Sc LOG - (OILE CABL	ngi _E F	nee PER(ring Lte	d. N	Hole II N12	D. A of 2
ontract No. roject lient consultant	F12080 Greenwich Village English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cable Danc IP AH	e Percus	ssion	Coordinates Ground Level Orientation Date Started Date Complete	539413 179298 2.77m Vertica 03/09/2 d 05/09/2	3.7 E 3.6 N AOD al 2001 2001	
	Description of Strata		Legend	Depth Below	O.D. Level	Sampling	SPT N & (U blows)	SPT type & depth	Inst
Brown sandy GRAVEL. Sa	v subangular-rounded, fine-coarse	e flint		13.00	-10.23				
Grey brown	silty SAND (THANET SAND)			15.00	-12.23				
Cable Perci	ussion boring complete at 15.00 r	n.							
NB All depth See hea	s in metres, all diameters in millir der sheet for details of drilling, pr	netres. ogress and wate	r				Form	ARIAL	. CP LC
strikes. S	See legend sheet for key to symb	ols.					Versi	1.00	

TQ37NG. 1999 368-95 EASTREN! 270 7037/289 CHELANSKH PANASCHA (ILOCUATION) Owner Enklight Patricksthe Licence No 7937 NE " Nat. Gold Red. TQ 3936 7934 Occupier IGS Ref. No. Status Ground Level m 00 ft. OD Aquiter UAVEL CHALK Level of Well Top 2.80 m 00 ft. 00 6.50 Rest Water Level m bwt ft. bwt ary of Geological Section m OD ft. 00 Construction Linings (below well top) Depti Fromas Type tin 61.8 07.0 56:0 lotte Abstraction Rate Type of Pump pph Chem./Bacs. Anal YES NO opd Well Driller Ser les hoy: Carmolal Temp lond If insufficient space has been allo

ATRY 7	SELISH	Padone	estil un	ence No.		Not. Grid Ref. 7Q 393	6 793	4
Decupier	- outer e.		165	Ref. No.		Status		
Ground Level			m 00		ft. OD	Aquiter 1/ DE1 MAR	K	
Level of Well	Top 2	-80	m 00		ft. OD	unge um	1.0	Death
Rest Water L	evel 6	.50	m bwt		fi. bwt	Summary of Geological Section	Inclose	100m
One 19/	2/99)		3ex 023 col	ogical Survey	ft. OD	Made Gound	65	6.5
Construction	()					Poal Huguran	3.8	10.5
Derth		Linings (belo	w well top)			Konzeni Chay	21.10	280
Dwit A	um	Froma	TANA	open	Туре	aks	18.0	200
5	3/Sun	0	172	610	dein_	There sugs	202	840
17.6	762	0	61-8	305	plain	appor challe	27%	1910
61.86	444	56.0	97.0	250	fotted	11		
100.22	300							
eclosical Su	vév					hote		British Ge
Abstraction	Rates		Type of Pump					
	gph		Chem./Bact. A	inal.	TES NO			
	opd		Well Driller	oil Mech	www.		1 7	<u> </u>



Norwest Holst Soil Engineering Ltd.

Hole ID.	
N13A	•
Header	

BOREHOLE LOG - CABLE PERCUSSION

Contra	act No	F120	080 Powich Vi	llage	Meth	hod	Ca	able P	ercu	Issi	on C	Coordinates	5	539 170	540.4 E	
110,00	·L	0100		llage	Drilli	ng Rig	Pi	lcon 1	500		C	Ground Lev	/el	1.2	1m AOD	
Client		Eng	lish Partn	erships	Drille	er	G	W			C	Drientation		Ver	tical	
Concu	ltopt	14/2	Atking		Logo	ged by	Ał	Η				Date Starte	d lotod	21/	09/2001	
Consu	mann	003	AIKIIIS								L		leted	221	09/2001	
		<u> </u>		PROGRESS								DR			All S	
Date	•	Time	Hole	Casing	Water		Remar	rks		ĮНа	ardbore	Hardbore	Chise	elling	R	Remarks
			depth	depth	depth				-	Tro	m deptn	to depth	nou	ırs		
21/09/2	001	0800 1730	0.00 10.00	0.00 10.00	Dry 3.00	SOH EOS										
22/09/2	001	0800 1300	10.00 15.00	10.00	3.00	SOS										
										1						
		CAS	SING)	\	WATER S	STRIKES	L		<u>_</u>	
Hole	Max	k depth of	Casing	Max depth of	Date	Time	Strike	e at	Rise	to	Time tal	ken Flo	w	Casir at str	ig depth	Casing depth
250		3 00	250	3 00					ucp		10 1130			at 51	ike unie	10 3021 1101
150		15.00	150	11.50				1								
								1								
			GENE	RAL NOTES								SPT D	ETAIL	S		
1.	Hand	dug servio	e pit to 1.	.20m. nd redrilled 1.4	0-3.40m		0	Depth	Туре	e I	Incremen	tal blow co	unt/pe	netrat	ion in m	m
3.	Stand	pipe instal	led upon	borehole comp	letion.											
4.	vvate	r strike at 3	s.um.													
										*	Seating	blows only				
NB A	All dep	oths in met	res, all di	ameters in milli	metres,									Fo	rm A	ARIAL CP HEADER
V	vater	strike rise	time in m	inutes, chisellir	ng time in	hours.								Ve	ersion 1	.00
														Re	evised 0	01/10/1998

	Norwest H BOREHOLE	olst Sc	OILE CABL	ngi _E F	nee PER	eri Cl	ing Lt JSSIO	d. N	H N She	lole ID. 113A eet 1 of 2
contract No. roject client consultant	F12080 Greenwich Peninsula English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cable Pilco GW AH	e Percu	ssion		Coordinates Ground Level Drientation Date Started Date Complete	53954 17900 1.21m Vertic 21/09 ed 22/09	0.4 E 0.1 N a AOD al /2001 /2001	
	Description of Strata	201 <u>7</u> (1977)	Legend	Depth Below	O.D. Level		Sampling	SPT N & (U blows)	SPT & de	type Ir epth
MADE GROU fine-coarse s brick. MADE GROU black slightly Sand coarse	JND: Dark brown very clayey, gr and with rootlets. Gravel is fine-r UND: Green grey mottled brown sandy clay with occasional root brick.	avelly medium and lets.		1.00	0.21	J	0.50 0.50 1.40 1.40			1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
Dark brown a remains	amorphous PEAT with occasiona	al plant	کی میلاد میلاد میلاد میلاد میلاد م میلاد میلاد میلاد د میلاد میلاد میلاد میلاد میلاد د میلاد میلاد	2.40	-1.19	J	2.40			11111
Grey brown	very sandy organic CLAY. Sand	is	-sile	0.00	- 0.40	J	3.00			1
GRAVEL) Soft green g laminations	prey slightly sandy CLAY with thi of brown fine-coarse sand.	nly spaced		9.20	-7.99	D	9.20 9.20	r en el constructione el constructione el constructione el constructione el constructione el constructione el c		
	in metres, all diameters in millir	netres.				D	10.00	F		
ND All ucpuis	 A second sec second second sec							Form		ARIAL GP L

N	Norwest Ho BOREHOLE	olst So LOG - (о <mark>і</mark> l Е Сав	ngi LE F	nee PER	er C	ing Lt USSIO	d. N	Hole II N13/ Sheet 2 c	D. A of 2
Contract No. Project Client Consultant	F12080 Greenwich Peninsula English Partnerships WS Atkins	Method Drilling Rig Driller Logged by	Cab Pilco GW AH	le Percu on 1500	ission		Coordinates Ground Level Drientation Date Started Date Complete	53954 17900 1.21m Vertic: 21/09/ d 22/09/		
	Description of Strata		Legend	Depth Below	O.D.	•	Sampling	SPT N &	SPT type	Ins
Soft green gr laminations of Cable Percus	rey slightly sandy CLAY with thinly of brown fine-coarse sand.	/ spaced		<u>G.L.</u> 15.00	Level	J	10.00	(U blows)	& depth	atu
NB All depths i See heade	n metres, all diameters in millimet r sheet for details of drilling, progr	tres. ress and water						Form	ARIAL CP	LOG
strikes. See	e legend sheet for key to symbols							Version	1.00	

Drilling	g Metho	d Cable Hole	e Percussion/(pen		Bor 40	ehole D Omm to	biameter 8.90m	Casing Diameter 400mm to 8.90m	BORE	IOLE No.		BH1C	6
Equipn	nent	Dando	2000/Knebel			30 20	0mm to 0mm to	o 15.20m o 19.00m	a 300mm to 15.20m a 200mm to 15.80m					
Drill Fl Orienta	uid ation (°)	Polym 90	ner GS550			17 Log	<u>'0mm to</u> ged by	<u>60.00</u> m Com	piled by Checked by					
Dates	Drilled	Start End	28/11/2005 08/12/2005			CM 30/	11/200	an)5 06/1	2/2005					
Date	Casing	Water Depth	Sample/Co	re Rec	overy		SPT Blows					Depth (Thick		logond
& Time	Depth (m)	(III) (Flush	Depth (m)	тся	NO.	ROD	Core		Description of	of Strata		(Thick- ness)	Levei	Legena
	(,	%	From To	%	%	%	(mm)					(m)		
28/11			0.30	D	1				MADE GROUND: Red tarmad	z.		(0.15) 0.15 (0.15)		
			- 0.30	CD/K D	: 2 3				Sand is fine to coarse, angular, fine to coarse	, gravel of vari	is ous	(0.13) 0.30 (0.15)		
			0.45	CD/K	: 4				lithologies.			0.45		
									MADE GROUND: Mesh over slightly clayey sandy o	r grey br gravel wi	own, th			
			1.50	D	5				coarse, gravel is angul	lar to ro	unded, and			
			1.50	CD/K	Ē				concrete fragments. Col concrete fragments.	obles are	brick and	ł		
			-						MADE GROUND: Soft, grey	, slight	ly sandy,	-		
			-						hydrocarbon odour. San coarse gravel is angul	, with a nd is fin lar to ro	e to unded	(3 75)		
			-						fine to coarse of flint concrete. Occasional fr	, brick	and of black	-		
28/11	3.00	DRY	-						organic material.	-				
29/11	3.00	DRY												
			3.50	D	7									
			3.50	CD/K	: 8									
			-									-		
			4.20 4.20	D CD/K	9 : 10				Soft to firm grey sligh	ntly sand	y CLAY	4.20		
			-						with occasional fragmer organic material. Stro	nts of bl ong hydro	ack carbon	-		
			-						odour. (ALLUVIUM)					
			-											
			-									(2.30)		
			-											
			_ 6.00 - 6.00	D CD/K	11 12							E		
			6 50		10							6 50		
			6.50	D CD/K	13				Dark brown, spongy PEAT	r with a	strong	- 6.50		
			-						(ALLUVIUM)					یالد الد عالد
			-											
			-									(2 40)		
			-									(2:10)		
			-									F		
			8 50	П	15									, , , , , , , , , , , , , , , , , , ,
			8.50	CD/K	16							Ē		M72 - SM - SM72
			8.90	D	17				Dark grey slightly silt	y slight	ly sandy	8.90		
			8.90	CD/F	. 18				is subrounded, fine to	coarse o vith hyd	, gravel f flint. rocarbon			• • • •
		(0)	-						(RIVER TERRACE GRAVELS)	ur.		‡		• • • • •
			-									ŧ.		· · · · ·
			_									<u> </u>		6 G
(See note & keyshe	KS 1 es eets)	Prior depth	to boring a (and rescanned	able lusir	Avoid ng the	dance CAI	Tool	(CAT) s neck for	services. Services were	An inspec e not loc	tion pit was ated.	hand-dug	g to 1.	20m
	2	in red	er protection luced diameter	was c casi	ng.		it by s	sealing	the base of the hole at a	a depth o	E 8.90m and	15.20m ai	na cont	inuing
	3 4 5	Ground	water was not	appa	arent	duri duri	ng bor	ing.	means to 19.00m and then	Progress	eu ny open h	ore mean	5 10 61	
Scale 1:5	5 50	see in	GEALLALLON DE	Lails		idl	. sueet	-•						
		-Fu	GRO			Pro	oject BF	1106. DC	ME ARENA, GREENWICH	C	ontract No.	WAL	050168	
							Si	ir Rober	t McAlpine Ltd t McAlpine Ltd					
			\sim									BH106 (1	of 7)	304/02

Drilling	g Metho	d Cable Hole	Percussion/C	pen		Bor 40	ehole D	iameter	Casing Diameter	BORE	IOLE No.		BH10)6
Equipn	nent	Dando	2000/Knebel			30	0mm to	5 15.20m	300mm to 15.20m					
Drill Fl	uid ation (°)	Polym	er GS550			17	<u>omm to</u>	<u>60.00m</u>	niled by Checked by	_				
Dates	Drilled	Start	28/11/2005			CM 30/	11/200	an 15 06/1	2/2005					
		Water	Sample/Co	ore Rec	overy	307	SPT	JJ 00/1	2/2005			Denth		
Date &	Casing Depth	(m)	Denth (m)	Туре	No.		Blows /N		Description of	of Strata		(Thick-	Level	Legend
Time	(m)	(Flush Return)	From To	TCR	SCR	RQD	Core Size		Decemption	or officiate		ness)		
		%		70	70	70	(mm)		Dark grev slightly silt	tv slight	lv sandv	(m)		0 0 0
			-						GRAVEL (See Previous St	heet)		ŧ		· · · · · ·
			-									+		• • • • •
			-											· · · · ·
			_ 11.00	В	19							} ₽		• • • • • • •
			-									-		° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °
			-											• • • • • • •
			-									(5.00)		• • • •
			-											° ° /
				1								Į		• • • • •
			-	1								ŧ		· · · · ·
			_ 13.00	в	20							F		
				1								ţ		0 0 0
			-											· · (· /
			-									+		• • • • • •
			- -									F		· · · ·
														• • • •
			-		01									° ° ° °
			14.70	D CD/H	21 22				Stiff possibly extremel	ly closel	y fissured	14.70		
			-						(LONDON CLAY)					
29/11	15.50	DRY												
30/11	15.50	9.00	-											
			-									-		
												F		
			-											
			-									(4.30)		
			-									-		
			-									ļ ļ		
			-											
			- -											
			-									ł		
			-									+		
30/11	15.80	DRY	18.70	D	23							Į		
08/12	15.80		- -	1					Brown green and grey cl	layey fin	e and	_19.00		
			• •	1					medium SAND, with rare (LAMBETH GROUP)	sħell fr	agments.	ŧ		·
			-	1								ł		
			-									-		
Remar	ks		<u> </u>	<u> </u>								<u> </u>		
(See note & keyshe	es eets)													
Scale 1:5	50						-1							
		-Fu	GRO			Pr	oject BF	1106, DO	ME ARENA, GREENWICH	C	ontract No.	WAL	050168	
		Į					Si	ir Rober Ir Rober	t MCAlpine Ltd t MCAlpine Ltd	F	igure No.		of 7	
											1	зн⊥06 (2	oi 7)	304/03

Drilling	g Metho	d Cable Hole	Percussion/C	pen		Bor 40	ehole D	ameter	Casing Diameter 400mm to 8,90m	BOREHOLE No.		BH10)6
Equipr	nent	Dando	2000/Knebel			30	0mm to 0mm to	5 15.20m 5 19.00m	300mm to 15.20m 200mm to 15.80m				
Drill Fl Orient	luid ation (°	Polym 90	ner GS550			17 Log	<u>'0mm to</u> ged by	<u>60.00</u> m Com	piled by Checked by				
Dates	Drilled	End	28/11/2005 08/12/2005			CM 30/	11/200	an)5 06/1	2/2005				-
Date	Casing	Depth (m)	Sample/Co	Type	No.		Blows /N		Description of	of Strata	Depth (Thick-	Level	Legend
Time	(m)	(Flush Return) %	From To	TCR %	SCR %	RQD %	Core Size (mm)		Description		ness) (m)		
			-										
			- - -								ļ ļ		
											Į		
			-								F		
			- - -								t t		
			-								ŧ		
			- 								F		
			-										
			-								ţ		
			- -						Between 23.00m and 23.5 gravel layer. (Drillers	50m; Possible s observation)	F		
			- - -								Ē		
			-								ŧ		
			- -								(10.00)		
			- - -								ļ		
			-								Ę		
			- 								È.		
			- - -						At 25 50m. Doggible or	corrol (Drillorg			
			- - -						observation).	aver (Driffers	F.		
											F		
			- - -								ţ		
			- - -								F F		
			-								F		
			- -								ţ		
			- - -								F F		
			- -								F		
			-										
			- - -										
			- - -						Vellow mottled green	clavey fine SAND			
			-						(LAMBETH GROUP)		ł		
			- - -						nodule? (Drillers obse	ravel or calcrete ervation)			
			_ 0.00-60.00	OH							F_		
Remar (See not	es				·		·					<u> </u>	
a reyso													
Scale 1:	50					. D-	niect			Contract No		050160	
		T	GRO				BH BH Si	H106, DC ir Rober	ME ARENA, GREENWICH t McAlpine Ltd	Contract NO.	WAL	891050	
		•					Si	ir Rober	t McAlpine Ltd	Figure No.	BH106 (3	of 7)	

304/03

Drilling Method Equipment		d Cable Hole	e Percussion/C	pen		Bore 40	ehole D	iameter 8.90m	Casing Diameter 400mm to 8.90m	BOREHOLE No.		BH1C)6
Equipn	nent	Dando	2000/Knebel			30 20	0mm to 0mm to	0 15.20m 0 19.00m	300mm to 15.20m 200mm to 15.80m				
Drill Fl Orienta	uid ation (°)	Polym 90	er GS550			Log	<u>'0mm to</u> ged by	<u>60.00m</u> Com	piled by Checked by				
Dates	Drilled	Start End	28/11/2005 08/12/2005			CM 30/	11/200	an)5 06/1	2/2005				
Date	Casing	Water Depth	Sample/Co	re Rec	overy		SPT Blows				Depth (Thick		Logond
& Time	Depth (m)	(Flush Return)	Depth (m)	TCR	SCR	RQD	Core		Description of	of Strata	ness)	Levei	Legenu
	(,	%	From To	%	%	%	(mm)				(m)		
			-										
			-								(3.00)		
			-								ŧ.		
			-										
			-										
			—						Grev clavev fine SAND	with rare gravel	32.00		
			-						(LAMBETH GROUP) Between 32.00m and 32.7	/5m; Possible	ł		
			-						gravel layer. (Driller	rs observation)	E I		
			L -								t		
			- -								ŧ		
			- -								ŧ		
			-								I I		
			-								F		
			-										
			-										
			-								-		
			-										
			- - -										
			- - -						Potuson 26 00m and 29 0	m. Doggihlo			
									gravel layer. (Driller	s observation)	(0.00)		
			-										
			-										
			- -								F		
			-								I I		
			-										
			[-								E		
											Ŧ		
			-								F		
			-						Between 39 NOm and 41 ()0m; Possible	Ŧ		
			-						gravel and cobble layer observation)	. (Drillers	F		
		(100)	[-								Ę		·
			-										
											_40.00		
Remarl (See note & keyshe Scale 1:5	KS es sets)					De	niest			Contract No.			
								1106, DO	ME ARENA, GREENWICH	Contract No.	WAL	U50168	
		!	Â				Si	.r корег .r Rober	t McAlpine Lta t McAlpine Ltd	Figure No.	BH106 (4	of 7)	304/03

	g Metho	d Cable Hole	Percussion/()pen		Bor 40	ehole D Omm to	iameter 8.90m	Casir 400	n g Diameter mm to 8.90m	BOREHOLE No.		BH10	6
Equip	ment	Dando	2000/Knebel			30	0mm to	15.20m	1 300 200	mm to 15.20m				
Drill F	luid	Polym 90	er GS550			17	<u>'Omm to</u>	60.00m	niled by	Checked by	4			
Dates	Drilled	Start	28/11/2005			CM	11/200	an 5 06/1	2/2005	Checked by				
		Water	Sample/Co	ore Rec	covery	30/	SPT	5 00/1	2/2005			Denth		
Date &	Casing Denth	(m)	Denth (m)	Туре	No.		Blows /N			Description	of Strata	(Thick-	Level	Legend
Time	(m)	(Flush Return)	Erom To	TCR	SCR	RQD	Core Size			Decemption		ness)		
		%		70	70	70	(1111)		Grev S	AND. Sand is fi	ne and medium.	+ 40.00		
			-						(THÂNE	r sands)		I		
			-									+		
			-									ŧ		
			-									+		
												Ē		
			-									F		
			-									ŧ		
			-									+		
			-									ŧ		
			-									F		
			-									Į		
			-									F		
			-									Ī		
			-									F F		
			-									Ę		
			-									Ē		
			-									ŧ		
			-									Ē		
			-									(10,00)		
			-									10.007		
			-									ł		
			-									Ē.		
			-									Ł		
			-									ţ		
			• • -									Ē		
			-									ł		
			-									I -		
												t		
												Į.		
			-									ŧ		
			-									È.		
												ŧ		
			-									ŧ		
												Ŧ		
												F		
												ŧ		
			•									ŧ		
			-									ŧ		
												_50.00		·····

Drilling	g Metho	d Cable Hole	Percussion/	Open		Bor 40	ehole D	Diameter	Casing Diameter 400mm to 8.90m	BOREHOLE N	о.	BH10)6
Equipr	nent	Dando	2000/Knebel			30 20	0mm to	o 15.20m o 19.00m	a 300mm to 15.20m 200mm to 15.80m				
Drill Fl Orient	luid ation (°)	Polyn 90	er GS550			17 Loa	omm to	5 60.00m Com	piled by Checked by	1			
Dates	Drilled	Start End	28/11/2005 08/12/2005			CM 30/	11/200	an)5 06/1	.2/2005				
Data	Casing	Water	Sample/C	ore Rec	covery		SPT				Depth		
&	Depth	(m) (Elush	Depth (m)	Туре	No.	l	/N		Description of	of Strata	(Thick- ness)	Level	Legend
Time	(m)	Return)	From To	TCR %	SCR %	RQD %	Size (mm)				(m)		
Time	(m)	Return)	From To		SCR %	RQD	Size (mm)		Possibly flint GRAVEL A intermixed with commina (BULLHEAD BEDS?) White CHALK with occasi	AND COBBLES uted chalk.	(m) 50.00 (1.50) 51.50 (8.50)		
Remar	ks		L				L	<u> </u>	End of Bor	rehole			<u> </u>
Scale 1:5	rə es eets) 50	-6.	RDN			Pre	oject	-105		Contract N	Io. WAL	050168	
		Γ					BH	H106, DC ir Rober ir Rober	ME ARENA, GREENWICH t McAlpine Ltd t McAlpine Ltd	Figure No.	BH106 (6	of 7)	
												' /	304/03

Drilling Method Cable Percussion/Open	Borehole Diameter	Casing Diameter	BOREHOLE No. BH106
Equipment Dando 2000/Knebel	300mm to 15.20m	300mm to 15.20m	
Polymer GS550 90	170mm to 60.00m Logged by Compi	led by Checked by	_
Dates Drilled Start 28/11/2005 End 08/12/2005	CM an 30/11/2005 06/12/	2005	
Description		Depth Level	
Concrete /		0.50	Flush lockable stopcock box cover with
Cement/Bentonite Grout			gas tap. Pipe diameter 75mm to 60.00m.
Bentonite Seal		48.50	_
		51.50	
Gravel Filter			
	2000 200 2000 2		
Remarks		60.00	Base of Hole
(See notes & keysheets)			
Not to Scale	Project		Contract No. WAL050168
	BH106, DOME Sir Robert	ARENA, GREENWICH McAlpine Ltd	-
	SIT Robert	MCALPINE LTO	Figure NO. BH106 (7 of 7)

Projec	t Nam	ne Silve	ertown Tun	nel						Evolore	torul			Н	ole ID.
Projec	t No.	TA7	510							Explora	litory F	TOIP L	ug	F	B3A
Engine	er	Atki	ns Ltd												-Dort
Emplo	yer Level	Atki 2.92	ns Ltd m OD			Coord	dinates	5392	40.99 E, 1796	56.52 N Natio	onal Grid			неа	uer Page 1
Date Sta	arted	25/1	1/2014			Date	Completed	10/12	2/2014			Inclina	tion Vert	ical	
Тор	Base	Туре	Date Started	Date Ended	Crew	Section Logged By	Core Barrel	Core Bit	Equip	oment	Sho Su	oring / pport	Ì	Remarks	
0.00	1.20	IP CP	25/11/2014 26/11/2014	26/11/2014 10/12/2014	CC/EC CC/EC	IH IH	NA NA	NA NA	Hand Dande	Tools o 3000					
4.00	15.45	CP	09/12/2014	10/12/2014	DAVMI	ІН	INA	NA	Dande	0 3000					
			2												
			CABLE	PERCUSSION	DETAILS		dn.				V	VATER STR	IKES		(*)
from	Strata	Depth to	Start time	selling F Duration hhmm	lemarks				Date	Time	depth	depth	taken to rise	at str tim	ike to seal e flow
9.10 11.30	6	9.30 11.50	NA NA	0030 C 0030 C	hiselling hiselling				26/11/2014 26/11/2014	1300 1300	1.80 1.80	1.70 1.70	5 10	1.8	0 NS 0 NS
									26/11/2014 26/11/2014 09/12/2014	1300 1300 1715	1.80 1.80 5.70	1.70 1.70 4.80	15 20 5	1.8	0 NS 0 NS
									09/12/2014 09/12/2014	1715 1715	5.70 5.70	4.30 4.20	10 15	5.7	0 NS 0 NS
									09/12/2014	1715	5.70	4.10	20	5.7	0 NS
										1	HOLE I	i Diameter	/ CASING		
	2								Hole diameter	Depth hole	of	Casing diameter	Dept casi	h of ing	
									200 150	5.00 15.4	5	200 150	5.0 13	00 .10	
			DOT!												
Fror	ņ	To	ROTA	Flush	AILS F	lush	Flush		Тор	Base		Diameter		ne	Recovery
depi	:n	aeptn		туре	re	%	COIOUT			_				imss	70
						=									
									8						
	INST/	ALLATIO	N DETAILS		 PIF	PE CONSTR	UCTION				BAC	CKFILL DET	AILS		
Distance from G	ID	Ту	e Respo	nse zone ID	Pi	pe de	Dia. Typ	be of	Top of section	Base of section	Ma	terial		Re	emarks
4.00	02	SF	2 1.00 2 6.00	4.00 02	0.00 1.00	1.00	50 Pl 50 Slo	ain tted	0.00	0.50	Con	tonite			
10.00	01	1	0.00	01	0.00	6.00 15.00	50 Pl 50 Slo	ain tted	1.00 4.00	4.00 6.00	Gravel	backfill tonite			
									6,00	15,45	Grave	backfill			
NOTES		nthe in	matres all dis	maters in mill	motros										
NOTES	Wate For de	r strike r etails of	ise time in mi abbreviations	inutes, hard s see kev	trata time	e in hhmm									
FINAL	, or ut		Log	Print Date And	l Time: 1	8/06/2015	5 15:19:22		Log checke	ed by: SF 03/0	06/2015		soi	Lend	Sineerir
	EXPHOLE	HDR P1		Issue.Revision No. 1	06	ls	sue Date 07/10/	2013					Part	of the Bach	y Soletanche group

Project	Nam	ne Silve	ertown ⁻	Tunnel							Exploratory Hole Log		Ho	le ID.	
Project Enginee	No. er	TA7 Atki	510 ns Ltd										E	B3A	
Employ	er	Atki	ns Ltd										Head	er Page 3	2
Ground L	evel ted	2.92r	n OD 1/2014				Coordinates	53924	40.99 E, 1	17965	6.52 N National Grid	n Ver	tical		
Date Star	teu	23/1	1/2014	PROGE	RESS		bate completed	10/12	., 2014		SPT DETAILS				
Date	Т	ïme	Hole depth	Casing depth	Water depth		Remarks	-	Depth	Туре	Incremental blow count / Penetration in mm	łamme No.	er Energy ratio %	Casing depth	Water depth
25/11/20; 25/11/20; 26/11/20; 09/12/20; 09/12/20; 10/12/20;	14 14 14 14 14 14 14 14 14 14 14 14	0730 1800 0730 1800 0730 1800 0730 1800	0.00 0.80 4.00 0.00 6.00 15.45	4.00	NR DRY 3.50 NR 4.90 5.10 9.70	Start of Hole End of Shift Start of Shift End of Shift Start of Shift Start of Shift End of Hole			1.30 2.50 7.00 8.50 10.00 11.50 15.00	SPT(C SPT(C) SPT(C SPT(C SPT(C SPT(C SPT(C)) N=20 (3,5,5,4,5,6) N=11 (2,2,4,2,2,3) N=26 (3,4,5,6,7,8) N=18 (2,3,3,4,5,6) N=17 (2,2,3,4,5,5) N=30 (2,4,6,7,8,9)	AR89 AR538 AR538 AR538 AR538 AR538 AR538 AR538	% 67 75 75 75 75 75 75	1.30 2.50 5.60 7.00 8.50 10.00 11.50 13.60	DRY 2.30 DRY 4.70 4.30 4.60 9.70
			DEP	TH RELA	TED REN	/ ARKS									
Top E Depth D	Base epth	Remark	s												
4.00 4 5.00 6 6.00 1	4.00 6.00 3.10	Contamir Bentonit Water ad	nation ence e seal. ded to ass	ountered? I ist drilling.	nole suspe	nded.									
				ENERAI	NOTES		8								
1. Insp tern 2. Bord 3. Hold	oectic ninat ehole e red	on pit ex ed. e abando rilled 0.	cavated oned an 00m to	l to 1.20 d backfil 4.00m.	m. Conc	rete encou	intered and pit				t Section blows colu				
NOTES:	All de	epths in l	netres, a	II diamete	ers in mill key	imetres.					secting states only.		6		
FIR 121	FOLG	icialis of	annieviai	Les D	NCy	J Times 40 "	00/2015 45 40 00		1	ack-	1 by SE 02/00/2015	50	IL enc	inee	BIUG
FINAL Form No. SIE	EXPHOL	EHDR P2		LOG Print Issue.R	L Date An Revision No. 1	u Time: 1870 .06	Issue Date 12/03/	2014	Log cr	IECKE(1 DY. OF 00/06/2010	Pa	rt of the Bachy	Soletanche	group

Project No.	TA7C10				1	EXD	orain	rv r			
Engineer	Atkins Ltd					слр	ionator	'y '	lole Log	EB3A	
Employer /	Atkins Ltd					_				Sheet 1 of 4	4
Ground Level 2 Hole Type I	2.92m OD IP+CP	Coordinates 53 Inclination Ve	9240.99 E, ertical	179656.5	2 N Nat	ional G	rid				
	Description of Strata		Legend	Depth	Datum		Sampling		Blow Count And Sample Recovery	In Situ Test	Install-
				(Thickness)	Level	(Details	Dia.	TCR SCR RQD	Details	ation
MADE GROUND: Block (Hardstanding)	c paving.			0.05 0.10 -	2.87 2.82						
MADE GROUND: Yellor (Made Ground)	w fine to coarse sand.			-		ES 1	0.30				
MADE GROUND: Brow size fragments of brick Occasional subangular (Made Ground)	n silty sandy angular to subrounded fine t, flint, concrete, granite and rare ceramic cobble size fragments of concrete and b	to coarse gravel s. rick.		(0.90) 		B2 ES3 B4	0.30 0.50 0.50				
MADE GROUND: Black sized fragments are an concrete, flint, ceramic	clocally clayey gravelly fine to coarse san gular to subrounded fine to coarse of bri is and rare possible ACM.	d. Gravel ck,		1.00	1.92 1.72	ES 5 B 6	1.00 1.00			-	
(Made Ground) MADE GROUND: Black sized fragments are su flint, concrete, coal, sla gravel sized pockets of and brown tar staining (Made Ground)	c locally clayey gravelly fine to coarse san bangular to subrounded fine to coarse of g and possible ACM. Occasional fine to c f grey and black clay. Hydrocarbon odour	d. Gravel brick, oarse and black				D 7 B 8	1.30 1.30-1.80			SPT(C) N=20 - 1.30 1.75 -	
						ES 9	2.00				
F						D 10 B 11	2.50 2.50-3.00			SPT(C) N=11 - 2.50 2.95	
				- 330 -	-0.38	ES 12	3.00			-	
MADE GROUND: Grey gravelly clayey silt. Fre- hydrocarbon/tar. Grave to coarse of brick, slag hydrocarbons. (Made Ground) at 3.50m 1 No	and black locally stained brown slightly s quent pockets of brown staining and low el sized fragments are subangular to sub and concrete. Strongly contaminated wit o cobble of subangular granite	andy slightly viscosity rounded fine h		(0.70)	0.00	UF B14	3.50-4.00 3.50-4.00	100	7 Blows, 0% Recovery		
PEAT. (Driller's descrip (Alluvium)	tion)		24. 10. 10 10. 1	4.00 — 	-1.08	ES 15 D 16	4.00 4.00				
			بالد بالان بالد بالد بالد بالا بالا بالد بالا بالد بالد بالد بالد بالد بالد بالد بالد بالد بالد بالد بالد	(0.80) 		UTF	4.50-4.95	100	20 Blows, 0% Recovery		
Soft grey locally black	slightly gravelly silty organic rich CLAY. G	ravel		4.80 -	-1.88	D 17 ES 18 B 19	4.80 4.80 4.80-5.30				
NOTES: All depths i See header For details FINAL	in metres, all diameters in millimetr r sheet for details of boring, progre of abbreviations, see key Log Print Date And	es. ss and water. Time: 18/06/2015	5 15:19:29	Log) checke	d by: S	F 03/06/2	2015	SO		

Project No. TA7510 EBS	3A		
Ligineer Alkins Lu	•••		
Employer Atkins Ltd Sheet 2	of 4		
Ground Level2.92m ODCoordinates539240.99 E, 179656.52 N National GridHole TypeIP+CPInclinationVertical			
Description of Strata Legend Depth Datum Sampling Blow Count And In Situ T	est	In	stall
Thickness Level Details Dia. TCR SCR RQD Detail	S	at	ion
is subangular to subrounded fine to coarse of flint. Occasional medium to coarse gravel sized pockets of peat. (Alluvium)			
Medium dense multicoloured slightly silty very sandy subangular to subrounded fine to coarse gravel of flint and chert (locally with hydrocarbon contamination). Sand is fine to coarse. (River Terrace Deposits) 5.70 -2.78 W21 5.70 D 23 6.00	5.05		
= B 24 6.70-7.20 = B 24 6.70-7.20 = D 25 7.00 F 536 7.00 F 53			
below 7.70m no silt	7.45		
below 8.50m occasional fine to medium gravel pockets of brown hydrocarbon staining	3.95		
below 9,50m sand and gravel			
NOTES: All doubles in motors all dismosters in millimotros			
See header sheet for details of boring, progress and water. For details of abbreviations, see key			
FINAL Log Print Date And Time: 18/06/2015 15:19:32 Log checked by: SF 03/06/2015 SOIL ETIGIT Form No. SIEXPHOLELOG Issue Revision No. 1.06 Issue Date 20/09/2013 Part of the Bachy Sole	etanche	e Gr	oup

Project Name	Silvertown Tunnel					Ехр	lorato	ry F	lole Loa	Hole ID.	
Project No. Engineer	TA7510 Atkins Ltd					P		<i>,</i> .		EB3A	
Employer	Atkins Ltd	Condition	F202/0.00 F	170000	2 NI NI-1					Sheet 3 of 4	4
Ground Level Hole Type	2.92m OD IP+CP	Inclination	Vertical	179656.5	oz in inat	ional G	na				
	Description of Strata		Legend	Depth	Datum Level		Sampling		Blow Count And Sample Recovery	In Situ Test	Install-
	5 93 90 N2995-32 500 N2 580		11.12.4.7	(Thickness)	Lever	1	Details	Dia.	TCR SCR RQD	507(c) N 19	
Medium dense multi subrounded fine to co hydrocarbon contami (River Terrace Deposi	coloured slightly silty very sandy subangu oarse gravel of flint and chert (locally with ination). Sand is fine to coarse. its)	lar to		. "a la a gladaa dafa ja a afamadafa a a fa a a a a a a a a a a a a a a a		B 34 D 35 ES 36 B 37 D 38	10.50-11.00 10.50 10.50			SPT(C) N=18 10.00 10.45 	
				e e e e e e e e e e e e e e e e e e e		8 39 D 40 ES 41	12.50-13.00 12.50 12.50				
Stiff fissured thinly la Fissures are extremel planar rough and smo	minated brownish grey slightly sandy CLA ly closely to very closely spaced randomly soth. Occasional thin partings of grey fine to lock to initia on first to co	Y. orientated sand.		13.10 - - -	-10.18	D 42 B 43	13.10 13.10-13.60				
sized selenite crystals silt. (London Clay Formati	s. Frequent 2mm x 5mm burrows infilled v on)	with grey				ES 44	13.40				
				 (2.35)		D 46	14.05	100	из ыожэ, 6/% Recover	-	
below 14.50	m sandy clay. No fissures					D 47	14.50				
NOTES: All depths See heade For details FINAL	s in metres, all diameters in millimeters er sheet for details of boring, progre s of abbreviations, see key Log Print Date And	res. sss and water. d Time: 18/06/20	015 15:19:34	Log) checke	d by: S	F 03/06/2	015	sc		ering
FINAL Form No. SIEXPHOLELOG	Log Print Date And Issue.Revision No. 1	d Time: 18/06/20 1.06	15 15:19:34 Issue Date 20/09.	Loç 2013) checke	d by: S	F 03/06/2	015	SC Pa	IL ENGINEE	erin he Group

Party Rot TAY 510 End of the Lud EB3A Employeer Akins Lud 323:00 Constrained and State Stat	Project Name	Silvertown Tunnel		Fyplorato	γŀ	lole Log	Hole ID.					
Employee Alfain Lid Sheet A of A Second Level Sign 200 F 179656 27 Matched Grid Sign 200 F 179656 27 Matched Grid Note Type ProP Documentary 200 F 179656 27 Matched Grid Documentary 200 F 179656 27 Matched Grid State Transch Binky instance Instanting or digity andy CLVP Documentary 200 F 17966 200 F 200 F 179666 200 F 200 F 179666 200 F 200 F 17966 200 F 17966 200 F 200 F 17966 200 F 1796 200 F 1796 200 F 17966 200 F 17	Project No. Engineer	TA7510 Atkins Ltd			уı	loie Log	EB3A					
Gravet Let 2.37: 00 Contract 532:400 pt 1976552.91 Multipaid Gifd The hype IP-CP inclusion Vertical The hype IP-CP inclusion Vertical The hype IP-CP inclusion Vertical Inclusion	Employer	Atkins Ltd									Sheet 4 of	4
Description of Status Legent Description of Status Bitwee Status (Status (St	Ground Level Hole Type	2.92m OD IP+CP	Coordinates	539240.99 E, Vertical	179656.5	2 N Nat	ional Grid					
During and uncerted, brown bit metrics of graph from each, and the second and uncerted, brain bit metrics of graph from each, and the second bit metrics of graph from each and the second bit metrics of graph from each and the second bit metrics of graph from each and the second bit metrics of graph from each and the second bit metrics of graph from each and the second bit metrics of graph from each and the second bit metrics of graph from each and the second bit metrics of graph from each and the second bit metrics of graph from each and the second bit metrics of graph from each and the second bit		Description of Strata		Legend	Depth (Thickness)	Datum Level	Sampling Details	Dia.	Blow Count A Sample Recov	And very	In Situ Test Details	Install- ation
NOTES: All depths in metres, all diameters in millimetres. See header sheet for details of boring, progress and water. For details of abbreviations, see key FINAL Log Print Date And Time: 18/06/2015 15:19:35 Log checked by: SF 03/06/2015 FORM No. SIEXPHOLELOG Issue Revision No. 106 Issue Date 20/09/2013	Stiff fissured thinly la Fissures are extreme planar rough and sm Occasional patches c sized selenite crystal silt. (London Clay Format below 15.00 Exploratory hole con	aminated brownish grey slightly sandy CL ly closely to very closely spaced randomi ooth. Occasional thin partings of grey fin iblack staining on fissures. Rare fine to o is. Frequent 2mm x 5mm burrows infilled ion) Im clay is very stiff 	AY. y orientated e sand. coarse sand i with grey			-12.53	Details				SPT N-30 15.00 15.45	
NOTES: All depths in metres, all diameters in millimetres. See header sheet for details of boring, progress and water. For details of abbreviations, see key Image: Constraint of the second secon					-							
See header sheet for details of boring, progress and water. For details of abbreviations, see key Solution FINAL Log Print Date And Time: 18/06/2015 15:19:35 Log checked by: SF 03/06/2015 SOIL ENGINEERING Form No. SIEXPHOLELOG Issue Revision No. 1.06 Issue Date 20/09/2013 Part of the Bachy Soletanche Group	NOTES: All depth	s in metres, all diameters in millime	tres.									
Form No. SIEXPHOLELOG Issue Revision No. 106 Issue Date 20/09/2013 Part of the Bachy Soletanche Group	See head For detai	er sheet for details of boring, progr Is of abbreviations, see key Loo Print Date Ar	ress and water.	015 15:19:35	Loc	checke	ed by: SF 03/06/2	015		SOIL	enginee	RING
	Form No. SIEXPHOLELOG	Issue Revision No.	1.06	Issue Date 20/09	/2013					Part of	the Bachy Soletanc	he Group

Appendix E. Silvertown Site Borehole logs

Project Name Silvertown Tunnel Exploratory Hole Log									og	Hole ID. G16							
Engine	er	Atki	ns Ltd											Hee	dar Page 1		
Employer Atkins Ltd Ground Level 2.29m OD Coordinates 540009.95 E, 180356.33 N National Grid													Header Page 1				
Date Sta	rted	08/0	1/2015			Date	/2015	15 Inclination Vertical									
Тор	Base	Туре	Date Started	Date Ended	Crew	Section Logged By	Core Barrel	Core Bit	Equip	oment		Remarks					
0.00 1.20	1.20 30.30	IP CP	08/01/2015 08/01/2015 DA/RA JR NA NA Hand Tools 08/01/2015 13/01/2015 DA/RA JR NA NA Dando 3000														
Hard	Strata I)enth	CABLE I Chis	PERCUSSION	DETAILS				Date	Time	V Strike at	VATER STR Rise to	IKES Time	C	asino depth		
from	Strata	to	Start time	Duration	Remarks				Date		depth	depth	taken to rise	at str tim	ike to seal e flow		
7.60 9.00 23.90 29.50		7.70 9.20 25.60 30.00	NR NR NR NR	0030 0030 0400 0100	Chiselling Chiselling Chiselling Chiselling				08/01/2015 08/01/2015 08/01/2015 08/01/2015 08/01/2015	0815 1345 1345 1345 1345 1345	1.40 6.60 6.60 6.60 6.60	1.40 3.10 2.70 2.50 2.30	NR 5 10 15 20	NI 6.3 6.3 NI 6.3	R NR O NR O NR R NR O NR		
HOLE DIAMETER / C												/ CASING	CASING				
									Hole diameter	Depth hole	of	Casing diameter	Dep	oth of sing			
									200 150	20.0	0	200 150	1	2.70			
					74110												
Fron		,To,	ROTA	Flush	FAILS	lush	Flush	_	Тор	Base		Diameter		ime	Recovery		
dept	h	depth		type	re	turn %	colour		19 MAN	012/01583	· · · · · · · · · · · · · · · · · · ·		hh	mmss	%		
Distance from G L	ID	Тур	e Respor	ise zone II Base) Pi	pe Base of	Dia. Typ	e of	Top of Base of Material section					Remarks			
10.50 26.00	02 01	SP	7.00 E 24.50	10.50 02 26.50 02	0.00 7.00 0.00	7.00 10.50 26.00	50 Pla 50 Slot 19 Pla	ain ted ain	0.00 0.50 7.00 10.50 24.50 26.50	0.50 7.00 10.50 24.50 26.50 30.30	Con Ben Gravel Ben Sand Ben	crete tonite backfill tonite backfill tonite					
NOTES:	All dep Water For de	ths in i strike r tails of	metres, all dia ise time in mi abbreviations,	meters in mi nutes, hard see key Print Date Ar	limetres. strata time	in hhmm	5 16:20:50						50				
Form No. St	EXPHOLEH	DR P1	LUG	ssue.Revision No.	1.06	5, 50, 201 I	isue Date 07/10/	2013					Par	t of the Bach	y Soletanche group		

Project No. TA/S10 Conditions Scotting of the second o	Project Name Silvertown Tunnel										Exploratory Hole Log	9	Hole ID.					
Employee Attins Lid Tender Page 2 Genul Leel 2300 D Coordinates 50000935 E, 1003563 N National Grid Date Startel 0807/0315 Date Completed 1302/2015 Indiantor Writer Date Startel 1302/2015 Date Completed 1302/2015 Endoator Writer Date Startel 1302/2015 Date Startel	Project No. TA7510 Engineer Atkins Ltd												G16					
	Employer Atkins Ltd Ground Level 2,29m OD Coordinates 54000								9.95 E.	18035	Header Page 2							
UTUE UTUE <th< td=""><td>Date Started</td><td>08/0</td><td>01/2015</td><td></td><td></td><td>Inclinatio</td><td>n Vert</td><td>ical</td><td></td><td></td></th<>	Date Started	08/0	01/2015			Inclinatio	n Vert	ical										
Oute Inm Opp Opp <td>Data</td> <td>Time</td> <td>Links</td> <td>PROGE</td> <td>RESS</td> <td></td> <td>Demarke</td> <td colspan="9">SPT DETAILS</td>	Data	Time	Links	PROGE	RESS		Demarke	SPT DETAILS										
0800/0231 090	Date	time	depth	depth	depth		Kemarks		Depth	туре	penetration in mm	No.	ratio %	depth	depth			
NOTES: All depths in metres, all diameters in millimetres. For details of abbreviations, see key Unchecked Log Print Date And Time: 03/06/2015 16:39:53 -	08/01/2015 09/01/2015 09/01/2015 12/01/2015 12/01/2015 13/01/2015 13/01/2015 13/01/2015 13/01/2015 13/01/2015 13/01/2015 13/01/2015	0730 1800 0730 1800 0730 1800 0730 1800 0730 1800 1800 1800	0.00 11.50 20.00 27.50 27.50 30.30 DEP ks dded to ass dded to ass	TH RELA sist drilling. sist drilling. sist drilling.	NR 9.60 2.50 7.10 2.50 5.46 5.20 NR	Start of Hole End of Shift Start of Shift End of Shift Start of Shift End of Shift End of Hole MARKS			1.20 3.50 5.00 6.90 8.50 13.50 13.50 13.50 23.00 24.50 23.00 24.50 29.00 30.00	SPT SPT SPT(C SPT(C SPT) SPT SPT SPT SPT SPT SPT SPT SPT SPT	N=6 (1.2.2,1,1.2) N=3 (1,0,1,0,1,1) N=2 (1,0,1,0,1,0) N=20 (3,5,5,4,5,6) N=11 (2,4,4,5,4,6) N=21 (2,3,4,5,6,6) N=21 (2,3,4,5,6,7) N=30 (4,5,7,8,9) N=36 (4,6,7,9,10,10) N=37 (4,5,7,9,10,11) SO/135mm (7,10,24,26/60) SO/125mm (2,11,3/60,50) SO/215mm (8,11,13,17,20/65) SO/215mm (8,11,13,17,20/65)	AR538 AR538 AR538 AR538 AR538 AR538 AR538 AR538 AR538 AR538 AR538 AR538 AR538 AR538	% 75 75 75 75 75 75 75 75 75 75 75 75 75	1.20 3.50 4.50 6.40 10.00 12.70 12.70 12.70 12.70 21.50 21.50 21.50 26.00 26.00	DRY DRY DRY 180 5.20 9.90 0 8.90 7.10 0 R NR NR NR NR NR NR NR			
NOTES: All depths in metres, all diameters in millimetres. For details of abbreviations, see key Unchecked Log Print Date And Time: 03/06/2015 16:39:53											* Seating blows only.							
Unchecked Log Print Date And Time: 03/06/2015 16:39:53 - SOIL ENGINEER	NOTES: All For	depths in details of	metres, a f abbrevia	II diamete tions, see	ers in mill key	imetres.					- every stove only.	11.000.000						
	Unchecked			Log Print	Date An	d Time: 03/0	06/2015 16:39:53	2014	×			SOI	Leng	Inee	RING			
Particle Total Conductory Profect Conductory Profeet Conductory Profect Conductory Profect Conductory Profec	Project Name	Silvertown Tunnel					Evo	orato	av L	lole Loo		Hole ID.						
--	---	---	--------------------------------	--	------------------------------	----------------	---------------------------	--	------	-------------------------------	-------------	------------------------------------	-------------------					
Employee Akkins tot Seet 1 of 7 Cound Level 2.25m CD Cound Level 2.25m CD Total State Vertical See Cound Level 2.25m CD Cound Level Vertical Vertical Vertical Mode Cound Cound Cound Level Vertical Cound Level Sec Cound Cound Cound Level In State Total State For Mode Cound Cound Level Sec Cound Cound Cound Level In State Total State Sec Cound Cound Level In State Sec Cound Cound Level	Project No. Engineer	TA7510 Atkins Ltd					cxb	orator	уг	IOIE LOY		G16						
	Employer	Atkins Ltd										Sheet 1 of 7	,					
Description of Strata Legend Description of Strata Legend Description of Strata In Sta Text Description of Strata <td>Ground Level Hole Type</td> <td>2.29m OD IP+CP</td> <td>Coordinates 5 Inclination V</td> <td>540009.95 E, /ertical</td> <td>180356.3</td> <td>3 N Nat</td> <td>ional G</td> <td>rid</td> <td></td> <td></td> <td></td> <td></td> <td></td>	Ground Level Hole Type	2.29m OD IP+CP	Coordinates 5 Inclination V	540009.95 E, /ertical	180356.3	3 N Nat	ional G	rid										
NUMC Clinication Group with fixed cancers. Difference of the server shallow days years of the server shallow days wears of the server shallow days wears we		Description of Strata		Legend	Depth	Datum Level		Sampling		Blow Count A Sample Recove	nd I ery	n Situ Test Details	Install- ation					
Purchased on the server and year opticity coveres and this loss of the server and year opticity coveres and this loss of the server and year opticity coveres and this loss of the server and year opticity coveres and this loss of the server and year opticity coveres and this loss of the server and year opticity coveres and this loss of the server and year opticity coveres and this loss of the server and year opticity coveres and this loss of the server and year opticity coveres and this loss of the server and year opticity coveres and this loss of the server and year opticity coveres and the server	MADE GROUND: Gre	av reinforced concrete		*****	(Thickness)		_	Details	Dia.	TCR SCR RQD			जाला ह					
MACE GROUND Durk haves highly charged with and the laborated fits to carrie and the laborate laborate and the laborate laborate and the laborate and the laborate and the laborate and the laborate laborate and the laborate and the laborate laborate and the laborate laborate and the laborate and the laborate laborate and the laborate laborate and the laborate laborate laborate and the laborate laborate and the laborate laborate and the laborate laborate and the laborate labor	(Hardstanding)				(0.40) -	190		0.00				-						
MOCE GROUND. Dark torown sody very cloyer, outsangular to subscanded fire to coarse. 0.04	MADE GROUND: Dan fine to coarse gravel sized fragments are (Made Ground)	rk brown slightly clayey sandy subangular sized fragments of bricks, concrete and fi fine to coarse.	to subrounded int. Sand		-	1,65	B2 ES3	0.40-0.90 0.40				-						
Mode Specific Dire to come and yeary days usuagular to subnounded fine to transformed within the come. 140 0.89 10 1.40.1 0.99 10 1.00 State date you allow come and yeary days usuagular to subnounded fine to come. 0.49 0.10 1.00 1.00 1.00 State date you allow come of film. Sand is fine to come. 0.49 0.10 1.00	ł.				(1.00) -		D 4 8 5 ES 6 D 7	0.90 0.90-1.20 1.00 1.20-1.65			SP1 1.2	- 						
United with any organity starts and y lightly gravely CLAY Gravel is subargular to assume of files. Said is file to cearse. 130 0.49 0.11 1.80 100	MADE GROUND: Dat coarse gravel sized fi fragments are fine to	rk brown sandy very clayey subangular to ragments of brick, concrete and flint. Sand o coarse.	subrounded fine to I sized		- 1.40 - - (0.40) -	0.89	B 10 D 8 ES 9	1.40-1.80 1.40 1.40			1000	1.65 - - -						
Soft data (prey sindy sightly gravely CLW. Gravel is subangular to without of fine to coarse of flint. Said is line to coarse. Image: Soft data (pre) sind (pre) si	(Made Ground)				1.80 -	0.49	D 11	1.80				1						
Durk grey and black clayey spongy pseudo fibrous PEAT. 0.10 - 0.11 2.55 100 150 3.65 Durk grey and black clayey spongy pseudo fibrous PEAT. 0.10 - 0.11 1.55 3.55 100 150 3.65 Durk grey and black clayey spongy pseudo fibrous PEAT. 0.10 - 0.11 1.55 3.55 100 150 8.55 3.55 100 150 8.55 100 150 8.55 100 150 8.55 100 150 8.55 100 150 8.55 100 150 8.55 100 150 8.55 100 150 8.55 100 150 8.55 100 150 8.55 100 150 8.55 100 150 8.55 100 150 8.55 100 150 100 150 100 150 100 150 100 150 100 150 100 150 100 150 100 150 100 150 100 150 100 150 100 150 100 150 100 <	Soft dark grey silty si subrounded fine to c (Alluvium)	andy slightly gravelly CLAY. Gravel is suba coarse of flint. Sand is fine to coarse.	ngular to		-		UT 12	2.00-2.45	100	10 Blows, 100% Reco	very							
Durk grey and black dayey spongy pseudo fibrous PEAT.					1							6						
Dark grey and black clayey spongy pseudo fibrous PEAT. 0 16 0 17 350-355 100 14 Blows, 100% Recovery Dark grey and black clayey spongy pseudo fibrous PEAT. 0 17 3.50-355 100 14 Blows, 100% Recovery 100 NOTES: All depths in metres, all diameters in millimetres. See header sheet for details of boring, progress and water. For details of abbreviations, see key 100 14 Blows, 100% Recovery 100 14 Blows, 100% Recovery 100 Unchecked Log Print Date And Time: 03/06/2015 16:39:59 - - - - - Inclusion Log Print Date And Time: 03/06/2015 16:39:59 - - - - - - - - Inclusion Log Print Date And Time: 03/06/2015 16:39:59 -							D 13 ES 14 B 15	2.45 2.50 2.50-3.00				1						
Dark grey and black clayey spongy pseudo fibrous PEAT. 3.90 - -1.61 D18 3.90 -					(2.10) _ 		D 16	3.00										
Dark grey and black clayey spongy pseudo fibrous PEAT. Image: Construction of the state o							D 17	3.50-3.95			SPT 3.5	- N=3 -) 3.95 - -						
NOTES: All depths in metres, all diameters in millimetres. See header sheet for details of boring, progress and water. For details of abbreviations, see key UT 20 4.50-4.95 100 14 Blows, 100% Recovery	Dark grey and black (Alluvium)	clayey spongy pseudo fibrous PEAT.			3.90 -	-1.61	D 18 B 19	3.90 3.90-4,40										
NOTES: All depths in metres, all diameters in millimetres. See header sheet for details of boring, progress and water. For details of abbreviations, see key Image: Constraint of the sector of the				44, 46, 47, 47, 47, 47, 47, 47, 47, 47, 47, 47	- - - (1.80) -		UT 20	4.50-4.95	100	14 Blows, 100% Recc	very .							
NOTES: All depths in metres, all diameters in millimetres. See header sheet for details of boring, progress and water. For details of abbreviations, see key Unchecked Log Print Date And Time: 03/06/2015 16:39:59 - Soil engineering Form No. SIEXPHOLELOG Issue.Revision No. 106 Issue Date 20/09/2013 Part of the Bachy Soletanche Group				ale ale ale			D 21	4.95										
Unchecked Log Print Date And Time: 03/06/2015 16:39:59 - SOIL Engineering Form No. SIEXPHOLELOG Issue.Revision No. 1.06 Issue Date 20/09/2013 Part of the Bachy Soletanche Group	NOTES: All depth See head For detail	s in metres, all diameters in millimet er sheet for details of boring, progre s of abbreviations, see key	res. ess and water.										1.111					
	Unchecked Form No. SIEXPHOLELOG	Log Print Date An Issue.Revision No.	d Time: 03/06/201	15 16:39:59 Issue Date 20/09	- /2013					S	Part of the	IGINEE Bachy Soletanch	e Group					

Project Name Silvertown Tunnel				Exp	lorato	ry F	lole	Lo	g	Ho	le l
Project No. TA7510 Engineer Atkins Ltd										C	<u>3</u> 1(
Employer Atkins Ltd		_				_	_			Shee	et 2
Ground Level 2.29m OD Coordinates Hole Type IP+CP Inclination	540009.95 E Vertical	, 180356.3	33 N Nati	onal G	rid						
Description of Strata	Legend	Depth	Datum Level		Sampling		Blov Sam	w Cou ple R	unt And ecovery	In Sit	u To tails
	ale. atc. atc.	(Thickness)	20101	D 22	Details	Dia.	TCR	SCR	RQD	SPT N=2	
Dark grey and black clayey spongy pseudo fibrous PEAT. (Alluvium)	alter aller alter			B 23 UT 24	5.00-5.50	100	20 Blov	vs, 100	% Recovery	5.00	5
Firm dark grey slightly sandy silty CLAY. Sand is fine to coarse. (Alluvium)			-3.41	D 25	5.95						
		(0.90)		UT 26	6.40-6.85	100	25 Blov	ws 100	1% Recover		
		- 660 -	-4 31	W 29	6.60	1920			00-01-07.00 FL		
Medium dense grey slightly clayey sandy subangular to subrounded fine to coarse GRAVEL of flint. Sand is fine to coarse. (River terrace deposits)		-									
		-		D 27 D 28 B 30	6.85 6.90-7.35 6.90-7.40					SPT N=20 6.90	7
				B 32 D 31	7.90-8.40 8.00						
										SPT(C) N=	18
		(4.10) 		8 33 D 34	8.90-9.40 9.00						8
				B 35	9.90-10.40		1				
NOTES: All depths in metres, all diameters in millimetres. See header sheet for details of boring, progress and water. For details of abbreviations, see key											
	015 16.40.00								- 60	IL POG	in

Project Name	Silvertown Tunnel					Evp	lorato			Hole ID.	
Project No.	TA7510					схр	Iorator	уг	TOTE LOG	G16	
Engineer Employer	Atkins Ltd									Sheet 3 of 7	,
Ground Level	2.29m OD	Coordinates	540009.95 E,	180356.3	3 N Nat	ional G	rid				
Hole Type	IP+CP	Inclination	Vertical								
	Description of Strata		Legend	Depth	Datum		Sampling		Blow Count And Sample Recover	In Situ Test	Install-
	• • • • • • • • • • • • • • • • • • •			(Thickness)	Level		Details	Dia.	TCR SCR RQD	Details	ation
Remaining Detail : 9.9 clayey very sandy	90m - 10.00m : from 9.90m to 10.00m k	ocally slightly				D 36	10.00			SPT(C) N=21 10.00 10.45 -	
				-	0.74	0.47	10.20			1	
Stiff fissured thinly la spaced randomly orie	minated dark grey sandy CLAY. Fissures entated. Sand is fine to coarse.	are closely		10.70 -	-8.41	037	10.70			-	
(London Clay Formati from 10.70n	ion) n to 11.45m firm silty sandy gravelly clay	,		-		1000210107					
from 10.70n	n to 11.45m with occasional rootlets			-		UT 38	11.00-11.45	100	60 Blows, 100% Recove	ry	
				÷		D 39 B 40	11.45 11.50-12.00				
				1 5						, in the second s	
				-							
				-		D 41 B 42	12.00-12.45 12.00-12.50			SPT N=18	
				4						-	
										y	
				-		UT 43	12.50-12.95	100	60 Blows, 100% Recove	ry -	
				Ē.							
				5						-	
				1		D 44	12.95				
2			197			UT 45	13.00-13.45	100	60 Blows, 100% Recove	ry -	
				-						1	
				4		D 46 D 47	13.45 13.50-13.95			SPT N=21 -	
						D 40	13.50*14.00			13.95 -	
				i n		117.40		100	60 Plane 100% Passar		
				-		0149	14.00-14.45	100	60 Blows, 100% Recove	'y -	
				-		12-12-12-12-12-12-12-12-12-12-12-12-12-1	1 10000				
				-		D 50 UT 51	14.45 14.50-14.95	100	60 Blows, 100% Recove	ry -	
				10 (A							
				-		D 52	14.95				
		1000							1	- Second	
NOTES: All depths See head For detail	s in metres, all diameters in millime er sheet for details of boring, prog s of abbreviations, see key	tres. ress and water.									5.15
Unchecked	Log Print Date A	nd Time: 03/06/2	2015 16:40:02	-					s	DIL enginee	RING
Form No. SIEXPHOLELOG	Issue.Revision No	1.06	Issue Date 20/09	2013	_		_	_	1	rart of the Bachy Soletanch	ie Group

Project Name	Silvertown Tunnel					Evn	lorato	a. L		Hole II	D.
Project No. Engineer	TA7510 Atkins Ltd					Ехр	iorator	уг	tole Log	G16	
Employer	Atkins Ltd							_		Sheet 4 o	f 7
Ground Level Hole Type	2.29m OD IP+CP	Coordinates Inclination	540009.95 E, Vertical	180356.3	3 N Nat	tional G	rid				
1	Description of Strata		Legend	Depth	Datum	1	Sampling		Blow Count An Sample Recover	d In Situ Tes	t Install-
	Description of Strata		Logena	(Thickness)	Level		Details	Dia.	TCR SCR RQD	Details	ation
Stiff fissured thinly la spaced randomly oria (London Clay Format	aminated dark grey sandy CLAY. Fissures entated. Sand is fine to coarse. ion)	are closely				D 53 B 54 ES 55 UT 56	15.00-15.45 15.00-15.50 15.50 15.50-15.95	100	70 Blows, 100% Recov	SPT N-21 15.00 15.4	5
				21 21 21 21 21 21 21 21 21 21 21 21 21 2		D 57 UT 58	15.95 16.00-16.45	100	70 Blows, 100% Recov	ery	
				- - (12.00) -		D 59 D 60 B 61	16.45 16.50-16.95 16.50-17.00			SPT N=22 16.50 16.9	
						UT 62	17.00-17.45	100	75 Blows, 100% Recov	ery	
						D 63 UT 64	17.45 17.50-17.95	100	75 Blows, 100% Recov	ery	
						D 65 D 66 B 67	17.95 18.00-18.45 18.00-18.50			SPT N=30 18.00 18.4	5
						UT 68	18.50-18.95	100	75 Blows, 100% Recov	ery	
						D 69 UT 70	18.95 19.00-19.45	100	75 Blows, 100% Recov	ery	
		i.				D 71 D 72 B 73	19.45 19.50-20.00 19.50-20.00			SPT N-36 19.50 19.9	
NOTES: All depth See head For detail	s in metres, all diameters in millime ler sheet for details of boring, prog ls of abbreviations, see key	tres. ess and water. nd Time: 03/06/	2015 16%0*04						s		ering
Form No. SIEXPHOLELOG	Issue,Revision No.	1.06	Issue Date 20/09	/2013			_			Part of the Bachy Soleta	nche Group

Project Name	Silvertown Tunnel					Evn	lorator	γL		Hole ID	
Project No. Engineer	TA7510 Atkins Ltd					схþ	Iorator	уг	IOLE LOG	G16	
Employer	Atkins Ltd	CS7 185	C MEMORY DENAM	o to governo e			335			Sheet 5 of	7
Ground Level Hole Type	2.29m OD IP+CP	Coordinates Inclination	540009.95 E, Vertical	180356.3	3 N Nat	ional G	rid				
	Description of Strata		Lecond	Depth	Datum		Sampling		Blow Count And	d In Situ Test	Install
	Description of Strata		Legend	(Thickness)	Level		Details	Dia.	TCR SCR RQD	Details	ation
Stiff fissured thinly la spaced randomly ori (London Clay Formati	iminated dark grey sandy CLAY. Fissu entated. Sand is fine to coarse. ion)	ires are closely		-		B 74 UT 75 D 76	20.00-20.50 20.50-20.95 20.95	100	100 Blows, 100% Recov	very	
				-		UT 77	21.00-21.35	100	100 Blows, 100% Recov	ery -	
				1 1		D 78	21.35			1	
				-		D 79 B 80	21.50-21.95 21.50-22.00			SPT N=37 21.50 21.95	
			fondaloga Fondaloga Fondaloga			UT 81	22.00-22.45	100	100 Blows, 100% Recov		
Mauri dana amin' alar		fina to conver		- - 22.70 -	-20.41	D 82 UT 83	22.45 22.50-22.95	100	100 Blows, 100% Recov	very	
GRAVEL of flint. With (Harwich Formation)	ey sandy subangular to subrounded a abundant shell fragments. Sand is fi	ine to coarse				D 84 B 85	22.95 23.00-23.50			SPT 50/135mm - 23.00 23.29	
				- - - - (2.90) - -		D 86	24.00			-	
from 24.50r	n to 25.00m locally very clayey grave	lly sand				B 87	24.50-25.00			SPT 50/125mm 24.50 24.76	
NOTES: All depth: See head For detail Unchecked Form No. SIEXPHOLELOG	s in metres, all diameters in mill ler sheet for details of boring, p ls of abbreviations, see key Log Print Dat isue.Revisio	imetres, rogress and water, e And Time: 03/06/; 1 No. 106	2015 16:40:05 Issue Date 20/09	- /2013					50		PRING

Project Name					Explo	ratory	/ Ho	ole L	og	Hole IL
Engineer	Atkins Ltd									G16
Employer Ground Level	Atkins Ltd	540009 95 F	180356 3	3 N Nati	ional Grid		-			Sheet 6 o
Hole Type	IP+CP Inclination	Vertical	100000.0			743				A. 2010. 194
	Description of Strata	Legend	Depth (Thickness)	Datum Level	Sa	ampling tails I	Dia. To	Blow C Sample	ount And Recovery	In Situ Tes — Details
Very dense grey cl GRAVEL of flint. Wi (Harwich Formation)	ayey sandy subangular to subrounded fine to coarse th abundant shell fragments. Sand is fine to coarse. n)		-							
Very dense dark gr (Woolwich Formati	ey clayey to very clayey fine to medium SAND. on - Laminated Beds)		25.60 - - - - - - -	-23.31	D 88 2 D 89 2 B 90 2	5.60 6.00-26.30 6.00-26.50				SPT 50/90mm 26.00 26.24
		en e			D 91 2	7.00				
below 27.	50m sand is dark brown		 (3.20) - - -		D 92 2 B 93 2	7.50-27.80 7.50-28.00				SPT 50/75mm 27.50 27.70
			-	1	094 2	8.50				
Very stiff dark grey to pinkish orange s (Woolwich Formati	sandy CLAY with abundant up to 2mm x 2mm x 3mm white hell fragments. Sand is fine to coarse. on – Lower Shelly Beds)		28.80 	-26.51	D 95 2 D 96 2 B 97 2	8.80 9.00-29.30 9.00-29.50				SPT 50/225mm 29.00 29.31
			(1.50) - - - -		D 98 2	9.50				
NOTES: All dept	hs in metres, all diameters in millimetres.					1	389			
See hea For deta	der sheet for details of boring, progress and water. ails of abbreviations, see key								- 50	
Unchecked	Log Print Date And Time: 03/06/20	15 16:40:06	-				_		SU	t of the Bachy Soletar

Project Name	Silvertown Tunnel					Evplorato	v F	lole Loc		Hole ID.	
Project No.	TA7510					LAPIOIALOI	yr	iole Log		G16	
Engineer Employer	Atkins Ltd Atkins Ltd									Sheet 7 of 7	£
Ground Level	2.29m OD	Coordinates 540	009.95 E,	180356.3	3 N Nat	tional Grid					
Hole Type	IP+CP	Inclination Ver	tical								
	Description of Strata		Legend	Depth	Datum	n Sampling		Blow Count Sample Reco	And very	In Situ Test	Install-
	· · · · · · · · · · · · · · · · · · ·			(Thickness)	Level	Details	Dia.	TCR SCR RQD		Details	ation
Very stiff dark grey s to pinkish orange sho (Woolwich Formation	andy CLAY with abundant up to 2mm x 2r ell fragments. Sand is fine to coarse. n - Lower Shelly Beds)	nm x 3mm white		-		D 99 30.00-30.30				SPT 50/215mm 30.00 30.37 -	
Exploratory hole com	plete at 30.30 m.			30,30 -	-28.01					27 17	
				-						7	
				-						-	
				-						-	
				1 <u>11</u>						-	
				2						1	
				-							
				5							
				-						in T	
				-						-	
				_						-	
				-						-	
				2						2	
				-						-	
				5							
										5	
				-						-	
				-						-	
				-						-	
				4		11				-	
				-							
				5						10- 10-	
				-							
				8						-	
				-						-	
				4						-	
				7 <u>2</u> 12						7 <u>0</u>	
				-						2	
			×								
NOTES AN LOS	- 1 II										
NOTES: All depth See head For detai	s in metres, all diameters in milliner ler sheet for details of boring, progr ls of abbreviations, see key	ess and water.									
Unchecked Form No. SIEXPHOLELOG	Log Print Date An Issue.Revision No.	d Time: 03/06/2015	16:40:07	- 9/2013					SOIL Part o	f the Bachy Soletanch	e Group

Project Na	ame Silv	rtown	Tunnel	1					Exploratory Hole Lo	0	Ho	ole ID.	
Project No	о. ТА7	510								9	C	615	
Engineer Employer	Atk Atk	ins Ltd ins Ltd									Head	er Page 2	2
Ground Leve	el 2.68	m OD			Coordinates	53989	1.47 E,	18026	6.12 N National Grid		(29 4		
Date Started	16/0	51/2015	PROG	DESS	Date Completed	02/02/	/2015			s vert	cal		
Date	Time	Hole depth	Casing depth	Water depth	Remarks		Depth	Туре	Incremental blow count / penetration in mm	Hammer No.	Energy ratio %	Casing depth	Water depth
19/01/2015 19/01/2015 20/01/2015 26/01/2015 27/01/2015 28/01/2015 29/01/2015 29/01/2015 30/01/2015 30/01/2015 02/02/2015 02/02/2015	0700 1800 0730 1800 0700 1800 0700 1800 0700 0700 1800	0.00 10.00 13.60 13.60 13.60 17.60 25.40 30.00 33.00 33.00 36.00	0.00 10.00 12.00 12.00 12.00 12.00 17.60 25.40 30.00 33.00 33.00 36.00	DRY 3.00 1.60 4.80 2.20 2.80 3.00 0.50 2.30 1.80 1.80 1.30 4.00 2.00	Start of Hole End of Shift Start of Shift End of CP Start of Shift End of Shift Start of Shift End of Shift Start of Shift End of Shift End of Shift End of Shift End of Shift End of Shift End of Hole		7.70 9.50 11.00 13.10 14.60 15.60 19.10 20.60 22.40 23.40 24.40 25.40 26.20 27.20 28.70 30.00 31.60 33.00 34.60	SPT SPT(C SPT(C SPT) SPT SPT SPT SPT SPT SPT SPT SPT SPT SPT	N=14 (1,1,2,3,4,5) N=20 (2,2,3,5,6,6) N=18 (2,2,3,4,5,6) N=22 (3,3,5,5,6,6) N=42 (4,6,7,9,11,15) N=34 (4,6,7,6,9,12) N=51 (6,10,10,12,13,16) N=51 (6,10,10,12,13,16) N=51 (6,10,10,12,13,16) S0/230mm (7,10,11,13,13,13/65) S0/235mm (8,17/35,50/25) S0/25mm (8,17/35,50/25) S0/25mm (12,15/20,22,8/15) S0/165mm (10,15,20,22,8/15) S0/125mm (10,15,20,22,8/15) S0/125mm (11,15,20,22,8/15) S0/125mm (11,15,20,22,8/15) S0/125mm (11,15,20,22,8/15) S0/125mm (11,15,20,22,8/15) S0/125mm (11,15,20,22,8/15) S0/125mm (11,15,20,22,8/15) S0/125mm (11,15,20,22,8/15) S0/120mm (10,15/55,30,20/45)	AR538 AR538 AR538 CD61 CD61 CD61 CD61 CD61 CD61 CD61 CD61	% 75 75 75 58 58 58 58 58 58 58 58 58 58 58 58 58	7.60 9.50 NR 12.00 14.60 15.60 16.60 22.40 23.40 24.40 25.40 26.20 27.20 28.70 31.60 31.60 31.60 34.60	DRY 2.10 NR 4.80 3.00 2.30 3.00 2.00 2.00 2.00 2.00 1.80 1.80 1.80 1.80
Top Base	Pemar	DEP	TH RELA	TED REN	/ARKS								
Depth Dept	h Watar	ddad to acc	let delline										
7:90 10.00 10.00 10.00 12:00 12:00	0 Water a D Falling P Falling t	dded to ass nead test, lead test, G	ENERAL	NOTES					* Seating blows only.				
NOTES: All For	depths in details of	metres, a abbrevia	ll diamete tions, see	ers in milli key	imetres.				Justicity and was only.		1		
Unchecked			Log Print	Date And	Time: 03/06/2015 16:47:13		-			sol	Leng	ineer	RING
Form No. SIEXPHO	OLEHDR P2		Issue.R	evision No. 1.	06 Issue Date 12/03/2	2014				Part	of the Bachy	Soletanche ç	roup

Project Name	Silvertown Tunnel					Evplorato	n l	Jol		0		Hole ID.	
Project No. Engineer	TA7510 Atkins Ltd					Explorato	i y i	101	e Lu	g		G15	
Employer	Atkins Ltd			_								Sheet 1 of 8	
Ground Level Hole Type	2.68m OD TP+CP+RC	Coordinates 5 Inclination V	39891.47 E, /ertical	180266.1	2 N Nat	tional Grid							
	Description of Strata		Legend	Depth	Datum	Sampling		B Sa	ow Co mple F	ount a	And verv	In Situ Test	Install-
				(Thickness)	Level	Details	Dia.	TCR	SCR	RQD	IF	Details	ation
MADE GROUND: Rei (Hardstanding)	nforced concrete layers.			(0.50) - -									
MADE GROUND: Rec coarse gravel sized fr cobble content. Cob brick. Sand sized fraç (Made Ground)	Idish brown clayey sandy subangular to su ragments of brick, concrete and flint. With ble sized fragments are subangular to sub ments are fine to coarse.	ibrounded fine to high rounded of		(0.50)	2.18								
Soft dark grey silty C (Alluvium)	LAY with abundant red stained rootlets.			1.00	1.68								
			1월 1	- - - - - - - - - - - - - - - - - - -									
Firm dark brown clay (Alluvium)	rey sandy fibrous PEAT. Sand is fine to coa	rse.	and a star	4.00 — - (0.50) — - 4.50 —	-1.32	D1 450						and the second se	
Firm dark grey/black (Alluvium)	r pseudo fibrous PEAT.			-	- 4:196	B 2 4.50-5.00							
NOTES: All depths See head For detail	s in metres, all diameters in millimet er sheet for details of boring, progre s of abbreviations, see key	res. ss and water.									20102		
Unchecked	Log Print Date And	d Time: 03/06/201	.5 16:47:17	-					_		SOIL	. enginee	Group
FORM NO. STEAPHOLELOG	Issue.Revision No. 3	in the second se	sade Date 20/09	AVA.							ratto	a the bachy soletanche	aroup

Project Name	Silvertown Tunnel					Eve	orato	ny l		Hole ID.	
Project No. Engineer	TA7510 Atkins Ltd					схр	orato	i y r	IOIE LOG	G15	
Employer	Atkins Ltd	<u></u>			000000	7 N/189	27			Sheet 2 of	8
Ground Level Hole Type	2.68m OD TP+CP+RC	Coordinates 5 Inclination V	39891.47 E, ertical	180266.1	z N Nat	ional G	rid				
	Description of Strata		Legend	Depth	Datum		Sampling		Blow Count A Sample Recove	nd In Situ Test	Install-
				(Thickness)	Level	1	Details	Dia.	TCR SCR RQD	IF Details	ation
Firm dark grey/black (Alluvium)	pseudo fibrous PEAT.		مالار ماله ماله مراجع ماله ماله ماله ماله ماله ماله ماله ماله ماله ماله ماله	(1.20) -		UT 3 8 5	5.00-5.45 5.00-5.50	100	15 Blows, 100% Recc	very	
below 5.45n	n clayey		ب معادر معادر م معادر معادر معادر معادر معادر معادر معادر معادر م معادر معادر معادر معادر معادر معادر			D4	5.45				
Soft dark grey moder (Alluvium)	ately organic CLAY.		34. 34. 34.	5.70 -	-3.02	D 7 B 8	5.70 5.70-6.00				
				-		UT 9	6.00-6.45	100	15 Blows, 100% Reco	very	
				1 1 1		D 10	6.45				
				- - (2.20) -		B 11	6.70-7.20				
7				-		an ou status					
						D 12 D 13	7.50 7.70-8.15			SPT N=14 7.70	
Medium dense dark fine to coarse GRAVE (River terrace denosit	grey brown and white very sandy subang L of flint. Sand is fine to coarse.	ular to rounded		7.90 -	-5.22	W 14 D 15 B 16	7.90 7.90 7.90-8.40			8.15 -	
(niver terrace deposit	aj			-							
						D 17 B 18	8.90 8.90-9.40				
				-						- SPT(C) N=20 9.50 9.95 -	
				(3.70)		D 19	9.90				
		.126-2								Stevensor	
NOTES: All depths See heade For details Unchecked	in metres, all diameters in millimeters er sheet for details of boring, progre s of abbreviations, see key Log Print Date And	res. ess and water. d Time: 03/06/201!	5 16:47:19						s		RING
Form No. SIEXPHOLELOG	Issue.Revision No. 1	1.06 It	ssue Date 20/09/	2013	_	_		_		Part of the Bachy Soletanch	e Group

Project Name	Silvertown Tunnel					Evo	lorato					Hole	ID.	
Project No. Engineer	TA7510 Atkins Ltd					схр	iorator	уг	TOTE	LO	9	G1	15	
Employer	Atkins Ltd							-	_			Sheet	3 of 8	_
Ground Level Hole Type	2.68m OD TP+CP+RC	Coordinates Inclination	539891.47 E, Vertical	180266.1	2 N Nat	tional G	rid							
	Description of Strata		Legend	Depth	Datum	۱	Sampling		Blo Sam	w Cou ple Re	nt And covery	In Situ	Test	Install
	1. 			(Thickness)	Level		Details	Dia.	TCR	SCR	RQD IF	Detai	IS	ation
Medium dense dark fine to coarse GRAVE (River terrace deposi	: grey brown and white very sandy sul EL of flint. Sand is fine to coarse. its)	bangular to rounded				B 20	10.00-10.50							
from 11.00r flint	m to 11.50m with occasional grey sub	rounded cobbles of				D 21 B 22	10.90 11.00-11.50					SPT(C) N=18 11.00		
Stiff grey sandy CLA (London Clay Format	Y. Sand is fine to medium. tion)				-8.92	D 23 B 24 UT 25	11.60 11.60-12.00 12.00-12.45	100	70 Blov	vs, 1009	% Recovery			
at 12.45m v	with a vertically orientated planar smo	oth fissure		(1.35)		D 26 UT 27	12.45 12.50-12.95	100	70 Blov	vs, 100	% Recovery			
Stiff thinly laminated randomly orientated (London Clay Format	l grey sandy CLAY. Fissures are closel planar smooth. Sand is fine to coarse tion)	y spaced		12.95 	-10.27	D 28 D 29 B 30	12.95 13.10-13.55 13.10-13.60		ť			SPT N=22 13.10		
Stiff fissured greyish white silt tubes (<1n 70-90 degrees and spaced planar locally (London Clay Format from 13.60) from 14.43r closely to ve	n brown slightly sandy silty micaceous nm x 4mm). Sand is fine. Fissures are 0-20 degrees very closely locally extra v curved smooth locally polished. tion) m to 14.00m drilling disturbed. Recov m to 14.75m indistinctly fissured sand ery closely spaced predominantly 0-2	CLAY with some predominantly emely closely ered as soft to firm ly with extremely 0 degrees. Fine to		13.60 - - - - - - - - - - - - - - - - - - -	-10.92	13	.60 14.60	102	100	NA	NA			
coarse grav from 14.60	el sized lenses of brown silty fine san m to 14.77m assumed zone of core lo	d				-0.32-			à	14	New Second	SPT N=42 14.60		
NOTES: All depth See head	s in metres, all diameters in milli ler sheet for details of boring, pr Is of abbreviations, see key	metres. ogress and water.												
Unchecked	Log Print Date	e And Time: 03/06/2	2015 16:47:21	2	_	_		_			so	Lengi	nee	RING
Form No. SIEXPHOLELOO	a Issue,Revision	No. 1.06	Issue Date 20/09	/2013				_		_	Part	t of the Bachy So	letanche	Group

Project Name	e Silvertown Tunnel					Explor	ator	vF	lole		0		Ho	le ID.	
Project No. Engineer	TA7510 Atkins Ltd					Lypion	ator	уr	1016		'y		G	15	
Employer	Atkins Ltd	and the second	an anna an								_		Sheet	t 4 of 8	
Ground Level Hole Type	2.68m OD TP+CP+RC	Coordinates Inclination	539891.47 E, Vertical	180266.1	L2 N Nat	tional Grid									
	Description of Strata		Legend	Depth	Datum	n Sar	npling		Blo Sarr	w Co ple R	unt A Recove	nd ery	In Situ	i Test	Install-
				(Thickness)	Lever	Deta	ails	Dia.	TCR	SCR	RQD	IF	Dett		u karrer
Stiff fissured greyi white silt tubes (< 70-90 degrees as spaced planar loca (London Clay Form at 15.25m from 15.6 x 6mm) from 15.7 from 15.7	sh brown slightly sandy silty micaceous Imm x 4mm). Sand is fine. Fissures are p d 0-20 degrees very closely locally extre ly curved smooth locally polished. ation) 1 No coarse gravel sized pocket of darl 0m to 15.75m assumed zone of core los 0m to 16.05m locally with frequent whi 5m to 16.08m drilling disturbed. Recove 7m to 17.32m sandy 1 No fragment of black lignite (2mm x 4m to 16.60m assumed zone of core los 0m to 17.05m very stiff with a few grav	CLAY with some predominantly emely closely k grey sandy clay s te silt tubes (<1mm ered as soft to firm 6mm) is el sized fragments	મિત્ર મિત્ર મિત્ર મિત્ર મિત્ મિત્ર મિત્ર મિત	- - - - - - - - - - - - - - - - - - -		14.60 	15.60 90-16.05 16.20 16.60	102 102 102	83	NA NA NA	NA NA		SPT N=34 15.60 SPT N=47 16.60	16.05	
of black li	gnite («Smm x 2mm)	second 0. 10 degraps				16.60	17.60	102	90	NA	NA			17.05	
from 1/.3 lenses of Stiff fissured greyi tubes (<1mm x 5m degrees planar loc (London Clay Form from 18.1 polished	2m to 17.50m with extremely closely sp light grey silt (<6mm x 20mm) sh brown silty micaceous CLAY with rar im). Fissures are predominantly 70-90 d ally curved smooth locally polished. ation) 2m to 18.47m fissures randomly orienta 0m to 19.25m assumed zone of core los	e white silt legrees and 0-20 ated frequently		17.60 - - - - - - - - - - - - - - - - - - -	-14.92	- D 38 17. 17.60 - D 41 19.	19.10 19.10	102	97	NA	NA		SPT N=51 17.60 SPT N=37 19.10		
from 19.1 18mm) from 19.7	Om with a few burrow, infilled with ligh	t grey silt (2mm x planar smooth highly				19.10	19.70	102	75	NA	NA			19.55	
NOTES: All dep See he	ths in metres, all diameters in milli ader sheet for details of boring, pr	metres. ogress and water.	Production Produc										4		
For det Unchecked	ails of abbreviations, see key Log Print Date	And Time: 03/06/2	2015 16:47:23								\$	5011	engi	nee	RING
Form No. SIEXPHOLEL	og Issue.Revision	140. 1.06	Issue Date 20/09	v 2013				_	_	_		rarte	/ the bachy s	soretanche	aroup

Project Name	Silvertown Tunnel					Evalore	ston						Hole ID.		
Project No.	TA7510					Explora	atory	уг	1016	e LC	g		G15		
Engineer	Atkins Ltd Atkins Ltd												Sheet 5 of 8	2	
Ground Loval	268m OD	Coordinates 53	0901 47 E	190266 1	2 NI Niat	tional Grid		-				-	Sheet 5 of t		-
Hole Type	TP+CP+RC	Inclination Ve	ertical	180200.1	2 14 144	donar driu									
	Description of Strata		Legend	Depth	Datum	n Sam	pling		Blo San	ow Co	ount A Recov	und erv	In Situ Test	In	stall-
	Desemption of ottata			(Thickness)	Level	Detai	ls I	Dia.	TCR	SCR	RQD	IF	Details	ati	ion
Remaining Detail : 1	9.78m - 19.85m : polished		X			0.0100			1283	2005	1000			11111	
Stiff to very stiff fiss	ured arevish brown slightly sandy to sandy	ility	Xan Angel	20.20 -	-17.52	19.70	20.60	102	77	NA	NA				
micaceous CLAY with	h rare white silt tubes (<1mm x 8mm) and ra	are burrow	× 2										-	Const.	
Fissures are predom	inantly 70-90 degrees and 0-20 degrees pl	anar locally	X X											1121	
sand on fissure surfa	ices.	ynne	x	-		-D-43-20:60	-21.04	_					SPT 50/290mm -	altra St	
(London Clay Format from 20.60r	ion) n to 20.81m assumed zone of core loss												20.60 21.04 -	Solution in	
			X	-										10.00	
			XX											1112	
			× × ×	1 (F		20.60	21.60	102	79	NA	NA			165 11	
			X	-									÷	Ser.	
			X	(2.14) -										11111	
			XX											11212	
			X	-				_			-			11111	
from 21 72	n to 22.24m with some locally frequent whi	to cilt tuboc	XXX	-									-	115.00	
(<1mm x 8r	nm)	të ant tubea	x											10.00	
at 21.90m 1	No coarse gravel sized fragment of black li	gnite	XXX			21.60	22.40	102	100	NA	NA		4	1011	
			XX	i 4		100000	-0130/200		10000	(81)3	0.000			102.01	
			X X 2	1										Course of	
from 22.28r	n to 22.31m 1 No very thin bed of weak ligh	nt brown	×	22.34	-19.66	D 46 22 40	-22 69		_	_			- SPT 50/135mm -	11211	
Very stiff grey CLAY	with abundant fine to coarse gravel sized w	hite		-	-49.74	0.00							22.40 22.69 -	1000	
shells and shell frage (Harwich Formation)	ments.		355										-	112-211	
Assumed zone of co	re loss. (Logged from SPT) Firm to stiff brow	n CLAY with		(0.70)										11112	
some fine to coarse greyish brown fine t	gravel sized fragments of white shell and po o medium sand.	ockets of		-		22.40	23.40	102	30	NA	NA			111111	
(Harwich Formation)			372	9 H									÷.	11211	
Firm to stiff grey CL/	AY with a few fine to coarse gravel sized wh	ite	- 523	23.10 -	-20.42									a series	
shells and shell frag (Harwich Formation)	ments.		222	(0.27)									-	11211	
from 23.10 from 23.15	n to 23.15m drilling disturbed. Recovered a n very sandy with frequent fine to coarse gi	s soft to firm ravel sized	1.18/18	23.37 _	-20.69	D 48 23.40	0-23.55	-	_				SPT 50/20mm -	CH12	
white shells	and shell fragments			23.55	-20.87								23.40 23.55 -	11214	
Very dense greyish I (Woolwich Formation	prown slightly silty fine to medium SAND. n - Laminated Beds)			-	ORIGANA.									10101	
Assumed zone of co	re loss.			(0.45) _									-	111111	
(woolwich Formatio	n - Laminated Bedsj		100	8		23.40	24.40	102	40	NA	NA			2442	
Very dense greyish l	prown silty fine to medium SAND.		XXXX	24.00-	-21.32								-	1111	
(Woolwich Formatio	n – Laminated Beds)		X 2 2	-										111201	
				(0.55) _										1000	
			X	-		-D 50-24.40	9-24.55		·	_			SPT 50/25mm - 24.40	inter a	
Assumed zone of co	re loss.		N/1235/0	24.55	-21.87								24.55 -	1100	
(Woolwich Formatio	n – Laminated Beds)			÷									4	111211	
				(0.53) -				222		20	195				
						24.40	z5.40	102	32	NA	NA			1212	
NOTES: All depth	s in metres, all diameters in millimetre	25.			-			-	_						
See head For detai	er sneet for details of boring, progres Is of abbreviations, see key	ss and water.													
Unchecked	Log Print Date And	Time: 03/06/201	5 16:47:27	÷							!	501	enginee	R	ING
Form No. SIEXPHOLELOO	3 Issue.Revision No. 1.	06 Is	sue Date 20/09	/2013								Part	of the Bachy Soletanch	ie Gr	oup

Project Name	Silvertown Tunnel					Evolor	ator	vF	اما		0		Hole ID.	
Project No. Engineer	TA7510 Atkins Ltd					слрю	ator	y ı	1010		g		G15	
Employer	Atkins Ltd	124 T 12 T 3			Statesti i					_		_	Sheet 6 of 8	3
Ground Level Hole Type	2.68m OD TP+CP+RC	Coordinates ! Inclination	539891.47 E, /ertical	180266.1	2 N Nat	ional Grid								
	Description of Strata		Legend	Depth	Datum	San	npling		Blo San	ow Co nple I	ount A Recove	nd ery	In Situ Test	Install-
				(Thickness)	Level	Deta	ils	Dia.	TCR	SCR	RQD	IF	Details	ation
Assumed zone of cor (Woolwich Formation	e loss. 1 - Laminated Beds)			25.08 _	-22.40								-	
Brownish grey locally (Woolwich Formation	/ brown silty fine to medium SAND. - Laminated Beds)			(0.46)		-D 51-25.4	i 0-25.54 -						- 	
Assumed zone of cor (Woolwich Formation	e loss. - Laminated Beds)			25.54 -	-22.86								-	
				(0.60) – –		25.40	26.20	102	8	NA	NA			
			a stand	2614	22/6							NA		
Very dense greyish b SAND. Gravel is subro brown). Weelwich Formation	rown slightly gravelly slightly silty fine to ounded fine of multicoloured flint (white,	medium grey and		26.30 -	-23.62	-D-52-26.2	0-26.30						SPT 100/40mm - 26.20 26.30 -	
Assumed zone of cor (Woolwich Formation	e loss. - Laminated Beds)													
				(0.84) - - -		26.20	27.20	102	6	NA	NA			
													1	
Very dense greyish b (Woolwich Formation at 27.14m 1 flint	rown slightly silty fine to medium SAND. Laminated Beds) No subrounded coarse gravel sized fragr	nent of black		27.14 -	-24.46	-D-53-27.2	:0-27:40			2004	-		SPT 75/50mm - 27.20 27.40 -	
from 27.20n Firm thinly to thickly closely to very closely	n to 27.33m assumed zone of core loss laminated greyish brown silty CLAY with / spaced thin and thick laminations of bro	extremely wwn silty		- (0.64)		27.20	27.70	102	90	NA	NA		1	
fine sand. (Woolwich Formation from 27.60n	- Laminated Beds) n to 27.70m 1 No thin bed of light brown	silty fine sand				raitago:	1940	.5345	10-1	76951	5985			
Stiff to very stiff thin! CLAY with some loca and shell fragments.	y laminated fissured brownish grey and o ly frequent fine to coarse gravel sized wi Fissures are randomly orientated very clo	lark grey hite shells sely spaced		27.97	-25.29	27.70	28.20	102	79	NA	NA			
Woolwich Formation from 27.97n from 28.08n	smooth rarely poisnes predominantly /C – Lower Shelly Beds) n to 28.08m very sandy n to 28.32m assumed zone of core loss	- 90 degrees.		-		28.20	28.70	102	79	NA	NA			
						-D 56	0-29.03							
from 28.70n	n to 28.84m assumed zone of core loss			-		19405-1827 - 18678 1	an States						28.70 29.03 -	
from 29.00n	n to 29.70m hard light brown clay			-		28.70	29.30	102	92	NA	NA		-	
from 29.18n	n locally with abundant shells and shelly f	ragments	200	(2.83) _						-			-	
from 29.58n claystone at 29.62m 1 from 29.66n	n to 29.61m 1 No very thin bed of weak li No coarse gravel sized articulated bivalv n with occasional fine to coarse gravel siz	ght brown s ed brown shell				29.30	30.00	102	96	NA	NA			
NOTES: All depths See head For detail Unchecked	OTES: All depths in metres, all diameters in millimetres. See header sheet for details of boring, progress and water. For details of abbreviations, see key nchecked Log Print Date And Time: 03/06/2015 16:47:30 -													
Form No. SIEXPHOLELOG	Issue.Revision No.	1.06	Issue Date 20/09	/2013		_		_		_		Part o	of the Bachy Soletanch	e Group

Project Name	Silvertown Tunnel					Evolorat		Hol		20		Hole ID.	
Project No. Engineer	TA7510 Atkins Ltd					Explorate	луі	101	5 L(Jy		G15	
Employer	Atkins Ltd						_					Sheet 7 of 8	
Ground Level Hole Type	2.68m OD TP+CP+RC	Coordinates Inclination	539891.47 E, Vertical	180266.1	.2 N Nat	ional Grid							
	Description of Strata		Legend	Depth	Datum	Samplir	9	Bl Sar	ow Co nple I	ount A Recove	nd erv	In Situ Test	Install-
			Logena	(Thickness)	Level	Details	Dia	TCR	SCR	RQD	IF	Details	ation
Remaining Detail : 2	9.66m - 30.80m : fragments			-		D 58 30.00-30	32					SPT 50/165mm 30.00 30.32 -	
from 30.34 medium cla	m to 30.37m 1 No very thin bed of ligh yey sand	t grey fine to				30.00 31	00 102	90	NA	NA			
at 30.79m 1 fine sand	No coarse gravel sized pocket of grey	ish blue silty		30.80 -	-28.12							-	
Medium strong to st core (angular fine to (Reading Formation	trong creamish white CALCRETE. Recov coarse gravel and angular cobble sized - Lower Mottled Clay)	vered as non intact I fragments).		(0.46)								-	
Very stiff greyish bro (Reading Formation	own sandy silty CLAY. Sand is fine to mo - Lower Mottled Clay)	edium.		(0.41)	-28.58	31.00 31	60 102	83	NA	NA			
from 31.50	m to 31.67m assumed zone of core los			31.67	-28.99	-D 61 31.60-31	82					SPT 50/115mm	
Stiff light bluish grey subangular fine to co calcrete. Sand is fine (Reading Formation	y mottled orangish brown very sandy C parse gravel sized nodules of creamish to medium. - Lower Mottled Clay)	LAY. With a few white		-		31.60 31	85 102	68	NA	NA		31.82 - - -	
from 32.04 strong crea	m to 32.13m 1 No subangular cobble s mish white calcrete	ized nodule of medium		-		D 62 32.05-32 31.85 32	35 40 102	100	NA	NA		5	
				 (1.46) 									
						32,40 33	00 102	92	NA	NA		-	
from 33.00	m to 33.13m assumed zone of core los	3		-		D 63 33.00-33	29					SPT 50/135mm	
Very dense light blui locally clayey fine to (Reading Formation from 33.13r silty clay	ish green mottled orangish brown sligh medium SAND. - Lower Mottled Clay) m to 33.20m and 33.31m to 33.50m 2 l	tly clayey No thin beds of sandy		33.13 - - -	-30.45	33.00 33	50 102	74	NA	NA		-	
				(0.99) — —									
grey, orang sandy silty o MOTTLED E	n to 33.93m 1 No thin bed of stiff mult ish brown, yellowish brown and brown clay. Sand is fine to medium (LAMBETH 3EDS)	ish green) slightly GROUP - LOWER			.91.44	33.50 34	60 102	98	NA	NA			
Very dense light gre clayey fine to mediu (Upnor Formation)	enish grey orangish brown mottled cla m glauconitic SAND.	yey locally very			-31,44								
						-D 65 - 34.60+34.	85					SPT 50/120mm - 34.60 34.85 - -	
NOTES: All depth See head For detai	s in metres, all diameters in millin ler sheet for details of boring, pro ls of abbreviations, see key	netres. gress and water.									с		a. 96
Unchecked	Log Print Date	And Time: 03/06/20	015 16:47:33	-					_	Ş	SOIL	enginee	RING
Form No. SIEXPHOLELOO	a Issue,Revision f	No. 1.06	Issue Date 20/09	/2013							Part o	f the Bachy Soletanch	e Group

Project Name	Silvertown Tunnel	-				Explorato	rv I			20		Hole ID.	
Project No. Engineer	TA7510 Atkins Ltd					Explorato	. y i	1010		9		G15	
Employer	Atkins Ltd											Sheet 8 of 8	
Ground Level Hole Type	2.68m OD TP+CP+RC	Coordinates 539 Inclination Vert	891.47 E, ical	180266.1	2 N Na	tional Grid							
	Description of Strata		Legend	Depth	Datum	n Sampling	l.	Blo San	ow Co	ount A Recov	nd erv	In Situ Test	Install-
				(Thickness)	Level	Details	Dia.	TCR	SCR	RQD	IF	Details	ation
Very dense light gree clayey fine to mediur (Upnor Formation)	nish grey orangish brown mottled clayey n glauconitic SAND.	locally very		(1.88) 		34.60 36.00	102	100	NA	NA		-	
	-1			- 36.00-	-33.32							-	
Exploratory hole com	plete at 36.00 m.												
				-								-	
												- - - - -	2
NOTES: All depths See head For detail	s in metres, all diameters in millimet er sheet for details of boring, progra is of abbreviations, see key	res. ess and water.											
Unchecked	Log Print Date An	d Time: 03/06/2015	16:47:37	-			_			5	Part of	f the Bachy Soletanch	
South Star Stear HOLELOG	rasue revision No.	1550	- sense 20/01				_		_			site eacity solution	- an a ala

17 I C 1 B

Project	t Nam	e Silve	ertown Tuni	nel						Explora	tory H	lole Lo	oq	Н	ole ID.
Project	t No.	TA7	510 ns 1 td								1		5	8	G22
Employ	yer	Atki	ns Ltd											Head	der Page 1
Ground	Level	3.54	n OD			Coor	dinates	5398	13.74 E, 1806	21.84 N Natio	nal Grid	Inclina	tion Va	rtical	
Top	Base	Type	Date Started	Date Ended	Crew	Section	Core	Core	Equip	oment	Sho	ring /		Remarks	
0.00	1.20	IP	17/11/2014	17/11/2014	DJ/JK	Logged By	y Barrel	Bit	Breaker /	hand tools	Sup	pořt			
1.20	10.00	CP	04/12/2014	05/12/2014	DA/DB	јк			Dando	8 2000					
·															
			CABLE	PERCUSSION	DETAILS					l	Ŵ	ATER STR	IKES	1 -	
Hard from	Strata I	Depth to	Chis Start time hhmm	elling Duration hhmm	Remarks				Date	Time S	strike at depth	Rise to depth	Time taken to rise	at stri time	ike to seal e flow
									05/12/2014 05/12/2014	0845 0845 0845	7.10 7.10 7.10	5.50 5.45 5.45	5 10 15	7.0	O NR O NR
									05/12/2014	0845	7.10	5.40	20	7.0	NR NR
									Hole	Depth	HOLE D	DIAMETER Casing	/ CASING	i oth of	
									diameter 250	hole 10.00		diameter 250	ca 1	sing 0.00	
										100000				22.7791	
			DOTA								DVALA				
Fror	2	To	ROTA	Flush	F	lush	Flush		Тор	Base		Diameter		ime	Recovery
dept	.n	depth		туре	re	%	colour						nn	mmss	70
	INSTA	LLATIO	N DETAILS		PIF	PE CONSTR	RUCTION				BAC	KFILL DET	AILS		
Distance from G.L	ID	Тур	e Respoi Top	nse zone ID Base	Pi Top	pe Base of	Dia. Typ f pipe pi	e of pe	Top of section	Base of section	Mat	erial		Re	emarks
3.00 8.50	02 01	SF SPI	E 1.00 8.00	3.00 02 9.00 02	0.00	1.00 3.00	50 Pla 50 Slot	ain tted	0.00 0.50	0.50	Conc Bent	onite			
11111111				01	0.00	8.50	19 19		3.00 8.00	8.00 9.00	Bent Sand b	onite			
									9.00	10.00	Bent	onite			
NOTES	: All dej Water	oths in i strike r	netres, all dia ise time in mi	meters in mil nutes, hard s	limetres. trata tim	e in hhmm									
Uncher	For de	tails of	abbreviations,	see key Print Date An	d Time: (3/06/201	5 16:41:21		-				so	IL end	Sineering
Form No. S	EXPHOLE	IDR P1	9	ssue.Revision No.	1.06	1	ssue Date 07/10/	2013				_	Pa	rt of the Bachy	/ Soletanche group

Project Na Project No Engineer Employer	ime Silv b. TA7 Atk	vertown 7510 Lins Ltd	Tunnel					Exploratory Hole Lo	g	Ho (ole ID. G22	2
Ground Leve	1 3.54	im OD			Coordinates 5	39813.74 E, 1	18062	1.84 N National Grid	22.0		or ruge	
Date Started	17/	11/2014	PROGE	DECC	Date Completed 0	5/12/2014		Inclinati SPT DETAIL	on Vert	ical		
Date	Time	Hole depth	Casing depth	Water depth	Remarks	Depth	Туре	Incremental blow count / penetration in mm	Hammei No.	Energy ratio	Casing depth	Wate
17/11/2014	0730 1800	0.00	1	NR NR	Start of Hole End of IP	1.50 3.00	SPT SPT	N=13 (3,1,2,2,4,5) N=7 (2,1,1,2,2,2)	AR87 AR87	% 60 60	1.50 3.00	DRY
77/11/2014 14/12/2014 15/12/2014 15/12/2014	1800 0730 1800 0730 1800	1.20 1.20 5.50 5.50 10.00	- 4.40 4.40 10.00	NR NR NR 4,00	End of IP Start of Shift / CP End of Shift Start of Shift End of Shift / Hole	3.00 6.00 7.50 9.00	SPT SPT(C) SPT(C)	N=7 (2,1,1,2,2,2) N=2 (0,0,0,0,0,0) N=11 (2,3,2,3,3) N=15 (3,3,4,3,4,4)	AR87 AR87 AR87 AR87		3.00 4.40 7.50 9.00	DR\ DR\ 5.90 4.90
		DEP	TH RELA	TED REN	/ARKS							
Top Base Depth Depth	Remark	G	ENERAL	NOTES								
								* Castlas blows esta				
NOTES: All o For	depths in details of	metres, a 'abbrevia	ll diamete tions, see	rs in milli key	metres.	RI		- searing blows only.	501			
Unchecked	LEHDR P2		Log Print Issue.Re	Date And evision No. 1.	1 Fime: 03/06/2015 16:41:24 06 Issue Date 12/03/201	- 4			Part	of the Bachy	Soletanche	roup

2.34	ES1 ES3 B2 ES3 B4 ES3 B4	arid Sampling Details 0.30 0.50	Dia.	Blow Co Sample R TCR SCR	unt And lecovery RQD	G Sheet	22 1 of 2 Test ils - - - - - - - - - - - - - - - - - - -	Installation
2.34	ES1 ES3 B4 D5 B6 ES7 D5	arid Sampling Details 0.30 0.50	Dia.	Blow Co Sample R TCR SCR	unt And lecovery RQD	Sheet In Situ Deta	1 of 2 Test ils 	Installation
2.34	ES 3 B 2 ES 3 B 4 ES 3 B 4	Sampling Details 0.30 0.50	Dia.	Blow Co Sample R TCR SCR	unt And lecovery RQD	In Situ - Deta	Test ils 	Installation
Datum Level 3.29 2.34	ES1 B2 ES3 B4 D5 B6 ES7 D5	Sampling Details 0.30 0.50	Dia.	Blow Co Sample R TCR SCR	RQD	In Situ Deta	Test ils 	Installation
Datum Level 3.29 2.34	ES1 B2 ES3 B4 D5 B6 ES7 C5 B6 ES7 C7 C7	Sampling Details 0.30 0.50	Dia.	Blow Co Sample R TCR SCR	unt And lecovery RQD	In Situ - Deta	Test ils 	Installation
2.34 1.69	ES 1 B 2 ES 3 B 4	0.30 0.50 1.00 1.00	Dia.	TCR SCR	RQD	- Deta	ils 	ation
3.29 2.34 1.69	ES1 B2 ES3 B4 D5 B6 ES7 C7	0.30 0.50 1.00 1.00					1.1.1.1.1.1.1	
3.29 2.34 1.69	ES 1 B 2 ES 3 B 4 D 5 B 6 ES 7 S 7	0.30 0.50 1.00 1.00						
2.34	ES 3 B 4 D 5 B 6 ES 7	1.00 1.00					1 1 1 1	1 0.0.0.0. 0.0.0.0. 0.0.0.0. 0.0.0.0.0. 0.0.0.0.0. 0.0.0.0.0.0. 0.0.0.0.0. 0.0.0.0.0.0.0.0.0.0.0. 0.0.0.0.0.0. 0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.
2.34	ES 3 B 4 D 5 B 6 ES 7 D 5	1.00 1.00					1 1	
2.34	B 4 D 5 B 6 ES 7	1.00						2 1 1 C 1
1.69	D 5 B 6 ES 7	10/22/0					-	
1.69	B6 ES7	1.50				SPT N=13		
1.69	0.0	1.50 1.50 1.50-1.95				1.50	1.95 - -	
	B 10	2.00					1 1 1	
	ES 11 D 9	2.00 2.00					1 1 1	
	D 12 B 13 ES 14	2.50 2.50 2.50					1 1 1 1 1	
	D 15 B 16 ES 17 D 18	3.00 3.00 3.00 3.00-3.45				SPT N=7 3.00	- 3.45 - -	
0.14	D 19 B 20 ES 21	3,50 3,50 3,50					1 1 1 1 1	
	D 22 B 23 ES 24	4.00 4.00 4.00					1 1 1 1	
-0.76	D 25 B 26 ES 27 U 28	4.50 4.50 4.50 4.50-4.95	100	8 Blows, 100	% Recovery			
	0.14	0.14 0.14 0.14 0.14 0.14 0.22	0.14 0.14 0.14 0.14 0.14 0.14 0.14 0.14 0.14 0.14 0.14 0.14 0.19 0.13 0.14 0.19 0.350 0.350 0.13 0.00 0.18 0.00-3.45 0.18 0.00-3.45 0.18 0.00-3.45 0.18 0.00-3.45 0.18 0.00-3.45 0.18 0.00-3.45 0.18 0.00-3.45 0.18 0.00-3.45 0.18 0.00-3.45 0.18 0.00-3.45 0.18 0.00-3.45 0.22 4.00 0.22 4.00 0.22 4.00 0.22 4.00 0.25 4.50 0.25 0.2	D 12 2.50 B 13 2.50 E5 14 2.50 D 15 3.00 B 16 3.00 E5 17 3.00 D 18 3.00-3.45 D 18 3.00-3.45 D 18 3.00 E5 21 3.50 E5 21 3.50 E5 21 3.50 E5 24 4.00 E 23 4.00 E 24 4.00 E 24 4.00 E 24 4.00 E 24 4.00 E 27 4.50 E 27 4.50	D 12 2.50 B13 2.50 E5 14 2.50 D 15 3.00 B 16 3.00 E5 17 3.00 D 18 3.00-3.45 D 19 3.50 B 20 3.50 E5 21 3.50 D 22 4.00 B 23 4.00 E5 24 4.00 E5 24 4.00 E5 24 4.00 E5 27 4.50 B 26 4.50 E5 27 4.50 D 28 4.50-4.95 1.00 8 Blows, 100	D 12 2.50 B13 2.50 E5 14 2.50 D 15 3.00 B 16 3.00 E5 17 3.00 D 18 3.00-3.45 D 19 3.50 B 20 3.50 E5 21 3.50 D 22 4.00 B 23 4.00 E5 24 4.00 E5 24 4.00 E5 24 4.00 E5 27 4.50 U 28 4.50-4.95 100 B Blows, 100% Recovery	D 12 2.50 B 13 2.50 E 5 14 2.50 D 15 3.00 B 16 3.00 E 5 17 3.00 D 18 3.00-3.45 D 19 3.50 B 20 3.50 E 5 21 3.50 D 22 4.00 B 23 4.00 E 5 24 4.00 E 5 24 4.00 E 5 24 4.00 E 5 24 4.00 E 5 27 4.50 B 26 4.50 E 5 27 4.50 D 25 4.50 B 26 4.50 E 5 27 4.50 D 28 4.50-4.95 D 00 B Blows, 100% Recovery	D 12 2.50 B 13 2.50 E 5 14 2.50 D 15 3.00 B 16 3.00 E 5 17 3.00 D 18 3.00-3.45 D 19 3.50 E 5 21 3.50 E 5 24 4.00 C 5 24 4.00 E 5 27 4.50 U 28 4.50-4.95 U 28 4.50-4.95 D 00 8 Blows, 100% Recovery

Project Name	Silvertown Tunnel					Evn	lorato	ny F			0		Hole	e ID.		
Project No.	TA7510					схр	iorator	iy i	1016	5 LU	y		G	22		
Engineer Employer	Atkins Ltd Atkins Ltd										_		Sheet	2 of 2		_
Ground Level	3.54m OD	Coordinates	539813.74 E,	180621.8	4 N Nat	ional G	rid									
Hole Type	IP+CP	Inclination	Vertical													
	Description of Strata		Legend	Depth	Datum		Sampling		Blo San	ow Co nple F	unt A Recov	And	In Situ	Test	Ins	tall-
				(Thickness)	Level		Details	Dia.	TCR	SCR	RQD		Deta	ils	atio	on
Soft grey slightly san (Alluvium)	dy silty CLAY. Sand is fine.		X X A	-		D 29	5.00						1	-		
р — М			×	-										1 4		
				1		D 20	6.60									
				-		B 31 ES 32	5.50 5.50							-		
				(2.80) -												
	₹		X X X	1		-2122	0.000.000							-		
				20 2		D 33	6.00-6.45						5PT N=2 6.00	6.45 -		
				_										1 1		
				-		-								-		
				-		D 34 B 35 D 36	6.50 6.50 6.50									
			XX-	-		0.031								-		
				-										1		
				7.10 -	-3.56									4		
Medium dense multi to coarse GRAVEL of	i-coloured slightly sandy subangular to su chert and flint. Sand is fine to coarse.	brounded fine			1993/9983											
(River terrace deposi	ts)			-										-		
						B 37 ES 38	7.50 7.50						SPT(C) N=11 7.50	7.05		
				-		B 39	7.50-8.00							1.95		
				1												
				-										-		
				{]										-	100000	
				- 1												
at 8.50m loc	cally sandy			(2.90) -		D 40 B 41	8.50 8.50									
				-										1		
				-										3		
] [B 42	9.00-9.50						SPT(C) N=15	-		
				1									9.00	9.45 -		
				-										-		
				1		D 43	9.50							1 1		
						B 44 ES 45 B 46	9.50 9.50 9.50-10.00							-		
				-												
				1												
Exploratory hole com	nplete at 10.00 m.															
					_					_		_		iler		
NOTES: All depths See head	s in metres, all diameters in millimet er sheet for details of boring, progr	res. ess and water.														
Unchecked	Log Print Date Ar	d Time: 03/06/20	015 16:41:29						_		-	sol	Lengi	nee	RI	ng
Form No. SIEXPHOLELOG	issue.Revision No.	1.06	Issue Date 20/09	/2013						_		Part	of the Bachy Se	oletanche	Gro	up

Drilled PW/NE Logged CT/JC Checked SV

orilled PW/NE ogged CT/JC Checked SV	Start 14/02/2011 End 17/03/2011	Equipment, Methods Dando 175 / Beretta T5 Machine excavated insp percussion boring from using 175mm dia Symm using the technique	and Remarks 1 / Magnetometer bection pit from GL to 3.5 3.50m to 8.90m depth. F antix system from 8.90r	50m depth. Cable Rotary open-hole drilling n to 14.50m. Rotary coring	Depth from to 0.00m 8.90m 8.90m 14.50m 14.50m 60.75m	Diameter 250mm 175mm 146mm	Casing Depth 8.90m 14.50m 60.75m	Ground Level Coordinates National Grid Chainage	+: E 5 N 1	5.28 mOD 539972.81 180437.25
Samples a	nd Tests			Strata	•			1		
Depth	Type & No	Records	Date Time Casing Water		Description			Depth, Level (Thickness)	Legend	Backfill/
-				For geology and samp trial pit NIT TP01	les refer to					
			14/02/2011 0800	(MADE GROUND) CLAY and BRICK (DRILLERS DESCRIP	TION)		-	3.60 +1.68 (0.40)		
- 4.00-4.85 4.00 4.00-4.45 4.00 4.00-4.45	SPT S D 1 D 2 ES 200 B 3	 N=2 (1,-/1,-,1,-) SW=400 4 samples taken 	4.00 1.20	(MADE GROUND) Soft brown slightly sar gravelly CLAY with str Gravel is angular to su fine to coarse of flint, t	idy slightly ong odour. bangular, prick,	4.00 m P		4.00 +1.28		

- 4.00 - 4.00-4.45 	ES 200 B 3 ES 201	4 samples taken 4 samples taken			gravelly CLAY with strong odour. Gravel is angular to subangular, fine to coarse of flint, brick, quartz and rare ceramic and wood fragments (up to 5 x 20 x 40mm). Sand is fine to coarse. 4.50 m PID = 0.0 ppm	(1.00)		
Depth	Type & No	Records	Date T Casing W	lime Vater				
Groundwater Ent No. Struck Por (m) None observed (ries st strike beha see Key She	viour eet)	Depth seal	led (m)	Depth Related Remarks * From to (m) 0.00 4.00 Magnetometer scan during excavation 0.00 8.00 Driller notes water seepages 4.00 8.90 SPT Hammer ID SM 10 (Er 54%); rod type B 4.00 8.90 Magnetometer scan	Chiselling Depths (m)	Time Tools u	Jsed
Notes: For explanation abbreviations see ke levels in metres. Stra in depth column. Scale 1:25	on of symbols by sheet. All de atum thickness 408.	and pths and reduced given in brackets NIT BH02	Project Project No. Carried out f	for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NI Sh	T BH02 eet 1 of 13	



Drilled PW/NE Logged CT/JC Checked SV	Start 14/02/2011 End 17/03/2011	Equipment, Methods ar 14.50m to 60.75m depth.	nd Remarks		Depth from to Diameter Casing Depth 0.00m 8.90m 250mm 8.90m 8.90m 14.50m 175mm 14.50m 14.50m 60.75m 146mm 60.75m	Ground Level Coordinates National Grid Chainage	+5 E 5 N 1	.28 mOD 39972.81 80437.25
Samples a	and Test:	ـــــــــــــــــــــــــــــــــــــ			Strata	1		
Depth	Type & No	Records	Date	Time	Description	Depth,Level	Legend	Backfil
5.00-5.75	SPT S	N=2 (1,-/1,-,1,-) SW=300	5.00	1.80	(Continued from Sheet 1) (MADE GROUND) 5.00 m PID = 0.0 ppm	(Thickness) 5.00 +0.28		
5.00-5.45 5.00-5.45 5.00	D 5 B 6 ES 202	4 samples taken			Soft brown slightly sandy slightly gravelly CLAY with strong odour. Gravel is angular to subangular, fine to coarse of flint, brick, quartz and rare ceramic and wood fragments (up to 5 x 20 x 40mm). Sand is fine to coarse.			
- 6.00-6.95 6.00 6.00-6.45 6.00-6.45 6.00	SPT S D 7 D 8 B 9 ES 203	N=1 (1,-/-,1,-,-) SW=500 4 samples taken	6.00	2.70	6.00-7.00 m Occasional dark greyish black and possible hydrocarbon odour. 6.00 m PID = 0.0 ppm - - -	(3.00)		
7.00-7.58 7.00 7.00-7.45 7.00-7.45 7.00	SPT S D 10 D 11 B 12 ES 204	N=4 (1/1.1,1,1) SW=200 4 samples taken	7.00	3.30	- - - iight grey brown sitty clay, 7.00 m PID = 0.0 ppm - - -			
- 8.00 8.00 8.00-8.45 8.00	D 13 D 14 B 15 ES 205	8.00 SPT attempted/terminated due to hammer bouncing on concateteamples taken			(MADE GROUND) Soft dark brown slightly sandy gravelly CLAY with low cobble content. Gravel is angular to subangular, fine to coarse of brick and flint. Cobbles are angular of slate (150 x 120 x	8.00 -2.72 (0.40) 8.40 -3.12		
8.90-8.92 8.90 - 8.90	— SPT S — D 16 ESW 17	50 (25 for 15mm/50 for 3mm) *	14/02/2011 8.90 16/02/2011 8.90	1800 5.00 0800	10mm). Sand is fine to coarse. ////////////////////////////////////	(0.50) 8.90 -3.62		
8.90-10.40	0 N/A N/A	Flush: 8.90-10.40 Air-Mist, 99 %				(1.50)		
Depth	Type & No	Records	Date	Time Water	Stratum continues to 10.40 m	╂────	· • · . a	
Groundwater Ent No. Struck Po (m) None observed	tries >st strike beha (see Key She	viour ∋et)	Depth sea	aled (m)	Depth Related Remarks * From to (m) 8.90 60.75 SPT Hammer ID SM31 (Er 55%); rod type B 8.90 11.70 114mm Dia down hole hammer used in concrete	Chiselling Depths (m) T 8.90 -8.90 6	ime Tool	is used अ
otes: For explanati bbreviations see ke evels in metres. Str a depth column.	ion of symbols ey sheet. All de ratum thickness	and pths and reduced given in brackets NIT BH02	Project Project No. Carried out	t for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NI	Г ВН02	2



Drilled PW/NE Logged CT/JC Checked SV	Start 14/02 End 17/03	t 2/2011 3/2011	Equipment, Methods ar Dando 175 / Beretta T51 / Machine excavated insper percussion boring from 3. using 175mm dia Symmai using wireline techniques	1d Remarks Magnetometer 2tion pit from GL to 3.5 50m to 8.90m depth. F hits system from 8.90r and self-horing press	Depth from to Diameter Casing Depth 0.00m 8.90m 250mm 8.90m stary open-hole drilling 14.50m 175mm 14.50m no 14.50m. Rotary coring 14.50m 60.75m 146mm 60.75m	Ground Level Coordinates National Grid Chainage	+5.28 mOD E 539972.81 N 180437.25
Samples a	nd T	ests	\$		Strata	1	
Depth	TCR SCR RQD	lf	Records/Samples	Date Time Casing Water	Description (Continued from Sheet 2)	Depth,Level (Thickness)	Legend Backfill/
 	0 N/A N/4		Flush: 10.40-11.10 Air-Mist, 99 %	16/02/2011 1700 8.90 24/02/2011 0700 11.10	(MADE GROUND) CONCRETE and STEEL (DRILLERS DESCRIPTION) (MADE GROUND) CONCRETE with rebar (DRILLERS DESCRIPTION)	10.40 -5. <i>12</i>	
- - - - - - - - -	0 N/A N/A		Flush: 11.10-11.60 Air-Mist, 99 %		(MADE GROUND) BRICK FILL - (DRILLERS DESCRIPTION) -	11.10 -5.82 - - - - - - (0.50)	
- 11.70 	0 N/A N/A	NA NA NA	EW 300 Flush: 11.60-13.80 Air-Mist, 99 %		(MADE GROUND) Greyish sandy GRAVEL (DRILLERS DESCRIPTION)	(2.20)	
- - - - - - - - - - - - - - - - - - -	0 N/A N/A		Flush: 13.80-14.50 Air-Mist, 99 %		(MADE GROUND) Grey CLAY (DRILLERS DESCRIPTION)	13.80 -8.52 (0.70)	
- - - - -		NA NA NA		24/02/2011 1700 14.50 25/02/2011 0700 14.50	Firm fissured grey brown slightly fine to medium sandy CLAY. Fissures are very closely spaced, randomly orientated, predominantly horizontal to subhorizontal, smooth, mainly polished and rarely	14.50 -9.22 - - - -	
Depth	TCR	lf	Records/Samples	Date Time	Stratum continues to 15.46 m		
Groundwater En No. Struck Po (m) None observed (Notes: For explanat abbreviations see k	RQD tries st strike (see Ke ion of sy ey shee	e beha ey She ymbols t. All de	riour et) and phs and reduced	Casing verter	Depth Related Remarks * From to (m) Cable Car for London - Ground Investigation	Chiselling Depths (m) T Borehole	ime Tools used
in depth column.	atum thi	.ckness	given in brackets NIT BH02	Project No. Carried out for	D1002-11 Transport for London	NI I Shr	BH02



Drilled PW/NE Logged CT/JC Checked SV	Sta 14/0 End 17/0	rt 02/2011 d 03/2011	Equipment, Methods at 14.50m to 60.75m depth.	nd Remark	S	Depth from to Diameter Casing Depth 0.00m 8.90m 250mm 8.90m 8.90m 14.50m 175mm 14.50m 14.50m 60.75m 146mm 60.75m	Ground Level Coordinates National Grid Chainage	+5.28 mOD E 539972.81 N 180437.25
Samples a	nd	Tests				Strata	1	
Depth	TCR SCR	lf	Records/Samples	Date	Time	Description	Depth, Level	Legend Backfill/
14.50-15.75 	100 N/A N/A		Flush: 14.50-15.75 Water, 99 %		Water	matt. Sand is fine to coarse. Occasional brown staining on fissure surfaces. Micaceous. (LONDON CLAY FORMATION - A2)		
- 15.65-15.75 - - -		NA NA NA	D 100	25/02/2011 15.75 28/02/2011 15.75	1 1700	Stiff to very stiff fissured grey slightly sandy CLAY. Fissures are very closely spaced to locally extremely closely spaced, randomly orientated, smooth, matt and rarely polished. Frequent brown staining on fissure surfaces. Sand is fine to medium.	15.46 - <i>10.18</i>	
- - - 15.75-17.25 - -	100 N/A N/A	NA NA NA	Flush: 15.75-17.25 water, 99 %			Very stiff fissured grey slightly sandy CLAY. Fissures are very closely spaced, predominatly vertical to subvertical, rarely subhorizontal, smooth, matt and polished. Sand is fine to medium. Rare foram fossils (up to 2 x 1mm). (LONDON CLAY FORMATION - A2)	- 16.22 -10.94 - - - - - - - - - - - - - - - - - - -	
 			D 101	28/02/201 ⁻ 17.25 02/03/201 ⁻	1 1700	ASSUMED ZONE OF CORE LOSS DUE TO SELF BORING PRESSUREMETER TEST POCKET	 17.25 -11.97 	
_ _ _ _ _ 17.25-18.75 _ _ _ _ _ 18.25-18.70 _ _ 18.25-18.70 _ _ _ _ _	0 N/A N/A	NA NA NA	• Flush: 17.25-18.75 water, 99 % SPT S N=38 (4,5/8,8,9,13) D 18	17.23		AND SPT TEST POCKET		
- - - - - - - - - - - - - - - - - - -	95 N/A N/A	NA NA NA NA	Flush: 18.75-20.25 water, 99 %			Very stiff fissured grey CLAY. Fissures are very closely spaced, subhorizontal (0 - 13 degrees), subvertical (85 - 70 degrees), smooth, matt and rarely polsihed. (LONDON CLAY FORMATION - A2) Very stiff fissured grey slightly sandy CLAY. Fissures are closely spaced, randomly orientated, smooth and matt. Sand is fine to medium. Occasional foram fossils (up to 5 x 1mm). Rare pockets (up to 10 x 5mm) of black fine to 19.82-19.83 m 1no. pyritised [™] lignite (80 mm x 25 mm x 10 mm).	 18.82 -13.54 (0.58) 19.40 -14.12 (0.85) 	
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	Stratum continues to 20.25 m		
Groundwater Ent No. Struck Po (m) None observed (tries st stri (see k	ke beha Key She	viour et)	Depth se	ealed (m)	Depth Related Remarks * From to (m) 17.75 SBP test	Chiselling Depths (m) T	ime Tools used
Notes: For explanati abbreviations see ke levels in metres. Stra in depth column. Scale 1:25	on of s ey she atum t	symbols et. All de hickness 408.	and pths and reduced given in brackets NIT BH02 24 08/06/2011 14:00:05	Project Project No Carried ou	o. ut for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NI She	F BH02 eet 4 of 13



Drilled PW/NE Logged CT/JC Checked SV	Star 14/02 End 17/03	t 2/2011 3/2011	Equipment, Methods ar Dando 175 / Beretta T51 / Machine excavated inspec percussion boring from 3. using 175mm dia Symmar using wireline techniques	nd Remarks Magnetomet ction pit from 50m to 8.90m ntix system fro and self-borin	er GL to 3.5 depth. F om 8.90n	50m depth. Cable Rotary open-hole drilling n to 14.50m. Rotary coring remeter testing from	Depth from to Diameter Casin 0.00m 8.90m 250mm 8.9 8.90m 14.50m 175mm 14.5 coring 14.50m 60.75m 146mm 60.7				n G C N C	round Level oordinates ational Grid hainage	H E N	⊧5.28 mOD 539972.81 180437.25
Samples a		ests	6	Date	Time	Strata	Description				╞	Conth Level		Backfill/
Depth	SCR RQD	lf	Records/Samples	Casing	Water	(Coi	ntinued from	Sheet 4)				(Thickness)	Legend	Instrument
 20.15-20.25 			D 102	02/03/2011 20.25	1700	medium sand. (LONDO FORMATION - A2)	N CLAY							
-			•	03/03/2011 20.25	0800 1.10	ASSUMED ZONE OF C DUE TO SELF BORING PRESSUREMETER TE AND SPT TEST POCKE	ORE LOSS ST POCKET T	20.25-2 self	21.75 m 88.4 boring press	Imm diameter suremeter test pocket.	2().25 -14.97		
— 20.25-21.75	0 N/A N/A	NA NA NA	Flush: 20.25-21.75 water, 99 %								_	(1.50)		
- - 21.20-21.65 21.20-21.65			SPT S N=40 (8,6/10,10,10,10) D 19]			I A A
-						Voru stiff ficeured grov (-1 AV	21.	75-21.97 m	Slightly fine to	- - - - 21	1.75 -16.47		
- - 						Fissures are very closely closely spaced, randoml orientated, smooth, matt occasionally polished.	/ to y and	21.97 fine to burrow	redium sau forams -22.27 m Ra medium gr ws (up to 5 >	ndy clay. Rare s (up to 2mm). are partings of ey sand. Rare c 1mm) infilled	-			
						(LONDON CLAY FORM	ATION - A2	22.27	-22.28 m 1n	with grey silt. o. lignite (30 x				
- - 21.75-23.25	100 N/A	NA NA	Flush: 21.75-23.25 water 99 %					2 gravel	2.32-22.37 i size pyrite 30 :	5 x 5mm). m 6no. coarse nodules (up to x 20 x 10mm).	-	(1.50)		
-	N/A	NA	water, ee se								-	·		
-								22.0	04 00 05 m	O"	-		 	
 			D 103	03/03/2011 23.25	1800			mediu	m sandy cla fossils	y. Rare foram (up to 3mm).			 	
-				04/03/2011 23.25	0800 1.30	ASSUMED ZONE OF C DUE TO SELF BORING PRESSUREMETER TE AND SPT TEST POCKE	ORE LOSS ST POCKET	23.25-2 self	24.75 m 88.4 boring press	Imm diameter suremeter test pocket.		3.25 -17.97		
-			*								-			
 23.25-24.75	0 N/A N/A	NA NA	Flush: 23.25-24.75 water, 99 %								-	(1.58)		
- 24.20-24.65 24.20-24.65 -			SPT S N=40 (4,5/8,10,10,12) D 20								-			
-											-			
-		 					·· • •	2	24 83-25.05	m Sandy clay.	- - 	4.83 <i>-19.55</i>		
 Depth	TCR	lf	Records/Samples	Date	Time	Very stift fissured grey s	lightly	Occa	sional pocke 5mm)	of grey sand.	1			144
Groundwater Ent No. Struck Po (m) None observed (tries st strik	e beha ey She	viour et)	Depth se	aled (m)	Depth Related Remarks * From to (m) 20.70 SBP test 23.70 SBP test					C	Chiselling Depths (m) T	ime To	ols used
Notes: For explanati abbreviations see ke levels in metres. Stra in depth column.	on of sy ey shee atum th	/mbols t. All de ickness	and pths and reduced given in brackets	Project Project No.	t for	Cable Car for London - Gro D1002-11 Transport for London	ound Investi	gation			B	Borehole NIT	BHO)2



Drilled PW/NE Logged CT/JC Checked SV	d PW/NE Start 14/02/2011 End 17/03/2011 14.50m to 60.75m depth.					Depth from 0.00m 8 8.90m 14 14.50m 60	to Diameter 3.90m 250mm 4.50m 175mm 0.75m 146mm	Casing Depth 8.90m 14.50m 60.75m	Ground Level Coordinates National Grid Chainage	+5.28 mOD E 539972.81 N 180437.25	
Samples a	nd 7	Fest s	5			Strata	<u>.</u>				
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	(Col	Description	east 5)		Depth,Level (Thickness)	Legend Backfill/
- - - - 24.75-26.25 - -	95 N/A N/A	NA NA NA	Flush: 24.75-26.25 water, 99 %			sandy CLAY. Fissures a closely spaced to closely predominatly subhorizor occasionally subvertical, rough and matt. Sand is medium. Occasional fora (up to 2mm). (LONDON FORMATION - A2)	re very y spaced, ntal, , smooth, fine to am fossils V CLAY	25.40-25.80 r partings of bro 25.82-25.98 Fissures are extr	n Occasional = - 	(1.21)	
 26.15-26.25 			If NA/NA/NA D 104 If NA/NA/NA			Very stiff grey slightly sa slightly gravelly glauconi CLAY. Sand is fine to m Gravel is rounded, fine t of black flint. (HARWICH FORMATIO OLDHAVEN MEMBER)	andy itic .edium. .o medium NN - POSSIBLE	cic 26:03-26:04 r pockets (up to dark green slight) 26:14-26:18 m 1n 30 x 25mm) fragments (up 26:18-26:25 fragments (up t	osely spaced. n Occasional 10 x 4mm) of - y sandy clay. o. shell (57 x - composed of m Rare shell o 10 x 5mm).	26.04 -20.76 26.25 -20.97 26.35 -21.07	
- - - 26.25-27.75 - -	7 N/A N/A	NA NA NA	Flush: 26.25-27.75 water, 99 %			Black rounded, fine to co GRAVEL of flint. (HARWICH FORMATIO BLACKHEATH MEMBE ASSUMED ZONE OF C	oarse N - R) ORE LOSS				
- 27.75-27.93 - 27.75 - -			SPT S 50 (9,16 for 30mm/50) D 21					27.75 m SPT shoe o gravelly fine to n Gravel is rounde coarse o		(2.71)	
- 27.75-29.25 - -	13 N/A N/A	NA NA NA	Flush: 27.75-29.25 water, 99 %								
 		NA	D 105	04/03/2011 29.25 07/03/2011 29.25	1800	Black rounded medium t GRAVEL of flint. (HARW FORMATION - BLACKH MEMBER) Very stiff grey sandy CL is fine to medium. Rare : fragments (up to 20 x 15 (POSSIBLE HARWICH ASSUMED ZONE OF C	to coarse VICH HEATH AY. Sand shell 5mm). FORMATION) CORE LOSS	29.22-29.25 closely spaced thii (1mm thick) of blac	m Extremely⊏ n laminations k grey sandy clay.	29.06 -23.78 29.16 -23.88 29.25 -23.97 (1.10)	
<u>⊦</u> '		IN/A	1		ļ				-		i iaa
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	Stratum continues to 30.35 m	n				
Groundwater Ent No. Struck Po: (m) None observed (ries st strik see K	te behav	viour et)	Depth se	aled (m)	Depth Related Remarks * From to (m)				Chiselling Depths (m) T	ime Tools used
Notes: For explanation abbreviations see ke levels in metres. Stra in depth column.	on of sy y shee atum th	ymbols a t. All de iickness	and pths and reduced given in brackets NIT BH02	Project Project No Carried ou). It for	Cable Car for London - Gro D1002-11 Transport for London	ound Investiga	ition		Borehole NIT	• BH02



Drilled PW/NE Logged CT/JC Checked SV	Start 14/0: End 17/0	t 2/2011)3/2011	Equipment, Methods ar Dando 175 / Beretta T51 / Machine excavated insper percussion boring from 3.f using 175mm dia Symma using wireling techniques.	1d Remarks Magnetome stion pit from 50m to 8.90r ntix system 1 and self-bor	s eter n GL to 3.{ m depth. F from 8.901	50m depth. Cable Rotary open-hole drilling m to 14.50m. Rotary coring urremeter testing from	Depth from 0.00m 8 8.90m 14 14.50m 60	to Diameter (8.90m 250mm 4.50m 175mm 0.75m 146mm	Casing Depth 8.90m 14.50m 60.75m	Ground Level Coordinates National Grid Chainage	+5.28 mOD E 539972.81 N 180437.25
Samples a	nd T	Fest	S	<u>1111 - 2000</u>	<u> </u>	Strata				1	
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	(Cor	Description	neet 6)		Depth, <i>Level</i> (Thickness)	Legend Backfill Instrumer
29.25-30.75 	27 N/A N/A		Flush: 29.25-30.75 water, 99 %	0405	True.	ASSUMED ZONE OF C	ORE LOSS			-	
30.65-30.75		NA NA NA	D 106			Grey sandy to very sand Sand is fine to medium. fragments (up to 3mm). (POSSIBLE LAMBETH LAMINATED BEDS) ASSUMED ZONE OF C PARTLY DUE TO SELF PRESSUREMETER TE AND SPT TEST POCKE	iy CLAY. Rare shell GROUP - ORE LOSS BORING ST POCKET T	30.35-30.43 m laminations of black with org 30.45-30.48 m Fr fragments (30.65-30.75 30.75-32.25 m 88.4 self boring pressu	t Occasionali t k sandy clay [- ganic odour. E ganic odour. E quent shell 'up to 5mm) nm diameter remeter test pocket	30.35 -25.07 (0.40) 30.75 -25.47	
- - 30.75-32.25 -	0 N/A N/A		• Flush: 30.75-32.25 water, 99 %						-	-	
- 31.80-32.14 -		NA	3PT S 50 (3,6/9,25,16 for 40mm	P					-		
-		NA NA		07/03/201 32.25	1 1800 0.70					(2.55)	
				08/03/201 ⁻ 32.25	1 0800						
— 32.25-33.75 - -	30 N/A N/A		Flush: 32.25-33.75 water, 99 %							- 33.30 <i>-28.02</i>	
- - 33.65-33.75 -		NA NA NA	D 107			Very stiff fissurea grey of Fissures are very closely subvertical and subhoriz rough and matt. Occasic (up to 25mm thick) of ve clay. Shell fragments ar	CLAY. y spaced, contal, onal bands rry shelly e up to 10		-	(0.45)	
-						x 5mm. (LAMBETH GROUP - LA BEDS) ASSUMED ZONE OF C PARTLY DUE TO SELF PRESSUREMETER TE: AND SPT TEST POCKE	AMINATED ORE LOSS BORING ST POCKET T	33.75-35.25 m 88.4m self boring pressu	nm diameter remeter test pocket.	33.75 -20.41	
 - 33.75-35.25 -	0 N/A N/A	NA NA NA	Flush: 33.75-35.25 water, 99 %						-	- (1.50) -	
- 34.75-34.82 - 34.75 -		SP	T S 50 (25 for 35mm/50 for 30m D 22	m)				34.75 m SPT sh greenish yellow fine sandy clay and 34.7 <u>5-34.</u> 82 <u>m E</u>	noe contains e to medium – d grey shelly clay. Brown <u>vellow</u>		
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	Stratum continues to 35.25 m	1				
Groundwater Ent No. Struck Pos (m) None observed (ries st strike see Ke	t e beha ey She	viour et)	Depth se	ealed (m)	Depth Related Remarks * From to (m) 31.30 SBP test 34.25 SBP test				Chiselling Depths (m) Ti	me Tools used
Notes: For explanationabbreviations see ke levels in metres. Strain depth column. Scale 1:25	on of sy y sheet atum th	/mbols t. All de ickness	and pths and reduced given in brackets NIT BH02 AGS	Project Project N Carried o	o. ut for	Cable Car for London - Gro D1002-11 Transport for London	ound Investiga	ation		Borehole NIT She	• BH02

DrilledPW/NELoggedCT/JCCheckedSV	Start 14/02 End 17/03	t 2/2011 3/2011	Equipment, Methods an 14.50m to 60.75m depth.	nd Remarks	i		Depth from 0.00m 8.90m 1 14.50m 6	to Diameter 8.90m 250mm 4.50m 175mm 00.75m 146mm	Casing Depth 8.90m 14.50m 60.75m	Ground Coordi Nationa Chaina	d Level nates al Grid ge	++ E 5 N 1	5.28 mOE 39972.81 80437.25) 5
Samples a	nd T	ests	;			Strata								
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	(Con	Description tinued from SI	heet 7)		Depth (Thick	, <i>Level</i> kness)	Legend	Backfi Instrume	II. en
-						ASSUMED ZONE OF CC PARTLY DUE TO SELF PRESSUREMETER TES AND SPT TEST POCKE Very stiff greenish blue n brown and grey sandy C	DRE LOSS BORING ST POCKET T nottled LAY. Sand is	mottled gre sandy gravelly 5 subangular to su to medium of white calcrete 35.25-35.	en grey slightly SILT. Gravel is brounded fine medium strong e. Sand is fine. 42 m Frequent calcretes.	- 2 35.25	-29.97	· _ · ·		- / / / /
- 35.50-35.60 - - -	100	NA	D 108			fine to medium. Occasion (up to 40 x 20mm) of mo strong white calcrete. (LAMBETH GROUP - LC MOTTLED BEDS)	nal pockets derately)WER	35.60-35.62 m soft grey cla 35.73-35.79 m firm grey cla 35.93-35.95 m	1no. pocket of⊏ y (40 x 28mm). 2no. pocket of[y (60 x 20mm). 1no. pocket of⊏	-		· - · · · · · · · · · · · · · · · · · ·		
35.25-36.75 - - - -	N/A N/A	NA NA	Flush: 35.25-36.75 water, 99 %					firm grey cla	y (20 x 10mm)	- (1 - - -	.50)			
-				08/03/2011 36.75	1800			soft grey cla 36.52-36.7 pockets (50 m	y (40 x 25mm). 5 m Frequents m x 30 mm) of calcrete.	- 36.75	-31.47	·; -		, , ,
- 				09/03/2011 36.75	0800	ASSUMED ZONE OF CC DUE TO SELF BORING PRESSUREMETER TES AND SPT TEST POCKE	DRE LOSS ST POCKET T	36.75-38.25 m 88. self boring pres	4mm diameter suremeter test pocket. -	-				/ / / /
37.25-37.48 - 37.25 -	0		•SPT S 50 (20,5 for 15mm/ 25,25 for 65mm) D 23					37.25-37.48 mottle dgre sandy SIL	m Brown grey en grey slightly T. Sand is fine	-				
- 36.75-38.25 - - - - - - - -	N/A N/A	NA NA NA	Flush: 36,75-38,25 water, 99 %						-	(1 38.37	.62)			///////////////////////////////////////
-			If NA/NA/NA			Very stiff greenish light g blue sandy CLAY. Sand medium. (LAMBETH GROUP - LC MOTTLED BEDS)	reyish is fine to)WER			- 38.50	-33.22	· · · · · · · · · · · · · · · · · · ·		1 1 1
_ 38.25-39.75 	92 N/A N/A	NA NA NA	Flush: 38:25-39.75 water, 99 %			Very stiff bluish green occasionally mottled dark and brown glauconitic sa Sand is fine to medium. (LAMBETH GROUP - LC MOTTLED BEDS)	k green ndy CLAY. WER		-	- (0 - 39.25	.75)	· · -		1 1 1
- - - 39.65-39.75 -		NA NA NA	D 109	09/03/2011 39.75	1800	Very stiff dark green occa mottled brown sandy CL/ fine to medium. (LAMBETH GROUP - LC MOTTLED BEDS)	asionally AY. Sand is OWER			- (0 - 30.75	.50)			/ / /
_	705			10/03/2011 39.75	0800	ASSUMED ZONE OF CO PARTLY DUE TO SELF	DRE LOSS BORING	self boring pres	suremeter test pocket.					
Depth	SCR RQD	lf	Records/Samples	Casing	Water	Stratum continues to 42.70 m								_
Groundwater Ent No. Struck Por (m) None observed (ries st strike see Ke	e behav ey She	viour et)	Depth se	aled (m)	Depth Related Remarks * From to (m) 37.25 SBP test				Chisel Depths	ling ; (m) T	ime Too	ls used	
Notes: For explanation abbreviations see ke levels in metres. Stra in depth column.	on of sy y sheet atum thi	mbols a . All dej ckness	and pths and reduced given in brackets	Project Project No.	t for	Cable Car for London - Gro D1002-11 Transport for London	ound Investig	ation		Bore	hole NIT	BH0	2	
Scale 1:25		408.2	24 08/06/2011 14:00:12	Jamed ou	. 101	Transport for London					She	et 8 of 13		

Soil Mechanics



Drilled PW/NE Logged CT/JC Checked SV	Star 14/02 End 17/02	t 2/2011 3/2011	Equipment, Methods ar Dando 175 / Beretta T51 / Machine excavated inspec percussion boring from 3.5 using 175mm dia Symmar using wireline techniques.	nd Remarks Magnetome ction pit from 50m to 8.90r ntix system f and self-bor	s eter i GL to 3.f n depth. F rom 8.90r	0m depth. Cable otary open-hole drilling to 14.50m. Rotary coring temeler testing from to 14.50m. Stary coring temeler testing from	Ground Level Coordinates National Grid Chainage	+5.28 mOD E 539972.81 N 180437.25
Samples a	nd T	este	3			Strata		
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	Description (Continued from Sheet 8)	Depth, <i>Level</i> (Thickness)	Legend Backfill/ Instruments
- - - - - - - - 40.75-41.25 - - - - 40.75 - - - - - - - - - - - - - - - - - - -	0 N/A N/A	NA NA NA	* Flush: 39.75-41.25 water, 99 % SPT S 50 (10,15 for 55mm/ 25,25 for 65mm) D 24			PRESSUREMETER TEST POCKET AND SPT TEST POCKET 40.75-41.02 m Stiff blue grey mottled orage brown slightly sandy slightly gravelly galuconitic CLAY. Gravel is subangular to well rounded fine to medium of brown and dark grey flint. Sand is fine to coarse.	- - - - - - - - - - - - - - - - - - -	
- - - - - - 41.25-42.75 - - - -	3 N/A N/A		Flush: 41.25-42.75 water, 99 %					
42.75-43.01 42.75 42.75 42.75 42.75-44.25 43.75-44.04 43.75	0 N/A N/A	NA NA NA S	If NA/NA/NA SPT S 50 (8,14/30,20 for 35mm D 29 Flush: 42,75-44,25 water, 99 % SPT S 50 (10,15/18,32 for 65mm D 26	, -)		Very stiff green slightly gravelly glauconitic CLAY with occasional fine to coarse gravel size pockets of fine to coarse black and brown sand. Gravel is subrounded to rounded, fine to coarse of black and brown flint. (LAMBETH GROUP - LOWER MOTTLED BEDS) ASSUMED ZONE OF CORE LOSS DUE TO SELF BORING PRESSUREMETER TEST POCKET AND SPT TEST POCKET AND SPT TEST POCKET 43.75 m SPT shoe contains greenish grey sandy glauconitic cay. Rare	- 42.70 -37.42 - 42.75 -37.47 	
 44.25-45.25 	100 N/A N/A		Flush: 44 25-45 25 water, 99 %			Very stiff green sandy to very sandy CLAY. Sand is fine to coarse. Occasionally local fine to coarse gravel of shells and shell fragments. (LAMBETH GROUP - POSSIBLE UPNOR FORMATION) 44.32 m 1no. rounded coarse black lint. 44.32 m 1no. rounded coarse black lint. 44.46 m 1no. subrounded fine gravel of flint. 44.64 m 1no. subrounded fine gravel of flint.	44.25 -38.97 	
Depth Groundwater Ent No. Struck Po: (m) None observed (st strik	If e beha ey She	Records/Samples viour æt)	Date Casing Depth se	Time Water ∋aled (m)	Stratum continues to 45.75 m Depth Related Remarks * From to (m) 40.25 SBP test 43.25 SBP test	Chiselling Depths (m) T	ime Tools used
Notes: For explanation abbreviations see ke levels in metres. Stra in depth column.	on of sy y shee atum thi	/mbols t. All de ickness	and pths and reduced given in brackets NIT BH02	Project Project No Carried of	o. ut for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NIT	BH02

Scale 1:25

Drilled PW/NE Logged CT/JC Checked SV	Start 14/02 End 17/03	2/2011 3/2011	Equipment, Methods an 14.50m to 60.75m depth.	nd Remarks	5		Depth from 0.00m 8.90m 14.50m	to Diameter 8.90m 250mm 14.50m 175mm 60.75m 146mm	Casing Depth 8.90m 14.50m 60.75m	Ground Level Coordinates National Grid Chainage	+{ E 5 N 1	5.28 mOD 39972.81 80437.25
Samples a	nd T	ests	6			Strata						
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	(Cor	Description tinued from S	Sheet 9)		Depth, <i>Level</i> (Thickness)	Legend	Backfill/ Instrument
45.15-45.25		NA NA NA	D 110	10/03/2011 45.25 15/03/2011	1800 0700	Very stiff green sandy to sandy CLAY. Sand is fin coarse. Occasionally loc coarse gravel of shells a fragments.	very e to al fine to nd shell	45.25-4	- - 45.40 m AZCL	(1.50)		
- - 45.25-45.75 -	70 N/A N/A		Flush: 45.25-45.75 water, 99 %	45.25		UPNOR FORMATION)	JOOIDLL	45.46 m 1n	□		·	
45.65-45.75			D 111					subrounded to r	ounded fine to coarse flint		$\left \begin{array}{c} \cdot & \cdot \\ \cdot & \cdot \end{array} \right $	ИИ
- 						ASSUMED ZONE OF C DUE TO SELF BORING PRESSUREMETER TES AND SPT TEST POCKE	ORE LOSS ST POCKET T	45.7447.25 m 88. self boring pres	4mm diameter suremeter test pocket. - - -	45.75 -40.47		
- 45.75-47.25 -	0 N/A N/A	NA NA NA	Flush: 45.75-47.25 water, 99 %						-	(1.50)		
46.85-46.98 - 46.85 			SPT S 50 (25/50 for 55mm) D 27					46.85 m Fine to clayey sand.	o medium grey Sand is fine to medium			
- - - -						Grey slightly silty to local silty fine to medium SAN (THANET SAND FORM/	lly D. ATION)			47.25 -41.97		
- 47.25-48.75 - - -	100 N/A N/A	NA NA NA	Flush: 47.25-48.75 water, 99 %	15/03/2011	1900					(1.50)		
48.65-48.75 - - - - -			D 112	48.75 16/03/2011 48.75	0700 3.20	ASSUMED ZONE OF C DUE TO SELF BORING PRESSUREMETER TES AND SPT TEST POCKE	ORE LOSS ST POCKET T	48.75-50.25 m 88. self boring pres	4mm diameter suremeter test pocket. pocket. - - -	48.75 -43.47		
- - 48.75-50.25 - -	0 N/A N/A	NA NA NA	Flush: 48.75-50.25 water, 99 %						-	(1.56)		
- 49.85-50.03 - 49.85			SPT S 50 (5,18/50 for 30mm) D 28	Dete	Time			49.85-50.03 m V to m	ery dense fine - nedium SAND			
Depth	TCR SCR RQD	lf	Records/Samples	Casing	Water	Stratum continues to 50.31 m						
Groundwater Ent No. Struck Pos (m) None observed (ries st strik see Ke	e behav ey She	viour et)	Depth se	aled (m)	Depth Related Remarks * From to (m) 46.35 SBP test 49.35 SBP test				Chiselling Depths (m) T	'ime Too	ls used
Notes: For explanations see ke	on of sy y sheet	mbols a	and pths and reduced	Project		Cable Car for London - Gro	ound Investig	jation		Borehole		_
evels in metres. Stra n depth column. Scale 1:25	itum thi	ckness	given in brackets NIT BH02	Project No Carried ou	t for	D1002-11 Transport for London				NI She	F BH0 2 et 10 of 13	2

Soil Mechanics



Drilled PW/NE Logged CT/JC Checked SV	Start 14/02 End 17/0:	t 2/2011 3/2011	Equipment, Methods ar Dando 175 / Beretta T51 / Machine excavated inspec percussion boring from 3.5 using 175mm dia Symmar using wireline techniques	nd Remarks Magnetomete ction pit from G 50m to 8.90m ntix system fro and self-boring	r 3L to 3.5 depth. F m 8.90r 1 pressi	Depth from to Diameter Casing Depth 0.00m 8.90m 250mm 8.90m 0tary open-hole drilling 14.50m 175mm 14.50m 10 4.50m Rotary coring 14.50m 60.75m 146mm 60.75m	Ground Level Coordinates National Grid Chainage	+5.: E 53 N 18	28 mOD 9972.81 0437.25
Samples a	nd T	ests	\$			Strata			-
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	Description (Continued from Sheet 10)	Depth, <i>Level</i> (Thickness)	Legend	Backfill/ nstruments
-						ASSUMED ZONE OF CORE LOSS DUE TO SELF BORING PRESSUREMETER TEST POCKET AND SPT TEST POCKET Grey slightly silty to locally silty fine to medium SAND. (THANET SAND FORMATION)	 = 50.31 <i>-45.03</i> 	· · · · · ·	
- - - - 50.25-51.75 -	96 N/A N/A	NA NA NA	Flush: 50 25-51.75 water, 99 %			-			
-						ASSUMED ZONE OF CORE LOSS 51.75-53.25 m 88.4mm diameter self boring pressuremeter test	 51.75 -46.47		
- - - - - 51.75-53.25	0 N/A N/A	NA NA NA	• Flush: 51.75-53.25 water, 99 %			PRESSUREMETER TEST POCKET AND SPT TEST POCKET	- - - - - - - - - - - - - - - - - - -		
- 52.85-52.95 - 52.85 -			SPT S 50 (25/50 for 25mm) D 29			52.85-52.95 m Very dense fine to medium silty SAND. _	-		
- - - - - - - - - - - - - - - -	95 N/A N/A	NA NA NA	Flush: 53,25-54,75 water, 99 %	16/03/2011 54.75	1900	Grey slightly silty to locally silty fine to medium SAND. (THANET SAND FORMATION)	54.75 -49.47		
- - 	ŢĊŖ	16	Pasarde/Campios	17/03/2011 54.75 Date	0700 Time	ASSUMED ZONE OF CORE LOSS DUE TO SELF BORING 			18
Groundwater Ent No. Struck Por (m) None observed (Răb ries st strik see Ke	e beha	viour	Casing V Depth sea	Vater Iled (m)	Depth Related Remarks * From to (m) 52.35 SBP test	Chiselling Depths (m) T	ime Tools	used
Notes: For explanation abbreviations see ke levels in metres. Stra in depth column. Scale 1:25	on of sy y sheet atum thi	/mbols t. All de ickness 408	and pths and reduced given in brackets NIT BH02	Project Project No. Carried out	for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NIT	F BH02 et 11 of 13	,



Drilled PW/NE Logged CT/JC Checked SV	Star 14/0 End 17/0	t 2/2011 3/2011	Equipment, Methods an 14.50m to 60.75m depth.	ıd Remarks	Depth from to Diameter Casing Depth 0.00m 8.90m 250mm 8.90m 8.90m 14.50m 175mm 14.50m 14.50m 60.75m 146mm 60.75m	Ground Level Coordinates National Grid Chainage	+5.28 mOD E 539972.81 N 180437.25
Samples a	nd 7	Lests	2		Strata	•	
Depth	TCR SCR	If	Records/Samples	Date Time	Description	Depth, Level	Legend Backfill/
-	RQU			Casing water	(Continued from Sneet 11) PRESSUREMETER TEST POCKET AND SPT TEST POCKET	(I hickness) - -	
- 54.75-56.25 - -	4 N/A N/A	NA NA NA	Flush: 54.75-56.25 water, 99 %			- (1.44) -	
55.85-56.08 - 55.85 			SPT S 50 (11,14/50) D 30		55.85-56.08 m Very dense fine to medium silty SAND	- - 	
-		NA NA NA			(THANET SAND FORMATION) -	- - (0.60) -	
- 56.25-57.75 	100 58 54		If NA/NA/NA Flush: 56.25-57.75 water, 99 %		Very stiff grey slightly sandy gravelly CLAY. Sand is fine. Gravel is angular to subrounded, fine to coarse of black and brown flint. Rare fine gravel size pockets of green sandy clay. (BULLHEAD BED - THANET SAND FORMATION)	= 56.79 -51.51 - 56.88 -51.60	
- - 57.75-58.04 - 57.75 - - 58.00-60.00		- s	PT S 50 (11,14/23,27 for 65mm D 31 KRH k=2.5E-7 m/s)	Very weak to weak medium density white CHALK. Fractures are closely to medium spaced (60/300/940), closed with rare grey brown staining. Occasional fine to coarse gravel size pockets of grey wispy marl. (WHITE CHALK SUBGROUP - CIRIA GRADE A2) Very weak to weak medium density 57.47 m 1no. 90 x 60mm flint. 57.56 m 1no. coarse gravel of 57.75-57.86 m AZCL 57.86-58.18 m Zone of SPT recovered as gravel of fine to weak of fine to coarse gravel		
- - - - 57.75-59.25 -	93 93 85	940	Flush: 57 75-59.25 water, 99 %		size tragments. Locally rare fine to medium gravel of filnt. 		
-					58.82 m Rare fine gravel size	- - - 59.25 -53.97	
					CHÁLK. One subvertical fracture observed (530/-/970). Infilled 59.25-59.48 m AZCL (0/<1/<1) with light brown clay with black speckling. Occasional fine to coarse gravel size pockets of grey wispy marl. 59.68-59.71 m Occasional fine (WHITE CHALK SUBGROUP - CIRIA 59.73 m 1no. 70 x 70mm nodular GRADE B2 / B1)	-	
Depth	TCR SCR RQD	lf	Records/Samples	Date Time Casing Water	Stratum continues to 60.75 m		
Groundwater Ent No. Struck Pos (m) None observed (ries st strik see Ko	e behav	viour et)	Depth sealed (m)	Depth Related Remarks * From to (m) 55.35 SBP test	Chiselling Depths (m) T	ime Tools used
Notes: For explanationabbreviations see ker levels in metres. Stratin depth column. Scale 1.25	on of sy y shee atum th	/mbols a t. All der ickness	and pths and reduced given in brackets	Project Project No. Carried out for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NIT	BH02



Drilled PW/NE Logged CT/JC Checked SV	Star 14/0 End 17/0	t 2/2011 3/2011	Equipment, Methods ar Dando 175 / Beretta T51 / Machine excavated insper percussion boring from 3.1 using 175mm dia Symmai using wireline techniques	nd Remarks Magnetomete ction pit from (50m to 8.90m ntix system fro and self-borin	er GL to 3.t depth. I om 8.90⊨ og press	50m depth. Cable Rotary open-hole drilling m to 14.50m. Rotary coring uremeter lesting from	Depth from 0.00m 8.90m 14.50m	to 8.90m 14.50m 60.75m	Diameter 250mm 175mm 146mm	Casing Depth 8.90m 14.50m 60.75m	Ground Level Coordinates National Grid Chainage	+! E 5 N 1	5.28 mOD ;39972.81 180437.25
Samples a	ind T	lest:	S			Strata							
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	(Con	Description) heat 12)			Depth,Level (Thickness)	Legend	Backfill/
59.25-60.75	85 85 85	530 970	Flush: 59.25-60.75 water, 99 %	17/03/2011 60.75	1900	Very weak medium dens CHALK. One subvertical observed (530/-/970). Int (0/<1/<1) with light brown with black speckling. Occ fine to coarse gravel size of grey wispy marl. (WHITE CHALK SUBGF GRADE B2 / B1)	ity white fracture filled n clay casional ∋ pockets {OUP - CIRI,	А _{60.53-6}	30.63 m Ful	 diameter flint.	(1.50)		
60.75-61.03 60.75 - - - - - - - - - - - - -		If	- SPT 6 50 (13,12 for 65mm) 23,27 for 65mm) D 32	Date Casing	Time Water	EXPLORATORY HOLE END	S AT 60.75 m				60.75 -55.47		
No. Struck Po (m) None observed (st strik	e beha	viour et)	Depth sea	aled (m)	From to (m)					Depths (m) T	ime Too	Is used
Notes: For explanati abbreviations see ke levels in metres. Stra in depth column. Scale 1:25	on of sy ey shee atum th	/mbols t. All de ickness	and pths and reduced given in brackets NIT BH02 AGS	Project Project No. Carried out	t for	Cable Car for London - Gro D1002-11 Transport for London	ound Investig	gation			Borehole NI She	BH0	2

Start 15/02/201

02/03/201

Type & No

ES 200 D 1

B 2 ES 201

ES 202

D 3

B 4

SPT C

Β5

ES 203

ES 204 D 6

SPT S

D7 B8

ES 205

EW 12 ES 206

D 9

SPT S

D 10 B 11

End

Samples and Tests

PW

Logged CT/JC/NI

Depth

0.35 0.40

0.50 0.50

1.00

1.00-1.10

1.20-1.65

1.20-1.65

1.50

2.00

2.00

2.20-2.65 2.20-2.65 2.20-2.65

2.50

3.00 3.00 3.00

3.20-3.65

3.20-3.65

3.20-3.65

Checked SV

Drilled

Soil Mechanics Casing Depth 15.00m 16.10m 45.24m Equipment, Methods and Remarks JCB 3CX / Dando 175 / Beretta T51 / Magnetometer Ground Level Depth from 0.00m Diameter 250mm +5.16 mOD **to** 15.00m Coordinates National Grid E 539850.35 CAT scanned for services. Broken out from GL to 0.30m depth. Hand dug 15.00m 16.20m 16.20m 45.24m 200mm 146mm N 180286.36 inspection pit from 0.30m to 1.20m depth. Cable percussion boring from 1.20m to 16.20m depth. Rotary cored drilling using wireline techniques from 16.20m to 45.24m depth. Chainage Strata Time Depth,Level Backfill/ Date Description Records Legend Water (Thickness) Casing strum (MADE GROUND) ^ Reinforced CONCRETE. (0.30) ⊿ ` 0.30 +4.86 (MADE GROUND) 4 samples taken Brown very gravelly fine to coarse 4 SAND. Gravel is angular to 0.50 m PID = 0.0ppm subangular, fine to coarse of 5 samples taken ⊿ concrete and brick. (0.70) 16/02/2011 1800 4 samples taken 1.00 +4.16 1.00 m PID = 0.6ppm (MADE GROUND) dr Medium dense dark greyish black gravelly fine to coarse SAND of ash with pockets (up to 250mm) of soft brown slightly sandy slightly N=11 (1,3/4,3,2,2) 17/02/2011 0800 gravelly clay. Sand is fine to dr coarse. Gravel is angular to 5 samples taken subrounded, fine to coarse of brick, 1.50 m PID = 3.5ppm (1.10)quartz and coal. 5 samples taken 2.00 m PID = 4.9ppm 2.00-2.10 m Slightly clayey. 2 10 +3.06 (MADE GROUND) N=3 (1,1/1,1,1,-) 2.20 dr Soft brown slightly sandy slightly gravelly CLAY. angular to subrounded, fine to coarse Gravel of quartz, brick and chalk. Sand is fine to coarse. With pockets of 5 samples taken 2.50 m PID = 6.4ppm brownish grey occasional reddish (0.90) brown slightly clayey gravelly fine to medium sand 3.00 +2.16 3.00 m PID = 13.9ppm (MADE GROUND) 5 samples taken Soft greenish grey slightly sandy clayey SILT with rare subangular to N=11 (1,1/3,3,2,3) 3.20 2.80 3.20-3.65 m Pockets of sandy very clayey angular to subangular, fine to coarse gravel of slag. Sand is fine to coarse. 3.20-6.00 m Hydrocarbon odour. subrounded, fine to coarse quartz gravel. Rare wood fragments (up to 20mm). Sand is fine to coarse.

Scale 1:25	408.	NT BH02 AGS	Carried o	out for	Transport for London	Sh	eet 1 of 10
Notes: For explanation abbreviations see ke levels in metres. Stra in depth column.	on of symbols y sheet. All de atum thickness	and pths and reduced given in brackets	Project Project N	lo.	Cable Car for London - Ground Investigation D1002-11	Borehole N	Г ВН02
Groundwater Ent No. Struck (m) Po 1 3.00 Ro 2 3.00 Ro 3 3.00 Ro 4 3.00 Ro	ries st strike beha se to 2.90 m se to 2.85 m se to 2.80 m se to 2.80 m	viour after 5 minutes. after 10 minutes. after 15 minutes. after 20 minutes.	Depth s	sealed (m) - - 3.50	Depth Related Remarks * From to (m) 0.30 1.00 Driller notes material compact. 1.20 16.20 SPT hammer ID SM10 (Er 54%); rod type B 1.20 Magnetometer scan 2.20 Magnetometer scan 3.50 Magnetometer scan	Chiselling Depths (m)	Time Tools used
Depth	Type & No	Records	Date Casing	Time Water	Stratum continues to 6.00 m		
- 4.20-4.65 4.20-4.65 4.20-4.65 - 4.50 - 4.50	SPT S D 15 B 16 ES 209	N=1 (1,-/-,1,-,-) 5 samples taken	4.20	damp	- - 4.50 m PID = 0.5ppm - - - - - -	(3.00)	x x
- 4.00 4.00	D 14 ES 208	5 samples taken			gravelly. Gravel is angular to subangular, fine to coarse of slag. - 4.00 m Wood fragment found.		
- 3.50 3.50	D 13 ES 207	* 5 samples taken			3.50 m PID = 0.4ppm - 3.65-6.00 m Becomes slightly ⁻		




Drilled PW Logged CT/JC/NE Checked SV	Start 15/02/2011 End 02/03/2011	Equipment, Methods an JCB 3CX / Dando 175 / Br CAT scanned for services inspection pit from 0.30m 1.20m to 16.20m depth. F 16.20m to 45.24m depth.	nd Remarks eretta T51 / M . Broken out fi to 1.20m dept totary cored d	agnetorr rom GL 1 th. Cable rilling us	neter to 0.30m depth. Hand dug e percussion boring from ing wireline techniques from	Depth from to 0.00m 15.00 15.00m 16.20 16.20m 45.24	Diameter m 250mm m 200mm m 146mm	Casing Depth 15.00m 16.10m 45.24m	Ground Level Coordinates National Grid Chainage	+5.16 mOD E 539850.35 N 180286.36
Samples a	nd Tests	<mark>غ</mark>	Date	Time	Strata	Description			Depth / eve/	Backfill/
Depth	Type & No	Records	Casing	Water	(Cor	ntinued from Sheet	1)		(Thickness)	Legend Instruments
- 5.00 - 5.20-6.05 5.20-5.65 - 5.20-5.65 -	ES 210 SPT S D 18 B 19	5 samples taken N=2 (1,-/1,-,1,-) SW=400	5.20	damp	(MADE GROUND) Soft greenish grey slight clayey SILT with rare su subrounded, fine to coar gravel. Rare wood fragm 20mm). Sand is fine to c	Ily sandy bangular to rse quartz nents (up to coarse.	5.00 m	PID = 0.0ppm		
	D 20 ES 211	5 samples taken			Soft dark greenish grey	slightly	6.00 m	PID = 0.0ppm	6.00 -0.84	
- 6.20-6.65 6.20-6.65 - 6.20-8.65 -	SPT S D 22 B 23	N=1 (1,-/-,1,-,-)	6.20	damp	wood fragments (up to 5 slight organic odour. (ALLUVIUM)	i0mm) and		- - -	(0.70)	
- 6.70 6.70-7.00 - 7.00	D 24 B 25 FS 212	5 samples taken			Firm brown pseudofibrou plant remains of wood, g rootlets. (ALLUVIUM)	us PEAT with grass and			6.70 -1.54	
- 7.20-7.65 7.20-7.65 - 7.20-7.65	SPT S D 26 B 27	N=4 (1,-/1,1,1,1)	7.20	dry			7.20-9.30 m R gree	PID = 0.0ppm 		
- - - - - - 8.00	ES 213	• 5 samples taken					8.00 m	- - - - 1 PID = 0.0ppm		
- 8.20-8.65 8.20-8.65 - 8.20-8.65 - 8.20-8.65 	SPT S D 29 B 30	N=5 (1,-/1,1,1,2)	8.20	dry				- - - - -	(3.30)	
	ES 214 D 31	5 samples taken					9.00 m	- PID = 0.0ppm -		
- 9.20-9.65 9.20-9.65 9.20-9.65 - - - - -	SPT S D 32 B 33	N=2 (1,-/1,-,1,-)	8.20	dry		9.30	-10.00 m Frequ gree	ient pockets of nish grey clay 		
Depth	Type & No	Records	Date Casing	Time Water						
Groundwater Ent No. Struck Pos (m)	ries st strike beha	viour	Depth sea	aled (m)	Depth Related Remarks * From to (m) 5.50 Magnetome 7.50 Magnetome 9.50 Magnetome	eter scan eter scan eter scan			Chiselling Depths (m)	Time Tools used
Notes: For explanationabbreviations see ke levels in metres. Strain depth column.	on of symbols y sheet. All de atum thickness	and pths and reduced given in brackets NT BH02 L24, 2206/2011 12:16:24	Project Project No. Carried out	t for	Cable Car for London - Gro D1002-11 Transport for London	ound Investigation	n		Borehole NT Sh	Г ВН02 ееt 2 of 10



Drilled PW Logged CT/JC/NE Checked SV	Start 15/02/2011 End 02/03/2011	Equipment, Methods at JCB 3CX / Dando 175 / B CAT scanned for services inspection pit from 0.30m 1.20m to 16.20m depth. F 16.20m to 45.20m depth	nd Remarks eretta T51 / Mi 3. Broken out fi to 1.20m dept Rotary cored di	agneton rom GL th. Cable rilling us	teter Depth f 0.00r to 0.30m depth. Hand dug 15.00r percussion boring from 16.20r ing wireline techniques from	rom to Diameter m 15.00m 250mm m 16.20m 200mm m 45.24m 146mm	Casing Depth 15.00m 16.10m 45.24m	Ground Level Coordinates National Grid Chainage	+5.7 E 539 N 180	16 mOD 9850.35 0286.36
Samples a	ind Tests	5			Strata			L		
Depth	Type & No	Records	Date Casing	Time Water	Descri (Continued fr	ption rom Sheet 2)		Depth,Level (Thickness)	Legend	Backfill/
10.00 10.00	ES 215 D 34	5 samples taken	47/02/2011	1900	Soft dark grey CLAY.	10.00 m	PID = 3.4ppm	10.00 -4.84		
- 10.20-10.65	SPT S	* N=15 (2,3/3,3,4,5)	8.20	dry	(ALLUVIUM)		-	(0.30)	5	
10.20-10.65 10.30	D 35 EW 37	5 samples taken	10.30	3.90	Medium dense to dense brown v		0-11.00 m Silty.	10.30 -5.14	6	
10.30 10.30-10.50 - - -	D 38 B 39		18/02/2011 10.30	0800 4.30	sandy GRAVEL. Gravel is angul subrounded, fine to coarse of qu and flint. Sand is predominately medium to coarse with occasion fine. (RIVER TERRACE DEPOSITS)	ar to lartz al				
	ES 216 D 40	5 samples taken				11.00 m 11.00-12.00	n PID = 0.0ppm m Slightly silty.			
- - 11.20-11.65	SPT C	N=17 (3,3/4,5,4,4)	11.20	7.20			-		0 0 0 0 0 0	
_ 11.20-11.65 _ _ _ _ _ _	B 41							(1.90)		
12.00 12.00	ES 217 D 42	5 samples taken				12.00 m	n PID = 0.0ppm		°	NK
- - 12.20-12.65 12.20-12.65 -	SPT C B 43	N=31 (3,8/7,9,7,8)	12.20	7.30	Medimum dense to dense brown GRAVEL. Gravel is angular to subrounded fine to coarse of Q	ı sandy		12.20 -7.04	· · · · · · · · · · · · · · · · · · ·	
- - - - - 13.00 13.00	ES 218 D 44	 5 samples taken 			(RIVER TERRACE DEPOSITS)	13.00 m		(1.70)		
- 13.20-13.65 - 13.20-13.65 	SPT C B 45	N=21 (3,5/6,5,5,5)	13.20	6.90						
	ES 219 D 46 B 47	5 samples taken			Firm greenish grey slightly grave slightly sandy CLAY. Sand is fin coarse.	비y e to 14.00 m	n PID = 0.0ppm	13.90 -8.74		
— 14.20-14.65 —	UT 48	34 blows 350 mm rec	14.20	5.50	LONDON CLAY FORMATION)	031137	-	(0.55)		
	ES 220 D 49	5 samples taken			Stiff thinly laminated brownish gr CLAY with fine sand and silt	rey 14.50 m		14.45 -9.29		
	SPT S D 50	N=20 (2,3/4,5,5,6)	49/02/2011	1900	partings. (LONDON CLAY FORMATION)		-	-		
-			14.70	5.50			-			
 Depth	Type & No	Records	Date	Time Water	Stratum continues to 16.20 m			┣───-		
Groundwater Ent No. Struck Pos (m) 5 10.30 Rc 6 10.30 Rc 7 10.30 Rc 8 10.30 Rc	tries st strike beha ose to 5.10 m ose to 4.65 m ose to 4.40 m ose to 4.30 m	viour 1 after 5 minutes. 1 after 10 minutes. 1 after 15 minutes. 1 after 20 minutes.	Depth sea	aled (m) - - -	Depth Related Remarks * From to (m) 10.20 Magnetometer scan 13.00 15.00 Bentonite environmer	ntal seal		Chiselling Depths (m) T	ime Tools	used
Notes: For explanational abbreviations see ke levels in metres. Strain depth column.	on of symbols and	and pths and reduced given in brackets NT BH02	Project Project No. Carried out	t for	Cable Car for London - Ground Inv D1002-11 Transport for London	estigation		Borehole NT	BH02	



Drilled PW .ogged CT/JC/NE Checked SV	Start 15/02 End 02/0:	2/2011 3/2011	Equipment, Methods an JCB 3CX / Dando 175 / Be CAT scanned for services. inspection pit from 0.30m 1 1.20m to 16.20m depth. R	Id Remarks Bretta T51 / M Broken out f to 1.20m dep totary cored c	, /agneton from GL oth. Cable drilling us	neter 0.30m depth. Hand dug percussion boring from 5 percussion boring	Ground Level Coordinates National Grid Chainage	+5.16 m0 E 539850. N 180286.
Samples a	nd T	ests	16 20m to 45 24m deptn			Strata	1	
Depth	Туре	& No	Records	Date	Time	Description	Depth, Level	Legend Bac
15.00	ES	221	5 samples taken	Casiliy	Water	Stiff thinly laminated brownish grey 15.00 m PID = 0.0ppm	(THICKNESS)	
	1		, I	21/02/2011	0800	CLAY with fine sand and silt	1	- N
	1		1	14.70	6.70	(LONDON CLAY FORMATION)	1	$\vdash - \mid \upharpoonright$
	1	ļ	1	1	ļ		(1.75)	
	1	ļ	1	1	ļ	-	-	
	1	ļ	, I	1	ļ	-	4	
	1	ļ	, I	1	ļ	-	-	
	1	ļ	1	1	ļ	-	1	$\vdash - \mid \mid \mid \mid$
	1		1	01/00/0011	1200	-	1	- N
	1		1	21/02/2011	1800 6.70]	
	L		1.*		0.10	16 20-16 40 m AZCI	16.20 -11.04	
			1	22/02/2011	0800	Very stiff fissured brownish grey siltv CLAY. Fissures are very		
	67		1	16.10	7.00	closely spaced predominatly	4	I — 1 NI
16.20-16.80	N/A	N/A N/A	Flush: 16.20-16.80 water, 95 %	1	ļ	Rare to occasional localised	- (0.60)	[] N
		IN/A	, I	1	ļ	foraminifera present. Occasional fine to coarse gravel sized pockets 16.62 m 1no. subhorizontal	-	$\vdash _ $ \upharpoonright
	1 1	(1	ļ	of fine black sand. (1 ONDON CLAY FORMATION 42) 16.68 m 1no. subbrizontal	1	$ - \rangle$
16.80-17.25 16.80-17.25	[]		SPT S N=30 (4,5/6,7,8,9) D 51	1	ļ	(LUNDON CLAY FORMATION - A2) TISSURE (5 aegrees) 16.71 m 100. subbrizontal 16.71 m 100. subbrizontal	7 16.80 -11.64	
	1 1	(, I	1	ļ	ASSUMED ZONE OF CORE LOSS IIssure (5 acgrees) - 16.71-16.73 m 1no. subvertical frouve (95 00 degrees) -	1	
	1 1	N/A	, I	1	ļ	ופאטע פינסט אינע פינט אינע פינע פינע פינע פינע פינ 16.80-17.25 m Stiff fissured brown sliphtly sandy CI AV – Fissures -		
	1 1	N/A N/A	, I	1	ļ	are extremely closely to very closely snaced randomly	(0.65)	
	1 1	(, I	1	ļ	orientated rough and matrix	_	
	1 1	(1	1	ļ	fine sand.		
40 00 40 30	67		Fluck, 16 90 18 35	1	ļ	Stiff to very stiff brownish grey	- 17.45 <i>-12.29</i> -	FIN
16.80-18.30	N/A N/A	(water, 95 %	1	ļ	slightly sandy CLAY. Sand is fine. Occasional to locally frequent fine	4	- N
	1 1	(1	1	ļ	to coarse gravel size pockets of	4	$\vdash - \mid \bigwedge \mid$
	1 1	(1	1	ļ	present. Mica present.	1	
	1 1		1	1	ļ	(LONDON CLAY FORMATION - A2)	1	
	1 1	N/A N/A	1	1	ļ	-	(1.20)	
	1 1	IN/A	, I	1	ļ	-		=
	<u> </u>		*	1	ļ	-	4	$\vdash - \mid \mid \mid \mid$
18.35-18.80 18.35-18.80	1 1	(SPT S N=33 (4,6/7,7,9,10) D 52	1	ļ	18.35-18.65 m AZCL	4	-
	1 1	(, I	1	ļ	-	-	$\vdash - \mid \upharpoonright$
	1 1	\square	, !	1	ļ		- 18.65 <i>-13.4</i> 9	
	1 1	(, I	1	ļ	Very stiff fissured brownish grey		
	1 1	(, I	1	ļ	CLAY. Fissures are closely spaced, sub-prizontal to sub-perficiel emonth. 18.87 m Locally very sandy clay.	1	
		(, I	1	ļ	and matt. Occasional to locally with frequent forams.		
18.30-19.85	N/A	(Flush: 18.35-19.85	1	ļ	frequent pockets of tine black and grev sand. Occasional to locally 19.08 m 1no. 10 mm x 5 mm lignite. = -	4	$\vdash - \mid \mid \mid$
		N/A	water, oo to	1	ļ	frequent foraminifera. (I ONDONI CLAY FORMATION - A2)	(1.20)	- N
	1 1	N/A N/A	1	1	ļ	-	- (1.20)	$\vdash - \mid \mid \mid$
	1 1	(1	1	ļ	-	-	
	1 1	(1	1	ļ	-	1	
	1 1	(, I	1	ļ	19.58 m 1no. subhorizontal – fissure (10 degrees)	1	<u> </u>
	1 1	(1	1	ļ	19.69-19.76 m 1no. subverticai fissure (85 - 90 degrees)]	
19.85-20.30 19.85-20.30	┌──┤	\square	SPT S N=38 (5,6/9,8,10,11) D 53	1	ļ	19.85-20.30 m Stiff brown slightly	- 19.85 -14.69	
10.00 20.00	TCR	<u> </u>		Date	Time	partings of light brown fine sand.	┫ ┤	
Depth	SČŘ RQD	lf	Records/Samples	Casing	Water	Stratum continues to 20.40 m		
roundwater Ent o. Struck Pos (m)	ries st strike	3 behav	<i>r</i> iour	Depth se	aled (m)	Depth Related Remarks * From to (m) 16.20 45.00 SPT hammer ID SM29 (Er 60%); rod type B 18.30 18.35 TCR 100, SCR NR, RQD NR	Chiselling Depths (m) Ti	me Tools used
otes: For explanation breviations see ke vels in metres. Stra	on of sy y sheel atum th	mbols a 1. All de ickness	and pths and reduced given in brackets	Project		Cable Car for London - Ground Investigation	Borehole	BH02
depth column.			NT BH02 AGS	Carried ou	It for	Transport for London	She	et 4 of 10



Drilled PW Logged CT/JC/NE Checked SV	Start 15/02 End 02/03	t 2/2011 3/2011	Equipment, Methods at JCB 3CX / Dando 175 / Bd CAT scanned for services inspection pit from 0.30m 1.20m to 16.20m depth. R 16.20m to 45.24m depth	nd Remarks eretta T51 / N . Broken out f to 1.20m dep otary cored d	lagnetom from GL t th. Cable rilling usi	heter 0.30m depth. Hand dug percussion boring from ng wireline techniques from	Ground Level Coordinates National Grid Chainage	+5.16 mOD E 539850.35 N 180286.36
Samples a	nd T	est	6			Strata		
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	Description (Continued from Sheet 4)	Depth, Level (Thickness)	Legend Backfill/
-		N/A N/A N/A				ASSUMED ZONE OF CORE LOSS Occasional white foraminifera (1mm)	(0.55)	
- - 19.85-21.35 - - -	63 N/A N/A	N/A N/A N/A	Flush: 19.85-21.35 water, 95 %			Very stiff fissured grey slightly sandy to locally sandy CLAY. Fissures are closely spaced, subhorizontal to subvertical, smooth and matt. Occasional to locally frequent pockets of fine black and grey sand. Occasional to locally frequent foraminifera. (LONDON CLAY FORMATION - A2) 20.40-21.53 m Abundant fissures extremely closely spaced, randomly orientated, smooth and matt. Occasional to locally frequent foraminifera. (LONDON CLAY FORMATION - A2)	20.40 -15.24	
- 21.35-21.80 - 21.35-21.80 -			SPT S N=42 (5,6/8,10,12,12) D 54			Very stiff fissured grey silty CLAY. Fissures are very closely to closely spaced, predominatly subhorizontal, rarely vertical, smooth and realished Conscience to headly rare	21.27 -16.11	
- - - - - 21.35-22.85 - -	93 N/A N/A	N/A N/A N/A	Flush: 21.35-22.85 water, 95 %			polished. Occasional to locally rare fine to medium gravel size pockets of fine black sand. Rare light grey fine sandy clay infilled burrows. Rare forams. (LONDON CLAY FORMATION - A2) (LONDON CLAY FORMATION - A2) 21.36 m 1no. subhorizontal fissure (5 - 10 degrees). 21.46 m 1no. subhorizontal fissure (5 - 10 degrees). 21.46 m 1no. subhorizontal fissure (5 - 10 degrees). 21.46 m 1no. subhorizontal fissure (5 - 10 degrees). 21.53 m 1no. subhorizontal fissures, smooth and polished approx 5 mm spacing (5-10	(1.58)	
- - - 22.85-23.28 - 22.85 - -		SF	T S 50 (5,7/9,12,15,14 for 50m D 55	22/02/2011 22.85 m) 23/02/2011 24.35	1800 0.00 0800 1.20	Very stiff slightly sandy to sandy CLAY. Sand is fine to medium. Occasional fine to coarse gravel size pockets of fine black and grey sand. Rare foraminifera. (LONDON CLAY FORMATION - A2)	22.85 -17.69	
- - 22.85-24.35 - -	100 N/A N/A	N/A N/A N/A	Flush: 22.85-24.35 water, 95 %			23.43 m 1no. coarse gravel of pyratised lignite approx 50 mm x 40 mm.	- - (1.50) -	
 24.35-24.77 24.35 		SF N/A	'T S 50 (5,7/8,12,15,15 for 40m D 56	m)		ASSUMED ZONE OF CORE LOSS 4.21 m Possible subhorizontal fissure approx 10 degrees rough and undulated 24.35-24.77 m Stiff to very stiff brown slightly sandy CLAY. Sand is fine to medium. Rare pockets of grey clay. Rare white foraminfera (1mm).	24.35 -19.19	
- - -	TCR	N/A N/A		Date	Time		(0.75)	
Depth	ŚČŔ RQD	lf	Records/Samples	Casing	Water	Stratum continues to 25.10 m		
Groundwater Ent No. Struck Po (m)	ries st strik	e beha	viour	Depth se	aled (m)	Depth Related Remarks * From to (m)	Chiselling Depths (m) T	ime Tools used
abbreviations see ke levels in metres. Stra in depth column. Scale 1:25	tes: For explanation of symbols and previations see key sheet. All depths and reduced les in metres. Stratum thickness given in brackets depth column. NT BH02 at 1:25 408.24 22002011 12:16:29			Project Project No. Carried out	t for	Cable Car for London - Ground Investigation D1002-11 Transport for London	NT She	BH02 et 5 of 10



Drilled PW Logged CT/JC/NE Checked SV	Start 15/02 End 02/02	2/2011 3/2011	Equipment, Methods ar JCB 3CX / Dando 175 / Bc CAT scanned for services. inspection pit from 0.30m i 1.20m to 16.20m depth. R 16.20m to 45.24m depth	nd Remarks eretta T51 / . Broken out to 1.20m de totary cored	S Magneton t from GL pth. Cable drilling us	neter to 0.30m depth. Hand dug > percussion boring from ing wireline techniques from	Depth from to Diameter Casing Depth 0.00m 15.00m 250mm 15.00m 15.00m 16.20m 200mm 16.10m 16.20m 45.24m 146mm 45.24m	Ground Level Coordinates National Grid Chainage	+5.16 mOD E 539850.35 N 180286.36
Samples a	nd T	ests	>			Strata]	
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	(Co	Description ntinued from Sheet 5)	Depth, <i>Level</i> (Thickness)	Legend Backfill/ Instruments
- 24.35-25.85	50 N/A		Flush: 24.35-25.85			ASSUMED ZONE OF C	ORE LOSS	25.10 -19.94	E. NN
-	N/A	N/A N/A N/A	Water, 95 %			Very stiff slightly sandy t CLAY. Sand is fine to m Occasional fine to coars size pockets of fine blac sand. Rare foraminifera. (LONDON CLAY FORM	to sandy edium. .e gravel .k and grey .ATION - A2) 25.47 m 1no. 5 - 10 mm dark ⁻	- (0.37) - 25.47 -20.31	
			If N/A/N/A/N/A			Very stiff locally firm gre CLAY. Sand is fine. Occ	y sandy sasional finante 25 72-25 77 m Grev possible	- (0.30) -	
- 25.85-26.18 - 25.85 		s	If N/A/N/A/N/A PT S 50 (5,8/15,21,14 for 30mn D 57	•)		medium black and brown Occasional to locally free speckling of glauconite. shells and shell fragmer medium gravel size.	n gravel. fissured slightly sandy clay. quent black Occasional tts up to –	_ 25.77 -20.61 25.85 -20.69	
25.85-26.60 	0 N/A N/A		Flush: 25.85-26.60 water, 95 %			SWANSCOMBE MEMB	N - POSSIBLE ER) Idy CLAY.	+	
-		N/A N/A N/A				Sand is fine to coarse. S up to medium gravel siz subrounded fine to medi black flint. (HARWICH FORMATIO OLDHAVEN MEMBER)	Shells are e. Rare ium gravel of IN - POSSIBLE	(1.30)	
26.60-27.35	27 N/A N/A		Flush: 26.60-27.35 water, 95 %			ASSUMED ZONE OF C	ORE LUSS -	-	
		\vdash			ļ	Very dense grey slightly	' silty	- 27.15 <i>-21.99</i>	$ \dots $
27.35-27.73	\square	Ę	If N/A/N/A/N/A PT S 58 (2.4/8.16,25,9 for 5mn	n)	ļ	slightly gravelly SAND. S to medium. Gravel is su	Sand is fine brounded to 27 25 27 73 m Slightly eilty SAND [27.35 -22.19	
- 27.35 - - 27.35-28.10	33 N/A N/A	N/A N/A N/A	D 58 Flush: 27.35-28.10 water, 95 %	,		rounded, fine to medium rarely brown flint. Rare f medium gravel of light g sandstone. Rare fine to sized shell fragments. (HARWICH FORMATIO OI DHAVEN MEMBER)	i of black and ^{27,00-27,10} in oligina only of the ine to rey fine coarse sand IN - POSSIBLE	(0.50)	
 !			í ^I		ļ	ASSUMED ZONE OF C	, CORE LOSS	A 27.85 -22.69	
-		N/A N/A	If N/A/N/A/N/A			Very dense grey slightly to medium SAND. Rare sand sized shell fragmer (LAMBETH GROUP - P HARWICH FORMATION	clayey fine fine to coarse nts. OSSIBLE N)	- 28.10 -22.94 (0.56)	
28.10-28.85 	25 N/A N/A	N/A	Flush: 28.10-28.85 water, 95 %			ASSUMED ZONE OF C	ORE LOSS		
 28.85-29.10 28.85			SPT S 50 (3,7/20,30 for 20mm) D 59			Very dense grey slightly to medium SAND. Rare sand sized shell fragmer (LAMBETH GROUP - P HARWICH FORMATION	silty fine fine to coarse nts. 28.85-29.25 m AZCL OSSIBLE N)	_ 28.66 -23.50 - - -	
- 28.85-29.60 - -	47 N/A N/A		Flush: 28.85-29.60 water, 95 %				29.25-30.35 m Occasional fine to coarse gravel size pockets of clayey sand. 29.40 m 1no. subrounded medium black fint gravel.	-	
-	60	N/A N/A N/A					29.40 III LUCansed tim meyr pockets of very weak lignite (up to 40 x 1mm). 29.60-29.90 m AZCL	(1.98) 	
	N/A N/A	\square	Flush: 29.60-30.35 water, 95 %		T:			1	
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	Stratum continues to 30.64 m	1	<u> </u>	
Groundwater Ent No. Struck Pos (m)	ries st strike	e behav	viour	Depth se	ealed (m)	Depth Related Remarks * From to (m)		Chiselling Depths (m) T	ime Tools used
Notes: For explanation abbreviations see ke levels in metres. Stra in depth column.	on of sy y sheet atum thi	/mbols a t. All de ickness	and pths and reduced given in brackets NT BH02	Project Project N Carried o	o. out for	Cable Car for London - Gro D1002-11 Transport for London	ound Investigation	Borehole NT	• BH02



Drilled PW Logged CT/JC/NE Checked SV	Start 15/02 End 02/02	t 2/2011 3/2011	Equipment, Methods ar JCB 3CX / Dando 175 / Be CAT scanned for services inspection pit from 0.30m i 1.20m to 16.20m depth. R 16.20m to 45.24m depth	nd Remark eretta T51 / . Broken out to 1.20m de otary cored	s Magneton from GL pth. Cable drilling us	neter to 0.30m depth. Hand dug percussion boring from ing wireline techniques from	Depth from to 0.00m 15.00m 15.00m 16.20n 16.20m 45.24n	Diameter Casi n 250mm 15. n 200mm 16. n 146mm 45.	ng Depth 00m 10m 24m	Ground Level Coordinates National Grid Chainage	+5 E 5 N 1	.16 m(39850 80286	OD .35 .36
Samples a	nd T	ests	5	Bata	Time	Strata				Brith I aval	1	Der	. 201/
Depth	SCR RQD	lf	Records/Samples	Date Casing	Water	(Co	Description ntinued from Sheet 6	6)		(Thickness)	Legend	Bac Instru	.ktili/ ments
- - - 30.35-30.68 - 30.35		- S	PT S 50 (4,6/11,20,19 for 30mn D 60	n)		Very dense grey slightly to medium SAND. Rare sand sized shell fragme (LAMBETH GROUP - P HARWICH FORMATION	silty fine fine to coarse nts. OSSIBLE N)	30.35-30.52 r					
- 30.35-31.10 - - -	77 N/A N/A	N/A N/A N/A	Flush: 30.35-31.10 water, 95 %			Very dense grey thickly interlaminated grey sligh fine SAND with grey slig fine SAND. Occasional I gravel size shell fragmer (LAMBETH GROUP - L/ BEDS)	htly silty htly clayey up to medium nts. AMINATED	31 10 <u>-31 14 r</u>		30.64 <i>-25.48</i> (0.42) 31.06 <i>-25.90</i>			11111
- - _ 31.10-31.85 -	95 N/A N/A		Flush: 31.10-31.85 water, 95 %	23/02/201 31.85	1 1800 0.60	Very stiff fissured grey s CLAY. Fissures are extr spaced, randomly orient and polished. Shells are coarse gravel size. (LAMBETH GROUP - Lo BEDS)	helly emely closely ated, smooth fine to OWER SHELLY		 			.///////	
31.85-32.24		SP	T S 50 (7,13/13,13,15,9 for 10m D 61	m)				31.85-32.15 r				\mathbb{N}	
- 31.85 - - 31.85-32.75 -	67 N/A N/A	N/A N/A N/A	Flush: 31.85-32.75 water, 95 %	24/02/201 32.75	1 0800 2.00					(2.29)			
- - - 32.75-33.35	50 N/A N/A		Flush: 32 75-33 35 water, 95 %					32.75-33.05 r	 m AZCL 				
- 22 35 33 65									_	00.05 08.10		N	
- 33.35 	97	N/A N/A N/A	If N/AN/A/N/A			Very stiff fissured grey s with occasional fine to c gravel size pockets of fir coarse sand. Fissures a closely spaced, random smooth and polished. R. to 5 mm x 5mm of greer brown glauconite. Shell are fine to medium grave (LAMBETH GROUP - LO	helly CLAY parse he to y orientated, are pockets ug.74- a and orangishclaye fragments el size. 33.89 DWER SHELLY	33.35-33.40 r 33.89 m Becoming a 2y shelly fine to coars 1 m MID LAMBETH H	m AZCL	(0.54) 33.89 -28.73 (0.26)			
- 33.35-34.85 - - - -	N/A N/A	N/A N/A N/A	34.25-34.85 m 0.60 m CRF			Very dense bluish grey to orangish brown and rare to medium SAND with ra medium gravel size shel (LAMBETH GROUP - Lu MOTTLED BEDS) Strong to very strong gro	nottled ely grey fine are fine to I fragments. OWER			34.15 -28.99 (0.57)			
-			If N/A/N/A/N/A	Data	Tima	occasionally mottled ora and bluish grey CALCRI (LAMBETH GROUP - Lo	ngish brown ETE. OWER	·		34.72 -29.56 (0.30)	, χη × ο × ο × ο × ο × ο × ο × ο × ο × ο ×		
Depth	SCR RQD	lf	Records/Samples	Casing	Water	Stratum continues to 35.02 m	1						
Groundwater Ent No. Struck Pos (m)	ries st strik	e beha	viour	Depth s	ealed (m)	Depth Related Remarks * From to (m)				Chiselling Depths (m) T	ime Too	s used	:
Notes: For explanation	on of sy	/mbols	and	Project		Cable Car for London - Gr	ound Investigation	I		Borehole			
levels in metres. Stra in depth column. Scale 1:25	atum th	ickness	given in brackets NT BH02 24 22/06/2011 12:16:33	Project No Carried or	o. ut for	D1002-11 Transport for London				NT She	BH02 eet 7 of 10	;	



Drilled PW Logged CT/JC/NE Checked SV	Star 15/0 End 02/0	t 2/2011)3/2011	Equipment, Methods ar JCB 3CX / Dando 175 / Br CAT scanned for services inspection pit from 0.30m 1.20m to 16.20m depth. R 16.20m to 45.24m depth.	nd Remark eretta T51 / . Broken out to 1.20m de Rotary cored	t from GL pth. Cable drilling us	neter to 0.30m depth. Hand dug ϶ percussion boring from ing wireline techniques from	Depth from to Diameter Casing Depth 0.00m 15.00m 250mm 15.00m 15.00m 16.20m 200mm 16.10m 16.20m 45.24m 146mm 45.24m	Ground Level Coordinates National Grid Chainage	+5.16 mOD E 539850.35 N 180286.36
Samples a	nd 7	Fest :	S			Strata	•		
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	(Col	Description ntinued from Sheet 7)	Depth, Level (Thickness)	Legend Backfill/
- - _ 34.85-35.70 -	100 N/A N/A	N/A N/A N/A	Flush: 34.85-35.70 water, 95 %			34.15m - 34.72m : MOT 34.72m - 35.02m : Very mottled bluish grey bluis grey and orangish browi gravelly to gravelly SILT	TLED BEDS) stiff to hard h green n slightly . Gravel is	35.02 -29.86 (0.68)	
- - 35.70-35.98 - 35.70 -			SPT S 50 (6,6/21,29 for 55mm) D 63)		angular to subrounded, coarse of strong to mod calcrete. (LAMBETH GI MOTTLED BEDS) Bluish grey slightly silty medium SAND. Occasio predominantly fine to me coarse gravel size of me weak to moderately stro	fine to 53.40 III Kate subrounded of erately weak grey, black and brown flint. ROUP - LOWER 35.70-35.98 m Stiff brown mottled yellowish brown and tight grey sandy CLAY. Sand is fine to nal gravel edium, rarely derately pag calcrete	- - 35.70 <i>-30.54</i>	
 35.70-37.20 	35 N/A N/A	N/A N/A N/A	Flush: 35.70-37.20 water, 95 %			(LAMBETH GROUP - L(MOTTLED BEDS) ASSUMED ZONE OF C	ORE LOSS	(0.97) - - 	
- - - 37.20-37.43 37.20		N/A N/A N/A	37.00-37.20 m 0.20 m CRF SPT S 50 (6,16/50) D 64			Very dense biulsri grey i orangish brown green al slightly clayey fine to me (LAMBETH GROUP - Lo MOTTLED BEDS)	nottled nd grey . dium SAND. OWER	(0.79)	
 37.20-37.90 	100 N/A N/A		Flush: 37.20-37.90 water, 95 %			Dark bluish grey mottled orangish brown slightly f medium glauconitic SAN to rare medium to coars pockets of slightly claye medium sand.	I brown and ine to D. Occasional e gravel size y fine to	37.46 -32.30	
	95 N/A N/A	N/A N/A N/A	Flush: 37.90-38.70 water, 95 %			MOTTLED BEDS)		(1.94)	SPIE
- 38.70-39.40 -	69 N/A N/A		Flush: 38.70-39.40 water, 95 %					-	
- 39.40-39.74 - 39.40 - -		N/A N/A	PT S 50 (6,11/15,16,19 for 35mi D 65	m)		ASSUMED ZONE OF C	to rounded fine-gravel of brown fint. 39.40-39.74 m Very stiff dark greenish blue mottled brown and light grey sandy glauconitic CLAY. Sand is fine to coarse. Frequent pockets (upto 8mm) and partings of brown clay and grey yellow fine to coarse sand. Occasional rounded fine gravel of black and brown filmt. 39.40-40.42 m AZGL	39.40 -34.24	
Depth		lf	Records/Samples	Date	Time Water	Stratum continues to 40.42 m		<u> </u> ·	<u>├──</u>
Groundwater Ent No. Struck Por (m)	ries st strik	ce beha	viour	Depth s	ealed (m)	Depth Related Remarks * From to (m)	-	Chiselling Depths (m) 1	⊥i i
Notes: For explanation abbreviations see ke levels in metres. Stra in depth column.	on of sy y shee atum th	ymbols t. All de iickness	and pths and reduced given in brackets NT BH02	Project Project N Carried o	o. out for	Cable Car for London - Gro D1002-11 Transport for London	ound Investigation	Borehole N1	- BH02



Drilled PW Logged CT/JC/NE Checked SV	Star 15/0 End 02/0	t 2/2011 3/2011	Equipment, Methods and JCB 3CX / Dando 175 / Bo CAT scanned for services inspection pit from 0.30m 1.20m to 16.20m depth. R 16 20m to 45 24m depth	nd Remarks eretta T51 / M . Broken out to 1.20m dep otary cored o	5 Magneton from GL oth. Cable drilling us	heter 0.30m depth. Hand dug percussion boring from ng wireline techniques from	Ground Level Coordinates National Grid Chainage	+5.16 E 5398 N 1802	3 mOD 850.35 286.36
Samples a	nd 1	Fest s	5	-		Strata			
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	Description (Continued from Sheet 8)	Depth, <i>Level</i> (Thickness)	Legend Ins	Backfill strumer
	30 N/A N/A		Flush: 39.40-40.90 water, 95 %			ASSUMED ZONE OF CORE LOSS	-		
- - -			if N/A/N/A/N/A	24/02/2011	1800	Very stiff dark bluish grey mottled green and brown slightly gravelly sandy glauconitic CLAY. Sand is fine to medium. Gravel is subrounded to rounded fine to coarse of brown and	40.42 -35.26 (0.30) 40.72 -35.56		
- - 40 90-41 00			If N/A/N/A/N/A TCR 100, SCR NR, RQD NR	40.90	0.70	black flint. (LAMBETH GROUP - LOWER MOTTLED BEDS)	(0.28)		
- 41.00-41.26 - 41.00 -			SPT S 50 (8,14/21,29 for 30mm D 66	41.00 25/02/2011 40.90	0800 2.10	Very dense grey slightly silty fine to medium glauconitic SAND. Occasional fine to coarse gravel		!	
- _ 41.00-41.75 -	0 N/A N/A	N/A N/A N/A	Flush: 40.90-41.75 water, 95 %			ASSUMED ZONE OF CORE LOSS	(0.75)		
-						41.75-41.79 m AZCLI	- - 41.75 -36.59	,	
- 41.75-42.35 -	93 N/A N/A	N/A N/A N/A	Flush: 41.75-42.35 water, 95 %			to medium glauconitic SAND. Occasional fine to coarse gravel size pockets of clayey sand. (LAMBETH GROUP - LOWER MOTTLED BEDS)	(0.44)		$\left\{ \right\}$
- 42.35-42.58 - 42.35 -			SPT S 50 (7,18/35,15 for 5mm) D 67	42.35		Very dense grey slightly clayey fine to medium SAND with occasional medium to coarse gravel size pockets of slightly silty fine to medium sand. Rare shell fragmente un to			
- 42.35-43.10 -	100 N/A N/A	N/A	Flush: 42.35-43.10 water, 95 %			fine gravel size. (LAMBETH GROUP - POSSIBLE LOWER MOTTLED BEDS)	-		
- -		N/A N/A				43.10-43.58 m AZCL	_ (1.39) 		
- - _ 43.10-43.85 -	36 N/A N/A		Flush: 43.10-43.85 water, 95 %				- - - 43.58 -38.42		
- 43.85-44.18 - 43.85		- si	PT S 50 (6,11/14,15,21 for 30mi D 68	n)43.85		bell fragments. Rare subrounded to rounded, fine to coarse gravel of	-		
-	79	N/A N/A N/A	Elucit: 42 95 45 00			black flint. (LAMBETH GROUP - POSSIBLE LOWER MOTTLED BEDS / UPNOR FORMATION)	- (1.00)		
43.85-45.00 - -	N/A N/A		water, 95 %				(1.00)		
-				25/02/2011 45.00	1800 0.00		-		
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	Stratum continues to 45.24 m			
Groundwater Ent No. Struck Po (m)	tries st strik	ke beha	viour	Depth se	ealed (m)	Depth Related Remarks * From to (m)	Chiselling Depths (m) 1	ime Tools u	used
Notes: For explanati abbreviations see ke evels in metres. Stra in depth column	on of s ey shee atum th	ymbols et. All de iickness	and pths and reduced given in brackets	Project Project No).	Cable Car for London - Ground Investigation D1002-11	Borehole	F BH02	
20010 1:25			NT BH02 AGS	Carried ou	it for	Transport for London	She	eet 9 of 10	



Drilled PW Logged CT/JC/NE Checked SV	Start 15/02/2011 End 02/03/2011	Equipment, Methods ar JCB 3CX / Dando 175 / Bi CAT scanned for services inspection pit from 0.30m 1.20m to 16.20m depth. R 16.20m to 45.24m depth	nd Remarks eretta T51 / Magnetor . Broken out from GL to 1.20m depth. Cable Rotary cored drilling us	neter to 0.30m depth. Hand dug e percussion boring from sing wireline techniques from	epth Ground Level Coordinates National Grid Chainage	+5.16 mOD E 539850.35 N 180286.36
Samples a	nd Test	S		Strata		
Depth	TCR SCR If RQD	Records/Samples	Date Time Casing Water	Description (Continued from Sheet 9)	Depth, <i>Level</i> (Thickness)	Legend Backfill/ Instruments
Samples a Depth 45.00-45.24 45.00	ICR TCR ROD If If ries st strike beha	S Records/Samples PT S 50 (6, 14/21,29 for 15mm D 69 Records/Samples aviour	Date Time Casing Water	Strata Continued from Sheet 9) Very stiff grey sandy CLAY. Sand is :: 44.99m - 44.99 m 1m. 40.9 Iocal fine to coarse gravel size pockets of fine to medium sand and Shell fragments. Rare subrounded to rounded, fine to coarse gravel of Iblack fint. I/LOWER MOTTLED BEDS / UPNOR FORMATION) EXPLORATORY HOLE ENDS AT 45.24 m	Depth, Level (Thickness) nx of of ite. 45.24 -40.08 i	Legend Backfill/ Instruments
Notes: For explanationable of the set of the	on of symbols y sheet. All de tum thickness	and epths and reduced s given in brackets	Project Project No. Carried out for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NT	BH02

PID Readings

Hole ID	Hole Depth (mBGL)	PID Reading (ppm)	Remarks
NT BH02	0.50	0.00	
NT BH02	1.50	3.50	
NT BH02	2.00	4.90	
NT BH02	2.50	6.40	
NT BH02	3.00	13.90	
NT BH02	3.50	0.40	
NT BH02	4.50	0.30	
NT BH02	5.00	0.00	
NT BH02	6.00	0.00	
NT BH02	7.00	0.00	
NT BH02	8.00	0.00	
NT BH02	9.00	0.00	
NT BH02	10.00	3.40	
NT BH02	11.00	0.00	
NT BH02	12.00	0.00	
NT BH02	13.00	0.00	
NT BH02	14.00	0.00	
NT BH02	14.50	0.00	
NT BH02	15.00	0.00	
NT BH02	1.00	0.60	





Equipment, Methods and Remarks Hydraulic Breaker / JCB 3X / Dando 175 / Magnetometer Depth from 0.00m Diameter Casing Depth Ground Level Start 24/02/201 +4.87 mOD **to** 12.45m Drilled PW/SM 200mm 150mm 12.45m 25.00m Coordinates National Grid E 540152.83 Logged CT CAT scanned for services. Hydraulic breaker from GL to 0.05m depth. Hand dug inspection pit from 0.05m to 1.00m depth with breaking out from 0.20m to 0.50m depth. Machine dug trial pit from 1.00m to 2.00m depth with breaking out from 1.30m to 1.60m depth. Cable percussive boring from 2.00m to 12.45m 25.00m N 180702.12 End Checked SV 02/03/201 Chainage Samples and Tests Strata Backfill/ Date Time Description Depth.Level Depth Type & No Records Legend Water Casing (Thickness) strume (MADE GROUND) 0.05 +4.82 ~ CONCRETE slab. 0.20 Δ +4.67 (MADE GROUND) Yellowish brown fine to medium SAND (0.30) (MADE GROUND) . Reinforced CONCRETE. 0.50 +4.37 0.50 m Bl lue plastic membrane. 0.55 m PID = 0.1 ppm ES 200 D 1 4 samples taken 0.55 0.60 (MADE GROUND) Brown sandy GRAVEL with medium cobble content. Gravel is angular to subrounded, fine to coarse of quartz, brick, concrete, flint and wood. Cobbles of brick, concrete and 0.90 B 2 (0.80) wood. Sand is fine to coarse. ES 201 1.00 4 samples taken 1.00 m PID = 0.4 ppm 1.25 W 4 2 samples taken 1.30 +3.57 (MADE GROUND) Reinforced CONCRETE (0.30)24/02/2011 1800 1.20 1.60 +3.27 (MADE GROUND) 1.65 ES 202 D 3 4 samples taken 1.65 m PID = 0.0 ppm_ 1.70 m Dark grey. 1.70 Brown very gravelly fine to coarse SAND with medium cobble content. 25/02/2011 0800 Gravel is angular to subrounded, 1.90-2.10 B 5 1.2 fine to coarse of quartz, brick, 1.90-3.00 m High cobble content. concrete, flint and wood. Cobbles of 2.00 ES 203 4 samples taken 2.00 m PID = 0.0 ppm brick, concrete and wood. (1.40) 2.50 ES 204 4 samples taken 2.50 m PID = 0.0 ppn 3.00-3.45 SPT C ES 205 N=0 (1,-/-,-,-) 4 samples taken 3.00 1.45 3.00 +1.87 3.00 m PID = 0.7 ppm (MADE GROUND) 3.00 3.00 D 6 B 7 Very soft dark brownish grey silty sandy GRAVEL. Gravel is angular to 3.00-3.45 subrounded, fine to coarse of flint, brick, siliceous slag pumice. Sand is fine to coarse. Slight sweet (?) odour. ES 206 3.50 4 samples taken 3.50 m PID = 0.0 ppm 4.00 ES 207 D 8 4 samples taken 4.00 m PID = 0.0 ppm 4.00 4.50-4.95 4.50-4.95 4.50 4.50-4.95 SPT S N=1 (1,-/-,1,-,-) 4.50 2.40 4.50-5.00 m Greenish grev B 10 ES 208 D 9 4 samples taken Date Casing Time Water Depth Type & No Records Stratum continues to 7.00 m Depth Related Remarks * **Groundwater Entries** Chiselling No. Struck Post strike behaviour Depth sealed From to (m) Depths (m) Time Tools used (m) (m) SPT hammer ID SM10 (Er 54%); rod type B Magnetometer scan 1.20 2.00 3.00 4.00 12.45 1.25 water seepage Magnetometer scan Magnetometer scan

Soil Mechanics

		5.00 Magnetometer scan	
Notes: For explanation of symbols and abreviations see key speet. All denths and reduced	Project	Cable Car for London - Ground Investigation	Borehole
levels in metres. Stratum thickness given in brackets in depth column.	Project No.	D1002-11	NS BH01A
NS BH01A AGS	Carried out for	Transport for London	Sheet 1 of 6



Drilled PW/SM Logged CT Checked SV	Start 24/02/2011 End 02/03/2011	Equipment, Methods a 25.00m depth.	and Remarks	5	Depth fromtoDiameterCasing Depth0.00m12.45m200mm12.45m12.45m25.00m150mm25.00m	Ground Level Coordinates National Grid	+4.87 mOE E 540152.83 N 180702.12
Samulas :	and Test				Strata	Glamage	
Depth	Type & No	Records	Date	Time	Description	Depth,Level	Legend Backfi
5.00	D 11	*	Casing	Water	(Continued from Sheet 1) (MADE GROUND) 5.00-5.45 m Silty very gravelly	(Thickness) (4.00)	
_ 5.00 - - - 5.50-5.95 5.50-5.95	ES 209 SPT C B 12 F 200	4 samples taken	5.50	3.00	Very soft dark brownish grey silty sandy GRAVEL. Gravel is angular to subrounded, fine to coarse of flint, brick, siliceous slag pumice. Sand is fine to coarse. Slight sweet (?) odour. 5.50-6.00 m Low cobble content. Cobbles angular of concrete.		
- 5.50 -	ES 210	4 samples taken	25/02/2011	1800			
	D 13 ES 211	* 4 samples taken	28/02/2011	0800	6.00-6.50 m Greenish grey and brownish grey. - - -	-	
- 6.50-7.25 6.50-6.95 -	SPT C B 14	* N=1 (1,-/-,1,-,-) SW=300	6.50	1.90		-	
- 7.00 _ 7.00 _	D 15 ES 212	4 samples taken			(MADE GROUND) Dark grey sandy angular to subangular fine to coarse GRAVEL of slag, brick and quartz with occasional pockets of soft greenish grey slightly sandy gravelly clay	7.00 -2.13 (0.50)	
- 7.50-8.25 - 7.50-7.95 	SPT C B 16	* N=1 (1,-/-,1,-,-) SW=300	7.50	3.10	(up to 100 mm). (MADE GROUND) Very loose black slightly clayey sandy angular to subangular fine to coarse GRAVEL of slag and quartz with occasional pockets of soft black slightly sandy gravelly clay	7.50 -2.63	
8.00 - 8.10 8.10-8.50 -	D 17 ES 213 D 18 B 19	4 samples taken			(up to 120 mm). Strong odour. (MADE GROUND) Very soft black slightly sandy gravelly CLAY with strong odour. Sand is fine to coarse. Gravel is angular to subangular, fine to coarse of slag and brick.	8.00 -3.13 8.10 -3.23	
- 8.50-9.15 8.50-8.95 - 8.50-8.95 -	SPT S D 20 B 21	* N=0 (1,-/-,-,-) SW=200	8.50	5.20	Plastic brown very clayey amorphous ^{8.50-8.95} m Soft brown silty clay. PEAT. Rare grass and wood fragments. (ALLUVIUM)	(1.45)	
- 9.00 9.00 	ES 214 D 22	4 samples taken					
- 9.50-11.40 9.50-9.88 9.50-9.95 9.50-9.95	KRH SPT S D 23 D 24	* k=0.0E+0 m/s 49 (1,6/10,17,22)	9.50	6.10	Dense grey SAND and GRAVEL. Sand is fine to coarse. Gravel is angular to subrounded, fine to coarse of flint and quartz. (RIVER TERRACE DEPOSITS)	9.55 -4.68	
Depth	Type & No	Records	Date Casing	Time Water	Stratum continues to 11.40 m		
Groundwater En No. Struck Po (m)	tries ost strike beha	viour	Depth se	aled (m)	Depth Related Remarks *Fromto (m)5.00Magnetometer scan6.00Magnetometer scan6.50Magnetometer scan7.50Magnetometer scan8.50Magnetometer scan9.50Magnetometer scan	Chiselling Depths (m) 1	Tools used
Notes: For explanati abbreviations see ke levels in metres. Str in depth column.	ion of symbols ey sheet. All de atum thickness	and apths and reduced given in brackets NS BH01A	Project Project No Carried ou). It for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NS	BH01A

Soil Mechanics

Drilled PW/SM Logged CT Checked SV	Start 24/02/2011 End 02/03/2011	Equipment, Methods an Hydraulic Breaker / JCB 3 CAT scanned for services dug inspection pit from 0.0 0.50m depth. Machine du out from 1.30m to 1.60m	nd Remarks X / Dando 17 Hydraulic br 5m to 1.00m g trial pit from denth_Cable	5 / Magn eaker frc depth w 1.00m to percussio	etometer m GL to 0.05m depth. Hand th breaking out from 0.20m to 2.00m depth with breaking e boring from 2.00m to	Ground Level Coordinates National Grid Chainage	+4.87 mOD E 540152.83 N 180702.12
Samples a	nd Tests	\$			Strata		
Depth	Type & No	Records	Date Casing	Time Water	Description (Continued from Sheet 2)	Depth, Level (Thickness)	Legend Backfill/ Instruments
10.00 - 10.00 	ES 215 D 25 SPT C B 26	4 samples taken N=49 (4,8/10,12,13,14)	10.50	6.30	Dense grey SAND and GRAVEL. Sand is fine to coarse. Gravel is angular to subrounded, fine to coarse of flint and quartz. (RIVER TERRACE DEPOSITS)	(1.85)	
- 11.00	D 27					-	
- - 11.50-11.95 - 11.65	UT 28 ES 216	• 25 blows 400 mm rec 4 samples taken	11.50	6.50	Stiff grey slightly sandy slightly 11.40 m Firm garvelly CLAY. Gravel is subangular to angular fine to coarse of quartzite and flint.	- 11.40 -6.53 - -	
- 11.90 11.95-12.00 12.00-12.45	ES 217 D 29 SPT S	4 samples taken N=17 (2,3/4,4,4,5)	11.50	6.50	12.00-12.45 m Slightly gravelly	-	
12.00-12.45 12.00-12.45 	D 30 B 32		28/02/2011	1800 6 50	sandy clay. Sand is fine to coarse.	-	
- 12.40 - -	ES 218	4 samples taken	01/03/2011	0800	-	(2.10)	
- - 13.00-13.45 -	UT 33	27 blows 350 mm rec	13.00	damp	13.00-13.20 m Slightly sandy. Sand is fine.	-	
- 13.45-13.50 13.50-13.95 13.50-13.95 - -	D 34 SPT S D 35	N=22 (2,3/4,5,6,7)	13.00	dry	Stiff to very stiff fissured grey brown and brown CLAY. Fissures are extremely closely spaced to very closely spaced, randomly orientated, rough and matt with occasional	- - 13.50 -8.63 - -	
-					dustings of brown silt on insure surfaces. Occasional partings of light brown fine sand and burrows infilled (upto 4mm) with dark grey clay. Rare white foraminifera (1mm). (LONDON CLAY FORMATION)	-	
- 14.50-14.95 - 14.50 	UT 37 D 36	31 blows 400 mm rec	13.00	dry		-	
Depth	Type & No	Records	Date	Time Water	Stratum continues to 17.95 m	<u></u> − ·	
Groundwater Ent No. Struck Po (m)	tries st strike beha	viour	Depth sea	aled (m)	Depth Related Remarks * From to (m) 11.40 12.45 Bentonite environmental seal	Chiselling Depths (m)	fime Tools used
Notes: For explanati abbreviations see ke levels in metres. Stra in depth column. Scale 1:25	on of symbols and symbols and symbols and symbols atum thickness	and pths and reduced given in brackets NS BH01A .24, 22/06/2011 11:55:25	Project Project No. Carried ou	t for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NS	BH01A neet 3 of 6



Drilled PW/SM Logged CT Checked SV	Start 24/02/2011 End 02/03/2011	Equipment, Methods a 25.00m depth.	nd Remark	KS	Depth from to Diameter Casing Depth 0.00m 12.45m 200mm 12.45m 12.45m 25.00m 150mm 25.00m	Ground Level Coordinates National Grid Chainage	+4 E 54 N 1	87 mOD 40152.83 80702.12
Samples a	and Tests	s			Strata			
Depth	Type & No	Records	Date	Time	Description	Depth,Level	Legend	Backfill/
15.00-15.45 _ 15.00-15.45 _ _ _ _ _ _	SPT S D 39	N=25 (2,4/5,6,7,7)	13.00	dry	Stiff to very stiff fissured grey brown and brown CLAY. Fissures are extremely closely spaced to very closely spaced, randomly orientated, rough and matt with occasional dustings of brown silt on fissure surfaces. Occasional partings of light brown fine sand and burrows infilled (upto 4mm) with dark grey clay. Rare white foraminifera (1mm). (LONDON CLAY FORMATION)	(4.45)		
16.00-16.45 - 16.00 	UT 41 D 40	36 blows 350 mm rec	13.00	dry				
10.40-10.30 16.50-16.95 - - - - - - - - - - - - -	D 42 SPT S D 43	N=26 (3,4/5,6,7,8)	13.00	dry	- - - - - - - -			
- 17.50-17.95 - 17.50 	UT 45 D 44	44 blows 400 mm rec	13.00	dry				
17.95-18.00 18.00-18.45 18.00-18.45 - - - - - - - - -	D 46 SPT S D 47	N=30 (3,5/6,7,8,9)	13.00	dry	Very stiff brown locally structureless rarely fissured slightly sandy CLAY. Fissures are very closely spaced randomly orientated rough and mattt. Sand is fine. Locally frequent white foraminifera (1mm). (LONDON CLAY FORMATION)	17.95 -13.08		
19.00-19.45 - 19.00 -	UT 49 D 48	50 blows 400 mm rec	13.00	dry				2
- 19.45-19.50 - 19.50-19.95 _ 19.50-19.95 	D 50 SPT S D 51	N=34 (4,6/7,8,9,10)	13.00	damp	19.45-19.50 m Frequent[extremely closely spaced fine sand and silt partings. 			
Depth	Type & No	Records	Date	Time	Stratum continues to 22.00 m	(<u>4.05)</u>	<u> </u>	+
Groundwater Ent No. Struck Po (m) 2 19.45 Si	tries ist strike behav ight water see	viour spage	Depth s	sealed (m)	Depth Related Remarks * From to (m)	Chiselling Depths (m) T	ime Tool	is used
Notes: For explanati abbreviations see ke levels in metres. Str in depth column.	ion of symbols a sy sheet. All de atum thickness	and pths and reduced given in brackets	Project Project N Carried (√o. out for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NS	BH01	A



Drilled PW/SM Logged CT Checked SV	Start 24/02/2011 End 02/03/2011	Equipment, Methods a Hydraulic Breaker / JCB 3 CAT scanned for services dug inspection pit from 0. 0.50m depth. Machine du out from 1.30m to 1.60m	nd Remarks 3X / Dando 17 5. Hydraulic bi 05m to 1.00m g trial pit from depth_Cable	i 75 / Magn reaker frc 1 depth w 1 1.00m t percussi	Depth from to Diameter Casing Depth 0.00m 12.45m 200mm 12.45m 0.00m 12.45m 25.00m 12.45m 12.45m 25.00m 150mm 25.00m ve boring from 2.00m to to 10.00m 10.00m	Ground Level Coordinates National Grid Chainage	+4. E 54 N 18	.87 mOD 40152.83 30702.12
Samples a	and Tests	\$			Strata	l		
Depth	Type & No	Records	Date Casing	Time Water	Description (Continued from Sheet 4)	Depth,Level (Thickness)	Legend	Backfill/
- - - - 20.50-20.95 - 20.50	UT 53 D 52	52 blows 380 mm rec	13.00	damp	Very stiff brown locally structureless rarely fissured slightly sandy CLAY. Fissures are very closely spaced randomly orientated rough and matt. Sand is fine. Locally frequent white foraminifera (1mm). (LONDON CLAY FORMATION)			
20.95-21.00 21.00-21.45 21.00-21.45 21.00-21.45 - -	D 54 SPT S D 55	N=36 (3,5/7,9,10,10)	13.00	damp				
- 22.00-22.45 - - -	UT 57 D 56	58 blows 350 mm rec	13.00	damp	Stiff to very stiff possibly structureless dark brown slightly sandy CLAY. Sand is fine. Occasional partings of light brown fine sand and dark grey silt.	22.00 -17.13		
22.45-22.50 22.50-22.95 22.50-22.95 	D 58 SPT S D 59	N=37 (3.5/7,9,10,11)	13.00	damp	Occasional white foraminifera (1mm). Occasional pockets (up to 10mm) of light grey silt. (LONDON CLAY FORMATION)			
- 23.50-23.95 23.50 - -	UT 61 D 60	63 blows 400 mm rec	13.00	damp	-	(3.00)		
- 23.95-24.00 - 24.00-24.45 24.00-24.45 	D 62 SPT S D 63	N=39 (4,6/8,9,10,12)	13.00	damp 1 1800				
-		I	13.00	damp	-			Y//,
Donth	Tuno 8 No	Basarda	Date	Time		┨────-	[···+	
Groundwater Ent No. Struck Po (m)	tries Ist strike behav	viour	Casing Depth se	Water aled (m)	Depth Related Remarks * From to (m)	Chiselling Depths (m) T	ime Tools	s used
Notes: For explanati abbreviations see ke levels in metres. Str. in depth column.	ion of symbols a ey sheet. All de atum thickness	and pths and reduced . given in brackets NS BH01A	Project Project No Carried ou). Jt for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NS	BH01/	4



Drilled PW/SM Logged CT Checked SV	Start 24/02/2011 End 02/03/2011	Equipment, Methods a 25.00m depth.	and Remarks	Depth from to Diameter Casing Depth 0.00m 12.45m 200mm 12.45m 12.45m 25.00m 150mm 25.00m	Ground Level +4.87 Coordinates E 5401: National Grid N 1807 Chainage	mOD 52.83 02.12
Samples a	Ind Tests	s		T Strata	1	
Depth	Type & No	Records	Date Time	Description	Depth, Level Legend	ackfill/
25.00	D 64		Casing water	(Continued from Sheet 5)	(Thickness) - Inst 25.00 -20.13	rumen
	D 64			EXPLORATORY HOLE ENDS AT 25.00 m ⁻² 25.00 m Slightly sandy. Sand is fine. Rare shell fragments.		
Denth	Turna & No	Baparda	Date Time	-	-	
Deptn Groundwater En No. Struck Po (m)	tries Ist strike beha	viour	Casing Water Depth sealed (m)	Depth Related Remarks * From to (m)	Chiselling Depths (m) Time Tools us	sed
Notes: For explanat	ion of symbols	and	Project	Cable Car for London - Ground Investigation	Borehole	
abbreviations see ke levels in metres. Str	ey sheet. All de ratum thickness	pths and reduced oven in brackets	Project	Cable Carlor London - Ground Investigation		
in depth column.		NS BH01A	Project No. Carried out for	D1002-11 Transport for London		

PID Readings

Hole ID	Hole Depth (mBGL)	PID Reading (ppm)	Remarks
NS BH01A	0.55	0.10	
NS BH01A	1.00	0.40	
NS BH01A	1.65	0.00	
NS BH01A	2.00	0.00	
NS BH01A	2.50	0.00	
NS BH01A	3.00	0.70	
NS BH01A	3.50	0.00	
NS BH01A	4.00	0.00	





Soil Mechanics

Drilled PW/NE Logged CT/JC Checked SV	Start 02/03/2011 End 05/04/2011	Equipment, Methods a Dando 175 / Beretta T51 CAT scanned for services dug inspection pit from 0. 1.20m to 12.10m depth. F 12.10m to 60.70m depth	nd Remarks / Magnetome 3. Concrete co 38m to 1.20m Rotary cored co SPT from 60	ter bred from depth. C drilling us 70m to 6	GL to 0.38m depth. Hand able percussive boring from 10.20m bit 250mm 10.20m 10.20m 12.10m 200mm 11.95m 12.10m 60.70m 146mm 60.70m 10.20m 12.10m 146mm 60.70m	h Ground Level Coordinates National Grid Chainage	+2.76 mOD E 539868.52 N 180300.77
Samples a	nd Tests	6	Dete	T	Strata	- Denth / secol	
Depth	Type & No	Records	Casing	Water	Description	(Thickness)	Legend Instruments
0.00-0.14 	CC 1 CC 2		02/03/2011	1800 dry	(MADE GROUND) Strong black MACADAM. 70 to 80% aggregate angular fine to medium pink granite and blue green possible limestone gravel. 5% small to medium voids 0.25 m 5 mm rebar mesh 0.33 m Honeycomb voids	- 0.14 +2.62 	
- 0.50 0.50-0.60 - 0.50 - 0.75	D 1 B 2 ES 200 ES 201	4 samples taken 4 samples taken	03/03/2011	0800 dry	(MADE GROUND) Strong light grey CONCRETE. 70 to 80% aggregate subangular to rounded fine to medium flint gravel. 10 to 15% small to medium voids.	(0.62)	
- - 1.00 - 1.00 - 1.20-1.65	ES 202 D 3 SPT S	4 samples taken * N=1 (1,-/-,1,-,-)		damp	(MADE GROUND) Soft brownish grey slightly sandy slightly gravelly CLAY. Sand is fine to coarse. Gravel is angular to subangular, fine to coarse of brick, quartz, concrete and rare metal (up	- 1.00 +1.76 1.20 +1.56	
1.20-1.65 - 1.20-1.65 - 1.50	D 4 B 5 ES 203	• 4 samples taken			to 30 mm). Soft brownish grey CLAY with occasional shell fragments. (ALLUVIUM)		
-					Soft grey CLAY with rare black speckles. (ALLUVIUM)	-	
2.00 2.00 -	ES 204 D 6	4 samples taken				-	-
- 2.20-2.65 - -	UT 7	8 blows 310 mm rec	1.70	dry			
2.65 	D 8					- - (3.00) -	
3.00 3.00	ES 205 D 9	4 samples taken					
- - 3.20-3.65 3.20-3.65 - 3.20-3.65 - - -	SPT S D 10 B 11	• N=5 (1,-/1,1,1,2)	3.20	dry			
- 4.00 - 4.00 - 4.20-4.65 - 4.20-4.65 -	D 12 ES 206 UT 13 B 15	4 samples taken • 8 blows 400 mm rec	3.20	dry	Plastic brown clayey amorphous PEAT interbedded with soft grey clay. Occasional wood and grass fragments.	- - - - - 4.20 -1.44	
- 4.65-4.70 -	D 14				(ALLUVIUM)	-	
Depth	Type & No	Records	Date Casing	Time Water	Stratum continues to 6.10 m	+	
Groundwater Ent No. Struck Po (m)	tries st strike beha	viour	Depth se	ealed (m)	Depth Related Remarks * From to (m) 1.20 9.10 SPT hammer ID SM10 (Er 54%); rod type B 1.20 Magnetometer scan 2.20 Magnetometer scan 3.20 Magnetometer scan 4.20 Magnetometer scan	Chiselling Depths (m) 1	lime Tools used
Notes: For explanati abbreviations see ke levels in metres. Stra in depth column. Scale 1:25	on of symbols by sheet. All de atum thickness 408.	and pths and reduced given in brackets NT BH01 24 22/06/2011 12:15:23	Project Project No Carried ou	It for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NT	F BH01 eet 1 of 13



Drilled PW/NE Logged CT/JC Checked SV	Start 02/03/2011 End 05/04/2011	Equipment, Methods a Dando 175 / Beretta T51 CAT scanned for service dug inspection pit from 0 1.20m to 12.10m depth.	and Remarks / Magnetomet s. Concrete cc .38m to 1.20m Rotary cored c	; ∷er ∋red from ⊨depth. C drilling us	Depth from to Diameter Casing Depth 0.00m 10.20m 250mm 10.20m GL to 0.38m depth. Hand 10.20m 12.10m 200mm 11.95m able percussive boring from na wireline techniques from 12.10m 60.70m 146mm 60.70m	Ground Level Coordinates National Grid Chainage	+2.76 mOD E 539868.52 N 180300.77
Samples a	I and Tests	12 10m to 60 70m depth	SPT from 60	70m to 6	107m depth		
Depth	Type & No	Records	Date	Time	Description	Depth, Level	Legend Backfill/
5.00	D 16	4	Casing	Water	Plastic brown clayey amorphous PEAT	(Thickness)	
_ 5.00	ES 207	4 samples taken	4.90	dry	interbedded with soft grey clay. Occasional wood and grass fragments.	(1.90)	511/2
5.20-5.65	D 17	* 14-3 (1,1/1,1,1,2)	4.90	ury	(ALLUVIUM)	4	
_				ĺ		-	
-						1	
						1	
_				ĺ			
_						-	
6.00 6.00	D 18 ES 208	4 samples taken			-	-	
6.10	D 19	N=2 (1 (1 1 1)	4.00	den	Soft grey silty CLAY.	6.10 -3.34	
- 0.20-0.05 6.20-6.65 - 6.20-6.65	D 20 B 21	* N=3 (1,-/1,-,1,1)	4.90	ary	(ALLUVIUM) 6.20-6.65 m Sandy organic [®] material.		
-	52.			ĺ		-	
				ĺ		-	× ×
_					L	1	××××
-						(1.40)	
_				ĺ			
	ES 209 D 22	4 samples taken		ĺ	-	4	××××
- 7 20 7 65	edt e	N-E (1 1/1 1 1 2)	6.40	den		1	
7.20-7.65	D 23	* N=5 (1,1/1,1,1,2)	6.40	úi y		1	
F						-	
Γ					Plastic brown clavev amorphous PEAT.	7.50 -4.74	
					Occasional wood and grass fragments.	-	
Ĺ					(ALLOVION)	1	
-						4	
- 8.00	ES 210	4 samples taken				(1 10)	
						- (
- 8.20-8.oo	0125	* 13 blows 370 mm rec	8.00	ary		1	
-						4	
-			03/03/2011	1800		-	
8.65-8.70	D 26		8.70	1.50	Medium dense grey SAND and GRAVEL.	8.60 -5.84	P
- 8.70	EW 27		07/03/2011	0800	Sand is fine to coarse. Gravel is angular to subrounded, fine to	1	° • • •
-			8.70	1.80	coarse of quartz and flint. (RIVER TERRACE DEPOSITS)	4	· · · · · ·
	ES 211	5 samples taken			9.00 m PID = 0.4 ppm.	-	
				C 40		(1.10)	0.00
9.20-9.65 9.20-9.65	B 29	* N=11 (3,3/3,2,3,3)	9.20	2.40		1	
-						-	
_						-	0 0 0 0
							P
- 9.70	D 30			ĺ	Soft grey slightly sandy slightly	9.70 -6.94	
_					coarse. Gravel is angular to	4	· - • •
Depth	Type & No	Records	Date	Time	Stratum continues to 10.20 m	+	
Groundwater Er	Itries		Casing	Water	Depth Related Remarks *	Chiselling	
No. Struck Po (m) 1 8.60 Ro	ose to 1.50 m	viour nafter 20 minutes. Fast	Depth se Inflow	aled (m) -	Fromto (m)5.206.20Magnetometer scan6.207.20Magnetometer scan7.208.20Magnetometer scan8.209.20Magnetometer scan9.2010.20Magnetometer scan	Depths (m) T	ïme Tools used
Notes: For explanat	tion of symbols	and	Project		Cable Car for London - Ground Investigation	Borehole	
abbreviations see k levels in metres. Str in depth column	ey sheet. All de ratum thickness	s given in brackets	Project No		D1002-11	NT	⁻ BH01
Scale 1:25		NT BH01 AGS	Carried ou	t for	Transport for London	She	pot 2 of 13



Drilled PW/NE Logged CT/JC Checked SV	Start 02/03 End 05/04	3/2011 4/2011	Equipment, Methods ar Dando 175 / Beretta T51 / CAT scanned for services. dug inspection pit from 0.3 1.20m to 12.10m depth. R 12.10m to 60.70m denth. 3	1d Remarks Magnetomet . Concrete cc 38m to 1.20m .otary cored c	ter pred from depth. C drilling us	GL to 0.38m depth. Hand able percussive boring from g wireline techniques from 0.7m double	Ground Level Coordinates National Grid Chainage	+2.76 mOD E 539868.52 N 180300.77
Samples a	nd T	ests	<u> </u>	<u></u>	/////	Strata		
Depth	Туре	& No	Records	Date Casing	Time Water	Description (Continued from Sheet 2)	Depth,Level (Thickness)	Legend Backfill/ Instruments
10.00 10.00	ES 2	212 31	5 samples taken			10.00 m PID = 0.1 ppm.	(0.40)	
- 10.20-10.65 10.20 10.20 10.20-10.50	UT D B	33 32 35	* 29 blows 300 mm rec	9.20	2.40	9.70m - 9.80m : subrounded, tine to coarse of quartz and flint. (RIVER TERRACE DEPOSITS)	- 10.20 <i>-7.44</i>	
10.45 	ES :	213	5 samples taken			9.80m - 10.20m : Grey SAND and GRAVEL. Sand is fine to coarse. Gravel is angular to subrounded, first to subrounded,	+	
	D S ES S	34 214	5 samples taken			(RIVER TERRACE DEPOSITS) Stiff fissured grey brown CLAY. 10.70 m PID = 0.8 ppm.	-	
	D ;	36				Fissures are extremely to very closely spaced, randomly orientated, smooth, matt and occasionally polished with rare dustings of fine brown sand and dark brown staining. U ONDON CLAY EORMATION	(1.90)	
- 11.20 -	ES 2	215	5 samples taken			(LONDON CLAT FORMATION) 11.20 m PID = 0.0 ppm	-	
-							-	
				07/03/2011 11.95	1800 damp		12 10 -9 34	
- - - 12.10-12.70	100 N/A	NA NA	*	23/03/2011 11.95	0800	Firm to stiff fissured grey slightly sandy CLAY. Fissures are very closely spaced, randomly orientated, smooth, matt and rarely polsihed.	(0.60)	
- - 12.60-12.70	N/A	NA	D 100	23/03/2011 12.70	1800	(LONDON CLAY FORMATION - A2)		
-			•	24/03/2011 12.70	0800	ASSUMED ZONE OF CORE LOSS DUE TO SELF BORING PRESSUREMETER TEST POCKET	12.70 -0.07	
12.70-14.20 - -	0 N/A N/A	NR NR NR	Flush: 12.70-14.20 water, 99 %				(1.80)	
-				24/03/2011 14.20	1700	-	-	
-			*	25/03/2011 14.20	0800	Verv stiff grey slightly sandy CLAY. 14,50-14,96 m Occasional	- - 14.50 <i>-11.74</i>	
- - 14.80-15.25 - 14.80 - 14.20-15.70	80 N/A N/A	NA NA NA	SPT S N=30 (3,4/6,6,9,9) D 37 Flush: 14.20-15.70 water, 99 %			Sand is fine to medium. Occasional foram fossils (up to 3 mm x 1 mm). Occasional pockets (up to 20 mm x 104.74 m 1no. pocket (20 mm x 20 mm) of fine to medium black sand. (LONDON CLAY FORMATION - A2) 14.74-14.75 m 1no. pyrite nodule (20 mm x 30 mm).	(0.60)	
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	Stratum continues to 15.10 m		
Groundwater Ent No. Struck Po: (m)	tries st strike	e behav	viour	Depth se	⊧aled (m)	Depth Related Remarks *Fromto (m)10.2011.20Bentonite environmental seal12.1061.07SPT hammer ID SM31 (Er 55%); rod type NWY13.20SBP test.14.30SBP test.	Chiselling Depths (m) T	ïme Tools used
Notes: For explanation abbreviations see ke levels in metres. Stra in depth column.	on of sy y sheet atum thi	mbols a All de ckness	and pths and reduced given in brackets NT BH01	Project Project No Carried ou	It for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NT	BH01



Drilled PW/NE Logged CT/JC Checked SV	Start 02/03 End 05/0	3/2011 4/2011	Equipment, Methods ar Dando 175 / Beretta T51 / CAT scanned for services dug inspection pit from 0.: 1.20m to 12.10m depth. F	Magnetome Magnetome Concrete cc 38m to 1.20m Cotary cored (s ored from a depth. (drilling us	Depth from to Diameter Casing Depth 0.00m 10.20m 250mm 10.20m GL to 0.38m depth. Hand 10.20m 12.10m 200mm 11.95m able percussive boring from ing wireline techniques from 12.10m 60.70m 146mm 60.70m	Ground Level Coordinates National Grid Chainage	+2 E 5 N 1	2.76 mOD 39868.52 180300.77
Samples a	nd T	est	<u>12 10m to 60 70m depuis</u>	SPL from pu	70m to e	Strata	l		
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	Description (Continued from Sheet 3)	Depth,Level (Thickness)	Legend	Backfill/
		NA NA NA		Gasing	TTULC.	Very stiff grey slightly sandy CLAY. 14.77-14.82 m Rare lignite Sand is fine to medium. Occasional fragments (up to 15 mm x 10 mm). – foram fossils (up to 3 mm x 1 mm). Occasional pockets (up to 20 mm x 10 mm) of fine to medium black sand. (LONDON CLAY FORMATION - A2)	(0.46)		
- - 15.60-15.70 - - -			D 101			Very stiff fissured grey CLAY. Fissures are very closely to closely spaced, predominatly subhorizontal, smooth and matt. Occasional foram fossils (up to 3 mm x 1 mm). Rare partings of greyish brown fine to medium sand. (LONDON CLAY FORMATION - A2)	15.56 -12.80		
	100 N/A N/A		Flush: 15.70-17.20 water, 99 %			Very stiff grey slightly sandy CLAY. Sand is fine to medium. Occasional foram fossils (up to 3 mm). Rare to occasional partings of black and grey fine to medium sand. (LONDON CLAY FORMATION - A2)			
-		NA NA NA					(2.40)		
- 17.10-17.20 - - -			D 102						
- - 17.20-18.70	96 N/A N/A		Flush: 17.20-18.70 water, 99 %			Very stiff fissured grey CLAY. Fissures are very closely to closely spaced, predominatly subvertical, occasionally subhorizontal smooth.	17.96 -15.20		
- - - 18.53-18.63 - -		NA NA NA	D 103 18.63-18.70 m 0.07 m CRF			outside in any construction of the interview of the intervie	(1.24)		
- - - - - 19.40-19.50 18.70-20.20 -	100 N/A N/A		Flush: 18.70-20.20 104 water, 99 %			Very stiff fissured grey slightly sandy CLAY. Fissures are very closely to closely spaced, vertical to subvertical, rarely subhorizontal, smooth and matt. Sand is fine to medium. Rare to	19.20 -16.44		
- - Depth	TCR SCR RQD tries	If	Records/Samples	Date Casing	Time Water	occasional foram fossils (up to 2 mm). Rare pockets (up to 5 mm x 10 mm) of black fine to medium sand. (LONDON CLAY FORMATION - A2) (50 mm x 45 mm x 25 mm). Stratum continues to 22.24 m	Chiselling		
No. Struck Pos (m) Notes: For explanatic abbreviations see ke levels in metres. Stra in depth column.	on of sy y shee atum th	ymbols t. All de ickness	and apths and reduced s given in brackets	Depth se Project Project No	aled (m)	Cable Car for London - Ground Investigation D1002-11	Borehole	ime Too BH0'	ls used

Soil Mec	hanics

Drilled PW/NE Logged CT/JC	Start 02/03 End	t 3/2011	Equipment, Methods ar Dando 175 / Beretta T51 / CAT scanned for services dua inspection pit from 0.3	1d Remarks Magnetometer Concrete cored fr 38m to 1.20m depth	from C	Depth from to Diameter Casing Depth 0.00m 10.20m 250mm 10.20m SL to 0.38m depth. Hand 10.20m 12.10m 200mm 11.95m 12.10m 60.70m 146mm 60.70m 60.70m	Ground Level Coordinates National Grid	+2.76 mOD E 539868.52 N 180300.77
Checked 5v	05/04	4/2011	1.20m to 12.10m depth. Ro 12.10m to 60.70m depth	otary cored drilling	g usin to 61	g wireline techniques from 07m denth	Chainage	
Samples a	nd T	ests	;	Data Tim	_	Strata	Derth Loval	Beekfill/
Depth	SCR RQD	lf	Records/Samples	Casing Wate	ne ter	Uescription (Continued from Sheet 4)	(Thickness)	Legend Instruments
		NA NA NA				Very stiff fissured grey slightly sandy CLAY. Fissures are very closely to closely spaced, vertical to subvertical, rarely subhorizontal, smooth and matt. Sand is fine to medium. Rare to occasional foram fossils (up to 2 mm). Rare pockets (up to 5 mm x 10 mm) of black fine to medium sand. (LONDON CLAY FORMATION - A2)	(3.04)	
20.20-21.70 21.05-21.28 21.28-21.38	90 N/A N/A		Flush: 20.20-21.70 water, 99 % CS 105 D 106					
- - 21.79-22.01 - - 22.01-22.11 -			CS 107 D 108			21.55-21.70 m AZCL 		
21.70-23.20 	74 N/A N/A	NA NA NA	If NA/NA/NA Flush: 21.70-23.20 water, 99 %			Very stiff grey slightly sandy slightly gravelly slightly galuconitic CLAY. Sand is fine to medium. Occasional shell fragments (up to 20 mm x 10 mm). Gravel is rounded, fine of black flint. (HARWICH FORMATION - SWANSCOMBE MEMBER) Very stiff fissured grey slightly glauconitic CLAY. Fissures are very	22.24 -19.48 22.47 -19.71 (0.34) 22.81 -20.05	
-				25/03/2011 180 23.20 28/03/2011 080 23.20	300 	closely to closely spaced, subvertical, smooth and polished. Rare shell fragments (up to 10 mm). (HARWICH FORMATION - SWANSCOMBE MEMBER) ASSUMED ZONE OF CORE LOSS	(1.89)	
23.20-24.70 	0 N/A N/A	NA NA NA	Flush: 23.20-24.70 water, 99 %					
-	TCR		•	Date Time	10	ASSUMED ZONE OF CORE LOSS DUE TO SELF BORING PRESSUREMETER TEST POCKET	24.70 -21.94	
Depth	ŠČR RQD	lf	Records/Samples	Casing Wate	er	Stratum continues to 25.97 m		
Groundwater Ent No. Struck Pos (m)	ries st strike	e behav	<i>r</i> iour	Depth sealed (m) Project	i i)	Depth Related Remarks * From to (m) 24.90 25.70 Borehole caved in. Collapsed after SBP pulled from hole. Cable Car for London - Ground Investigation	Chiselling Depths (m) T	ime Tools used
abbreviations see ke levels in metres. Stra in depth column.	y sheet itum thi	t. All dej ickness	pths and reduced given in brackets NT BH01	Project No. Carried out for	ם ד י	01002-11 Transport for London	NT She	BH01



Drilled PW/NE Logged CT/JC Checked SV	Start 02/03 End 05/04	t 3/2011 4/2011	Equipment, Methods ar Dando 175 / Beretta T51 / CAT scanned for services dug inspection pit from 0.3 1.20m to 12.10m depth. R 12.10m to 60.70m depth	nd Remarks Magnetome Concrete or 38m to 1.20m totary cored of SPT from 60	s eter ored from n depth. C drilling us 0 70m to f	GL to 0.38m depth. Hand able percussive boring from ing wireline techniques from 10.7m depth	Ground Level Coordinates National Grid Chainage	+2.76 mOD E 539868.52 N 180300.77
Samples a	nd T	este	5			Strata		
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	Description (Continued from Sheet 5)	Depth, <i>Level</i> (Thickness)	Legend Backfill/ Instrument
24.70-26.20	15 N/A N/A		• Flush: 24.70-26.20 water, 99 %			ASSUMED ZONE OF CORE LOSS DUE TO SELF BORING PRESSUREMETER TEST POCKET	(1.27)	
25.97-26.07 			d 109 If NA/NA/NA			Grey clayey fine to medium SAND. Occasional bands (up to 10 mm thick) of very stiff grey clay. Rare lignite fragments (up to 8 mm x 5 mm). Rare shell fragments (up to 10	25.97 -23.21 - 26.20 -23.44	
26.20-27.70 27.26-27.36	77 N/A N/A	NA NA NA	Flush: 26.20-27.70 water, 99 % D 110			(HARWICH FORMATION - OLD HAVEN MEMBER) Laminated grey slightly clayey fine to medium SAND with extremely closely spaced thin laminae of light and dark grey sand, black lignite and rare clay. Rare shell fragments (up to 10 mm x 5 mm). (HARWICH FORMATION - OLD HAVEN MEMBER) 27.07 m 1no. pocket (25 mm x 5 ⁻ mm) of lignite. ASSUMED ZONE OF CORE LOSS	(1.16) 27.36 -24.60	
-		NA NA NA		28/03/2011 27.70 29/03/2011 27.70	1 1800	DUE TO SELF BORING PRESSUREMETER TEST POCKET 27.70-28.80 m 88.4mm diameter self boring pressuremeter test pocket.	(1.53)	
27.70-29.20 	27 N/A N/A		Flush: 27.70-29.20 water, 99 %			-	-	
- 28.80-29.22 28.80 29.10-29.20 		SP NA NA	F S 50 (6,9/11,15,14,10 for 45m D 38 D 111	im)		28.80 m Stiff locally fissured grey brown CLAY. Fissures are extremely closely spaced, are up to 25 mm x 15 mm. Frequent partings of fine grey sand and light brown silt. (LAMBETH GROUP - LAMINATED	28.89 -26.13	
_ 29.47-29.57 		NA	D 112			BEDS / POSSIBLE LOWER SHELLY BEDS) Delished fissures 29.31-29.37 m Light brown 29.44-29.57 m Very shelly clay. ASSUMED ZONE OF CORE LOSS	29.57 -26.81	
29.20-30.30 	34 N/A N/A			Date	Time			
Depth Groundwater Ent No. Struck Po (m)	st strik	lf e beha	Records/Samples	Depth se	Water ealed (m)	Stratum continues to 31.19 m Depth Related Remarks * From to (m) 25.20 SBP test. 28.30 SBP test.	Chiselling Depths (m) Ti	ime Tools used
Notes: For explanati abbreviations see ke levels in metres. Stra in depth column. Scale 1:25	on of sy y sheel atum thi	/mbols at t. All de ickness	and pths and reduced given in brackets NT BH01 .24 22/06/2011 12:15:31	Project Project No Carried ou	o. ut for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NT She	BH01

Drilled PW/NE Logged CT/JC Checked SV	Star 02/0 End 05/0	t 13/2011 14/2011	Equipment, Methods an Dando 175 / Beretta T51 / CAT scanned for services dug inspection pit from 0.3 1.20m to 12.10m depth. R 12 10m to 60 70m depth	nd Remarka Magnetome Concrete c 38m to 1.20m otary cored SPT from 60	s eter pred from depth. C drilling us 20m to 6	GL to 0.38m depth. Hand able percussive boring from 10.20m borne for the constraint of the constraint	Ground Level Coordinates National Grid Chainage	+2.76 mOD E 539868.52 N 180300.77	
Samples a	and T	<u>Fest</u>	5			Strata			
Depth	SCR RQD	lf	Records/Samples	Casing	Water	Continued from Sheet 6)	(Thickness)	Legend Instru	iment
		NA NA NA	Flush: 29.20-32.20 water, 99 %			ASSUMED ZONE OF CORE LOSS	(1.62)		
- 30.30-32.20 - 31.30-31.73 - 31.30 	53 N/A N/A		SPT S 50 (25/50 for 50mm) If NA/NA/NA D 39 If NA/NA/NA			Recovered as very soft shelly CLAY. Shells and shell fragments up to 20 mm x 10 mm. (LAMBETH GROUP - POSSIBLE LOWER SHELLY BEDS) Very strong light grey CAL CRETE	31.19 -28.43 31.44 -28.68		
- - - - 32.10-32.20 -		NA NA NA	D 113	29/03/201 ⁻ 32.20	I 1700	very stirft grey mottled brown and rarely red sandy CLAY. Sand is fine to medium. Occasional pockets (up to 20 mm x 20 mm) of	31.69 -28.93		
-				30/03/201	0700	white calcrete. (LAMBETH GROUP - LOWER	32.36 -29.60		И
- 32.42-32.72 	98 N/A N/A	NA NA NA	If NA/NA/NACS 114 D 115 Flush: 32.20-33.50 water, 99 %	32.20		MOTTLED BEDS) 32.42-32.71 m Clayey fine to[/ Very stiff blueish grey mottled medium sand. brown and rarely red very sandy . CLAY. Sand is fine to medium. . (LAMBETH GROUP - LOWER . MOTTLED BEDS) .	(1.12)		
- 33.50-33.80 - 33.50 		NA NA NA	SPT S 50 (9,13/20,30) D 40			ASSUMED ZONE OF CORE LOSS DUE TO SELF BORING PRESSUREMETER TEST POCKET 33.50-34.50 m 88.4mm diameter self boring pressuremeter test pocket.	33.48 -30.72 (0.57)		
- - 33.50-35.20 - - -	68 N/A N/A		Flush: 33.50-35.20 water, 99 %			Dark blueish green mottled brown and rarely red very clayey to clayey fine to medium SAND. (LAMBETH GROUP - LOWER MOTTLED BEDS)	34.05 -31.29		
Donéh	ICR	14	Pocordo/Comulas	Date	Time	Stratum continues to 36.40 m	 		
Groundwater En No. Struck Po (m)	tries	ke beha	viour	Casing	Water ealed (m)	Depth Related Remarks * From to (m) 30.80 SBP test. 34.00 SBP test.	Chiselling Depths (m) T	ime Tools use	d
Notes: For explanat abbreviations see k levels in metres. Str in depth column. Scale 1:25	ion of s ey shee atum th	ymbols et. All de iickness 408	and pths and reduced given in brackets NT BH01 _24_22/06/2011 12:15:32	Project Project No Carried ou). It for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NT	BH01	

NT BH01

ØD
Soil Mechanics



Drilled PW/NE Logged CT/JC Checked SV	Star 02/03 End 05/04	t 3/2011 4/2011	Equipment, Methods ar Dando 175 / Beretta T51 / CAT scanned for services dug inspection pit from 0.3 1.20m to 12.10m depth. R 12 10m to 60 70m depth	nd Remarks Magnetome Concrete or 38m to 1.20n totary cored SPT from 60	s eter ored from n depth. C drilling us 1 70m to 6	GL to 0.38m depth. Hand able percussive boring from ing wireline techniques from 1.07m denth	Depth from to Diameter Casing Dept 0.00m 10.20m 250mm 10.20m 10.20m 12.10m 200mm 11.95m 12.10m 60.70m 146mm 60.70m	n Ground Level Coordinates National Grid Chainage	+2.76 mOD E 539868.52 N 180300.77
Samples a	nd T	est	S			Strata	•	1	
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	(Co	Description ntinued from Sheet 7)	Depth, <i>Level</i> (Thickness)	Legend Backfill/ Instruments
35.07-35.17 		NA NA NA	D 116 35.17-35.20 m 0.03m CRF			Dark blueish green mott rarely red very clayey to fine to medium SAND. (LAMBETH GROUP - L' MOTTLED BEDS)	led brown and clayey OWER	- (2.35) -	
_ 35.58-35.88 - - 35.20-36.40 _ 35.88-35.98	100 N/A N/A		CS 117 Flush: 35.20-36.40 water, 99 % D 118					-	
-				30/03/2017 36.40	1 1700			- - - 36.40 -33.64	
-		NR NR NR	•	31/03/2011 36.40	I 0700	ASSUMED ZONE OF C DUE TO SELF BORING PRESSUREMETER TE	ORE LOSS 36.40-37.40 m 66.4mm daimeter self boring pressuremeter test ST POCKET	- - - - - - (1.03)	
- 36.40-38.20 - 37.40-37.70 - 37.40 -	43 N/A N/A		Flush: 36.40-38.20 water, 99 % SPT S 50 (6,10/18,32) D 41 If NA/NA/NA			Very dense green slight slightly sandy rounded f coarse GRAVEL of brov red flint. (LAMBETH GROUP - L MOTTLED BEDS)	ly clayey ine to vnish black and 37.67-37.87 m Occasional brown OWER mottling.	37.43 -34.67 - - - - - - - - - - - - - - - - - - -	
- 38.05-38.15 - - -		NA NA NA	37.87-38.20 m 0.33m CR⊢ D 119			Mottled blue and green to medium slightly glauc Rare rounded to subrou coarse gravel of brown f (LAMBETH GROUP - L MOTTLED BEDS)	clayey fine conitic SAND. nded, fine to flint. OWER 38.17 m 2no. shell fragments (up to 10 mm x 8 mm).	(0.91) 	
- - 38.20-39.40 38.86-39.16 - - - 39.16-39.26	100 N/A N/A	NA NA NA	Flush: 38.20-39.40 water, ^{99 %} CS 120 D 121			Grey mottled greenish g fine to medium glauconi Rare to occasional pock mm x 5 mm) of grey cla (LAMBETH GROUP - Li MOTTLED BEDS)	rrey clayey tic SAND. iets (up to 10 y. OWER	38.58 -35.82 - - - - (0.82) -	
-		NR NR NR				ASSUMED ZONE OF C DUE TO SELF BORING PRESSUREMETER TE	ORE LOSS 39.40-40.40 m 88.4mm diameter self boring pressuremeter test pocket. ST POCKET		
Depth	TCR	lf	Records/Samples	Date	Time	Stratum continues to 40.29 m	n	+	
Groundwater Ent No. Struck Po (m)	ries st strik	e beha	viour	Depth se	ealed (m)	Depth Related Remarks * From to (m) 36.90 SBP test. 39.90 SBP test.		Chiselling Depths (m) T	ime Tools used
Notes: For explanation abbreviations see ke levels in metres. Stra in depth column. Scale 1:25	on of sy y shee atum th	/mbols t. All de ickness	and pths and reduced given in brackets NT BH01	Project Project No Carried or). ut for	Cable Car for London - Gr D1002-11 Transport for London	ound Investigation	Borehole NT	BH01



Drilled PW/NE Logged CT/JC Checked SV	Start 02/03 End 05/0	t 3/2011 4/2011	Equipment, Methods ar Dando 175 / Beretta T51 / CAT scanned for services, dug inspection pit from 0.3 1.20m to 12.10m depth. R 12.10m to 60.70m depth	nd Remarks Magnetome . Concrete c 38m to 1.20r lotary cored SPT from 6(s eter ored from n depth. C drilling us) 70m to €	GL to 0.38m depth. Hand able percussive boring from ng wireline techniques from 10.70m to Diameter Casing Depth 0.00m 10.20m 250mm 10.20m 10.20m 12.10m 200mm 11.95m 12.10m 60.70m 146mm 60.70m 10.70m 40mh	Ground Level Coordinates National Grid Chainage	+2.76 mOD E 539868.52 N 180300.77
Samples a	nd 1	iests	S	Date	Time	Strata Description	Depth,Level	Backfill/
Depth	SCR RQD		Records/Samples	Casing	Water	(Continued from Sheet 8)	(Thickness)	Legena
- - - 39.40-41.20 - 40.40-40.68 - 40.40 -	51 N/A N/A		- Flush: 39.40-41.20 water, 99 % SPT S 50 (10,14/19,31 for 50mm D 42	;)		ASSUMED ZONE OF CORE LOSS DUE TO SELF BORING PRESSUREMETER TEST POCKET Very dense dark grey very clayey fine to medium glauconitic SAND. Rare shell fragments (up to 15 mm x 5 mm). Rare rounded fine to medium gravel of black flint. Occasional pockets (up to 25 mm x 5 mm) of grey	- - = 40.29 -37.53 - - -	
- 40.78-41.08 			CS 122			(LAMBETH GROUP - UPNOR FORMATION)	-	
- 41.08-41.20 		NA NA NA	D 123				- _ (1.77) - _ _	
- 41.20-42.40 - 41.96-42.06	72 N/A N/A		Flush: 41.20-42.40 water, 99 % D 125			ASSUMED ZONE OF CORE LOSS	- - - 42.06 -39.30	
-		NA		31/03/2011 42.40 01/04/201	1 1700 1 0800	PRESSUREMETER TEST POCKET 42.40-43.40 m 88.4mm diameter self boring pressuremeter test pocket.	-	
-		NA NA		42.40			(1.12) - - - - - - - - - - - - - - - - - - -	
- 42.40-44.20 - 43.40-43.59 - 43.40 	57 N/A N/A		Flush: 42.40-44.20 water, 99 % SPT S 50 (17,8/50 for 35mm) D 43 CS 126			Very dense grey to brownish grey slightly clayey fine to medium SAND. Rare wispy black mottling. (THANET SAND FORMATION)	-	
- 44.10-44.20 		NA NA NA	D 127			44.20-44.23 m AZCL	- - - - - - - -	
- 44.20-45.40 44.85-45.15 	98 N/A N/A	lf	Flush: 44.20-45.40 water, 99 % CS 128 Records/Samples	Date	Time	Stratum continues to 45.40 m		
Groundwater Ent No. Struck Por (m)	ries st strik	e beha	Iviour	Casing Depth se	ealed (m)	Depth Related Remarks * From to (m) 42.90 SBP test.	Chiselling Depths (m) T	ime Tools used
Notes: For explanationabbreviations see ke levels in metres. Strain depth column.	on of sy y sheel atum th	/mbols t. All de ickness	and apths and reduced s given in brackets NT BH01	Project Project No Carried o	o. ut for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NT	• BH01



Drilled PW/NE Logged CT/JC Checked SV	Start 02/03 End 05/04	3/2011 4/2011	Equipment, Methods ar Dando 175 / Beretta T51 / CAT scanned for services dug inspection pit from 0.2 1.20m to 12.10m depth. R 12.10m to 60.70m depth.	nd Remarks Magnetomet Concrete co 38m to 1.20m totary cored d SPT from 60	ter pred from depth. C drilling us 70m to f	GL to 0.38m depth. Hand able percussive boring from 10.7m depth	Ground Level Coordinates National Grid Chainage	+2.76 m0 E 539868 N 180300	DD 52 .77
Samples a	nd T	este	3	Data	Timo	Strata	Depth Loval	1 Pac	
Depth	SCR RQD	lf	Records/Samples	Casing	Water	Uescription (Continued from Sheet 9)	(Thickness)	Legend Instru	ktill/ ments
45.15-45.25 			D 129			Very dense grey to brownish grey slightly clayey fine to medium SAND. Rare wispy black mottling. (THANET SAND FORMATION)	- - - 45.40 -42.64		
- - - -		NA NA NA				ASSUMED ZONE OF CORE LOSS self boring pressuremeter test DUE TO SELF BORING pocket. PRESSUREMETER TEST POCKET	- - - - - - - - - - - - - - - - - - -		
- 45.40-47.20 - 46.40-46.55 - -	58 N/A N/A		Flush: 45.40-47.20 water, 99 % SPT S 50 (15.10 for 25mm/ 50 for 50mm) D 44			Very dense grey slightly silty fine to medium SAND. Rare to occasional black wispy mottling. (THANET SAND FORMATION)	- 40.15 -43.39 - - - -		
- 46.75-47.05 			CS 130 D 131			-	-		
-		NA NA NA				47.20-47.33 m AZCL	- (2.25) 		
47.65-48.00 - - 47.20-48.40 -	76 N/A N/A		CS 132 Flush: 47.20-48.40 water, 99 %				-		
- - - 48.30-48.40 -			D 133	01/04/2011 48.40	1700	ASSUMED ZONE OF CORE LOSS 48.40-49.40 m 88.4mm diameter	- - - 48.40 -45.64		
-		NA NA NA		02/04/2011 48.40	0700	DUE TO SELF BORING PRESSUREMETER TEST POCKET	- - (0.54) 		
- - 48.40-50.20 - 49.40-49.55 49.40 - 49.48-49.78 -	79 N/A N/A		Flush: 48.40-50.20 water, 99 % SPT S 50 (12,13 for 30mm/ 50 for 45mm) D 45 CS 134			Very dense grey slightly silty fine to medium SAND. Rare to occasional black wispy mottling. (THANET SAND FORMATION)			
49.78-49.90 			D 135				- - - -	SI	1 7E
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	Stratum continues to 51.40 m			
Groundwater Ent No. Struck Po (m)	ries st strike	e beha	viour	Depth se	aled (m)	Depth Related Remarks * From to (m) 45.90 SBP test. 48.90 SBP test.	Chiselling Depths (m) T	ïme Tools used]
Notes: For explanati abbreviations see ke levels in metres. Stra in depth column. Scale 1:25	on of sy y sheet atum thi	rmbols : t. All de ickness	and pths and reduced given in brackets NT BH01 AGS	Project Project No Carried ou	it for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NT	BH01 et 10 of 13	



Drilled PW/NE Logged CT/JC Checked SV	Star 02/03 End 05/04	t 3/2011 4/2011	Equipment, Methods and Remarks Dando 175 / Beretta T51 / Magnetometer CAT scanned for services. Concrete cored froi dug inspection pit from 0.38m to 1.20m depth. 1.20m to 12.10m depth. Rotary cored drilling u 12.10m to 60.70m depth. SPT from 60.70m to			GL to 0.38m depth. Hand able percussive boring from ing wireline techniques from 10 Zm deth	Ground Level Coordinates National Grid Chainage	+2.76 mOD E 539868.52 N 180300.77	
Samples a	nd T	ests	5			Strata	1		
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	Description (Continued from Sheet 10)	Depth, <i>Level</i> (Thickness)	Legend	Backfill/ Instrument
- - - -			CS 136			Very dense grey slightly silty fine to medium SAND. Rare to occasional black wispy mottling. (THANET SAND FORMATION)	(2.46)		
- - 50.20-51.40 - 50.91-51.01 	100 N/A N/A	NA NA NA	Flush: 50.20-51.40 water, 99 % D 137	02/04/2011	1700		-		
-				51.40 03/04/2011 51.40	0800	ASSUMED ZONE OF CORE LOSS 51.40-52.00 m 88.4mm diameter but to SELF BORING 51.40-52.00 m 88.4mm diameter self boring pressuremeter test pocket.	51.40 -48.64 (0.60)	· · · ·	
- 52.00-52.23 - 52.00 51.40-53.20 	67 N/A N/A		SPT S 50 (15,10 for 50mm/ 24,26 for 25mm) D 46 Flush: 51.40-53.20 water, 99 %			Very dense grey silty fine to medium SAND. Rare black wispy mottling. (THANET SAND FORMATION) 52.34-53.07 m Rare fine to medium gravel of flint and lignite.	- 52.00 -49.24 - -		
- - - 52.80-52.90 -			D 139			Grey clayey sandy GRAVEL. Gravel is	(1.07) - - 53.07 -50.31		
		380 380 380				angular to subangular fine to coarse of flint. Sand is fine to medium. Frequent pyrite nodules (up to 2 mm x 2 mm). 1no. cobble of flint. (BULLHEAD BEDS - THANET SAND FORMATION) 53.50-53.58 m Frequent flints (up to 65 mm x 35 mm x 30 mm).	53.20 -50.44 (0.38) 53.58 -50.82		
- - 53.20-54.70 - - -	25 25 25	NA NA NA	Flush: 53.20-54.70 water, 99 %			fractures observed. (WHITE CHALK SUBGROUP - CIRIA GRADE B1) ASSUMED ZONE OF CORE LOSS Driller notes: Core loss due to flint jammed in core barrel.	(1.12)		
- 54.70-55.11 - 54.70 -		SP	T S 49 (5,8/10,13,15,11 for 35m D 47	m)		Weak medium density white CHALK. 54.70-55.05 m Recovered as silt Fractures are widley spaced (70/780/1090), rarely closed and	- - 54.70 -51.94 -		
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water	Stratum continues to 61.07 m			
Groundwater En No. Struck Po (m)	tries st strik	e beha	viour	Depth se	ealed (m)	Depth Related Remarks * From to (m) 52.00 Unable to drill beyond with SBP due to ground conditions.	Chiselling Depths (m) T	ime Too	Is used
Notes: For explanate abbreviations see ke levels in metres. Str in depth column. Scale 1:25	on of sy ey shee atum th	/mbols t. All de ickness	and pths and reduced given in brackets NT BH01 24. 22/06/2011 12:15:39	Project Project No Carried ou	It for	Cable Car for London - Ground Investigation D1002-11 Transport for London	Borehole NT	BH0 ⁴	1 3

Drilled PW/NE Logged CT/JC Checked SV	Start 02/03 End 05/04	3/2011 4/2011	Equipment, Methods a Dando 175 / Beretta T51 / CAT scanned for services dug inspection pit from 0. 1.20m to 12.10m depth. R 12 10m to 60 70m depth	uipment, Methods and Remarks ndo 175 / Beretta T51 / Magnetometer T scanned for services. Concrete cored from GL to 0.38m g inspection pit from 0.38m to 1.20m depth. Cable percuss 0m to 12.10m depth. Rotary cored drilling using wireline t 10m to 60 70m depth. SPT from 60 70m to 61 07m depth Certada		GL to 0.38m depth. Hand able percussive boring from ng wireline techniques from 107m depth	Depth from to Diameter 0.00m 10.20m 250mm 10.20m 12.10m 200mm 12.10m 60.70m 146mm	Casing Depth 10.20m 11.95m 60.70m	Ground Level Coordinates National Grid Chainage	+; E 5 N 1	2.76 mOD 539868.52 180300.77
Samples a		ests	5	Date	Time	Strata	Description		Depth,Level	Lonond	Backfill/
Depth	SCR RQD	fl	Records/Samples	Casing	Water	(Con	tinued from Sheet 11)		(Thickness)	Legend	Instruments
- 55.10-55.43 - - 54.70-56.20 - -	73 73 73		CS 140 Flush: 54.70-56.20 water, 99 %			infilied (0/<1/<1) with occasional pockets x 30 mm) of light grey w (WHITE CHALK SUBGF GRADE B1)	alk Silt 55.05-55.10 m C (up to 60 mm(up to 85 mm x 65 ispy marl. ROUP - CIRIA 55.35-55.79 m C (up to 80 mm x 40 55.60 m 1no. she	ccasional flints [
- - - - -			00.44				55.80- 56.20-57.29 m O fragments (up to 40	-56.20 m AZCL			
56.55-56.88 56.20-57.70 	80 80 75		CS 141 Flush: 56.20-57.70 water, 99 %				56.88-57.03 m Fre to 85 mm x 75 mm associated	quent flints (up – x 65 mm) with core damage. – –			
- - - 57.70-58.06 - 57.70 -		70 780sf 1090	57.64-57.70 m 0.06m CRF YT S 50 (8,10/15,20,15 for 60m D 48	03/04/201 57.70 n) 04/04/201 57.70	1 1530 1 0700		57.34	57.64 m AZCL	(6.37)		
- 57.70-59.20 58.56-58.87 -	100 100 100		Flush: 57.70-59.20 water, 99 % CS 142				58.22-58.30 m Flint x 45 mm x 40 mm) 1	s (up to 65 mm vith associated core damage. 			
-							58.90-58.97 m Fre to 60 mm x 60 mm associated 58.98-59.00 m Fr (up to 10 mm x 5 m	quent flints (up x 50 mm) with core damage. quent pockets n) of grey clay.			
- - - - - - - - - - - - - - - - - - -	100 100 100 TCR RCD	lf	Flush: 59.20-60 <i>č</i> 용 143 water, 99 % Records/Samples	Date Casing	Time Water	Stratum continues to 61.07 m	59.54 m 2no. shel 59.64-59.68 m 1no 40	fragments (up [−] 5 mm x 5 mm). — 1 filnt (30 mm x 0 mm x 70mm). — - - - - - -			
Notes: For explana	tion of sy	mbols	and and reduced	Depth s Project	ealed (m)	From to (m)	ound Investigation		Depths (m)	Time Too	ols used

Notes: For explanation of symbols and	Project	Cable Car for London - Ground Investigation	Borehole
levels in metres. Stratum thickness given in brackets in depth column.	Project No.	D1002-11	NT BH01
Scale 1:25 408.24 22/06/2011 12:15:41 AGS	Carried out for	Transport for London	Sheet 12 of 13

Soil Mechanics



Drilled PW/NE Logged CT/JC Checked SV	Start 02/03/2 End 05/04/2	2011 2011	Equipment, Methods ar Dando 175 / Beretta T51 / CAT scanned for services, dug inspection pit from 0.3 1.20m to 12.10m depth. R 12 10m to 60 70m depth S	uipment, Methods and Remarks ndo 175 / Beretta T51 / Magnetometer T. scanned for services. Concrete cored from GL to 0.38m depth. Hand g inspection pit from 0.38m to 1.20m depth. Cable percussive boring fro 10m to 12.10m depth. Rotary cored drilling using wireline techniques fro 10m to 60.70m depth. SPT from 60.70m to 61.07m depth Strata			Depth from 0.00m 10.20m 12.10m	to 10.20m 12.10m 60.70m	Diameter 250mm 200mm 146mm	Casing Depth 10.20m 11.95m 60.70m	Ground Level Coordinates National Grid Chainage	+2 E 5 N 1	2.76 mOD 39868.52 80300.77
Samples a			Pocords/Samples	Date	Time	Strata	Descriptio	n			Depth,Level	Legend	Backfill/
Depth	RQD	п	Records/Samples	Casing	Water	(Con	tinued from s	Sheet 12)			(Thickness)	Legend	Instruments
- - - -						(WHATE CHALK SUBGF GRADE B1)	iced sed and alk silt (up to 60 m spy marl. COUP - CIR	60.25-6 Mpockets	0.38 m Free (up to 30 m	- quent marl and m x 20 mm) of grey clay -			
- 60.70-61.07 - 60.70		—sP	T S 50 (8,12/12,15,23 for 65mr D 49	n)						-			
-				04/04/2011 60.70	1700					-			
-						EXPLORATORY HOLE END	 S AT 61.07 m				61.07 -58.31	ļ	<u> </u>
-										-			
-										-			
-										-			
-										-			
-										-			
-										-			
-										-			
-										-			
-										-			
-										-			
-										_			
-										-			
-										-			
-										-			
-										-	-		
-										-			
╞										_			
F										-			
-										-	1		
-										-			
-										-			
-										-			
Depth	TCR SCR RQD	lf	Records/Samples	Date Casing	Time Water					_			
Groundwater Ent No. Struck Po (m)	ries st strike	behav	viour	Depth se	aled (m)	Depth Related Remarks * From to (m)					Chiselling Depths (m) T	ïme Too	ls used
Notes: For explanation abbreviations see ker levels in metres. Stra	on of sym sheet. atum thicl	nbols a All dep kness	and oths and reduced given in brackets	Project No		L Cable Car for London - Gr D1002-11	ound Invest	igation			Borehole	. BH01	
in depth column. Scale 1:25		408.2	NT BH01	Carried ou	t for	Transport for London					She	et 13 of 13	-

PID Readings



Hole ID	Hole Depth (mBGL)	PID Reading (ppm)	Remarks
NT BH01	9.00	0.40	
NT BH01	10.00	0.10	
NT BH01	10.45	0.00	
NT BH01	10.70	0.80	
NT BH01	11.20	0.00	

Notes:



Mark Marshall Atkins Limited Woodcote Grove Epsom KT18 5BW

Email: <u>Mark.Marshall@atkinsglobal.com</u> Telephone: 01372 75 6708

Appendix G. Construction Noise and Vibration Mitigation Scheme

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

Construction Noise and Vibration Mitigation Scheme

April 2017



Table of Contents

List o	of Abbreviations	i
Defir	nitions	i
1	Introduction	1
2	Eligibility	1
3	The Noise Insulation Package	2
4	Delivery of the Noise Insulation Package	4
5	The Temporary Rehousing Package	6
6	Temporary Rehousing and Noise Insulation	7
7	Procedure for noise insulation and temporary rehousing	8
8	Appendix A - Criteria for Eligibility for Noise Insulation or Temporary Re-housing	13
8.1	Noise Insulation	13
8.2	Temporary Rehousing	13
8.3	Interpretation of the trigger levels	14
9	Appendix B - Eligibility Assessment	14
9.1	Introduction	14
9.2	Procedure for Administering the Policy	15
9.3	Protocol for Determining Eligibility	15
List of Abbreviations

The following abbreviations are created:

Abbreviation	Definition
The Contractor	Transport for London

Definitions

The following terms are used within this document:

Term	Definition
A-weighted	Is the A-weighted level, expressed as "dB(A)", allows for the frequency dependent characteristics of human hearing. Corrections are applied for each octave frequency band, and the resultant values summed, to obtain a single overall level.
Best Practicable Means	This means that every practicable mitigation has been applied to reduce noise and vibration produced by the Works to as low a level as reasonably practicable. Best Practicable Means (BPM) is defined under Section 72 of the Control of Pollution Act 1974 as those measures which are 'reasonably practicable having regard among other things to local conditions and circumstances, to the current state of technical knowledge and to financial implications'
Claimant	Means an owner or occupier of an eligible building who accepts an offer made under this Construction Noise and Vibration Mitigation Scheme.
Construction	Includes demolition, excavation, and tunnelling
Contiguous façade	Means a façade of a building that is horizontally separated from other facades by a stairwell, corner or some other discontinuity.
Contractor	The Contractor will be the construction entity through which the Project Company would deliver the design and construction of the Scheme and includes anyone who directly employs or engages construction workers or manages construction work, including sub-contractors, or any individual self-employed worker or business that carries out, manages or controls construction work.

Term	Definition
decibel (dB)	Is the ratio of sound pressures which we can hear – a ratio of 10^6 (one million: one). For convenience, therefore, a logarithmic measurement scale is used. The resulting parameter is the 'sound pressure level' (L _p) and the associated measurement unit is the decibel (dB). As the decibel is a logarithmic ratio, the laws of logarithmic addition and subtraction apply.
eligible building	Dwelling lawfully used by Claimants for habitation within the limitations and subject to the exceptions set out in regulation 7 of the Regulations.
Eligible room	Means a living room or a bedroom having a qualifying door or a qualifying window in an eligible building.
Equivalent continuous sound pressure level (L _{eq})	Another index for assessing overall noise exposure is the equivalent continuous sound level, L_{eq} . This is a notional steady level which would, over a given period of time, deliver the same sound energy as the actual time-varying sound over the same period. Hence fluctuating levels can be described in terms of a single figure level. The A-weighted L_{eq} is denoted as L_{Aeq} .
Façade	Means an outer wall of a building.
Insulation work	Means work carried out to insulate an eligible room against noise which will include provision of adequate ventilation and may include blinds.
Pre-existing ambient noise	Means the level of ambient noise, expressed as a level of L_{Aeq} determined with respect to the relevant time period and the relevant L_{Aeq} averaging time, prevailing one metre in front of relevant windows or doors in a façade of a dwelling, immediately before commencing any works which could affect the prevailing noise climate.
Qualifying door	Means an external door opening directly into an eligible room which is in that part of the façade in respect of which the relevant noise level satisfies the requirements of Appendix A of this document or meets the criteria for a contiguous façade as set out in Appendix B.
Qualifying window	Means a window in an eligible room which is in that part of the façade in respect of which the relevant noise level satisfies the requirements of Appendix A of this document or meets the criteria for a contiguous façade as set out in Appendix B.
The Noise Insulation Package	Means the package of works set out in part 3 of this document.
The Regulations	Means the Noise Insulation Regulations 1975.

Term	Definition
The relevant specifications	Means the items in Part I of Schedule 1 to the Regulations except where they are amended by the provisions of this document, such of the items in Part II of Schedule 1 to the Regulations as may be approved by The Contractor and such of the specifications set out in Part III of Schedule 1 to the Regulations as are applicable in the circumstances of the case or items whose performance is equivalent thereto.
Section 61 agreement	Prior agreement from a council - under Section 61 of the Control of Pollution Act 1974 - for construction works where Best Practicable Means has been demonstrated to be applied.
	Scheme.

1 Introduction

- 1.1.1 The Contractor will seek to control the effects of construction noise and vibration arising from the Scheme using the Best Practicable Means (BPM) mitigation measures secured in the Noise and Vibration Management Plan. Nevertheless, there may be circumstances in which noise impacts arise that will need to be mitigated still further. In certain circumstances, explained below, the Contractor will offer to either provide and install free of charge, or provide grant aid for a claimant to install, noise insulation. In certain cases where the level of noise created by construction activity is predicted to exceed the qualifying criteria outlined within this document the Contractor will contact potential Claimants to offer to arrange temporary rehousing, or help Claimants to arrange it for themselves and recoup the costs from the Contractor.
- 1.1.2 The Contractor has adopted a set of noise and vibration thresholds in relation to the provision of noise insulation and, if appropriate, temporary rehousing. These thresholds follow the precedents established by recent and similar major transport schemes.
- 1.1.3 The purpose of this document is to explain both how the noise insulation and temporary rehousing schemes work, and what Claimants should do next if they think that they may be eligible for either scheme.

2 Eligibility

- 2.1.1 Construction noise insulation and temporary rehousing arrangements apply to dwellings and other buildings lawfully used for residential purposes.
- 2.1.2 To be eligible a Claimant must own or occupy a private dwelling and the dwelling must be one in which the predicted or actual construction noise exceeds the relevant 'noise trigger level' (as shown in Appendix A) for one or more of the following:
 - a period of 10 or more days of working in any 15 consecutive days;
 - a period of 3 or more nights of working in any 7 consecutive nights; or
 - for a total of 40 days or more in any 6 consecutive months.
- 2.1.3 The rooms to which this scheme applies, eligible rooms, are defined as living rooms or bedrooms having a qualifying door or a qualifying window in any eligible building.

- 2.1.4 Eligibility for the scheme depends on the predicted noise level following the construction noise assessment that will be carried out for that purpose once detailed construction plans are in place,. If those noise predictions indicate that a property is eligible, the offer of noise insulation will be made by the Contractor and, if accepted and all necessary approvals obtained, the insulation installed before the Works (with the potential to exceed the noise trigger levels) commence. Full details of the noise trigger levels, for both noise insulation and temporary rehousing are set out in Appendix A.
- 2.1.5 Some buildings and/or their occupants will be treated as special cases, and will therefore be considered on a case by case basis. These include:
 - Night workers;
 - those needing a particularly quiet home environment to work in, or
 - those that have a medical condition which will be seriously aggravated by construction noise,.

3 The Noise Insulation Package

- 3.1.1 The package will consist of:
 - Secondary glazing or thermal double glazing (see also sections 5.1, 5.2 and 5.5) for living room and bedroom windows on eligible facades, plus additional ventilation if required under the relevant specifications.
 - Blinds, for south facing windows (windows in an aspect falling anywhere within a 270 degrees arc between the bearings of 45 degrees and 315 degrees from true north).
 - Acoustic insulation treatment for external doors on eligible facades.
- 3.1.2 Depending on the type of window already existent at an eligible building, secondary glazing will usually comprise another pane of glass in its own frame (wood, metal or plastic) 100-200 mm inside the existing window. This can be opened for cleaning or ventilation.
- 3.1.3 Secondary glazing works best when closed so additional ventilation is usually required. The package includes an electric ventilator fan in a slim metal cover, fitted inside the room in question, to an outside wall (a 75-100 mm hole is drilled through the wall, through which the fan draws in air from the outside). Such ventilation shall comply with the specification required by the Regulations.
- 3.1.4 On a south facing window secondary glazing may make the room too hot. As set out under the relevant specifications, subject to the

agreement of the claimant, blinds will be fitted between the main window and the secondary glazing to minimise this effect. If the Claimant chooses not to accept blinds as part of the noise insulation package the possible impacts of this will be explained to them, blinds will not be retrofitted post installation of the noise insulation package should the claimant change their mind at a later date.

- 3.1.5 The Contractor may be able to install a "secondary" door to improve noise insulation. If the design of a Claimant's property prevents this, other methods can be used, such as sealing strip between the existing door and its frame.
- 3.1.6 There may be circumstances in which it is not possible to fit secondary glazing. Such cases will be considered on a case by case basis. Where eligibility is confirmed, appropriate mitigation measures will be adopted. The sorts of measures that will be considered include works management methods (e.g. adopting quiet times, rescheduling works, and imposing noise limits), or temporary rehousing even if the Temporary Rehousing thresholds (as set out below) are not exceeded.
- 3.1.7 Thermal double glazing can be provided instead of secondary glazing only if it is specifically requested by the Claimant. The Claimant will be made aware of the potential shortfall in sound insulation performance of the thermal double glazing compared to the secondary glazing.
- 3.1.8 There will be no obligation to repair, maintain or make any payments in respect of repairing or maintaining any equipment or apparatus installed under the application of this document or to pay for the running costs, which will be minimal for mechanical ventilation units. Notwithstanding this, should equipment such as the ventilation units fail after installation of the noise insulation package through no fault of the Claimant, and this occurs during the Works, the failed apparatus will be repaired or replaced as necessary.
- 3.1.9 If a Claimant is an occupier but not the owner of the property, it is the occupier's responsibility to obtain the owner's (landlord or freeholder) consent to install the Noise Insulation Package and demonstrate this to the Contractor.
- 3.1.10 If the occupier or owner (landlord or freeholder) of the private dwelling refuses the Noise Insulation Package or does not provide adequate access, temporary rehousing will not be offered, unless the Temporary Rehousing thresholds are exceeded.

4 Delivery of the Noise Insulation Package

- 4.1.1 Once the Contractor has conducted the assessment set out in paragraph 2.1.4 above and the details of the insulation for a property are agreed with a Claimant, the Contractor will either offer to deliver the Noise Insulation Package at its expense, or offer grant aid for the Claimant to carry it out.
- 4.1.2 The Contractor will ask Claimants to ensure that they provide adequate access for the survey and installation (as detailed in paragraph 4.1.8 below); and if they should incur expense in arranging access, the Contractor will reimburse them provided it has agreed the amount before the cost is incurred.
- 4.1.3 In the cases where the Contractor offer Claimants a grant so that they can undertake the noise insulation work themselves, the grant would be made on the following conditions:
 - i. 3 independent written quotations being obtained by the Claimant.
 - ii. The work must comply with the relevant specifications.
 - iii. the quote that represents the cheapest price for complying with point ii must be chosen.
 - iv. The amount of the grant will be for whichever is the lesser amount of the selected quote, or the actual cost of the installation.
 - v. The Contractor may pay 10% of the estimated cost in advance, and the balance when the work is satisfactorily completed.
 - vi. The work must be completed within 12 months of any advance payment, or before completion of the construction works for which insulation is needed, whichever is the earlier. If this condition is not complied with, no further grant will be paid, and any payments already made will have to be repaid to the Contractor or its agent.
 - vii. the consent of any other person or body that may be required to permit the carrying out of insulation work (e.g. a landlord or local authority) must be obtained prior to the work taking place.
 - The amount of the grant payable for the installation of thermal double glazing will be no more than the cost of installing the secondary glazing package specified in this

document. If the Claimant arranges for the work to be carried out themselves, the amount paid in reimbursement will be for no more than the cost that would have been incurred if the secondary glazing package specified in this document had been installed.

- 4.1.4 The Noise Insulation Package will not be able to be used for work needed to remedy existing building defects.
- 4.1.5 The Noise Insulation Package will cover the making good of the existing fabric and decoration (not including curtains) after the installation of secondary glazing, ventilation equipment, and second doors, including the adaptation of any existing pelmet and curtain rack.
- 4.1.6 A decision to accept an offer of a Noise Insulation Package must be made no later than 6 months after the date it is made in writing to a Claimant or one month before the Contractor intends to install the other noise insulation at eligible properties affected by the same construction works, whichever is the sooner. In the latter case, Claimants will receive notice of the cut-off date for acceptance at the time the offer is made or shortly thereafter. If a Claimant does not respond within the time-frame due to circumstances beyond their control, the Contractor will give due consideration to their case but the Works will continue as programmed. If a Claimant chooses not to accept the offer of a Noise Insulation Package there is no scope for a Claimant to change their mind at a later date. However, if the noise levels change during the course of the works such the Claimant would be eligible for noise insulation/temporary rehousing, then the process set out in paragraphs 7.1.3 to 7.1.8 will apply.
- 4.1.7 If upon determining that a property would be eligible for the Noise Insulation Package, it is discovered that a Claimant's property has already had insulation work carried out or a grant for such work in respect of another public works scheme (such as a road or earlier railway works) or privately, the Claimant's property will not be eligible for further work or grant under this scheme unless the existing noise insulation is not in a state adequate to attenuate the construction noise for the Scheme. If it is not, the Works will be carried out or a grant made to have them carried out to bring the installed noise insulation package up to the appropriate standard.
- 4.1.8 The carrying out of insulation work and the making of grants shall be dependent upon compliance with the following conditions;
 - (a) a Claimant who accepts an offer to carry out insulation work shall;

- (i) if he is the occupier of or entitled to occupy the building, afford to the Contractor such access to the building as it may reasonably require for the purpose of carrying out and inspecting the work to ascertain whether it complies with the relevant specifications; or
- (ii) if he is not the occupier of nor entitled to occupy the building, procure the occupier or person entitled to occupy the building to afford the Contractor such access;
- (b) a Claimant who accepts an offer to make a grant in respect of the cost of the noise insulation work shall carry out the work or secure that it is carried out in accordance with the relevant specifications and complete it before the expiration of 24 months from the date of acceptance and;
 - (i) if he is the occupier of or entitled to occupy the building, afford to the Contractor such access to the building as it may reasonably require for the purpose of inspecting the work to ascertain whether it complies with the relevant specifications; or
 - (ii) if he is not the occupier of nor entitled to occupy the building, procure the occupier or person entitled to occupy the building to afford the Contractor such access

5 The Temporary Rehousing Package

i.

5.1.1 If, following the assessment that will be carried out for that purpose once detailed construction plans are in place, the predicted construction noise level exceeds the trigger level for temporary rehousing as set out in Appendix A, the Contractor will notify Claimants that they are eligible for alternative temporary accommodation. There are two options:

Option A: The Contractor to arrange temporary alternative accommodation to meet the Claimant's agreed needs.

Option B: The Contractor to provide information and guidance to help the Claimant arrange temporary alternative accommodation

- 5.1.2 The acceptance of any offer of temporary rehousing is discretionary.
- 5.1.3 If the Claimant chooses Option A, the services provided by the Contractor will include arranging for:

- Temporary alternative accommodation (which, where appropriate, could be a local hotel or guest house).
- Relocation of personal effects.
- Storage and insurance of personal effects.
- Insurance for the vacated property.
- Where appropriate for pets to go into kennels, catteries etc.
- Where appropriate the disconnection and later reconnection of gas, water, electricity etc.
- 5.1.4 If a Claimant chooses Option B then, instead of actually identifying the alternative accommodation and making the arrangements for them, the Contractor will supply the Claimant with information and guidance on all the matters listed above, to enable them to make the arrangements themselves; and the Contractor will also help the Claimant ensure that the costs the Claimant incurs can be agreed and paid to them as soon as practicable.
- 5.1.5 Whether Option A or Option B is chosen, the Contractor will bear (or reimburse) the reasonable costs associated with temporary rehousing together with the continuing, unavoidable costs of the Claimant maintaining their own house whilst they are away. However, these will be paid less the costs that the Claimant would have paid if they had stayed in their own house over the same period.
- 5.1.6 The type of rehousing offered will depend on the duration of the relocation. For short durations hotel accommodation may be appropriate. For longer periods, alternative rented accommodation would be more suitable. In all cases account will be taken of a Claimant's existing accommodation as far as possible.
- 5.1.7 The offer of temporary accommodation will be in addition to a Claimant's current home. As such, they will still be responsible for the rent, bills and other outgoings at their current home and will still be a tenant there.
- 5.1.8 If a Claimant's tenancy agreement expires during the relocation it should be renewed with the landlord in the normal way. If a Claimant chooses not to renew their tenancy, the grant to meet the cost of the alternative accommodation will cease when the tenancy expires.

6 Temporary Rehousing and Noise Insulation

6.1.1 The noise generated by the Works will vary over the course of the Works. In some areas, the noise may mean that a eligible building qualifies for temporary rehousing for one period, and noise insulation

only for a different period. In these circumstances a Claimant would receive a temporary rehousing offer for one period and a noise insulation offer for the other period. In other areas, a Claimant's property may qualify for temporary rehousing for a given period, but outside that period the noise may not trigger a separate noise insulation offer. In such a case, a temporary rehousing offer only would be made and noise insulation would not be offered.

6.1.2 If a Claimant's property qualifies for temporary rehousing but not noise insulation, a Claimant does not have to accept the offer of rehousing and may request noise insulation instead. In such cases, a Claimant will be made aware of any shortfall in sound insulation performance of the noise insulation in relation to the thresholds presented in Appendix A and that the degree of disturbance could be high even with the noise insulation in place. If a Claimant chooses to adopt this approach and noise insulation is provided, they will not be able to later request temporary rehousing.

7 Procedure for noise insulation and temporary rehousing

7.1.1 The procedure comprises 7 steps.

- i. The Contractor will carry out a construction noise assessment based on the final construction plans, so as to predict what the noise levels will be and will discuss and agree the findings with the relevant local authority, as part of the Noise and Vibration Management Plan required by the CoCP.
- ii. The Contractor will then notify owners or occupiers of buildings which, on the basis of the assessment, the Contractor considers qualify, and accordingly which type of assistance (noise insulation or temporary rehousing) they are eligible for. The Contractor will also send an application form at this stage.
- iii. The notice and application form, should be completed and returned to the Contractor or its agent. The Contractor will then assess the application and if acceptable notify the Claimant in writing.
- iv. The Contractor will then arrange to visit a Claimant in order to discuss the application ; view the property and in the case of noise insulation take any necessary measurements; and identify any special issues or requirements (such as any other approvals that may be required in the case of noise insulation).
- v. The Contractor will then assess the Claimant's case in detail and,

if it is accepted, notify them of:

- any further survey likely to be needed at the Claimant's property, and (in insulation cases) the work the Contractor thinks should be done and its offer to do it; or
- (in rehousing cases) either its proposals to re-house the Claimant temporarily or the information and guidance the Claimant needs to make its own rehousing arrangements. In either case the proposals will be discussed with the Claimant and the Claimant will not be under any obligation to accept the offer.
- vi. Assuming the Claimant agrees, the noise insulation package or temporary rehousing plan (as the case may be) is then put into effect.
- vii. The Contractor reimburses the Claimant for any agreed costs, which they have incurred or (in grant cases) pays the balance of the grant. Alternatively, the Contractor pays for noise insulation or removal/rehousing costs where it or its agent has done the work. A noise insulation package will not be offered if the noise trigger level is only exceeded whilst the Claimant is in temporary alternative accommodation (however see section 7.6 above).
- 7.1.2 Further technical information describing the process for predicting eligibility, particularly in relation to cut-off points on long facades such as terraced housing, is presented in Appendix B.

Claims after the start of construction

7.1.3 The trigger levels for eligibility for noise insulation/temporary rehousing involve both noise levels and durations (temporal criteria). There are two possible cases that may arise:

i. The predictions do not identify that noise insulation/temporary rehousing thresholds will be exceeded, but in practice they are and this is expected to continue for a period of time sufficient to exceed the temporal criteria.

ii. The predictions identify that the noise insulation/temporary rehousing thresholds will be exceeded but will not carry on for a sufficient duration to trigger the temporal criteria. However, in practice the works go on for longer and the temporal criteria are triggered.

In both cases the approach will essentially be the same.

7.1.4 If a person claims, after the start of construction work, that the noise levels actually experienced are such as to cause eligibility for noise insulation/temporary rehousing where none was predicted, or that received noise levels are sufficient for eligibility for noise

insulation/temporary rehousing where this was predicted, and that the noise has continued, or seems to them likely to continue for longer than the temporal triggers where that had not been predicted, the claim will be considered by the Contractor according to the following process.

- 7.1.5 A claim after the start of construction will inevitably take the form of a complaint or formal representation to the nominated undertaker. On receipt of the claim, The Contractor will review the works being undertaken that have generated the claim and assess whether it is likely that the claim is valid. Where the Contractor considers there is a potentially valid claim short-term site monitoring will be undertaken to identify whether the noise insulation/temporary rehousing trigger levels are indeed being exceeded. Whether or not monitoring is undertaken the Contractor will discuss the results of the review with the claimant and explain the findings and any actions that have been taken.
- 7.1.6 At the same time, the Contractor will inform the local authority that approved the Noise and Vibration Management Plan about the claim and what actions are being taken to address it.
- 7.1.7 If the short-term noise monitoring identifies that the noise insulation/temporary rehousing thresholds are being exceeded, but that operations are being performed in accordance with the terms of the Noise and Vibration Management Plan, the Contractor will identify whether the activities causing those levels will carry on for longer than a period of 10 or more days of working in any 15 consecutive days or for 3 or more nights of working in any 7 consecutive nights or for a total of days exceeding 40 in any six consecutive months. If they are not, then no further action is required. The findings will be made known to the local authority who approved the Noise and Vibration Management Plan and discussed with them.
- 7.1.8 If the works causing noise levels above the noise insulation/temporary rehousing thresholds are projected to go on for longer than a period of 10 or more days of working in any 15 consecutive days or for 3 or more nights of working in any 7 consecutive nights or for a total of days exceeding 40 in any six consecutive months, but the construction works are being carried out in accordance within the terms of Noise and Vibration Management Plan, then the Contractor will require action to be taken to reduce the level of noise being caused, or offer noise insulation and/or temporary rehousing to the affected property as appropriate. Works will not cease during the organisation and installation of the noise insulation. However, if appropriate, temporary rehousing will be offered to cover the period

during which the noise insulation is installed. The temporary rehousing will be withdrawn:

• once the noise insulation is installed; or

• if the claimant unnecessarily delays obtaining any necessary consents in accordance with paragraph 4.1.3 (vii). If it is not possible to fit secondary glazing appropriate measures will be considered on a case by case basis (see paragraph 3.1.6).

If the complainant is not satisfied by the response of the Contractor following a claim under section 7.1.5 above, they may register a complaint in accordance with the project's complaints procedure (to be determined prior to the start of the main construction works).

Transport for London

8 Appendix A - Criteria for Eligibility for Noise Insulation or Temporary Re-housing

(To be read in conjunction with Section 2 above.)

- 8.1 Noise Insulation
- 8.1.1 A dwelling will be eligible where noise insulation does not already exist that is of a standard that will mitigate the impacts of the scheme as identified by the re-assessment of construction noise impacts; and the total noise level due to construction of the works (pre-existing ambient plus airborne construction noise), measured or predicted at a point one metre in front of the most exposed of any windows and doors in any façade of a building which is an eligible dwelling, exceeds whichever is the higher of either: a) any of the following criteria in Table 1:

Table 1: Noise Insulation Trigger Level Table Time	Relevant Time Period	Averaging Time T	Noise Insulation Trigger Level dB LAeq, T
Monday to Friday	07:00 – 08:00	1 hr	70
	08:00 – 18:00	10 hr	75
	18:00 – 19:00	1 hr	70
	19:00 – 22:00	3 hr	65
	22:00 - 07:00	1 hr	55
Saturday	07:00 - 08:00	1 hr	70
	08:00 - 13:00	5 hr	75
	13:00 - 14:00	1 hr	70
	14:00 - 22:00	3 hr	65
	22:00 - 07:00	1 hr	55
Sunday and	07:00 - 22:00	1 hr	65
Public	22:00 - 07:00	1 hr	55
Holidays			

or

(b) the pre-existing airborne noise level for the corresponding times of day (i.e. the Relevant Time Periods presented in column 2 of Table 1);

8.2 Temporary Rehousing

8.2.1 Occupiers of a dwelling will be eligible for re-housing where the total noise level due to construction of the works (pre-existing ambient plus airborne construction noise), measured or predicted at a point one metre in front of the most exposed of any windows and doors in any façade of an eligible dwelling, exceeds whichever is the higher of either (a) or (b) or if criteria (c) is met:

- (a) 10 dB above any of the noise levels in Table 1 above or
- (b) 10 dB above the pre-existing airborne noise level for the corresponding time of day (i.e. the Relevant Time Periods presented in column 2 of Table 1);

and in the case of both (a) and (b) this occurs for a period of 10 or more days of working in any 15 consecutive days or for a period of 3 or more nights (22:00-07:00) of working in any 7 consecutive nights or for a total number of days exceeding 40 in any six consecutive months.

8.3 Interpretation of the trigger levels

- 8.3.1 In interpreting and applying the trigger levels in Table 1, two conventions will be adopted. The first is that in interpreting the noise insulation/temporary rehousing policy where eligibility arises if noise levels in Table 1 are exceeded, a resolution of 0.1 dB will be applied. For example, a value of L_{Aeq,T} of 55 dB (with pre-existing ambient at least 5 dB lower) will not trigger eligibility. A value of 55.1 dB will trigger eligibility.
- 8.3.2 The second convention relates to the choosing of minimum one-hour L_{Aeq,T} levels at night to define the pre-existing ambient, given that a series of survey results often shows different minima over a series of nights. The approach will be to select a 7-day survey period during which favourable weather conditions existed (wind speed not more than 5,/s and no precipitation) and select the lowest one hourly value from that data set.

9 Appendix B - Eligibility Assessment

9.1 Introduction

- 9.1.1 As explained in the main body of this appendix , eligibility for noise mitigation arises under the Scheme when whichever is the higher of the following two requirements is met (i) the total predicted (or actual) noise level due to construction works (pre-existing ambient plus airborne construction noise) exceeds a trigger level or (ii) the margin between the construction noise level plus the pre-existing ambient and the pre-existing ambient is at least 5 dB and also (iii) the temporal requirements (10 out of 15 days of working etc) are met.
- 9.1.2 If the eligibility requirements were applied strictly this could lead to anomalies whereby some dwellings in a terrace might be included and not others or it might result in dividing the facades of apartment blocks into eligible and ineligible properties.

9.1.3 The procedure to be followed by the Contractor in implementing the Scheme so as to avoid dividing facades in a manner likely to be contentious for residents is set out below.

9.2 Procedure for Administering the Policy

- 9.2.1 While construction noise predictions made using a noise model such as SoundPlan can be presented using contours that will indicate a finite value for any location of interest, the same is not true of eligibility. The principal reason for this is that measured baseline noise levels are of necessity carried out at discrete locations. While interpolation between discrete values is possible in theory, it is in many circumstances impracticable.
- 9.2.2 The procedure will normally identify a single representative noise measurement location per façade, except for long facades. Sometimes a noise measurement location may serve as a surrogate for other comparable facades as well. Measurement locations should generally be towards the centre of the façade or façade section that they represent. The noise measurements from these locations may well be rounded.
- 9.2.3 The predicted noise including the contribution from the construction works will then be made for the worst affected window in the façade under consideration.
- 9.2.4 Whether a property is eligible for noise mitigation or not will then be determined using this predicted level. This determination will be applied to all the dwellings for which the measurement location was taken as representative.
- 9.2.5 In the case of a very long façade, it may be appropriate to utilise more than one noise measurement location. However, since measured values will vary slightly with quite small movements in position, a protocol needs to be established to avoid anomalous results as described above. The solution is to determine that more than one measurement location will be adopted for the same continuous facade only if the results from different noise measurement locations alongside the same façade differ by at least 3dB. For a façade at right-angles to a noise source such as a road or railway, this broadly means a doubling of distance from the source and would therefore normally only apply to long facades.

9.3 Protocol for Determining Eligibility

1) Establish baseline L_{Aeq} for relevant time of day for appropriate monitoring locations.

2) Assign monitoring results to facades according to the following rules:

a. Monitoring results to apply to whole façade where there is only one monitoring location for that façade. The monitoring location is to be as near as possible to the centre of the façade.

b. Monitoring results to apply to whole façade where another façade is used a surrogate.

c. Where more than one monitoring location exists for the same façade, only if the L_{Aeq} levels for any period differ by 3 dB or more shall the façade be divided, in which case façade areas around the location to be apportioned equally (i.e. as far as practicable each monitoring location to be in the centre of the area assigned to it).

d. The definition of a façade of a building is one that is horizontally separated from other facades by a stairwell, corner or some other discontinuity, as set out in section 2 of this document.

3) The predicted noise levels including construction noise to be utilised for the whole facade are those for the worst affected window/door in any façade.

Appendix H. Outline Ecology Management Plan

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

SILVERTOWN TUNNEL Outline Ecology Management Plan

TR010021

March 2017



MAYOR OF LONDON

THIS PAGE HAS INTENTIONALLY BEEN LEFT BLANK

Silvertown Tunnel

Outline Ecology Management Plan (OEMP)

Author:

Transport for London

Rev.	Date	Approved By	Signature	Description
0	06/03/2017	David Rowe (TfL Lead Sponsor)		Outline Ecological Management Plan

Contents

1.			3
1	1.1	Ecology Management Plan	3
2. LIM	IITS	SUMMARY OF THE ECOLOGICAL BASELINE WITHIN THE ORDER 7	
3.		GENERAL SITE PRESCRIPTIONS – PRE AND DURING CONSTRUCTION	1 3
3	3.1	Introduction	3
3	3.2	Governance / Other Relevant Documentation	3
3	3.3	Statutory Licenses	3
3	3.4	Pre-Construction Surveys	3
3	8.5	General Method of Working	9
4.		HABITAT MITIGATION10)
4	l.1	Habitats of Value10)
4	1.2	Trees)
5.		SPECIES MITIGATION	2
5	5.1	Black Redstarts	2
5	5.2	Terrestrial Invertebrates	2
5	5.3	Non-native invasive plant species	2
App ME	peno THC	dix A. PRE CONSTRUCTION, & CONSTRUCTION, SURVEY	4
Арр	penc	dix B. PRELIMINARY METHOD STATEMENT – TREES	3
Арр	pend	dix C. JAPANESE KNOTWEED ON CONSTRUCTION SITES	2

1. INTRODUCTION

1.1 Ecology Management Plan

- 1.1.1 This Outline Ecology Management Plan (OEMP) supports the application for a Development Consent Order (DCO) for the Silvertown Tunnel ("the Scheme"). It forms part of the Code of Construction Practice (CoCP) and sets out the broad ecological management measures which Transport for London (TfL) (the 'Applicant') will require the Contractor to adhere to and implement for the construction of the Scheme.
- 1.1.2 The detailed Ecology Management Plans (EMP) for each worksite will be developed by the Contractor to be in substantial accordance with this OEMP, but taking into account the detailed design and local ecological issues at each worksite (North and South). The detailed EMPs will also be prepared in consultation with Natural England (NE) and submitted for approval to the relevant planning authority north and south of the Scheme (London Borough of Newham and Royal Borough of Greenwich).
- 1.1.3 This OEMP, and the subsequent detailed EMPs, should be read in conjunction with Chapter 9 of the ES, Terrestrial Ecology (APP-031).

2. SUMMARY OF THE ECOLOGICAL BASELINE WITHIN THE ORDER LIMITS

- 2.1.1 The following ecological receptors have been identified as having the potential to be present and adversely affected within the worksites:
 - Habitats including: plantation woodland, scrub, trees, grassland, tall ruderal vegetation, standing water, bare ground and ephemeral short perennial vegetation (some of which are components of brownfield habitat which is present on site). In addition, potential impacts from dust on East India Dock Basin Site of Interest for Nature Conservation (SINC) have been considered; and
 - Species and species groups, specifically notable terrestrial invertebrates; breeding birds (including black redstart), and commuting and foraging bats.
- 2.1.2 Japanese Knotweed and other non-native invasive plant species are also known to be present within the worksites.

3. GENERAL SITE PRESCRIPTIONS – PRE AND DURING CONSTRUCTION

3.1 Introduction

3.1.1 Detailed method statements will be developed and implemented by the Contractor for mitigation where required as detailed in the detailed EMPs. For the purposes of this document, outline method statements have been provided for receptors.

3.2 Governance / Other Relevant Documentation

- 3.2.1 The Contractor will be required to follow general principles for good working practice and site operations as outlined in this document, the CoCP and any other supporting subsidiary plans required by the CoCP.
- 3.2.2 At the inception of the project, a suitably qualified ecologist (SQE) will brief the site team of the ecological restrictions on the site and inform the site team of any works which require supervision by SQE or are subject to a specific method statement.

3.3 Statutory Licenses

3.3.1 As agreed in the Statement of Common Ground (SoCG) with NE (REP3-009), there are no requirements for terrestrial protected species licenses based on the existing information provided in the ES (APP-031). However this position will be reviewed and updated as the detailed design progresses and the necessary licences obtained if required.

3.4 **Pre-Construction Surveys**

- 3.4.1 A pre-construction survey will be undertaken a maximum of two years prior to commencement of construction works by the Contractor. This should include an extended Phase 1 habitat survey followed by targeted surveys for protected species that may be using the worksites (if required). This could include (for example) updated reptile surveys should the status of any habitats present on site have changed. This will prevent impacts on mobile species in the unlikely event that new species move into the area and/or the status of the worksites change.
- 3.4.2 Non-native invasive species will also require updated mapping as they may have spread since the 2015 surveys (i.e. any occurrences of non-native invasive species listed on Schedule 9 of the WCA or listed on the London

Invasive Species Initiative). This also includes an updated tree survey to BS 5837:2012 (Trees in relation to design, demolition and construction) – Recommendations).

3.5 General Method of Working

- 3.5.1 This section includes generic best practice that will be implemented throughout the site, specific mitigation for dedicated habitats and species is presented in Section 4.
- 3.5.2 In addition to pre-construction surveys, all site staff will be informed about the species and habitats that may be present on site via toolbox talks provided by the SQE. Toolbox talks will be tailored to the specific ecological issues relevant to the Scheme including Black redstarts and will focus on sensitive receptors, their characteristics and mitigation requirements.
- 3.5.3 The SQE must be present onsite during the clearance of vegetation if it's undertaken during the breeding bird season (end of February to mid-August). The detailed EMPs will detail which construction activities will require a SQE to be present.
- 3.5.4 Any site clearance would take account of seasonal constraints and will be carried out in accordance with a Site Clearance Plan (which will form part of the detailed EMPs) which will be provided prior to carrying out the works.
- 3.5.5 Worksites must be screened where necessary to reduce adjacent disturbance.
- 3.5.6 To prevent a potential impact from dust on East India Dock Basin Site of Interest for Nature Conservation (SINC), which is within 50m of a construction traffic track-out route, dust attenuation measures will be used and agreed with local authorities as part of the Air Quality Management Plan, as is required by the CoCP.

4. HABITAT MITIGATION

4.1 Habitats of Value

- 4.1.1 All habitat, including trees, will be retained and protected where possible. Areas of temporary land occupation will be returned to their previous state, condition and owner following completion of construction.
- 4.1.2 The habitats listed below were identified in the Phase 1 Habitat surveys (October 2015) within the land to be temporarily occupied during construction of the Scheme.
 - Brownfield (open mosaic habitat) Habitat;
 - Plantation Woodland and Scattered Trees;
 - Dense Scrub;
 - Grassland; and
 - Standing Water
- 4.1.3 Site clearance will take account of seasonal constraints and as part of the EMP, the contractor will provide details of site clearance and restoration.
- 4.1.4 Following the completion of the works, all land temporarily occupied will be examined by a Suitably Qualified Ecologist (SQE) to ensure habitats have been returned to their previous state and condition.

4.2 Trees

- 4.2.1 All reasonably practicable measures will be implemented to minimise the loss of trees.
- 4.2.2 A detailed Arboricultural Impact Assessment (AIA) will be undertaken once the full details of the construction compounds and detailed construction methodology are known and will be based upon an updated tree survey if required. This assessment will lead to the production of a schedule of trees to be retained and removed with accompanying maps.
- 4.2.3 A Tree Impact and Protection Plan and Arboricultural Method Statement will be produced by the Contractor and will be appended in the detailed EMPs. This will confirm the location of the fencing required to protect the Root

Protection Areas (RPA's) of trees to be retained. An example of a preliminary Arboricultural Method Statement is presented in Appendix B .

5. SPECIES MITIGATION

5.1 Black Redstarts

- 5.1.1 Black redstart monitoring will be undertaken weekly, during the construction period in areas that are suitable for black redstart, from April to July.
- 5.1.2 If black redstart is recorded, a SQE will determine whether there is a need for additional mitigation in accordance with the scale of the potential impact, and will prescribe the measures to be taken.
- 5.1.3 Potential additional mitigation measures the SQE may suggest include, but are not limited to, more frequent monitoring during the breeding season and during construction, demarcation of exclusion zones or whether discrete elements of the works, proximate to the recorded sighting area and which may give rise to local disturbance, are required to stop temporarily until the birds have left the area (i.e. following the breeding period).

5.2 Terrestrial Invertebrates

5.2.1 In order to mitigate for notable invertebrates, the CoCP provides for other subsidiary plans to prevent adverse impacts from construction, such as the Air Quality Management Plan (AQMP) for dust pollution, timing recommendations to avoid core activity periods, and pollution prevention measures following Environment Agency guidelines.

5.3 Non-native invasive plant species

- 5.3.1 Areas containing non-native invasive species will be demarcated with an appropriate buffer to ensure no spread of these species.
- 5.3.2 If Japanese Knotweed, or any other WCA Schedule 9 invasive species (such as Virginia Creeper or Cotoneaster), is located within the Order Limits then special measures will be used to deal with this vegetation. These measures will include classification and disposal of the waste as a 'controlled waste' under the Environmental Protection Act 1990 (c. 43) (as amended in 1996 and 1999). Further to this, non-native invasive species will be prevented from spreading in accordance with the latest Government guidance. Buddleia/Butterfly-bush, while not a WCA Schedule 9 prohibited species, is listed by the LISI as undesirable, and has dominated much of the Scheme area. It will be removed from within the Order Limits during construction.

5.3.3

5.3.4 Appropriate measures, based on latest Government guidance, will be implemented to prevent the spread of non-native invasive species during construction. This will include appropriate tool box talks to communicate the potential presence and appearance of such species.

Appendix A. PRE CONSTRUCTION, & CONSTRUCTION, SURVEY METHODOLOGY AND TIMINGS

Table A-1 Survey Details

Habitats/Species	Survey Methodologies	Timing
Extended Phase 1 survey	In order to determine if the status of the site is broadly as reported in the ES, an extended Phase 1 habitat survey will be conducted if the construction commences more than 2 years subsequent to the surveys conducted in October 2015. It will follow the methodology set out in the 'Handbook for Phase 1 Habitat Survey - Technique for Environmental Audit' JNCC 2010. The locations of all habitats larger than 0.5ha will be mapped, alongside any smaller, but ecologically significant habitats. The habitat types mapped will be based on the colour coding system detailed in the handbook for the Phase 1 habitat survey (JNCC, 2010). Any incidental observations of the presence, or potential presence, of protected species would be recorded to inform the need for further survey (if the status of the site has changed). Target Notes will detail any unusual habitats, species signs or areas too small to map.	In accordance with the 'Handbook for Phase 1 Habitat Survey - Technique for Environmental Audit' JNCC 2010
	Invasive species will be mapped if observed.	
Trees	Prior to construction when details of working corridors are confirmed a detailed Arboricultural Impact Assessment will be undertaken to BS 5837:2012 Trees in relation to design, demolition and construction – Recommendations. The subsequent Tree Impact and Protection Plans will highlight the trees to be lost or retained. All trees to be retained will be protected by appropriate fencing as will be detailed in a dedicated Arboricultural Method Statement.	At any time of year (Prior to construction)

Habitats/Species	Survey Methodologies	Timing
Black redstart	 Methodology: Black Redstart Action Plan Working Group http://www.blackredstarts.org.uk/pages/sitesurvey.html In areas suitable for black redstart, the site will be monitored in the breeding season; from April to August (with the core period being May to July) during construction. At least once a week a survey should be undertaken under favourable weather conditions (warm, windless days) in the early hours of the morning. 	Monitoring to take place in areas with black redstart potential weekly April to August (during construction)
Appendix B. PRELIMINARY METHOD STATEMENT – TREES

B.1 Overview

B.1.1 This Preliminary Arboricultural Method Statement provides generic best practice measures to be adopted in order to protect retained trees during the development process. It has been prepared in order to inform the OEMP and should be updated once the construction methodology and detailed design are known.

B.2 Protective Fencing

- B.2.1 The purpose of this fencing is to provide protection to the RPA of retained trees/groups and to protect trees and hedgerows prior to their translocation. The details of the actual type of fencing to be used shall be appropriate to the level of adjacent construction activity and shall be agreed with the Local Authority. Weather-proof notices shall be attached to any protective fencing located adjacent to retained trees displaying the words "Construction Exclusion Zone" and listing restrictions which apply. All personnel must be made aware of these restrictions.
- B.2.2 It is anticipated that three specifications for fencing would be employed during construction. The below gives an indication of the potential types of fencing that could be used:
- B.2.3 Low-use areas: The system illustrated in Figure B-1 is one example of a type of adequate fencing which can define areas of protected vegetation and exclude traffic. This example comprises Cleft Chestnut Pale Fence in accordance with BS 1722 Part 4: Specification for cleft chestnut pale fences (British Standards Institution, 1991) supported by 150mm wooden stakes and assembled with galvanized 14 gauge (2 mm) wire, four strands per row, peeled and pointed one end and with approximate spacing of pales of 75 mm.



Figure B-1 Tree Protection fencing example for low use areas

B.2.4 Medium-use areas: An example of such a system could comprise anti-climb weldmesh panels ('heras') connected by clamps and supported by rubber or concrete bases and bracing struts. The system is illustrated in Figure B-2 and is based on BS 5837:2012 Trees in relation to design, demolition and construction – Recommendations (British Standards Institution, 2012) (Ref 1) guidelines. This kind of system is robust enough to withstand occasional knocks by plant machinery.



Figure B-2 Tree Protection fencing specification

a) Stabilizer strut with base plate secured with ground pins



B.2.5 High-use areas: An example of such a system could involve driving scaffold poles into the ground, onto which are affixed horizontal scaffold poles and diagonal bracing struts. Anti-climb weldmesh panels could then be secured to this scaffold framework using standard scaffold clips or wire. The system is illustrated in diagram Figure. B3 and is based on BS 5837:2012 Trees in relation to design, demolition and construction – Recommendations (British Standards Institution, 2012) guidelines.



Figure B-3 Tree Protection fencing specification (extract from BS 5837)

B.3 Construction Exclusion (CEZ)

- B.3.1 The Construction Exclusion Zone (CEZ) is the area identified by an arboriculturist in the AIA to be protected during development, including site clearance and construction work, through the use of barriers and/or ground protection fit-for-purpose to ensure the successful long-term retention of a tree. The area within the construction exclusion zone will be protected during the construction works.
- B.3.2 All areas excluded by protective tree fencing shall be treated as CEZs, and the following restrictions shall apply:
 - No construction activity whatsoever must occur within these areas.

- No tree works, without the written consent from the Local Authority.
- No alterations of ground levels or conditions.
- No chemicals or cement washings.
- No excavation.
- No temporary structures.*
- No storage of soil, rubble or other materials.
- No vehicles or machinery to be used or parked without appropriate ground protection measures as per BS5837 recommendations. This will require the use of a proprietary system of reinforced concrete slabs/steel road plates on a compressible layer, or side butting scaffold boards/ 18mm plywood sheets on a compressible layer. The type of ground protection used shall be appropriate for the likely loading applied.
- No fixtures (lighting, signs etc.) to be attached to trees.
- No fires within 10 metres of the canopies of any tree or hedgerow.

*Temporary structures are acceptable provided they are of the Jack Leg type, can be sited to act as ground protection for the duration of the construction.

B.4 General construction activity

- B.4.1 Since the canopies of retained trees may be in close proximity to areas of crane operation, the following restrictions will apply, all cranes will be sited outside the defined RPAs of retained trees / groups, and the Contractor will ensure all relevant personnel shall be briefed on the location of trees and the need to avoid causing damage to them.
- B.4.2 Should additional tree removal or pruning be required above and beyond those previously agreed the Local Authority Tree Officer shall be contacted and the scope of works agreed in writing.
- B.4.3 All materials will be stored within designated areas and no materials shall be stored within any RPA.

B.5 Hazardous Materials

B.5.1 Any mixing of cement-based materials is to take place outside the RPAs of all trees. Provision shall be made to ensure that the mixing area is

contained so that no water runoff enters the RPAs of any trees. All mixers and barrows shall be cleaned within this dedicated mixing area.

B.5.2 All other chemicals hazardous to tree health, including petrol and diesel, are to be stored in suitable containers as specified by the Control of Substances Hazardous to Health (COSHH) Regulations (2002) (Ref 4), and kept away from the RPAs.

Appendix C. JAPANESE KNOTWEED ON CONSTRUCTION SITES

- C.1.1 There are several species of plant that are considered to be invasive in the UK, including Japanese Knotweed. Japanese Knotweed can cause significant delays and costs to development of brownfield sites and can spread rapidly causing damage to hard surfaces and buildings. It is therefore important that Japanese Knotweed is identified as early as possible so that an effective control programme can be implemented.
- C.1.2 Japanese Knotweed is covered by two main pieces of legislation in the UK. The Wildlife and Countryside Act (Schedule 9) makes it unlawful to 'plant or otherwise cause Japanese Knotweed to grow in the wild' and under the Environmental Protection Act 1990 Japanese Knotweed is classified as a controlled waste and thus must be accompanied by the relevant Duty of Care documents when transported and, as such must be disposed of at a licensed landfill site.
- C.1.3 The vigorous growth can cause damage to buildings and hard surfaces. Once established underneath or around the built environment, it can be particularly hard to control. Japanese Knotweed produces a network of underground stems called a rhizome. Only a very small piece of Japanese Knotweed rhizome (less than a cm) is needed to start regrowth. It is important that the rhizome is not transported to other parts of the site or onto another site.

C.2 Demarcation and Exclusion Zones around Japanese Knotweed

- C.2.1 As an initial measure, any stands of Japanese Knotweed should be identified and demarcated by the Contractor to create an exclusion zone (a minimum of 7m) and an accompanying risk register/ toolbox talk provided to the team on the ground so that people are aware of its presence and understand that no ground breaking works can be undertaken in the vicinity without appropriate management and mitigation. If Japanese Knotweed is identified as being present in the area appropriate specialist management advice must be sought.
- C.2.2 Possible methods for the eradication of Japanese Knotweed that may be utilised by the Contractor include, but are not limited to:
 - Stockpiling of Excavated Material and Herbicide Treatment: Potentially suitable for larger development/ construction sites. The area

impacted by Japanese Knotweed can be excavated and stockpiled in another area of the site for subsequent herbicide treatment/ spraying. This allows the development of the originally contaminated area to proceed immediately.

- Excavation and Mechanical Segregation: Considered to be a more environmentally friendly treatment method as it does not require the use of herbicides to eradicate the Japanese Knotweed; this method comprises the excavation and mechanical segregation/ sifting of the excavated soil to remove the Japanese Knotweed material.
- Excavation and Disposal Onsite: Where significant construction activities are taking place there is an opportunity to bury Japanese Knotweed waste on site.
- Excavation and Disposal to Landfill: This treatment method is quick and effective at ensuring the Japanese Knotweed is removed from site. However, Japanese Knotweed is classed as 'controlled waste' under the Environmental Protection Act 1990 and the costs of haulage and disposal costs associated with this treatment method are high and is considered to be a less sustainable treatment method.
- **Prevention of Spread of** Japanese Knotweed : During any works around JK, rigorous contamination control must be implemented, that is, no cross contamination of machinery or personal protection equipment should be allowed. No soil material should be taken off site, wheel washing, spade cleaning and boot cleaning before movement of personnel or equipment off site must be undertaken.
- C.2.3 Further details around Japanese Knotweed management and eradication can be found in CIRIA report C679, 2008 'Invasive species management for infrastructure managers and the construction industry' and the Environment Agency 'Managing Japanese knotweed on development sites' (version 3 amended in 2013).